UNIVERSITY OF CALGARY

Modelling Markovian light-matter interactions for quantum optical devices

in the solid state

by

Stephen Christopher Wein

A THESIS

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

GRADUATE PROGRAM IN PHYSICS AND ASTRONOMY

CALGARY, ALBERTA MARCH, 2021

© Stephen Christopher Wein 2021

Abstract

The desire to understand the interaction between light and matter has stimulated centuries of research, leading to technological achievements that have shaped our world. One contemporary frontier of research into light-matter interaction considers regimes where quantum effects dominate. By understanding and manipulating these quantum effects, a vast array of new quantum-enhanced technologies become accessible. In this thesis, I explore and analyze fundamental components and processes for quantum optical devices with a focus on solid-state quantum systems. This includes indistinguishable single-photon sources, deterministic sources of entangled photonic states, photon-heralded entanglement generation between remote quantum systems, and deterministic optically-mediated entangling gates between local quantum systems. For this analysis, I make heavy use of an analytic quantum system, which I introduce as a photon-number decomposition. This approach allows for many realistic system imperfections, such as emitter pure dephasing, spin decoherence, and measurement imperfections, to be taken into account in a straightforward and comprehensive way.

Preface

The content of this thesis was either directly developed for projects or indirectly inspired by projects that I was part of. Some of the content has already been published, some of it has not yet been published, and other parts may not be published other than in this thesis. In addition, some of the published material of which I was the primary author is presented verbatim, although it may be rearranged and modified as appropriate to suit the flow of this thesis. Other published material from works where I was a co-author but not the primary author is presented from my own perspective and cited appropriately. In every case, I have diligently detailed my contributions to all these original works to provide full transparency.

The following is a full list of published and submitted papers that I have co-authored during my graduate studies. The list is presented in chronological order. For each entry, the PDF symbol provides a link to the corresponding arXiv manuscript and the chain symbol links to the journal article page if published and the arXiv abstract page if submitted. The asterisk symbol (*) indicates equal contribution.

- [ii] I. Dhand, A. D'Souza, V. Narasimhachar, N. Sinclair, S. Wein, P. Zarkeshian, A. Poostindouz, and C. Simon. "Understanding quantum physics through simple experiments: from wave-particle duality to Bell's theorem." arXiv:1806.09958 (2018). Solution Science Press.
 Note: being developed into a textbook for Cambridge University Press.

- [v] R. Ghobadi*, S. Wein*, H. Kaviani, P. Barclay, and C. Simon. "Progress toward cryogen-free spin-photon interfaces based on nitrogen-vacancy centers and optomechanics." Phys. Rev. A 99, 053825 (2019). ▲
- [vi] H. Ollivier, I. Maillette de Buy Wenniger, S. Thomas, S. C. Wein, A. Harouri, G. Coppola, P. Hilaire, C. Millet, A. Lemaître, I. Sagnes, O. Krebs, L. Lanco, J. C. Loredo, C. Antón, N. Somaschi, and P. Senellart. "Reproducibility of high-performance quantum dot single-photon sources." ACS Photonics 7, 1050 (2020). Sources
- [vii] F. Kimiaee Asadi, S. C. Wein, and C. Simon. "Cavity-assisted controlled phase-flip gates." Phys. Rev. A 102, 013703 (2020). S
- [viii] F. Kimiaee Asadi, S. C. Wein, and C. Simon. "Protocols for long-distance quantum communication with single ¹⁶⁷Er ions." Quantum Sci. Technol. 5, 045015 (2020).
- [ix] S. C. Wein, J.-W. Ji, Y.-F. Wu, F. Kimiaee Asadi, R. Ghobadi, and C. Simon."Analyzing photon-count heralded entanglement generation between solid-state spin qubits by decomposing the master-equation dynamics." Phys. Rev. A 102, 033701 (2020).

- [xii] J.-W. Ji, Y.-F. Wu, S. C. Wein, F. Kimiaee Asadi, R. Ghobadi, and C. Simon. "Proposal for room-temperature quantum repeaters with nitrogen-vacancy centers and optomechanics." arXiv:2012.06687 (2020). ▲ S
- [xiii] H. Ollivier*, S. E. Thomas*, S. C. Wein, I. Maillette de Buy Wenniger, N. Coste, J. C. Loredo, N. Somaschi, A. Harouri, A. Lemaître, I. Sagnes, L. Lanco, C. Simon, C. Antón, O. Krebs, and P. Senellart. "Hong-Ou-Mandel interference with imperfect single photon sources." Phys. Rev. Lett. 126, 063602 (2021). 3
- [xiv] S. C. Wein, J. C. Loredo, M. Maffei, P. Hilaire A. Harouri, N. Somaschi, A. Lemaître, I. Sagnes, L. Lanco, O. Krebs, A. Auffèves, C. Simon, P. Senellart, and C. Antón-Solanas. "Photon-number entanglement generated by sequential excitation of a two-level atom." In preparation (2021).

Throughout this thesis, I will be using the above roman numeral references whenever citing works where I am a co-author. However, not all of these papers are relevant to the present topic. The papers [iii] and [ix] are the only two that are entirely included in this thesis, and the author contributions for these works are summarized below. Please also see appendix C for copyright permissions.

Author contributions for [iii]—SW and CS conceived the idea. SW and RG developed the methods. SW performed the analysis and wrote the manuscript with guidance from NL and RG. NL and CS provided critical feedback. All authors contributed to editing the manuscript.

Author contributions for [ix]—SCW and CS conceived the idea. SCW developed the methods. SCW performed the analysis and wrote the manuscript with help from JWJ, YFW, and FKA. RG and CS provided critical feedback. All authors contributed to editing the manuscript.

A portion of supplementary theory material from [xiii] and [xiv] are also described in this thesis under fair dealing, as I was the primary author of that material within those experimental papers. Moreover, I will include unpublished research related to my contributions to [vii] and [viii]. Both of these latter papers have already appeared in the thesis of the primary author, Dr. Faezeh Kimiaee Asadi. Because of this, I will make clear what material deviates from the published material and present any material from those published papers in my own original words with the appropriate citations. The remaining original material in this thesis is tangentially related to my other publications and I will detail those relationships as they become relevant.

The content of [iii] appears in section 3.1 and appendix B. In addition, [ix] appears in chapter 2 and section 4.1. Some supplementary material related to [xiii] and [xiv] is included in chapter 3. The material related to [vii] and [viii] appears in section 4.2. Aside from the introduction, the remaining material that is not attributed to a paper is, to the best of my knowledge, original unpublished research that serves to aggregate my published works into a consistent framework.

Acknowledgements

I am genuinely grateful to everyone who has contributed to my personal and academic development, allowing for the completion of this milestone in my life. Although I cannot mention everyone, I hope to highlight those who have had the most significant impact.

I would like to express my heartfelt thanks to all the people who have mentored me through my studies. Most importantly, to my advisor Prof. Christoph Simon, whose wisdom and mindful guidance has undoubtedly shaped both my career and my life. To Dr. Dean Kokonas, whose words—"*you just need to find your groove*" said to me when I was failing grade 10 math— echo in my head whenever I struggle with a concept. To Mr. Bob Shoults, who never failed to pique my curiosity for physics. To Prof. Michael Wieser, whose lectures on quantum mechanics inspired me to switch undergraduate programs and study physics. And to Prof. David Feder, whose enthusiasm brought me into physics research.

This work would not be as it is without the influence from many of my collaborators, colleagues, and peers. I would like to thank all past and present members of Prof. Simon's theory group who have shaped my research environment; and also Prof. Pascale Senellart and all members of GOSS at the Centre for Nanosciences and Nanotechnologies for inspiring discussions and projects. In particular, I am very grateful to Prof. Khabat Heshami for guidance on my first research paper, to Prof. Sandeep Goyal for initiating my interest in open quantum systems, and Dr. Faezeh Kimiaee Asadi for the many productive discussions and successful collaborations.

The completion of my doctorate would not have been possible without my family and friends. I am foremost indebted to my parents, Beverly and Marcus Wein, who are the source of my confidence and continue to teach me to think critically. I cannot thank enough my partner in life and love, Hélène Ollivier, who inspires me everyday with her insight, fortitude, and wit. I am also grateful to my siblings, Carissa Murray and Michael Wein, whose encouragement and compassion give me the determination to pursue my dreams; to Florence and Yvonnick Ollivier for all their help and kindness; and to my closest friends, Levi Eisler and Emily Wein, for their moral support.

I am fortunate to have received financial help from many donors and institutions. Thank you to all the donors and estates thereof that contributed to scholarships and bursaries that I have received during my studies, including the TransAlta Corporation, the Don Mazankowski Scholarship Foundation, Gerald Roberts and Victor Emanuel Mortimer, Wilfred Archibald Walter, Alvin and Kathleen Bothner, William Robert Grainger, and the family, friends and colleagues of Dr. Robert J. Torrence. Thank you also to the University of Calgary's Faculty of Science, Faculty of Graduate Studies, Department of Physics and Astronomy, and Institute for Quantum Science and Technology for their support. My research would not have been possible without the early support from the University of Calgary's Program for Undergraduate Research Experience (PURE) and the Natural Sciences and Engineering Research Council of Canada (NSERC) Undergraduate Student Research Award. I am also thankful for the help from the Alberta Innovates-Technology Futures Graduate Student Scholarships, the NSERC Canadian Graduate Scholarships (CGS)-Master's, the NSERC Alexander Graham Bell CGS, and the SPIE Optics and Photonics Education Scholarships.

Lastly, I would like to thank all the members of my supervisory and exam committees during my graduate studies. Thank you to my supervisory committee members: Prof. Paul Barclay, Prof. Wolfgang Tittel, Prof. Daniel Oblak, and Prof. David Hobill, for their constructive feedback and guidance. Thank you to Prof. Belinda Heyne and to Prof. David Feder for serving on my candidacy exam committee, and to Prof. David Knudsen for serving on my thesis exam committee. Finally, thank you to Prof. Stephen Hughes who served as the external examiner for my thesis defense and provided critical feedback that helped me improve my work.

Dedication

To Wolfgang and Sophie, my two wonderful children who fill my life with happiness.

Table of Contents

Al	ostrac	et				ii
Pr	eface					v
Ac	: <mark>know</mark>	ledgem	ents			vi
De	edicat	ion				viii
Ta	ble of	f Conter	nts			ix
Li	st of I	Figures				xiii
Li	List of Tables xvi					
Li	st of A	Abbrevia	ations			xvii
1	Intro	oductio	n			1
	1.1	Motiva	tion	•••		1
		1.1.1	History			1
		1.1.2	Applications	•••		6
		1.1.3	Solid-state optical devices			8
		1.1.4	Decoherence	•••		10
		1.1.5	Outline			12
	1.2	Quantu	ım systems			13

	1.2.1	Measurements	15
	1.2.2	Closed quantum systems	19
	1.2.3	Open quantum systems	20
	1.2.4	Markovian master equations	22
	1.2.5	Multi-time correlations	25
1.3	Quant	um electrodynamics	27
	1.3.1	Spontaneous emission and pure dephasing	28
	1.3.2	Jaynes-Cummings Hamiltonian	32
	1.3.3	Input-output theory	34
	1.3.4	Dissipative cavity quantum electrodynamics	35
	1.3.5	Regimes of cavity quantum electrodynamics	41
	1.3.6	Cavity-enhanced one-dimensional atom	44
	1.3.7	Quasi-normal mode master equation	46
1.4	Figure	s of merit	48
	1.4.1	Cavity cooperativity	49
	1.4.2	Brightness	50
	1.4.3	Single-photon purity	52
	1.4.4	Indistinguishability	53
	1.4.5	Fidelity	57
	1.4.6	Entanglement and concurrence	57
1.5	Analyt	tic tools	59
	1.5.1	The Fock-Liouville space	59
	1 5 0	Quantum regression theorem	60
	1.5.2		00
	1.5.2 1.5.3	Non-Hermitian Hamiltonians	62

2	Phot	ton nun	nber decomposition	67
	2.1	Condit	tional propagation superoperators	69
	2.2	Photon	a counting measurements	72
	2.3	Condit	tional correlation functions	75
	2.4	Tempo	ral density functions	77
	2.5	Relatio	on to photon statistics	82
3	Sing	le photo	D n s	84
	3.1	Single	-photon sources at room temperature	87
		3.1.1	Indistinguishability in the critical regime	90
		3.1.2	A Markovian quenching model	92
		3.1.3	Phonon sideband corrections	94
		3.1.4	Proposal for silicon vacancy centers	95
		3.1.5	Generalized Purcell factor	99
	3.2	Hong-	Ou-Mandel interference	101
		3.2.1	Separable noise model	101
		3.2.2	Pulsed two-level emitter	105
		3.2.3	Self-homodyne measurements	111
	3.3	Photon	nic state from a purely-dephased emitter	114
		3.3.1	Temporal coherence and photon statistics	115
		3.3.2	Number coherence	124
		3.3.3	Single-photon entanglement	130
		3.3.4	Two-photon coherence	135
4	Spin	-spin eı	ntanglement	142
	4.1	Herald	ed entanglement generation	146
		4.1.1	System	146
		4.1.2	Spin-photon number entanglement	149

		4.1.3	Spin-time bin entanglement	. 158
		4.1.4	Spin-polarization entanglement	. 162
		4.1.5	Protocol comparison	. 166
	4.2	Determ	ninistic entanglement generation	. 172
		4.2.1	System	. 175
		4.2.2	Simple virtual photon exchange	. 176
		4.2.3	Subradiant virtual photon exchange	. 183
5	Con	clusion		195
	5.1	Summa	ary	. 195
	5.2	Outloo	k	. 197
Ro	feren	COS		201
NC		CC5		201
AI	PPEN	DICES		226
A	Prop	perties o	f conditional propagation superoperators	227
	A .1	Integra	l form of the master equation	. 227
	A.2	Subinte	ervals of time	. 230
	A.3	Invaria	nce of order	. 231
B	Crit	ical regi	me and phonon sideband corrections	233
	B .1	Efficier	ncy	. 233
	B .2	Indistin	nguishability	. 234
	B .3	Phonor	n sideband corrections	. 237
С	Сор	yright p	ermissions	241
	C.1	Permis	sions from journals	. 241
			sions from co-authors	

List of Figures

1.1	An illustration of an open quantum system within its environment
1.2	The emitter-cavity system coupled to a one-dimensional collection waveguide 37
1.3	Log-scale illustrations of emitter-cavity QED regimes
1.4	Characterization of a pulsed single-photon source
3.1	Product of the indistinguishability I and intrinsic cavity efficiency β for an emitter- cavity system in the critical regime
3.2	Indistinguishability and efficiency estimations for a nanodiamond negatively-charged silicon vacancy (SiV ^{$-$}) center enhanced by an ultrasmall mode volume cavity 97
3.3	Two specific models for an imperfect single-photon source
3.4	Single-photon figures of merit for a purely-dephased two-level emitter driven by a square π pulse
3.5	The relationship between Hong-Ou-Mandel visibility and the integrated intensity correlation for a purely-dephased two-level emitter
3.6	Constituent mean wavepacket overlap integrals illustrating the effect of re-excitation on the single-photon indistinguishability from a purely-dephased emitter
3.7	Photon emission statistics for pulsed excitation of a two-level system
3.8	Single-photon temporal density functions for a two-level system coherently driven with a square π pulse
3.9	Single-photon temporal density functions for a two-level system coherently driven by a square 7π pulse

3.10	Two-photon amplitude correlation components and intensity correlation for a purely-dephased two-level emitter coherently driven by a square π pulse		
3.11	A comparison of measures for the indistinguishability of photons emitted from a pulsed two-level emitter experiencing emitter pure dephasing		
3.12	A comparison of the HOM visibility from a driven two-level emitter and from a driven cavity-emitter source in the bad-cavity regime		
3.13	The self-homodyne signal amplitude for a purely-dephased emitter driven by co- herent pulses		
3.14	Bipartite time-bin decomposition of a single photon: density matrix elements, Bell-state fidelity, and concurrence		
3.15	Bipartite time-bin decomposition of the single-photon indistinguishability 134		
3.16	A comparison of the bipartite time bin density matrix elements and their estimates from indistinguishability measurements		
3.17	The concurrence of a $ \phi^+\rangle$ photonic Bell state generated by a pulsed two-level emitter experiencing pure dephasing		
3.18	A comparison of the bipartite time bin density matrix elements for a $ \phi^+\rangle$ photon number Bell state and their estimates from self-homodyne measurements 140		
4.1	Diagram of the system model used to analyze remote spin-spin entanglement 147		
4.2	Diagram illustrating remote entanglement generation between pulsed L-type systems.150		
4.3	The time dynamics of remote entanglement generation via spin-photon number entanglement		
4.4	The time dynamics of remote entanglement generation via spin-time bin entanglement. 160		
4.5	Diagram illustrating remote entanglement generation between pulsed Λ -type systems. 163		
4.6	The time dynamics of remote entanglement generation via spin-polarization entan- glement		
4.7	A comparison of the optical limits to fidelity for remote entanglement generation protocols		

4.8	A comparison of phase errors for remote entanglement generation protocols 169
4.9	A comparison of loss and distance for remote entanglement generation protocols 171
4.10	The effect of emitter pure dephasing on the simple photon exchange controlled phase gate fidelity for local entanglement generation
4.11	Purcell enhancement spectrum and effective dipole-dipole coupling rate for a dual quasi-normal mode model
4.12	Time dynamics of the virtual photon exchange phase gate fidelity with a Fano resonance enhancement
4.13	Subradiant virtual photon exchange phase gate fidelity in a perfect Fano dip 190
4.14	Susceptibility to deviations from the ideal phase conditions for the subradiant virtual photon exchange phase gate fidelity
4.15	The effect of emitter pure dephasing on the subradiant photon exchange controlled phase gate fidelity using a Fano dip
B .1	Estimated nanodiamond negatively-charged silicon-vacancy (SiV ⁻) center emission spectrum
B.2	Error estimates for the Markovian approximation using a phonon sideband correction.240
C .1	Permission to include APS papers in this thesis
C.2	Permission from Christoph Simon to include Ref. [iii] and Ref. [ix] in this thesis. 243
C.3	Permission from Roohollah (Farid) Ghobadi to include Ref. [iii] and Ref. [ix] in this thesis
C .4	Permission from Nikolai Lauk to include Ref. [iii] in this thesis
C.5	Permission from Faezeh Kimiaee Asadi to include Ref. [ix] in this thesis 245
C.6	Permission from Jia-Wei Ji to include Ref. [ix] in this thesis
C .7	Permission from Yu-Feng Wu to include Ref. [ix] in this thesis

List of Tables

List of Abbreviations

Abbreviation	Definition
BD	Bin detector
BS	Beam splitter
BSM	Bell-state measurement
CW	Continuous wave
EPR	Einstein-Podolski-Rosen
FWHM	Full width at half maximum
HOM	Hong-Ou-Mandel
LDOS	Local density of states
LO	Local oscillator
NV	Nitrogen-vacancy
PNRD	Photon-number resolving detector
POVM	Positive operator-valued measure
PSB	Phonon sideband
PVM	Projection-valued measure
QD	Quantum dot
QED	Quantum electrodynamics
QIP	Quantum information processing
QNM	Quasi-normal mode
Qubit	Quantum bit
SiV	Silicon-vacancy
SPDC	Spontaneous parametric down-conversion
SPS	Single-photon source
ZPL	Zero-phonon line

Chapter 1

Introduction

In this chapter, I will establish the context for this thesis. The first section will provide a brief overview of the history of research into light and matter without delving into the modern mathematical language. This will include a description of the contemporary applications of quantum light-matter interactions for fundamental research and technology and also a motivation for the systems of focus in this thesis. In the remaining of this chapter, I will then introduce the theoretical framework and basic mathematical background used throughout this thesis.

1.1 Motivation

1.1.1 History

The conceptual relationship between light and matter has been debated for millennia. In the era of ancient Greek philosophers, it was postulated that light behaved like a ray while matter was composed of indivisible pieces or 'atomos'. However, even in the ancient world, there were popular theories proposed that light rays were also composed of indivisible particles [1]. During the Renaissance when the foundations of classical science were being established, Christiaan Huygens and Isaac Newton debated whether light was a wave-like phenomenon or if it was composed of discrete corpuscles, respectively. At the time, a particle-like model was generally accepted because

it could explain the straight motion and polarization properties of light. This dominance shifted once the wave-like properties of interference and diffraction became well-understood by the likes of Thomas Young and Augustin-Jean Fresnel [1]. On the other hand, some evidence for the particle nature of matter arose with the atomic theory of John Dalton in chemistry at the end of the 18th century [2]. Regardless of their conceptual similarities and differences, the physics and theory of light and matter were largely segregated into very different fields of study.

The 19th century brought with it the industrial revolution instigated by discoveries in thermodynamics and statistical mechanics propelled by scientists such as Sadi Carnot and Ludwig Boltzmann, among many others. Alongside these advancements in understanding matter physics were the seminal electric and magnetic experimental observations by Michael Faraday that were later developed into the classical unified theory of electromagnetism and light founded by James Clerk Maxwell [3]. This fully established the concept that light was an electromagnetic wave. Furthermore, with the discovery of the electron by Joseph John Thomson [4], the particle nature of matter was gaining support. By the end of the 19th century, it may have appeared that physics was nearing completion.

Spawning from Max Planck's curiosity for black-body radiation [1] that led him to revive the idea of discretized light, the definitive counter-intuitive connection between light and matter began to emerge. Soon after, this hypothesis was re-enforced by Albert Einstein's explanation of the photoelectric effect [5], which was observed by Heinrich Hertz. The behaviour of light could not be fully described as a wave when it interacted with matter. The existence of the fundamental particle of light—the photon—was discovered. Around the same time, Jean Perrin experimentally demonstrated the atomic nature of matter by verifying Einstein's theory of Brownian motion [5]. Less than two decades after the particle nature of matter was unambiguously confirmed came the theoretical discoveries by Louis de Broglie that matter could also exhibit wave-like behaviour, which was itself confirmed shortly thereafter by observations of electron diffraction [5]. Within a few decades, the conceptual line separating light and matter blurred into a perplexing duality that challenged the brightest minds of the 20th century.

The desire to understand and reconcile the intricate relationship between light and matter was the driving force behind the development of a new field of study—quantum physics. The pioneering advancements made by Einstein and Planck stimulated great debates and progress throughout the first few decades of the 20th century. For a while, physicists such as Niels Bohr and Arnold Sommerfeld sought to patch ever-growing inconsistencies between classical explanations and new observations in atomic physics. Then, in the fall of 1925, Werner Heisenberg along with Max Born and Pascual Jordan brought consistency by introducing the concept that properties of quantum particles of light and matter can be described by matrices [1]. Just months after, Erwin Schrödinger published his seminal paper on the differential wave equation for quantum states—the famous Schrödinger equation—that reconciled wave-particle duality as a manifestation of quantum superposition [5]. Shortly after that, Schrödinger united his theory of wave mechanics with the matrix mechanics of Heisenberg, forming quantum mechanics—the foundations of modern quantum physics.

With a mathematically consistent framework, quantum mechanics was poised to stoke the fires of a new technological revolution paralleling that of the 19th century. However, quantum mechanics troubled many of its founding authors. Many of the steps advancing towards a quantum theory were taken reluctantly. Max Planck was unsatisfied to find that quantization explained blackbody radiation. Louis de Broglie did not accept the mathematically abstract formulation of wave mechanics introduced by Schrödinger. Most famously, Albert Einstein, Boris Podolski, and Nathan Rosen together published a paper in 1935 that challenged the extent to which quantum mechanics explained reality—the EPR paradox [6]. The paradox surrounded one peculiar consequence of quantum mechanics: the special case of *entanglement* where two particles can seemingly influence each other's behaviour, even when separated by arbitrarily large distances. This "spooky action at a distance" ("spukhafte Fernwirkung"), as Einstein called it, was completely counter-intuitive and it seemed to contradict his newly-developed theory of relativity. It was also somewhat ironic that Einstein had just resolved the problem of action at a distance in gravitational physics only to discover it appears in quantum physics. Unsurprisingly, this turn of events led Einstein and his

colleagues to conclude that quantum physics must be incomplete.

The unanswered questions and confusing interpretations surrounding quantum mechanics were nonetheless impotent to hinder its influence. Regardless of its subtleties, quantum mechanics explained physical observations to such an accurate degree that it heralded an unstoppable technological revolution. It influenced developments of devices in fields such as nuclear physics, optics, and electronics; and medical technologies such as magnetic resonance imaging (MRI) and positron emission tomography (PET). In the 1960s, the laser was demonstrated after significant theoretical development following quantum theory. This invention alone allowed for uncountable applications in communications, sensing, and medical procedures. It also brought excimer laser lithography, modern transistors and, through these, semi-conductor microelectronics leading to modern computers and smart phones.

Quantum theory itself continued to develop significantly through the mid 20th century. Notably, Paul Dirac developed relativistic quantum mechanics and introduced the contemporary Dirac notation for quantum physics. Mathematician and physicist John von Neumann developed the connection to functional analysis by pioneering the modern mathematical description in terms of Hilbert spaces of quantum states, quantum measurements, and operator theory. Still, the nagging question embodied by the EPR paradox lingered.

In 1964, almost 30 years after the EPR paradox was introduced, John Stewart Bell developed a theorem that brought with it a definitive proposal to test whether the quantum predictions arising from entanglement indeed explained reality in a way that no reasonable classical theory could [7]. The proposal, now known as a Bell inequality, was a statistical test based on measuring entangled particles and comparing the average outcomes to the fundamental bounds allowed by classically correlated outcomes. Bell predicted that certain entangled states of quantum particles could indeed give stronger correlations than any classical analog. If proven correct, Bell's theorem would finally resolve the EPR paradox.

With Bell's theorem, entanglement went from being a peculiar consequence of quantum physics

to a unique resource of quantum physics. It soon became a hard-driven goal to develop ways to prepare and manipulate quantum systems to generate entangled states, and then to measure those states with high accuracy. The concept of entanglement as a resource motivated a series of technological advances, setting the stage for a second quantum revolution.

Although Bell originally proposed to use electrons to test his theorem, it soon became apparent that photons were the better choice. In 1972, John Clauser and Stuart Jay Freedman provided strong evidence for the validity of Bell's theorem using entangled photons generated by cascaded emission from calcium atoms [8]. However, the important fundamental consequences of Bell's theorem required extreme care to ensure that the experiments leave no possible classical explanation, called loopholes [9]. Hence, there were a series of experiments stretching from 1972 to 2013 that made important incremental steps towards the first loophole-free experiments published in 2015 [10–12].

Among the Bell test experiments were those of Alain Aspect *et al.* [13], Wolfgang Tittel *et al.* led by Nicolas Gisin [14], and Gregor Weihs *et al.* led by Anton Zeilinger [15] that made use of spontaneous parametric down-conversion (SPDC) [16], a process based on non-linear light-matter interaction used to generate entangled photon pairs. Notably, the first loophole-free Bell test was accomplished by Ronald Hanson's group in Delft by exploiting light-matter interaction in single atomic defects in diamond to generate entanglement between two solid-state quantum systems separated by 1.3 kilometers [10]. Shortly after, two other loophole-free Bell tests published by independent groups in Vienna and Boulder utilized entangled photons generated by SPDC, yielding impressive statistical significance of violation [11, 12]. All these Bell test experiments paralleled the development of quantum communication technology, which is one major category of second-generation quantum technology that can exploit entanglement.

Another major category of technology was spurred by the concept of entanglement as a resource. In the 1980s, it became apparent that quantum phenomena, such as entanglement, could allow for the implementation of computational algorithms that vastly exceeded classical limitations. Pioneered by physicists Paul Beinoff [17] and Richard Feymann [18], the field of quantum computation has seen huge growth over the past four decades, and a massive increase in commercial interest over the last decade. Just within the last two years, the first experimental demonstrations have been published showing that, for some specific tasks, quantum devices can outperform the best-known classical algorithms run on a supercomputer [19, 20].

1.1.2 Applications

To this day, light-matter interactions in quantum physics remains a key topic that motivates both fundamental and technological advances. The three pillars of quantum technology spearheading the second quantum revolution are (1) quantum communication, (2) quantum computation, and (3) quantum sensing. The first two pillars have already profoundly impacted the fundamental development of quantum physics, but all three open the door to devices and applications that would otherwise be impossible.

Quantum communication focuses on using the principles of superposition and entanglement to perform tasks between spatially separated parties. The most advanced application is quantumsecure key distribution [21], where one uses the laws of quantum physics to ensure the security of a generated random key to be used for encrypting classical information. Current implementations rely on the transmission of photons through telecommunication fibre optic cables [22], but long-distance communication will require free-space transmission using satellites [23–26] and quantum repeater nodes along a network of fibre optics [27]. Quantum communication networks are currently being developed around the world [28–30]. These networks may become the basis for a quantum internet [31–33], where quantum resources can be distributed across the globe to enhance the operation of other novel devices and applications involving the other two pillars of quantum technology. Furthermore, quantum networks allow for a platform on which to test fundamental questions about the nature of quantum phenomena, such as Bell inequalities and quantum collapse models [34].

Quantum computation and simulation stand to provide the most technologically impressive results. Already, optical quantum systems are capable of performing information processing tasks

in mere minutes that would take current classical algorithms on super computers billions of years [20]. Although the demonstrated tasks are system-specific and not yet useful, there are plenty of proposed algorithms and applications that quantum computers may be able to tackle in the future. The seminal example by Peter Shor promises an exponential speed-up of prime factorization [35]. Quantum simulation could solve complicated protein folding problems [36] that would lead to advances in medicine, shed light on the process of nitrogen fixation allowing for improved agriculture practices [37], accelerate materials engineering, and improve solar energy conversion and power transmission [38]. Quantum computation could also enhance artificial intelligence and machine learning [39], efficiently solve weather and climate models for accurate forecasting [40], assist financial modelling [41], and optimize routing and traffic control [42]. With the help of quantum networks, quantum computing could also be performed over spatially distributed systems and allow for blind quantum computing [43].

Quantum sensing uses quantum effects to measure a system with accuracy and resolution exceeding the limits of classical technology [44, 45]. This ranges from magnetic field sensing using atomic defects in nanoparticles [46] to superresolution of thermal light sources [47]. Quantum effects can also be exploited to improve interferometric phase measurements [48], which can increase the sensitivity of gravitational wave detectors [49, 50]. Quantum light may also improve methods for imaging biological tissues without damaging them [51]. With the help of quantum networks it may be possible to implement global synchronized access to a single quantum timekeeping device [52] and improve the resolution of astronomical objects [53, 54].

In all three of these pillars, one can find the extensive involvement of light-matter interactions [55]. In particular, many technologies such as quantum networks, optical quantum computing, and biological imaging require the generation of non-classical light, such as single photons and entangled photonic states. These types of states are usually produced by the interaction of a laser with a solid-state material or atomic gas. For example, as mentioned before, SPDC can produce entangled photons. It is also one of the most common ways to generate high-quality single photons. This process occurs when laser light passes through a non-linear crystal. More recently, there has

been significant motivation to study how laser light can manipulate optically-active single quantum defects in the solid-state to produce non-classical light and light-matter entanglement [56, 57]. These solid-state emitter systems have great potential to accelerate the development of many, if not all, of the technological applications mentioned above.

1.1.3 Solid-state optical devices

The technological evolution of classical computing and the internet has shown a clear trend towards favoring compact, scalable, and robust devices. For example, vacuum tube transistors were replaced by contemporary solid-state silicon-based technology while fiber optics and satellites have now replaced many electrical transmission lines for long-distance communication. A similar trend could be expected for the development of quantum technology whereby solid-state materials and optical light become more favored over time.

In the previous section, I touched on the multitude of applications of quantum technology. In particular, optically-active solid-state defects will continue to play a huge role in the future of the ongoing quantum technology revolution [56, 57]. These defects could provide many advantages over other platforms for quantum technology such as trapped ions [58] or superconducting circuits [59]. In addition, optically-active defects have the potential to satisfy any future demand for scalability and robustness. However, there are still many challenges to overcome. In this section, I will highlight some of the advantages and disadvantages of this platform.

The optical spectrum of light combines the infrared, visible, and ultraviolet ranges of the electromagnetic spectrum, comprising wavelengths near 100s of micrometers to 10s of nanometers. In terms of frequency, this can range from 100s of GHz to 1000s of THz. The range covers two prominent telecommunication bands at 1300 nm and 1500 nm, where transmission losses are low through fibre optics. It also covers the operational wavelengths of commercially available lasers such as titanium-sapphire and helium-neon lasers. Many semi-conductor materials, such as silicon, gallium arsenide, and diamond, have optical band-gaps that allow for addressable defects, such

as quantum dots (QDs) [60] and nitrogen-vacancy centers (NV centers) [61]. Optical light also transmits well in free space and can be manipulated with many off-the-shelf components such as silvered or dielectric mirrors, half-silvered beam splitters, objectives, and lenses. There are also plenty of commercially-available efficient detectors for optical light.

Optical light (as with all light) is composed of photons, which generally interact only weakly with other particles. This makes it difficult to manipulate photons to perform logic gates, usually requiring probabilistic post-selection to accomplish [62]. They are also particles that travel quickly and are difficult to contain. This makes photons ideal for transmitting quantum information over long distances [57] but a poor choice for storage of quantum information in one location. That said, there are very promising proposals for all-optical quantum computing [63]. But even then, light-matter interactions are necessary to generate and detect the quantum states of light used to implement all-optical quantum information processing.

Solid-state quantum defects are microscopic or atomic-scale defects within a crystal lattice that can spatially confine electrons. This electron confinement is similar to how a nucleus can trap electrons to form an atom. These quantum systems are referred to as artificial atoms for this reason. Electrons trapped by a defect can exist in different quantum states, or electronic orbitals, dictated by the defect structure. When there are multiple electrons, they together occupy the molecular orbitals of the defect. Since electrons interact with light, laser pulses can be used to provide the right amount of energy to excite electrons from one orbital to another. Excited electrons may also spontaneously emit photons back into the environment. For this reason, I will refer to a single quantum defect, such as a QD or NV center, that is in an electronic configuration allowing for the emission of photons, as an emitter.

Each electron confined to a defect can itself exist in one of two spin states. Depending on the number of confined electrons and the defect symmetry, there may be optically-active spin states that can be manipulated by external static or dynamic electric and magnetic fields. This can provide a robust degree of freedom for manipulating and storing quantum information. Thus, optically active solid-state defects allow for a rich space of quantum states that are spatially localized, can

be externally manipulated, may contain spin degrees of freedom, that can coherently interact with photons, and generate quantum states of light [57].

Solid-state defects have some prominent advantages over other stationary quantum systems, such as confined atomic gases or trapped atom/ion systems. Because the defects are integrated into a solid-state lattice, they do not require external manipulation to keep them localized in space. In addition, there are many well-developed methods for fabricating solid-state systems that can be adopted from semi-conductor microelectronics. This latter advantage also allows for the natural integration of quantum defects with classical information processing architectures, potentially providing a smooth and scalable transition to quantum-enhanced devices.

By combining the benefits of optical light with those of optically-active solid-state defects that may contain spin degrees of freedom, we have a platform that can prepare, manipulate, store or transmit quantum information. This solid-state spin-photon platform satisfies all the main requirements for quantum technology to succeed [56, 64]. However, solid-state environments are hostile to delicate quantum phenomena. In the next section, I will discuss the biggest disadvantage of using solid-state materials for quantum information processing.

1.1.4 Decoherence

The primary challenge for developing quantum technology is decoherence. This occurs when a quantum system's environment perturbs the quantum system in an uncontrolled, probabilistic way [65]. The system then deviates from the desired quantum state, which can destroy the quantum properties such as superposition and entanglement that are required for quantum technology to operate. Decoherence occurs quickly for solid-state systems because there are many ways that the surrounding host environment can affect the state of the quantum defect. Thus, the main advantage of solid-state defects also gives rise to its biggest weakness. The three main culprits of decoherence in solid-state systems are thermal vibrations, electric noise, and magnetic noise [66].

Bulk thermal vibrations of atoms in the host lattice can displace the electronic orbitals of the

quantum defect, causing decoherence. The quantized description of these thermal vibrations are called phonons. Decoherence becomes significantly worse as the lattice temperature increases. However, even at very low temperatures, energy contained in the defect can also spontaneously dissipate into the environment as phonons, which also causes decoherence. The interaction between the confined electrons and lattice phonons can dampen the light-matter interaction, putting limits on the speed and fidelity at which the electron can be manipulated with laser pulses [67].

Electronic noise is often induced by charge fluctuations [68]. This can occur if there are uncontrolled electrons moving into and out of other nearby defects or moving across the material surface. As a consequence, the local electric field experienced by the electrons within the defect can change. The rate of charge fluctuations is also temperature dependent. However, the severity depends a lot on the surface quality (if near a surface) and the density of the surrounding defects.

Magnetic noise is usually caused by random changes in the spin states of surrounding atomic nuclei. If a nearby nucleus flips from one spin state to another, this can change the local magnetic field experienced by electrons within the defect [66]. The many surrounding nuclei form what is called a spin bath. Random fluctuations in the state of this nuclear spin bath can cause decoherence of the electronic spin state of the defect.

In some cases, the environmental noise can be minimized. For example, low-frequency phonons may be suppressed by cutting a pattern into the material to forbid some vibration modes [69]. However, this is not effective against high-frequency phonons. Charge noise can often be controlled by applying a static electric field to lock free charges in place [68]. Magnetic noise can sometimes be suppressed by isotopically engineering the material so that the nuclei do not have a spin degree of freedom at all [70]. Unfortunately, not all atoms have stable isotopes with zero nuclear spin. For some slower sources of noise, such as magnetic field fluctuations, rapid pulse sequences can also be used to dynamically decouple the quantum system from its environment [71].

Besides minimizing noise, decoherence can be overcome by operating the device much faster than the timescale of decoherence. This means that the system should be prepared, manipulated, and measured before decoherence can occur. For interacting quantum subsystems, it is sometimes possible to enhance the rate of interaction to complete the protocol before decoherence becomes a major hindrance to the operation of the device. Unfortunately, often a portion of decoherence occurs on a timescale that cannot be realistically overcome [72]. In some cases for large quantum systems, certain measurements and operations can be applied to correct for errors [73]. This may one day allow for protocols that can be implemented over arbitrary lengths of time. However, these error correction protocols require the system to already satisfy some threshold of quality to be successful [74]. Hence, understanding and circumventing the effects of decoherence is still essential for the advancement of quantum technology.

1.1.5 Outline

There are many different physical implementations of solid-state quantum optical devices for many different applications. Each implementation may be affected by its environment in a different way than the next. In this thesis, I analyze arguably the most simple type of model that can capture effects of decoherence on a quantum system: the phenomenological Markovian master equation. As I will discuss later in this chapter, this approach makes some simplifying assumptions about how a quantum system interacts with its environment. This allows for overarching and generalized results that capture the basic behaviour and limitations of many different devices. In addition, the methods I detail in chapter 2 and apply throughout the thesis can also be applied to devices with more specific types of Markovian master equations that take into account additional subtleties in the behaviour of a device. However, Markovian master equations usually cannot capture all subtle details of a particular system, especially those in solid-state environments, and may not capture limiting behaviour in some regimes of operation. For this reason, the Markovian master equation approach used in this thesis should be considered a first step in modelling decoherence that gives insight to further develop system-specific models that are more closely tailored for a particular device.

I will use the Markovian master equation approach to explore the effects of environment noise on two main types of applications of solid-state optical defects. The first type of application, discussed in chapter 3, are those related to the generation of single photons (sections 3.1 and 3.2) and other pulsed photonic states (section 3.3). The second type of application, discussed in chapter 4, is related to using photons to mediate the interaction between two different defects. In section 4.1, I cover a few protocols that can be used to generate entangled spin states between remote defects. Section 4.2 follows up with a discussion on using photons to mediate deterministic local interactions between spins states of different defects. Finally, I summarize the main results and conclude the thesis in chapter 5.

1.2 Quantum systems

In this section, I will introduce the basics of quantum mechanics that set the foundation for the analysis used in this thesis. In order to be concise and pertinent, I focus on using the Schrödinger picture to introduce discrete variable systems that do not contain degenerate states nor utilize degenerate measurements of those states. The majority of content in this section and those that follow is drawn from the open quantum systems textbook by H.-P. Breuer and F. Petruccione [75].

A quantum system is a system described by a Hilbert space \mathbb{H} of possible states. A state in this Hilbert space is represented by a complex-valued vector $|\psi\rangle$. This vector can be written as a linear superposition of orthonormal basis vectors $|\phi_n\rangle \in \mathbb{H}$ so that $|\psi\rangle = \sum_n c_n |\phi_n\rangle$. The complex amplitudes $c_n \in \mathbb{C}$ are given by the inner product $c_n = \langle \phi_n | \psi \rangle$, where $\langle \phi_n |$ is the covector of $|\phi_n\rangle$ that belongs to the dual space of \mathbb{H} . The amplitudes c_n have a particular physical interpretation as giving the probability $\Pr(\phi_n) = |c_n|^2$ of finding the quantum system in the state $|\phi_n\rangle$. Hence, the state $|\psi\rangle$ must be restricted to satisfy the normalization condition $\sum_n \Pr(\phi_n) = 1$. The stochastic behaviour of the quantum system under measurement is entirely quantum in that it cannot be described by reasonable classical models [7, 76, 77]. I will discuss measurements in more detail in the following section. A quantum system can change from one state to another. Changes in the system state are described by linear operators denoted $\hat{U} : \mathbb{H} \to \mathbb{H}$, which have a particular property that they maintain the normalization condition of quantum states. That is, they are unitary transformations so that $\hat{U}^{\dagger}\hat{U} = \hat{U}\hat{U}^{\dagger} = \hat{I}$, where \hat{I} is the identity operator. There are also linear operators \hat{A} acting on the Hilbert space that are not unitary. These operators can describe the time dynamics, average physical quantities, and measurements of the system.

Just like a state, every linear operator can be written in terms of an orthonormal basis $|\phi_n\rangle$ of the Hilbert space as $\hat{A} = \sum_{n,m} A_{n,m} |\phi_n\rangle \langle \phi_m|$, where $A_{n,m} = \langle \phi_n | \hat{A} | \phi_m \rangle$ are the complex-valued matrix elements of \hat{A} and $|\phi_n\rangle \langle \phi_m|$ is the outer product of states $|\phi_n\rangle$ and $|\phi_m\rangle$. The expectation value of a linear operator on the Hilbert space is determined by the quantum state $|\psi\rangle$ using $\langle \hat{A} \rangle = \langle \psi | \hat{A} | \psi \rangle = \sum_{n,m} c_n^* c_m A_{n,m}$.

An operator \hat{O} is said to be an observable if it is Hermitian $\hat{O}^{\dagger} = \hat{O}$. Observable operators correspond to physical quantities of the system that can be measured. By convention, we define the adjoint \hat{O}^{\dagger} as the conjugate transpose of the operator's matrix representation. This implies that all observables have real eigenvalues and that they are diagonalizable. Hence, we can always find a new basis $|\varphi_n\rangle$ such that an observable \hat{O} can be written as $\hat{O} = \sum_n \alpha_n |\varphi_n\rangle \langle \varphi_n|$, where α_n are the eigenvalues satisfying $\hat{O} |\varphi_n\rangle = \alpha_n |\varphi_n\rangle$.

In practice, our knowledge about the system state may be incomplete so that it is best described as a classical statistical mixture of possible quantum states. For example, it could be in the state $|\psi_1\rangle$ with probability $\Pr(\psi_1)$ and state $|\psi_2\rangle$ with probability $\Pr(\psi_2) = 1 - \Pr(\psi_1)$. This type of classical uncertainty in the state of the system is physically very different from the quantum uncertainty described at the beginning of this section. Rather than tracking all possible statistical outcomes of a quantum system individually, we can describe the statistically mixed state of a quantum system using an observable operator, the density operator $\hat{\rho}$. Suppose we want the expectation value of $\hat{\rho}$ for a given state $|\phi_n\rangle$ to be $\Pr(\phi_n)$, it is then intuitive that the operator $|\psi\rangle\langle\psi|$ is chosen so that $\Pr(\phi_n) = \langle \phi_n | \hat{\rho} | \phi_n \rangle = |\langle \phi_n | \psi \rangle|^2$. In general, we have $\hat{\rho} \equiv \sum_n \Pr(\psi_n) |\psi_n\rangle\langle\psi_n|$ where $\sum_n \Pr(\psi_n) = 1$. A quantum system that is in a state $|\psi\rangle$ where $\Pr(\psi) = 1$ is said to be in a pure quantum state. Otherwise, it is said to be in a mixed state.

Written in terms of the orthonormal basis $|\phi_n\rangle$, a general quantum state is

$$\hat{\rho} = \sum_{n,m} \rho_{n,m} |\phi_n\rangle \langle \phi_m|$$
(1.1)

where $\rho_{n,m}$ are called the density matrix elements. The diagonal elements $\rho_{n,n} = \langle \phi_n | \hat{\rho} | \phi_n \rangle = \Pr(\phi_n)$ are equal to the probability of finding the quantum system in state $|\phi_n\rangle$ while the off-diagonal elements $\rho_{n,m} = \langle \phi_n | \hat{\rho} | \phi_m \rangle$ ($n \neq m$) quantify the amount of quantum coherence between states $|\phi_n\rangle$ and $|\phi_m\rangle$. Hence, with this formalism, both the quantum and classical statistical behaviours of a quantum system are captured.

The density operator is Hermitian $\hat{\rho}^{\dagger} = \hat{\rho}$ and by definition we have $\text{Tr}(\hat{\rho}) = \sum_{n} \rho_{n,n} = 1$, where Tr is the trace. As a consequence of these two properties, the magnitude of the off-diagonal elements of the density matrix are bounded by their corresponding diagonal elements by $|\rho_{n,m}|^2 \leq \rho_{n,n}\rho_{m,m}$. In a related way, the trace of the density operator squared always satisfies $1/\text{dim}(\hat{\rho}) \leq \text{Tr}(\hat{\rho}^2) \leq 1$, where $\text{dim}(\hat{\rho})$ is the dimension of $\hat{\rho}$ [75]. This inequality is only saturated to 1 when $\hat{\rho}$ is a pure state and it is saturated at the lower bound when it is fully mixed. This quantity $\mathcal{P} = \text{Tr}(\hat{\rho}^2)$ is called the trace purity, and it gives a natural way to characterize how pure a quantum state is.

1.2.1 Measurements

In order to observe and exploit quantum properties of the system, it is necessary to measure the state of the system. At the beginning of the previous section, I alluded to the idea of 'finding' the quantum system in a particular state, but gave little indication to how that is mathematically achieved. I will first introduce the idea of an ideal quantum measurement and then expand this concept to include imperfections.

The ideal von Neumann approach [78] used for instantaneous measurements associates every measurement with an observable acting on the Hilbert space. As discussed before, the average

value of an observable operator is given by the state of the system $|\psi\rangle$ through the real-valued expectation $\langle \hat{O} \rangle = \langle \psi | \hat{O} | \psi \rangle$. This concept can be extended to a quantum system state represented by a density operator $\hat{\rho}$ by the relation $\langle \hat{O} \rangle = \text{Tr} (\hat{O} \hat{\rho})$.

Since every observable is diagonalizable, it can be written in terms of a set of operators $\hat{\Pi}_n = |\varphi_n\rangle\langle\varphi_n|$ so that $\hat{O} = \sum_n \alpha_n \hat{\Pi}_n$. The operators $\hat{\Pi}_n$ are projection operators because $|\varphi_n\rangle$ form an orthonormal basis making $\hat{\Pi}_n^2 = \hat{\Pi}_n$ idempotent. They are also complete: $\sum_n \hat{\Pi}_n = \hat{I}$. Thus, the action of \hat{O} becomes clear when applied to the system state. It projects the state $|\psi\rangle$ onto state $|\varphi_n\rangle$ with probability $\Pr(\alpha_n) = |\langle \psi | \varphi_n \rangle|^2$ and returns the associated eigenvalue α_n . The expectation value of \hat{O} is then interpreted as the average value of all possible measurement outcomes α_n weighted by the probability of projecting the quantum system onto the eigenstate $|\varphi_n\rangle$.

Once the quantum system has been ideally measured with the outcome α_n , it is in the state $|\varphi_n\rangle$. This occurs with the probability $\text{Tr}(\hat{\Pi}_n \hat{\rho} \hat{\Pi}_n) = \text{Pr}(\alpha_n)$. Hence, after an ideal measurement, the system is in the pure state

$$|\varphi_n\rangle\langle\varphi_n| = \frac{\Pi_n\hat{\rho}\Pi_n}{\mathrm{Tr}\left(\hat{\Pi}_n\hat{\rho}\hat{\Pi}_n\right)}$$
(1.2)

even if it was in a mixed state before the measurement. This type of idealized measurement directly implemented using projection operators is called a projection-valued measure (PVM).

To simplify notation, it is convenient to define an operator \mathcal{P}_n so that $\mathcal{P}_n \hat{\rho} = \hat{\Pi}_n \hat{\rho} \hat{\Pi}_n$. This operator is our first example of a superoperator, which is an operator that takes an operator on the Hilbert space and returns another operator on the Hilbert space. For clarity and consistency, I will notate all superoperators in this thesis using the calligraphic font. In addition, I assume that they act on all operators situated to their right, unless otherwise specified. Using this projection superoperator, we can write $|\varphi_n\rangle\langle\varphi_n| = \mathcal{P}_n\hat{\rho}/\text{Tr}(\mathcal{P}_n\hat{\rho})$. The projection superoperators are idempotent $\mathcal{P}_n^2 = \mathcal{P}_n$. However, in general, they are not complete. That is, $\sum_n \mathcal{P}_n$ is not necessarily the identity superoperator \mathcal{I} because the sum does not contain the superoperator projections onto $|\varphi_n\rangle\langle\varphi_m|$. This is because a projective measurement will remove quantum coherence between different eigenstates $|\varphi_n\rangle$ of the observable.

The power of measurements is now clear. They allow one to purify a quantum system by preparing it in a specific quantum state. In addition, a measurement gives a small piece of information about what state the system was in before the measurement. By repeatedly preparing and measuring a quantum system, it is possible to build up enough information to get a clear picture about how the quantum system behaves.

Not every measurement is ideal in that it can project the system onto a pure state. For example, in most realistic scenarios, there can be some classical noise in the device used to measure the quantum system. This noise may prevent the device from faithfully reporting the value α_n given that the quantum system was in the state $|\varphi_n\rangle$. Instead, it could erroneously indicate some other outcome $\alpha_m \neq \alpha_n$. In that case, it is not guaranteed that the state $\hat{\rho}$ is in the state $|\varphi_m\rangle$ if the device reports α_m . There is some remaining chance that it is actually in the state $|\varphi_n\rangle$ and hence the quantum system is in a mixed state of $|\varphi_m\rangle$ and $|\varphi_n\rangle$ after the measurement.

We can describe the effect of classical noise of this non-ideal measurement in terms of conditional probabilities. I will first describe a simple scenario so that when I write the result it becomes clear what it physically represents. Suppose we have a device that gives two possible outcomes 0 and 1, and projects onto two possible orthogonal states $|\downarrow\rangle$ and $|\uparrow\rangle$. If we characterize the device by supplying it with many copies of $|\uparrow\rangle$ (ignoring the fact that we need a source of perfect $|\uparrow\rangle$ states), we can determine the probabilities $Pr(1|\uparrow)$ ($Pr(0|\uparrow)$) that it reports 1 (0) given the input $|\uparrow\rangle$. Likewise, we can do the same for the input $|\downarrow\rangle$. Hence, after characterization, we have a set of four conditional probabilities $Pr(1|\uparrow)$, $Pr(0|\uparrow)$, and $Pr(0|\downarrow)$.

Using Bayes' theorem we can write the probability of the state being in either $|\uparrow\rangle$ or $|\downarrow\rangle$ given that we measure 0 or 1. For example, $\Pr(\uparrow|1) = \Pr(1|\uparrow) \Pr(\uparrow) / \Pr(1)$. Then, if we measure outcome 1, we know that the state after measurement is the mixed state $\hat{\varrho}_1 = \Pr(\uparrow|1) |\uparrow\rangle\langle\uparrow| +$ $\Pr(\downarrow|1) |\downarrow\rangle\langle\downarrow|$. Substituting in Bayes' theorem and applying our ideal measurement postulate gives us $\hat{\varrho}_1 = (\Pr(1|\uparrow) \mathcal{P}_{\uparrow}\hat{\rho} + \Pr(1|\downarrow) \mathcal{P}_{\downarrow}\hat{\rho}) / \Pr(1)$. We can now identify another important superoperator $\mathcal{F}_1 = \Pr(1|\uparrow) \mathcal{P}_{\uparrow} + \Pr(1|\downarrow) \mathcal{P}_{\downarrow}$ so that $\hat{\varrho}_1 = \mathcal{F}_1 \hat{\rho} / \Pr(1)$. Also, since $\operatorname{Tr}(\hat{\varrho}_1) = 1$, we can easily identify that $\operatorname{Tr}(\mathcal{F}_1 \hat{\rho}) = \Pr(1)$. Now, we can do the same procedure for $\hat{\varrho}_0$ to find the superoperator \mathcal{F}_0 . These two superoperators together form an implementation of what is called a positive operator valued measure (POVM).

Let us generalize this concept. Using a set of projection superoperators \mathcal{P}_n , we saw from above that we can write a POVM implementation as $\mathcal{F}_n = \sum_m \Pr(\alpha_n | \varphi_m) \mathcal{P}_m$. Then, the probability of measuring outcome α_n is $\Pr(\alpha_n) = \operatorname{Tr}(\mathcal{F}_n \hat{\rho})$ and the state after measurement is

$$\hat{\varrho}_n = \frac{\mathcal{F}_n \hat{\rho}}{\operatorname{Tr}(\mathcal{F}_n \hat{\rho})}.$$
(1.3)

Therefore, for the extension to imperfect measurements using a POVM implementation, the superoperator \mathcal{F}_n plays the role analogous to \mathcal{P}_n in the PVM implementation. However, in general, the set of operators \mathcal{F}_n are neither complete $\sum_n \mathcal{F}_n \neq \mathcal{I}$ nor idempotent $\mathcal{F}_n \neq \mathcal{F}_n^2$.

In both the PVM and POVM implementations described above, the basis of projection operators $\hat{\Pi}_n$ (or projection superoperators \mathcal{P}_n) determines how the measurement affects the state. However, for a given set of measurement outcome probabilities $\Pr(\alpha_n | \varphi_n)$, the set of superoperators \mathcal{F}_n is not uniquely determined for a POVM. In the simplified scenario above, I assumed *a priori* that the measurement performed projections onto the $\{|\uparrow\rangle, |\downarrow\rangle\}$ basis. However, this cannot be confirmed by only measuring the two quantum states $|\uparrow\rangle$ and $|\downarrow\rangle$. In practice, it is necessary to characterize the measurement device by measuring multiple sets of orthogonal states to fully determine the unique set of superoperators \mathcal{F}_n that form the POVM implementation. Furthermore, I have neglected to describe the generalization of \mathcal{F}_n where the comprising superoperators \mathcal{P}_m are not necessarily projection superoperators. In fact, a general POVM implementation is built from any set of positive operators $\hat{F}_m = \hat{M}_m^{\dagger} \hat{M}_m$ known as POVM elements that satisfy the property $\sum_m \hat{F}_m = \hat{I}$. Then, the general form of \mathcal{P}_m is the jump superoperator of \hat{M}_m : $\mathcal{P}_m \hat{\rho} = \hat{M}_m \hat{\rho} \hat{M}_m^{\dagger}$. The details of this generalization and the general properties of \mathcal{F}_n are given in Ref. [75].

1.2.2 Closed quantum systems

The time evolution of a quantum state is governed by the observable operator corresponding to the total system energy—the Hamiltonian $\hat{H}(t)$. This time evolution is described by the Schrödinger equation

$$i\hbar \frac{d}{dt} |\psi(t)\rangle = \hat{H}(t) |\psi(t)\rangle , \qquad (1.4)$$

where \hbar is Planck's constant. Here, I am using the total time derivative, as in Ref. [75], to illustrate that the equation is basis independent, as opposed to a partial derivative used in the common Schrödinger wave equation that additionally includes position as a variable. The solution to the Schrödinger equation is given by the unitary transformation acting on the Hilbert space $\hat{U}(t, t_0)$, where $|\psi(t)\rangle = \hat{U}(t, t_0) |\psi(t_0)\rangle$. This unitary operator also satisfies the Schrödinger equation

$$i\hbar\frac{\partial}{\partial t}\hat{U}(t,t_0) = \hat{H}(t)\hat{U}(t,t_0), \qquad (1.5)$$

with the initial condition $\hat{U}(t_0, t_0) = \hat{I}$. Here, I am using a partial derivative to emphasize that t_0 is held constant. Likewise, the evolution of the covector $\langle \psi(t) |$ is given by the adjoint propagator $\hat{U}^{\dagger}(t, t_0)$ and so $\langle \psi(t) | = \langle \psi(t_0) | \hat{U}^{\dagger}(t, t_0)$.

For time-independent Hamiltonians, and like any matrix differential equation, the propagation operator is readily computed by exponentiation $\hat{U}(t, t_0) = e^{-i(t-t_0)\hat{H}/\hbar}$, which can be solved by diagonalizing the Hamiltonian \hat{H} . For time-dependent Hamiltonians, it is possible to solve the system using a time-ordered integration, analytic coarse graining [79], or by various numerical integration techniques.

The evolution of the density operator is also governed by the propagator \hat{U} . Since each quantum state $|\phi_n(t_0)\rangle$ that form a basis evolves like $|\phi_n(t)\rangle = \hat{U}(t,t_0) |\phi_n(t_0)\rangle$ (and similarly for $\langle \phi_n(t)|$), we have that $\hat{\rho}(t) = \hat{U}(t,t_0)\hat{\rho}(t_0)\hat{U}^{\dagger}(t,t_0)$. In fact, by taking the derivative of $\hat{\rho}(t)$, applying the product rule, and using Eq. (1.5) along with its adjoint, we arrive at the von Neumann equation of
motion for the density operator

$$i\hbar \frac{d}{dt}\hat{\rho}(t) = \left[\hat{H}(t), \hat{\rho}(t)\right]$$
(1.6)

where $[\hat{A}, \hat{B}] = \hat{A}\hat{B} - \hat{B}\hat{A}$ is the commutator of operators \hat{A} and \hat{B} .

I will follow the convention of [75] and define a closed quantum system as one whose evolution is described by the Schrödinger equation or von Neumann equation for mixed states. It is also useful to distinguish a closed quantum system from an isolated quantum system, which is a closed system where the Hamiltonian is time-independent. This latter case implies that the average energy $\langle \hat{H} \rangle$ of the system remains constant with respect to time. However, a closed quantum system can still be externally manipulated, and hence experience a change in average energy.

One important property of a closed quantum system is that the trace purity \mathcal{P} remains constant. That is, an initial pure state will remain pure while an initial mixed state will stay just as mixed. The proof of this follows readily from the fact that \hat{U} is unitary, which implies $\hat{U}^{\dagger}\hat{U} = \hat{I}$, and the cyclic property of trace. Because the evolution of a closed quantum system is governed by a unitary transformation, it is also said to be time reversible. This means that an original state $|\psi(t_0)\rangle$ can be recovered by applying the inverse transformation $\hat{U}^{\dagger}(t, t_0) = \hat{U}(t_0, t)$, which implies $\hat{U}(t_0, t) |\psi(t)\rangle = |\psi(t_0)\rangle$ for $t_0 \leq t$. However, there are many quantum systems where reversible evolution seemingly does not occur because they are not closed systems—they inevitably interact with their environment.

1.2.3 Open quantum systems

In many cases, the size of the quantum system model would have to be very large to capture all the interacting systems affecting the device. This is particularly true for solid-state based quantum optical devices where there are many different interacting subsystems, such as confined electrons, charge noise, vibrations in the host lattice, a surrounding bath of nuclear spins, and electromagnetic cavity modes. This makes an exact mathematical description of the dynamics using the Schrödinger equation very difficult to solve. Even if a solution is managed, the vast



Figure 1.1: An illustration of an open quantum system within its environment. The open quantum system is affected by the noisy environment. It also leaks information into the environment by, for example, photon emission. This emission can be detected by an auxiliary measurement system to gain information about the state of the open quantum system.

number of degrees of freedom also makes it difficult to distill that solution into relevant results that are easily applicable to experiments. Open quantum systems theory tackles this problem by first reducing the state space of the total system using a series of approximations. However, the dynamics of the resulting reduced state space is no longer governed by the Schrödinger equation as irreversible state evolution becomes possible.

The first step towards obtaining the dynamics of the reduced system is to categorize the subsystems composing the total system. Some subsystems may interact only weakly with other subsystems such that they primarily introduce a quantum noise to the desired state evolution. Some other subsystems may interact strongly, contributing significantly to the quantum dynamics. Finally, some subsystems may be measured by an external system to gain information about the state.

I will proceed in the fashion outlined above by dividing subsystems into three categories (see Fig. 1.1). First, I call the *open quantum system* the subsystem of interest combined with all subsystems that strongly interact with it. The open quantum system is also generally referred to as the reduced quantum system or, plainly, the quantum system. This open quantum system interacts with its *environment*, which is composed of the remaining subsystems that interact only weakly with the open quantum system. Finally, we can have an auxiliary system which serves to measure one or more subsystems of the environment. This *measurement system* may be quantum

or classical. However, in this thesis, I will only ever consider a classical measurement system. Furthermore, if a subsystem in the environment has infinite degrees of freedom, such as a continuum of electromagnetic modes, it is called a *reservoir*. If a reservoir is in a thermal equilibrium state, I will refer to it as a *bath*.

For now, let us set aside the measurement system and consider only the open quantum system state $\hat{\rho}_{\rm S} = \text{Tr}_{\rm E}(\hat{\rho})$ and the environment state $\hat{\rho}_{\rm E} = \text{Tr}_{\rm S}(\hat{\rho})$, where $\text{Tr}_{\rm k}$ is the partial trace over subsystems within $k \in \{S, E\}$. By applying the Schrödinger equation, the total system density operator $\hat{\rho} = |\psi\rangle\langle\psi|$ was found to follow $i\hbar d\hat{\rho}(t)/dt = [\hat{H}(t), \hat{\rho}(t)]$. Then the reduced system state formally satisfies the Liouville-von Neumann first-order differential equation

$$i\hbar \frac{d}{dt}\hat{\rho}_{\rm S}(t) = \mathrm{Tr}_{\rm E}\Big[\hat{H}(t), \hat{\rho}(t)\Big].$$
(1.7)

1.2.4 Markovian master equations

For any closed quantum system, there exists a dynamical map $\hat{U}(t_{\rm f}, t_0)$ so that any future quantum state $|\psi(t_{\rm f})\rangle = \hat{U}(t_{\rm f}, t_0) |\psi(t_0)\rangle$ depends only on the current quantum state $|\psi(t_0)\rangle$. In general, this is not true for an open quantum system because the reduced system can alter its environment and the environment can, in turn, affect how the reduced quantum system evolves. This environment back-action means that the evolution of a quantum state $\hat{\rho}_{\rm S}(t_0)$ of the reduced system depends on the system state before time t_0 . However, if the environment is a bath that returns to its equilibrium state very quickly after a slight perturbation due to the evolution of the reduced quantum system, the characteristic timescale $\tau_{\rm B}$ of this back-action can be very short—meaning that the evolution of $\hat{\rho}_{\rm S}(t_0)$ only depends on its history up to $\sim \tau_{\rm B}$ backwards in time. This timescale is called the bath memory time [75].

If the dominant dynamics of the reduced quantum system occurs on a timescale $\tau_{\rm S}$ that is much longer than the bath memory time $\tau_{\rm B}$, the evolution of the state of the reduced quantum system approximately no longer depends on its history. In this case, there can exist a dynamical map such that any future state depends only on its current state. This property is known as Markovianity [75]. All systems studied in this thesis are assumed to satisfy this very useful property of Markovianity. Formally, a Markovian system is one where there exists a dynamical map $\mathcal{U}(t_{\rm f}, t_0)$ with the semigroup property $\mathcal{U}(t_{\rm f}, t')\mathcal{U}(t', t_0) = \mathcal{U}(t_{\rm f}, t_0)$ where $t_{\rm f} \ge t' \ge t_0$ such that the state of the open quantum system $\hat{\rho}_{\rm S}$ at time $t_{\rm f}$ is fully described by $\hat{\rho}_{\rm S}(t_{\rm f}) = \mathcal{U}(t_{\rm f}, t_0)\hat{\rho}_{\rm S}(t_0)$ for any $t_{\rm f} \ge t_0$.

The superoperator \mathcal{U} is analogous to the unitary transformation \hat{U} in that it serves as a propagator of the system state but with the very important difference that the propagtion superoperator \mathcal{U} is almost never unitary, meaning that it captures irreversible processes such as the spontaneous emission of a photon from an excited emitter. It can also capture basic aspects of other decoherence processes discussed in section 1.1.4.

For a fixed time t_0 , the superoperator $\mathcal{U}(t, t_0)$ forms a quantum dynamical semigroup [75]. The generator superoperator \mathcal{L} of this semigroup satisfies the partial differential equation

$$\frac{\partial}{\partial t}\mathcal{U}(t,t_0) = \mathcal{L}(t)\mathcal{U}(t,t_0)$$
(1.8)

for $t \ge t_0$. By multiplying both sides by $\hat{\rho}_{\rm S}(t_0)$, this gives rise to the most general form of the Markovian master equation for the reduced system dynamics

$$\frac{d}{dt}\hat{\rho}_{\rm S}(t) = \mathcal{L}(t)\hat{\rho}_{\rm S}(t), \qquad (1.9)$$

where the generator \mathcal{L} is called the Liouville superoperator. Comparing this result to that of Eq. (1.7), one can see that we should have

$$\mathcal{L}(t)\hat{\rho}_{\rm S}(t) = -\frac{i}{\hbar} \mathrm{Tr}_{\rm E} \Big[\hat{H}(t), \hat{\rho}(t)\Big].$$
(1.10)

However, the right-hand-side of this equation does not, in general, allow for a quantum dynamical semigroup. To derive $\mathcal{L}(t)$, it is necessary to apply approximations to satisfy Markovianity [75].

Not all physical systems can be successfully modeled using a Markovian master equation, and there are many interesting situations where Markovian behaviour breaks down [80]. One such case is when a reduced quantum system is manipulated faster than the bath memory time $\tau_{\rm B}$ to decouple it from its environment [71]. However, the phenomenological application of the Markovian master equation allows for computationally tractable solutions that often correctly predict the general behaviour of a device. It also allows for the derivation of simple bounds on figures of merit for devices or protocols, which can help guide experimental design and development.

The standard form of a Markovian master equation that preserves the properties of $\hat{\rho}_{S}$ was derived by Lindblad, Gorini, Kossakowski and Sudarshan [81] and is given by

$$\frac{d}{dt}\hat{\rho}_{\rm S}(t) = -\frac{i}{\hbar}[\hat{H}_{\rm S}(t), \hat{\rho}_{\rm S}] + \sum_{i,j} u_{i,j} \left[\hat{L}_i\hat{\rho}_{\rm S}(t)\hat{L}_j^{\dagger} - \frac{1}{2}\left\{\hat{L}_j^{\dagger}\hat{L}_i, \hat{\rho}_{\rm S}(t)\right\}\right]$$
(1.11)

where the Lindblad operators \hat{L}_i are a unitless set of orthonormal operators acting on the reduced system space, $\{\hat{A}, \hat{B}\} = \hat{A}\hat{B} + \hat{B}\hat{A}$ is the anti-commutator, the coefficients $u_{i,j} \ge 0$ correspond to the rate of system relaxation induced by the environment, and \hat{H}_S is the reduced system Hamiltonian. The exact form of these operators and corresponding coefficients depends on the approach taken to move from the Liouville-von Neumann form (Eq. (1.7)) to the Markovian form (Eq. (1.9)).

One popular approach to achieve Markovianity and derive the Lindblad operators depends on the application of two main assumptions, which together are referred to as the Born-Markov approximations [75]. The Born approximation assumes that the environment is a bath that is weakly coupled to the open quantum system and that the total system state is initially separable $\hat{\rho}(t_0) \simeq \hat{\rho}_{\rm S}(t_0) \otimes \hat{\rho}_{\rm E}(t_0)$. The Markov approximation then assumes the necessary requirement that $\tau_{\rm B} \ll \tau_{\rm S}$, so that the environment quickly returns to its initial equilibrium state. These two approximations reliably produce the effects of spontaneous emission and allow for the derivation of quantum optical master equations, as I will reference in the following section.

There are other approaches to solving open quantum system dynamics that do not require the Born approximation, such as the derivation by a coarse graining of time [79]. This coarse graining approach can also reproduce the same Markovian master equation as the Born-Markov approach for cavity quantum electrodynamics, which is the topic of section 1.3. Another approach that can capture additional dynamics neglected by the Born-Markov approximations is obtained by applying a weak-coupling approximation in a more mathematically rigorous way [80]. This approach provides Lindblad operators and corresponding rate coefficients that are often time dependent to capture the nontrivial interaction dynamics between the reduced system and the bath. Although I do not explore these forms of the Markovian master equation in this thesis, many of the methods I use may be directly applied to more detailed Markovian models.

1.2.5 Multi-time correlations

Using a Markovian master equation, we can compute the state of the reduced system at any time and also compute the expectation values of reduced system operators, which may be directly tied to measured quantities of the environment. The expectation value of an operator \hat{A} in the reduced system space in the Schrödinger picture is given by $\langle \hat{A}(t) \rangle \equiv \text{Tr}(\hat{A}\hat{\rho}(t))$, where I will denote $\hat{\rho}$ as the reduced system density operator moving forward and take the trace to be over the reduced system space unless otherwise specified. Applying the solution to the master equation gives

$$\langle \hat{A}(t) \rangle = \operatorname{Tr} \left(\hat{A} \mathcal{U}(t, t_0) \hat{\rho}(t_0) \right).$$
 (1.12)

It is convenient to define the adjoint propagation superoperator \mathcal{U}^{\dagger} , which is defined as the dynamical map satisfying $\operatorname{Tr}\left(\left(\mathcal{U}^{\dagger}(t,t_0)\hat{A}\right)\hat{\rho}(t_0)\right) = \operatorname{Tr}\left(\hat{A}\mathcal{U}(t,t_0)\hat{\rho}(t_0)\right)$. Then, we can define the operators $\hat{A}_{\mathrm{H}}(t)$ in the Heisenberg picture $\hat{A}_{\mathrm{H}}(t,t_0) = \mathcal{U}^{\dagger}(t,t_0)\hat{A}$. With this notion, it is straightforward to write out multi-time correlation functions in terms of the propagator \mathcal{U} . For example, consider the two-time correlation function:

$$\langle \hat{A}(t')\hat{B}(t)\rangle \equiv \operatorname{Tr}\left(\hat{A}_{\mathrm{H}}(t',t)\hat{B}_{\mathrm{H}}(t,t_{0})\hat{\rho}(t_{0})\right) = \operatorname{Tr}\left(\left(\mathcal{U}^{\dagger}(t',t)\hat{A}\right)\left(\mathcal{U}^{\dagger}(t,t_{0})\hat{B}\right)\hat{\rho}(t_{0})\right)$$

$$= \operatorname{Tr}\left(\hat{A}\mathcal{U}(t',t)\hat{B}\hat{\rho}(t)\right),$$

$$(1.13)$$

where $t' \ge t \ge t_0$. This result can be extended to higher order multi-time correlation functions. For more details on Heisenberg operators and the adjoint space, please again refer to Ref. [75]. For the solution to $\langle \hat{A}(t)\hat{B}(t')\rangle$, we can also use the useful Hermitian property that $\langle \hat{A}(t+\tau)\hat{B}(t)\rangle =$ $\langle \hat{A}(t-\tau)\hat{B}(t)\rangle^*$ where $\tau > 0$. Although I am using the Heisenberg picture to summarize the result above for multi-time correlations, all the derivations and computations in this thesis are performed in the Schrödinger picture. The above result is intuitive because it implies that two-time correlation functions can be computed by propagating the initial state $\hat{\rho}(t_0)$ forward until time t when the first operator \hat{B} is applied, then propagating the result forward again until time t' when the second operator \hat{A} is applied. It also leads to a useful theorem pertaining to the coupled differential equations of two-time correlation functions for Markovian systems called the quantum regression theorem, which I will present in section 1.5.2.

We can also formulate multi-time correlation functions in terms of superoperators, which provides physical intuition and a simple approach to computation in the Fock-Liouville space (to be presented in section 1.5.1). For example, consider the two-time second-order correlation function $G^{(2)}(t,t') = \langle \hat{A}^{\dagger}(t)\hat{A}^{\dagger}(t')\hat{A}(t')\hat{A}(t) \rangle$. Let us then define the jump superoperator \mathcal{J} , which is of the form $\mathcal{J}\hat{\rho} = \hat{A}\hat{\rho}\hat{A}^{\dagger}$ for an operator \hat{A} . Then the correlation is written in a transparent way:

$$G^{(2)}(t,t') = \operatorname{Tr}(\mathcal{JU}(t',t)\mathcal{JU}(t,t_0)\hat{\rho}(t_0)).$$
(1.14)

From this notation it is clear that the system propagates from time t_0 to time t when it jumps by the instantaneous action of \hat{A} . Then it propagates to time t' when it jumps again. We can then interpret $G^{(2)}(t,t')$ as being an unnormalized probability density function of the system jumping once at time t and again at time t'. This superoperator notation can be extended to arbitrary time-ordered multi-time correlation functions [75].

1.3 Quantum electrodynamics

The work presented in this thesis is foremost an application of quantum electrodynamics (QED) to study and propose optical devices and protocols. QED develops from the quantization of the general solution to the classical electromagnetic wave equation [82]. When subject to boundary conditions dictated by an arbitrary mode volume V, the classical free-space field solution to the wave equation is given by a summation of forward-propagating and backward-propagating plane waves of frequency $\omega_{\mathbf{k}}$ with discrete wave vectors \mathbf{k} and polarization $\mathbf{u}_{\epsilon}(\mathbf{k})$, where ϵ is one of two possible polarizations of the plane wave. By quantizing the plane-wave amplitudes of the electric field solution and subjecting them to the canonical commutation relations [83], the quantized electric field in a homogeneous dielectric material is found to be

$$\hat{\mathbf{E}}(\mathbf{r}) = -\sum_{\mathbf{k},\epsilon} \sqrt{\frac{\hbar\omega_{\mathbf{k}}}{2n^2\varepsilon_0 V}} \left(\mathbf{u}_{\epsilon}(\mathbf{k})\hat{a}_{\epsilon}(\mathbf{k})e^{i\mathbf{k}\cdot\mathbf{r}} + \mathbf{u}_{\epsilon}^*(\mathbf{k})\hat{a}_{\epsilon}^{\dagger}(\mathbf{k})e^{-i\mathbf{k}\cdot\mathbf{r}} \right),$$
(1.15)

where **r** is the position in space within V, n is the index of refraction, and ε_0 is the vacuum permitivity. The quantized amplitudes $\hat{a}_{\epsilon}^{\dagger}(\mathbf{k})$ ($\hat{a}_{\epsilon}(\mathbf{k})$) are the photon creation (annihilation) operators that satisfy the canonical commutation relations $[\hat{a}_{\epsilon}(\mathbf{k}), \hat{a}_{\epsilon'}^{\dagger}(\mathbf{k}')] = \delta_{\mathbf{k},\mathbf{k}'}\delta_{\epsilon,\epsilon'}$, where δ is the Kronecker delta function defined by $\delta_{ij} = 1$ if i = j and $\delta_{ij} = 0$ if $i \neq j$. The creation and annihilation operators are ladder operators of a quantum harmonic oscillator. When these operators act on the state $|n\rangle$ of the quantized mode containing n photons, the energy of the state is increased $\hat{a}^{\dagger} |n\rangle = \sqrt{n+1} |n+1\rangle$ or decreased $\hat{a} |n\rangle = \sqrt{n} |n-1\rangle$ by one quantum of energy $\hbar\omega_{\mathbf{k}}$. The creation and annihilation operators are not observables since $\hat{a} \neq \hat{a}^{\dagger}$; however, the electric field operator is an observable since $\hat{\mathbf{E}}^{\dagger} = \hat{\mathbf{E}}$.

Using the quantized electric field, we can obtain the Hamiltonian of the field as a summation of the total quantum harmonic oscillator energy [83] for each mode

$$\hat{H}_E = \sum_{\mathbf{k},\epsilon} \hbar \omega_{\mathbf{k}} \left(\frac{1}{2} + \hat{a}_{\epsilon}^{\dagger}(\mathbf{k}) \hat{a}_{\epsilon}(\mathbf{k}) \right), \qquad (1.16)$$

where often the divergent vacuum energy $\sum_{\mathbf{k}} \hbar \omega_{\mathbf{k}}/2$ is physically inconsequential and ignored by renormalization [75, 83].

If the mode volume is physically confined in such a way to support only a single mode with wavenumber k and polarization $\mathbf{u}(k)$ within V, we obtain the one-dimensional single-mode waveguide field

$$\hat{\mathbf{E}}(x) = -\sum_{k} \sqrt{\frac{\hbar\omega_{k}}{2n^{2}\varepsilon_{0}V}} \left(\mathbf{u}(k)\hat{b}(k)e^{ikx} + \mathbf{u}^{*}(k)\hat{b}^{\dagger}(k)e^{-ikx} \right),$$
(1.17)

where \hat{b}^{\dagger} and \hat{b} are the photon creation and annihilation operators of the waveguide.

1.3.1 Spontaneous emission and pure dephasing

One of the most simple quantum system models with non-trivial dynamics is the two-state system, or two-level system. These two states could represent different electronic or molecular orbitals of a defect, two different spin states, or any two substates that are decoupled from a larger Hilbert space.

A two-level dipole emitter is a two-level system of different electronic states with a ground state $|g\rangle$ and an excited state $|e\rangle$ separated by an average energy $\hbar\omega_o$. The Hamiltonian is given by $\hat{H}_o = \hbar\omega_o \hat{\sigma}^{\dagger} \hat{\sigma}$ where $\hat{\sigma}^{\dagger} = |e\rangle\langle g|$ is the raising operator and $\hat{\sigma} = |g\rangle\langle e|$ is the lowering operator. The defining characteristic of a two-level dipole emitter is that it has a non-zero electric transition dipole moment between the two levels. This can arise if the electronic orbitals have different symmetry so that there is a shift in charge distribution when moving from one state to another. The dipole operator is a vector operator $\hat{\mathbf{d}} = q\hat{\mathbf{r}}$ analogous to the classical dipole moment, where q is the electric charge and $\hat{\mathbf{r}}$ is the position operator. The transition dipole moment is then $\langle g|q\hat{\mathbf{r}}|e\rangle$, which is nonzero for an emitter.

Under the dipole approximation, the interaction Hamiltonian for an emitter at position \mathbf{r}_o in an electric field is given by $-\hat{\mathbf{d}} \cdot \hat{\mathbf{E}}(\mathbf{r}_o)$ [83]. Using the quantized electric field from the previous section, this gives us the total emitter-field Hamiltonian

$$\hat{H} = \hbar\omega_o \hat{\sigma}^{\dagger} \hat{\sigma} + \sum_{\mathbf{k},\epsilon} \hbar\omega_{\mathbf{k}} \hat{a}^{\dagger}_{\epsilon}(\mathbf{k}) \hat{a}_{\epsilon}(\mathbf{k}) + \sum_{\mathbf{k},\epsilon} \sqrt{\frac{\hbar\omega_{\mathbf{k}}}{2n^2\varepsilon_0 V}} \hat{\mathbf{d}} \cdot \left(\mathbf{u}_{\epsilon}(\mathbf{k}) \hat{a}_{\epsilon}(\mathbf{k}) e^{i\mathbf{k}\cdot\mathbf{r}_o} + \mathbf{u}^*_{\epsilon}(\mathbf{k}) \hat{a}^{\dagger}_{\epsilon}(\mathbf{k}) e^{-i\mathbf{k}\cdot\mathbf{r}_o}\right).$$
(1.18)

To obtain a Markovian master equation for the dynamics of the two-level emitter, we can follow Ref. [75] by applying the Born-Markov approximations to the Liouville-von Neumann equation corresponding to Eq. (1.18). If we also assume that the electric field is a bath with a temperature T, then we arrive at the quantum optical master equation for the reduced system $d\hat{\rho}/dt = \mathcal{L}\hat{\rho}$ given by the Liouville superoperator [75]

$$\mathcal{L} = -\frac{i}{\hbar}\mathcal{H} + \gamma^{+}\mathcal{D}(\hat{\sigma}^{\dagger}) + \gamma^{-}\mathcal{D}(\hat{\sigma}), \qquad (1.19)$$

where \mathcal{H} is the Liouville-von Neumann superoperator defined by $\mathcal{H}\hat{\rho} = [\hat{H}_{\rm S}, \hat{\rho}]$ corresponding to the reduced system Hamiltonian $\hat{H}_{\rm S} = \hbar\omega_o \hat{\sigma}^{\dagger}\hat{\sigma}$. The dissipation superoperator \mathcal{D} is given by $\mathcal{D} = \mathcal{J} - \mathcal{A}/2$, where \mathcal{J} is the jump superoperator defined by $\mathcal{J}(\hat{A})\hat{\rho} = \hat{A}\hat{\rho}\hat{A}^{\dagger}$ and \mathcal{A} is the amplitude damping superoperator defined by $\mathcal{A}(\hat{A})\hat{\rho} = \{\hat{A}^{\dagger}\hat{A}, \hat{\rho}\}$. The temperature-dependent rates associated with the dissipation superoperators are $\gamma^+ = \gamma_{\rm r}\overline{n}(T)$ and $\gamma^- = \gamma_{\rm r}(1 + \overline{n}(T))$, where

$$\gamma_{\rm r} = \gamma^- - \gamma^+ = \frac{n\omega_o^3 d^2}{3\pi\varepsilon_0 \hbar c^3},$$
 (1.20)

is the zero-temperature radiative rate [83], $d = |\mathbf{d}|$ is the dipole magnitude, $\overline{n}(T) = (e^{\hbar\omega_o/k_{\rm B}T} - 1)^{-1}$ is the average number of photons in the mode of the field that is resonant with the two-level system, and $k_{\rm B}$ is the Boltzmann constant. The reduced system resonance frequency ω_o is slightly different from the original frequency due to the Lamb shift induced by the vacuum fluctuations of the electromagnetic field and a possible thermal Stark shift if $\overline{n}(T) \neq 0$. However, since this only amounts to a shift in the zero-point energy, I will continue to write ω_o without specification.

The result of spontaneous emission is profound because it shows that an excited quantum dipole emitter will dissipate its energy into the environment by spontaneously emitting a single photon of frequency $\hbar\omega_o$ at the rate γ_r even if the electromagnetic field is in the vacuum state (T = 0). This passive coupling to the electromagnetic vacuum causes the emitter to experience decoherence. To see this, consider an arbitrary initial state of the two-level system $|\psi(0)\rangle = \alpha |g\rangle + \beta |e\rangle$, where α and β are complex amplitudes satisfying $|\alpha|^2 + |\beta|^2 = 1$. At low temperature where $\overline{n}(T) \simeq 0$, the Liouville superoperator in the frame rotating at the emitter resonance ω_o is $\mathcal{L} = \gamma_r \mathcal{D}(\hat{\sigma}^{\dagger})$ corresponding to the propagator $\mathcal{U}(t, t_0) = e^{\mathcal{L}(t-t_0)}$. Then the state of the reduced system at time tis computed by $\hat{\rho}(t) = \mathcal{U}(t, 0) |\psi(0)\rangle\langle\psi(0)|$ to give

$$\hat{\rho}(t) = \left(|\alpha|^2 + |\beta|^2 (1 - e^{-\gamma_{\rm r} t})\right) |g\rangle \langle g| + |\beta|^2 e^{-\gamma_{\rm r} t} |e\rangle \langle e| + (\alpha^* \beta |e\rangle \langle g| + \alpha \beta^* |g\rangle \langle e|) e^{-\gamma_{\rm r} t/2}.$$
(1.21)

From Eq. (1.21) for $\hat{\rho}(t)$, we can see that the coherence $\langle e|\hat{\rho}|g\rangle$ is degraded at the rate $\gamma_r/2$ while the excited state population decays at the rate γ_r . Another interesting observation is that the trace purity $\mathcal{P} = \text{Tr}(\hat{\rho}^2(t))$ of the state is degraded during emission, meaning that the coherence is not always saturated to the upper bound dictated by the populations. For any initial pure state $|\psi(0)\rangle$ of the emitter, the trace purity begins at 1 and reaches a minimum value of $\min(\mathcal{P}) = |\alpha|^4 + 2|\alpha|^2|\beta|^2 + |\beta|^4/2$ at the time $t_{\min} = \ln(2)/\gamma_r$ before increasing back to unity as the emitter settles to its ground state $|g\rangle$. This purity dip is related to the fact that the emitter briefly becomes entangled with the electromagnetic environment during spontaneous emission, which is the topic of section 3.3.3.

In some systems, especially those in a solid state environment, the total dissipative rate of decay γ can be larger than the radiative rate γ_r . An increased decay rate can be caused by non-radiative decay pathways that can emit phonons instead of photons. This additional non-radiative decay rate γ_{nr} can be accounted for in a phenomenological way by replacing γ_r with $\gamma = \gamma_r + \gamma_{nr}$.

In addition to an increased dissipative rate, solid-state environments can cause the rate of decoherence to be even larger than the $\gamma/2$ predicted by the total dissipation alone. This can occur when environmental fluctuations, such as charge noise, cause the emitter resonance ω_o to rapidly fluctuate in time on a timescale faster than its decay rate. Alternatively, this dephasing can be seen

as random elastic scattering of particles off of the emitter, such as phonons. These particles can carry information about the emitter state into the environment. From either perspective, the result is that the phase of emitter superposition states becomes dependent on (or entangled with) the state of the environment. Looking in the reduced state picture, this emitter-environment entanglement manifests as random phase fluctuations, which on average decrease the emitter coherence. This dephasing of the emitter coherence is referred to as *pure emitter dephasing*.

For some emitters, such as quantum dots, pure dephasing due to phonons can be modeled by rigorously including electron interactions with a continuum of longitudinal acoustic phonon modes [84, 85]. For example, this can be done using a polaron transformation and a subsequent Markovian approximation to obtain a master equation [84]. However, the exact quantum processes that account for all pure dephasing phenomena for any given solid-state system are often unidentified or very complicated to rigorously include. Regardless of its origin, we can still account for pure dephasing in a less-accurate phenomenological way [86] by considering the effect of the superoperator $\mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma})$. If we consider just the dynamics induced by the Liouville superoperator $\mathcal{L} = 2\gamma^{\star} \mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma})$, we can see that for an initial state $|\psi(0)\rangle$ as above, we get the solution $\hat{\rho}(t) = 0$ $|\alpha|^{2} |g\rangle \langle g| + |\beta|^{2} |e\rangle \langle e| + e^{-t\gamma^{\star}} (\alpha \beta^{\star} |g\rangle \langle e| + \alpha^{\star}\beta |e\rangle \langle g|).$ Hence, the coherence $\langle g|\hat{\rho}|e\rangle$ will degrade exponentially at an additional rate γ^* while leaving the population untouched. By including the additional term $2\gamma^* \mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma})$ along with $\gamma \mathcal{D}(\hat{\sigma})$, the total decoherence rate of the emitter becomes $\Gamma/2$ where $\Gamma = \gamma + 2\gamma^{\star}$. The rate Γ is also the spectrally-broadened full width at half maximum (FWHM) of the Lorentzian emission line of the fluctuating emitter. Note that there are two conventions for defining the pure dephasing rate γ^* that differ by a factor of two: (1) as decay rate of the *amplitude* of the coherence such that the FWHM is $\Gamma = \gamma + 2\gamma^{\star}$ and (2) as the decay rate of the squared magnitude of the coherence such that the FWHM is $\Gamma = \gamma + \gamma^{\star}$. This thesis uses the former convention with the exception of section 3.1 where the latter convention is used.

In the literature, the total decay rate γ , pure dephasing rate γ^* , and emitter linewidth Γ are often expressed as a decay time rather than a rate, similar to the T_1 and T_2 notation in the field of nuclear magnetic resonance [87]. Drawing from the notation convention used in Ref. [86] for optical systems, and being careful to note the factor of 2 difference in choice of definition for γ^* , we have that $T_1 = 1/\gamma$, $T_2^* = 1/\gamma^*$, and $T_2 = 2/\Gamma$. This implies that $1/T_2 = 2/T_1 + 1/T_2^*$ and $2T_2 \leq T_1$. In addition, it is common to specify the radiative component of the decay time $T_{1,\text{rad}} = 1/\gamma_r$. When appropriate, e.g. for spin qubits, we can also distinguish between spontaneous decay $T_1^- = 1/\gamma^-$ and incoherent (thermal) excitation $T_1^+ = 1/\gamma^+$ for a model utilizing the general form of Eq. (1.19).

The assumption that the coupling between the emitter and the bath is weak is implicit in the Born-Markov approximations used to derive spontaneous decay and thermal excitation. If V is small enough for a particular mode that is resonant with the two-level system, it is possible to violate the Born-Markov weak-coupling condition. This scenario can be described by the Jaynes-Cummings Hamiltonian, which is the topic of the next section.

1.3.2 Jaynes-Cummings Hamiltonian

Suppose that we confine the electromagnetic field to one dimension so that the modes along this direction are described as in Eq. (1.17). However, now let us assume that the modes are highly confined so that the quantized mode resonances are far separated in frequency (ω_k are far apart). This can occur if, for example, V is on the order of the cubic wavelength of the mode resonance. This confinement of an electromagnetic mode is called a cavity.

If a two-level emitter with oscillation frequency ω_o is placed inside this cavity, it will now primarily couple to the cavity mode resonance ω_k that is closest to ω_o , which I call ω_c . For simplicity, let us assume that the dipole is oriented parallel to the cavity mode polarization direction and that it is placed at the antinode of the electric field so that the coupling is maximized. Under these conditions, the cavity-emitter dipole interaction can be much stronger than the coupling between the emitter and the rest of the electromagnetic bath. This violates the weak-coupling assumption used in the previous section. Considering only this single cavity mode-emitter interaction, in the dipole gauge [88], we can write the quantum Rabi model Hamiltonian

$$\hat{H} = \hbar\omega_o \hat{\sigma}^{\dagger} \hat{\sigma} + \hbar\omega_c \hat{a}^{\dagger} \hat{a} + \hbar g (\hat{\sigma} + \hat{\sigma}^{\dagger}) (\hat{a} + \hat{a}^{\dagger})$$
(1.22)

where \hat{a} (\hat{a}^{\dagger}) is the cavity mode photon annihilation (creation) operator and

$$\hbar g = d \sqrt{\frac{\hbar \omega_{\rm c}}{2n^2 \varepsilon_0 V}} \tag{1.23}$$

is the vacuum cavity-emitter coupling rate and d is the dipole magnitude of the emitter. Here, I am choosing a phase convention where g is a positive real-valued rate. If the dipole is not oriented perfectly or placed at an antinode, the magnitude of the coupling coefficient g will be reduced from its maximum value.

For optical systems, the cavity-emitter coupling rate g is usually on the order of GHz at most. This is much smaller than $\omega_o \simeq \omega_c$, which generally both fall into the THz regime. Thus, this thesis deals exclusively with interaction strengths where the quantum Rabi model satisfies gauge invariance [88]. As an additional consequence, the eigenenergies of the system are only slightly perturbed in magnitude and are on the order of $\omega_o \pm g \simeq \omega_o$. This allows us to make an important simplification called the rotating wave approximation [83, 89]. To easily see how this approximation can be implemented, we must first move into the interaction picture [75] by transforming \hat{H} by the unitary transformation $\hat{U} = e^{-it\hat{H}_0/\hbar}$ given by the homogeneous part of the Hamiltonian $\hat{H}_0 = \hbar \omega_o \hat{\sigma}^{\dagger} \hat{\sigma} + \hbar \omega_c \hat{a}^{\dagger} \hat{a}$. In this interaction picture, we have $\hat{H}_I = \hat{U}^{\dagger} \hat{H} \hat{U} - \hat{H}_0$, which gives

$$\hat{H}_I = \hbar g \left(\hat{\sigma} \hat{a} e^{it(-\omega_o - \omega_c)} + \hat{\sigma}^{\dagger} \hat{a}^{\dagger} e^{it(\omega_o + \omega_c)} + \hat{\sigma} \hat{a}^{\dagger} e^{it(-\omega_o + \omega_c)} + \hat{\sigma}^{\dagger} \hat{a} e^{it(\omega_o - \omega_c)} \right).$$
(1.24)

The first two terms show a phase rotation at the rates $\pm(\omega_o + \omega_c)$ while the second two terms show a phase rotation at the rates $\pm(\omega_o - \omega_c)$. Hence, the first two terms will only contribute to the dynamics of the system if the coupling rate g is on the order of or bigger than the sum of ω_o and ω_c whereas the second two terms simply require g to be on the order of or bigger than the difference $\omega_o - \omega_c$. For optical systems, the former case is very difficult to achieve (see more on this in section 1.3.5). Thus, we can eliminate the first two terms for having a much faster phase rotation.

By using the rotating wave approximation to eliminate the terms with a fast phase rotation, and transforming back into the original picture, we arrive at the Jaynes-Cummings Hamiltonian for a cavity-emitter system

$$\hat{H} = \hbar\omega_o \hat{\sigma}^{\dagger} \hat{\sigma} + \hbar\omega_c \hat{a}^{\dagger} \hat{a} + \hbar g (\hat{\sigma} \hat{a}^{\dagger} + \hat{\sigma}^{\dagger} \hat{a}).$$
(1.25)

The physical interpretation of this Hamiltonian is clear. The emitter and the cavity mode both evolve with their respective homogeneous frequencies of ω_o and ω_c . The interaction term coherently converts a single excitation of the emitter into a single photon in the cavity or vice versa at the cavity-emitter coupling rate g.

The Jaynes-Cummings Hamiltonian has been very successful at explaining the quantum coherent interaction between an emitter and a cavity mode in the optical regime. However, its derivation makes a simplistic assumption about the spectrum of the cavity resonance that may be violated in some systems. Furthermore, the quantization of the electromagnetic field used here assumes that the entire field is confined perfectly to the mode volume V. Of course, this is never true because light inevitably leaks out of the cavity or becomes absorbed by the material forming the cavity. In section 1.3.7, I will briefly introduce a more general approach that can account for both of these failings. However, in the sections preceding that discussion, I will introduce a simpler and more common approach that only accounts for the lossy nature of a nearly-ideal single-mode cavity with defined boundaries.

1.3.3 Input-output theory

The cavity mode that gives rise to the Jaynes-Cummings interaction discussed in the previous section can inevitably lose energy into the environment. This cavity disspation can be modeled by considering that the electromagnetic field of the cavity mode must satisfy certain electromagnetic

boundary conditions at the edges of the cavity mode volume V. This gives rise to an additional coupling between the confined cavity mode and the continuum of unconfined modes surrounding the cavity. These boundary conditions are reflected by the input-output relations derived by Gardiner and Collett [90].

The input-output relation between the confined mode \hat{a} and the remaining 1-dimensional continuum of modes (or waveguide mode) is, $\hat{b}(t) - \hat{b}_o(t) = \sqrt{\kappa \eta_c} \hat{a}(t)$, where \hat{b} is the forward propagating waveguide mode and \hat{b}_o is the reverse-propagating input waveguide mode. This holds in the limit that the cavity has a very low dissipative rate, or that it is highly reflective. For optical light, we can assume that the input waveguide mode is in the vacuum state since the thermal occupation $\overline{n}(T)$ is very small for frequencies on the order of THz, even at room temperature. The collection efficiency η_c can be introduced by assuming that the cavity has two possible decay channels, one into the waveguide at the rate κ_c and another $\kappa - \kappa_c$ due to emission into other propagating modes or absorption losses. Then $\eta_c = \kappa_c/\kappa$ is the collection efficiency of the waveguide. We can further take into account the propagation phases of these fields to yield $\hat{b}(t)e^{ikx} = \sqrt{\kappa \eta_c}\hat{a}(t) + \hat{b}_o(t)e^{-ikx}$.

This thesis deals exclusively with photon counting measurements of one or more waveguide modes. I will not explore measurements where it is necessary to consider the direct impact of vacuum fluctuations, such as quadrature measurements. As a result, I quite often write the direct equivalence between system operators and waveguide operators, such as $\hat{b}(t) = \sqrt{\kappa \eta_c} \hat{a}(t) e^{-ikx}$, when computing quantities related to \hat{b} . This is common practice in the literature [91–93]. However, this proportionality should be seen as a short-hand notation because it violates the necessary commutation relations for the propagating mode \hat{b} .

1.3.4 Dissipative cavity quantum electrodynamics

By accounting for the input-output coupling of the cavity mode and the waveguide, and subsequently tracing out the waveguide states, we arrive at the master equation for the reduced cavity-emitter system given by the Liouville superoperator $\mathcal{L} = -i\mathcal{H}/\hbar + \gamma \mathcal{D}(\hat{\sigma}) + \kappa \mathcal{D}(\hat{a})$, where the cavity-

emitter Hamiltonian superoperator $\mathcal{H}\hat{\rho} = [\hat{H}, \hat{\rho}]$ is given by Eq. (1.25) [75]. Here, I have taken the temperatures of both environment baths of the emitter and cavity to be such that $\hbar\omega \gg k_{\rm B}T$ so that only spontaneous emission remains.

One last important ingredient that we need in order to arrive at a functional device model is the ability to externally control our two-level system within the cavity. This can be done by illuminating the two-level system with coherent laser pulses. If these pulses occupy electromagnetic modes that do not couple to the previously-described cavity mode or the collection waveguide mode, we can make a semi-classical approximation for the dipole interaction Hamiltonian and append it to our Jaynes-Cummings Hamiltonian with few negative consequences [75, 91]. In the semi-classical approximation of the dipole interaction $-\hat{\mathbf{d}} \cdot \mathbf{E}$, the electromagnetic field is taken to be classical while the dipole remains quantized. This gives the semi-classical driving Hamiltonian in the rotating wave approximation

$$\hat{H}_D = -\hat{\mathbf{d}} \cdot \mathbf{E}(t) \simeq \frac{\hbar}{2} \left(\Omega^*(t) \hat{\sigma} e^{i\omega_{\rm L}t} + \Omega(t) \hat{\sigma}^{\dagger} e^{-i\omega_{\rm L}t} \right), \qquad (1.26)$$

where $\Omega(t)$ is the Rabi frequency induced by the coherent driving of frequency $\omega_{\rm L}$. This direct emitter driving scenario occurs in practice using a cross-polarization setup where excitation and emission are orthogonally polarized [68][vi]. In addition, it is possible that the semi-classical drive is applied to the emitter via a cavity mode orthogonal to \hat{a} , which can increase Ω and may also distort pulses that are spectrally broad compared to the cavity linewidth.

Taking into account the semi-classical driving Hamiltonian and the phenomenological pure dephasing term introduced in section 1.3.1, we arrive at the driven dissipative cavity QED Markovian master equation model used in this thesis. This master equation is described by the Liouville superoperator

$$\mathcal{L}(t) = -\frac{i}{\hbar}\mathcal{H}(t) + \kappa \mathcal{D}(\hat{a}) + 2\gamma^* \mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma}) + \gamma \mathcal{D}(\hat{\sigma}), \qquad (1.27)$$

where

$$\hat{H}(t) = \hbar \Delta_o \hat{\sigma}^{\dagger} \hat{\sigma} + \hbar \Delta_c \hat{a}^{\dagger} \hat{a} + \hbar g \left(\hat{a}^{\dagger} \hat{\sigma} + \hat{a} \hat{\sigma}^{\dagger} \right) + \frac{\hbar}{2} \left(\Omega^*(t) \hat{\sigma} + \Omega(t) \hat{\sigma}^{\dagger} \right),$$
(1.28)



Figure 1.2: The emitter-cavity system coupled to a one-dimensional collection waveguide. The emitter with optical frequency ω_o , represented by the lowering operator $\hat{\sigma} = |g\rangle\langle e|$, experiences pure dephasing at the rate γ^* and decays from its excited state $|e\rangle$ to its ground state $|g\rangle$ at the rate γ . Part of the dissipated energy is radiated into the environment at the rate γ_r . The cavity mode with frequency ω_c , represented by the bosonic annihilation operator \hat{a} where $\hat{a} |n\rangle = \sqrt{n} |n-1\rangle$, decays exponentially into the environment at the rate κ . The interaction between the cavity mode and the optical transition of the emitter is described by the Jaynes-Cummings interaction with a coupling rate of g. The input-output relation describes the propagating mode \hat{b} of the one-dimensional waveguide mode, which depends also on the collection efficiency η_c , position along the waveguide x, and the wavenumber k. The vacuum input mode \hat{b}_o of the collection waveguide must be included to satisfy the commutation relations, although it does not contribute to intensity measurements of the waveguide mode.

and with a collection waveguide mode given by $\hat{b}(t) = \sqrt{\kappa \eta_c} \hat{a}(t) e^{ikx}$. In writing the Hamiltonian, I have moved into the rotating frame defined by the driving laser frequency ω_L to remove the explicit time-dependent phase. This is done by applying the unitary transformation $\hat{U} = e^{-i\omega_L t(\hat{\sigma}^{\dagger}\hat{\sigma} + \hat{a}^{\dagger}\hat{a})}$ to the original Hamiltonian. Hence, $\Delta_o = \omega_o - \omega_L$ is the detuning between the emitter resonance and the driving frequency and $\Delta_c = \omega_c - \omega_L$ is the detuning between the cavity resonance and the driving frequency. Fig. 1.2 shows a summary diagram of this quantum model for $\Omega(t) = 0$ and emitter-cavity detuning $\Delta = \omega_o - \omega_c$.

In many cases studied in this thesis, I assume that the emitter is initially perfectly prepared in the excited state $|e\rangle$. For the cavity QED model above, this can be achieved by applying a square π pulse with non-zero amplitude $\Omega = \pi/t_p$ during a duration of time t_p . Then, if t_p is much smaller than the timescale of the remaining system dynamics (so that the pulse is very fast), the initial emitter state $|g\rangle$ is excited perfectly to $|e\rangle$ after t_p . Although this perfect π pulse excitation is achievable using the Markovian master equation model, in practice there are limits to the regimes where it is valid. In particular, phonon damping of the solid-state emitter will prevent arbitrarily perfect excitation and the Markovian approximation breaks down when t_p is reduced to be on the order of the phonon bath memory time [67]. In addition, emission can occur during the excitation pulse if t_p is not much faster than the total emitter decay rate. This may cause re-excitation to occur, leading to multi-photon emission. These additional details generally hinder the perfect operation of a device. Hence, in the spirit of this thesis, the perfect excitation approximation serves to derive general bounds on figures of merit and explore the general regimes of operation for a device. That said, I do also extensively explore multi-photon effects and re-excitation processes in section 3.2.

To get insight into the emission dynamics in the idealized scenario where the emitter is perfectly excited, we can derive a set of differential equations known as the optical Bloch equations for when $\Omega = 0$. These describe the time dynamics of the expectation values of the operators $\hat{\sigma}$ and \hat{a} . In the Schrödinger picture, we have $d\langle \hat{A}(t) \rangle/dt = \text{Tr}(\hat{A}\mathcal{L}\hat{\rho}(t))$ for an operator \hat{A} in the Hilbert space. Applying this to $\hat{\sigma}$ and \hat{a} , where for convenience I will drop the explicit dependence on time, gives

$$\frac{d}{dt} \langle \hat{\sigma} \rangle = -i\Delta \langle \hat{\sigma} \rangle + ig \langle \hat{a}\hat{\sigma}_z \rangle - \frac{\Gamma}{2} \langle \hat{\sigma} \rangle$$

$$\frac{d}{dt} \langle \hat{a} \rangle = -ig \langle \hat{\sigma} \rangle - \frac{\kappa}{2} \langle \hat{a} \rangle,$$
(1.29)

where I have set $\Delta_c = 0$ so that the cavity frequency ω_c defines the reference frame, $\Gamma = \gamma + 2\gamma^*$ is the FWHM of the homogeneously broadened spectrum of the emitter. Recall from section 1.3.1 that the quantity $\Gamma/2$ here describes the total decoherence of the emitter two-level system. The correlation function $\langle \hat{a}\hat{\sigma}_z \rangle$ couples these two equations to an infinite series of equations of motion for higher-order correlation functions. However, if the system begins with at most one excitation, either in the emitter or cavity, then the state subspace accessible to the system reduces to $\{|g, 0\rangle, |g, 1\rangle, |e, 0\rangle\}$. In this subspace, the operator $\hat{a}\hat{\sigma}_z$ vanishes when acting on all states except $|g, 1\rangle$ where it returns $-|g, 0\rangle$. Hence, in this single-excitation approximation we have $\hat{a}\hat{\sigma}_z = -\hat{a}$, which decouples the above coupled differential equations from the higher-order correlations.

Using the master equation, we can also derive the equations of motion for the populations $\langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle$ and $\langle \hat{a}^{\dagger} \hat{a} \rangle$ as well as correlations $\langle \hat{a}^{\dagger} \hat{\sigma} \rangle$ and $\langle \hat{a} \hat{\sigma}^{\dagger} \rangle$ in the single-excitation approximation, which are

$$\frac{d}{dt} \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle = -ig \langle \hat{a} \hat{\sigma}^{\dagger} \rangle + ig \langle \hat{a}^{\dagger} \hat{\sigma} \rangle - \gamma \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle$$

$$\frac{d}{dt} \langle \hat{a} \hat{\sigma}^{\dagger} \rangle = i\Delta \langle \hat{a} \hat{\sigma}^{\dagger} \rangle - ig \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle + ig \langle \hat{a}^{\dagger} \hat{a} \rangle - \frac{1}{2} (\kappa + \Gamma) \langle \hat{a} \hat{\sigma}^{\dagger} \rangle$$

$$\frac{d}{dt} \langle \hat{a}^{\dagger} \hat{\sigma} \rangle = -i\Delta \langle \hat{a}^{\dagger} \hat{\sigma} \rangle + ig \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle - ig \langle \hat{a}^{\dagger} \hat{a} \rangle - \frac{1}{2} (\kappa + \Gamma) \langle \hat{a}^{\dagger} \hat{\sigma} \rangle$$

$$\frac{d}{dt} \langle \hat{a}^{\dagger} \hat{a} \rangle = ig \langle \hat{a} \hat{\sigma}^{\dagger} \rangle - ig \langle \hat{a}^{\dagger} \hat{\sigma} \rangle - \kappa \langle \hat{a}^{\dagger} \hat{a} \rangle.$$
(1.30)

If the emitter-cavity correlations decay very quickly compared to the coupling rate so that $2g < \kappa + \Gamma$, then the emitter-cavity system is said to be in the weak-coupling regime. I will present the different regimes of cavity QED in more detail in the following section. In the weak-coupling regime and when the initial system state does not give cavity-emitter correlations, the values of $\langle \hat{a}\hat{\sigma}^{\dagger} \rangle$ and $\langle \hat{a}^{\dagger}\hat{\sigma} \rangle$ can be adiabatically eliminated [94]. This is done by setting their time derivatives to zero so that the correlations depend on time implicitly through the populations $\langle \hat{\sigma}^{\dagger}\hat{\sigma} \rangle$ and $\langle \hat{a}^{\dagger}\hat{a} \rangle$. Solving the resulting linear system of equations for the correlations gives

$$\left\langle \hat{a}\hat{\sigma}^{\dagger}\right\rangle^{*} = \left\langle \hat{a}^{\dagger}\hat{\sigma}\right\rangle = \frac{2ig}{\Gamma + \kappa + 2i\Delta} \left(\left\langle \hat{\sigma}^{\dagger}\hat{\sigma}\right\rangle - \left\langle \hat{a}^{\dagger}\hat{a}\right\rangle\right).$$
(1.31)

Substituting this solution into the set of differential equations results in

$$\frac{d}{dt} \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle = -(\gamma + R) \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle + R \langle \hat{a}^{\dagger} \hat{a} \rangle$$

$$\frac{d}{dt} \langle \hat{a}^{\dagger} \hat{a} \rangle = -(\kappa + R) \langle \hat{a}^{\dagger} \hat{a} \rangle + R \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle,$$
(1.32)

where

$$R = \frac{4g^2(\kappa + \Gamma)}{(\kappa + \Gamma)^2 + 4\Delta^2} \tag{1.33}$$

is the effective rate of population transfer between the emitter and the cavity. This rate is maximized under the resonant condition $\Delta = 0$, giving $R = 4g^2/(\kappa + \Gamma)$. Although this quantity R only accurately approximates the rate of emitter-cavity population transfer when $2g < \kappa + \Gamma$, it is still a very useful quantity to discuss and explain various emitter-cavity behaviours. For this reason, the quantity R will be discussed even when $\kappa + \Gamma \ge 2g$. Furthermore, from Eq. (1.32), we can see that the effective decay rate of the two-level system becomes $\gamma + R$. This decay rate enhancement is called the Purcell effect [95], and it is quantified by the unitless quantity known as the Purcell factor.

The Purcell factor F_p is defined as the ratio of the cavity-enhanced radiative rate emitted through the cavity mode to the bare radiative rate γ_r of the emitter. In the weak-coupling regime [94, 96], this cavity rate is R and so

$$F_{\rm p} = \frac{R}{\gamma_{\rm r}} = \frac{4g^2(\kappa + \Gamma)}{\gamma_{\rm r}(\kappa + \Gamma)^2 + 4\Delta^2}.$$
(1.34)

Recall from section 1.3.2 that the cavity coupling rate can be written as $g = d\sqrt{\omega_c/2n^2\varepsilon_0\hbar V}$. In addition, in section 1.3.1, we found that $\gamma_r = n\omega_o^3 d^2/3\pi\varepsilon_0\hbar c^3$. Substituting these into F_p , we have

$$F_{\rm p} = \frac{4g^2}{\gamma_{\rm r}\kappa} F_{\rm inh} = \left(\frac{2\omega_{\rm c}d^2}{n^2\varepsilon_0\hbar V}\right) \left(\frac{3\pi\varepsilon_o\hbar c^3}{n\omega^3 d^2\kappa}\right) F_{\rm inh} = \frac{3}{4\pi^2} \left(\frac{\lambda}{n}\right)^3 \left(\frac{Q}{V}\right) F_{\rm inh},\tag{1.35}$$

where $Q = \omega_c/\kappa$ is the cavity quality factor and $\lambda = 2\pi c/\omega_o$ is the emission wavelength in the vacuum. The inhibition factor due to detuning and emitter dephasing is

$$F_{\rm inh} = \frac{\kappa(\kappa + \Gamma)}{(\kappa + \Gamma)^2 + 4\Delta^2}.$$
(1.36)

Note that I am still assuming that the dipole is oriented and positioned to maximize the cavityemitter coupling rate. The more general expression would have an additional inhibition factor F_{geom} that depends on the geometry of the emitter-cavity system. The above result is valid for a phenomenological pure dephasing model. It is also possible to derive Purcell inhibition factors that rigorously include the electron-phonon interactions giving rise to emitter dephasing, which consequently affect the cavity-emitter interaction [97]. These more detailed approaches are highly applicable to quantum dot systems, especially those that operate at warmer temperatures. The expression given by Eq. (1.35) illustrates that the Purcell factor is proportional to the ratio of the cavity quality factor Q to the cavity mode volume V. Thus, the decay rate enhancement experienced by the emitter is larger if the cavity is smaller for a given Q or if the cavity dissipates energy into the environment more slowly for a given V. However, the validity of the Purcell factor is restricted to the weak-coupling regime of cavity QED. If a cavity approaches the bounds of this regime, the Purcell factor expression derived above can over-estimate the decay rate enhancement. In section 3.1.5, I will generalize the Purcell factor to include this additional inhibition effect induced by strong cavity coupling. In the following section, I will expand on the different regimes of cavity QED.

1.3.5 Regimes of cavity quantum electrodynamics

Depending on the relative magnitudes of the optical emitter-cavity QED system parameters, the dynamics of their interaction can be categorized into various regimes that display particular properties. The names of these regimes vary in the literature and so for clarity I will define my choice of vernacular following the convention taken in Ref. [96], which is somewhat followed in Ref. [86] as well.

In this section, I will assume that the emitter is resonant with the cavity so that $\Delta = 0$. I will also not consider regimes related to ultra-strong or deep-strong coupling [98, 99], where the cavity coupling g is on the order of or greater than the frequency of the emitter ω_o , which invalidates approximations made to obtain the Jaynes-Cummings model. This requires g to be orders of magnitude larger than what has been demonstrated for a single optical emitter. However, effects of this regime can be observed in microwave systems where ω_o is much smaller or for ensembles of emitters where g is effectively enhanced to $g\sqrt{N_o}$ for N_o emitters and may one day be reached for a single optical emitter.

The cavity system can be first divided into the strong $(2g \ge \kappa + \Gamma)$ and weak $(2g < \kappa + \Gamma)$ coupling regimes (see Fig. 1.3 (a)). Note that it is equivalent to define them as $2g \le R$ for



Figure 1.3: Log-scale illustrations of emitter-cavity QED regimes as a function of the cavity decay rate κ and cavity coupling rate g, inspired by plots from Ref. [86]. (a) The division of the strong-coupling (blue) and weak-coupling (red) regimes by the curve R = 2g drawn in black, where $R = 4g^2/(\kappa + \Gamma)$, $\Gamma = \gamma + 2\gamma^*$, γ is the bare emitter decay rate, and γ^* is the emitter pure dephasing rate. (b) The division of the good-cavity (purple) and bad-cavity (orange) regimes separated by the line $R = \kappa$. The hatched region indicates the Purcell regime $(R/\gamma \ge 1)$ where the cavity can greatly enhance the emitter decay rate. Note that the entire region in the top right corner where κ , $R > \gamma$ (indicated by the dashed lines) can also allow for a Purcell enhancement, but the spectrum will show a vacuum Rabi splitting when 2g > R and then the standard Purcell factor is invalid. (c) The critical coupling (green) and incoherent-pumping (cyan) regimes that allow for spectrally/temporally pure emission from the cavity mode. The curve labeled by $R = \kappa$ shows the Goldilocks condition, which extends into both the green and cyan regions (red dashed lines). The faded gray hatched area shows the Purcell regime from panel (b) and the red hatched area shows the overlap between the critical regime and the Purcell regime. Panels (b) and (c) are drawn under the assumption that $\gamma \ll 2\gamma^*$ so that $\Gamma \simeq 2\gamma^*$.

strong coupling and 2g > R for weak coupling. The strong-coupling regime is attained when the cavity coupling rate exceeds the decay rate of the cavity-emitter correlations, causing vacuum Rabi oscillations. This is also characterized by a visible vacuum Rabi splitting in the emission spectrum. The spectrum provided in the weak-coupling regime remains Lorentzian in shape, albeit with a width and quantum characteristics that further depend on the parameters.

The cavity system is said to be in the good-cavity regime when the emission rate from the cavity is slower than the effective rate of population transfer from the emitter ($\kappa < R$) and otherwise it is said to be in the bad-cavity regime ($\kappa \ge R$) (see Fig. 1.3 (b)). It is also important to consider the regime where an enhancement of the emission rate is induced by the cavity. The enhancement occurs when the rates R and κ both exceed the emitter lifetime γ , shown by the upper right region bounded by the dashed lines in Fig. 1.3 (b). Within this regime of enhancement, the area that is also within the weak-coupling regime is the classic Purcell regime, where the Purcell factor is valid and larger than 1.

In the case that γ^* is very small, all the regimes discussed so far will allow for emission that is spectrally pure, meaning that emission is in a pure quantum state. For pulsed systems, the emission will also be temporally pure and sequentially indistinguishable (I will discuss indistinguishability more in section 1.4.4). However, for a pulsed resonantly excited emitter that experiences dephasing, there are two regimes where highly indistinguishable emission can be achieved. I will refer to these two regimes as the incoherent pumping regime $(R, \kappa < \gamma)$ and the critical regime $(R, \kappa > \Gamma)$ (see Fig. 1.3 (c)). The incoherent pumping regime is named so because the emitter inefficiently and incoherently excites the cavity mode so that the characteristics of emission are dominated by the cavity rather than the emitter. Since the cavity acts as the emitter, and it does not suffer from dephasing, indistinguishable emission can be achieved [86]. I refer to the other regime as the critical regime because it allows for system dynamics that are reminiscent of a critically damped oscillator, although not all points within this regime satisfy that analogy. In this regime, the emission rate is so fast that it overcomes the dephasing rate, allowing for indistinguishable emission.

These latter two regimes contain the condition $R = \kappa$ that defines the boundary between the good- and bad-cavity regimes, which is a 'Goldilocks condition' [100] for single-photon sources. This condition implies that the cavity mode decays at the same rate that it is effectively populated by the emitter. Here, the perfect balance is struck between having a cavity that is good enough to enhance the light-matter coupling but still dissipative enough to not prolong the light-matter interaction. In the critical regime, this condition reduces to $\kappa = 2g$, which is analogous to an impedance matching condition, allowing for an efficient transfer of emitter population into the environment. In the incoherent pumping regime, the condition reduces to $\kappa \Gamma = 4g^2$. Of these two regimes, only the critical regime is within the region where a significant rate enhancement can be achieved. It also lies on the boundary between the strong-coupling and the weak-coupling regimes. This boundary is defined by R = 2g, which incidentally converges to the Goldilocks condition

in the critical regime. The small region between this boundary and the curve $R = \Gamma$ is the only region where the standard Purcell factor is valid (there are no vacuum Rabi oscillations) and the emission is highly indistinguishable. Interestingly, reasonably efficient emission that is partially indistinguishable can be achieved along the Goldilocks condition between the incoherent pumping and critical regimes. This bridge between the regions arises due to the cavity funnelling effect [86], where the broader dephased emitter linewidth is funneled into the narrower cavity linewidth.

Although these different regimes are useful to keep in mind, the remainder of this thesis will focus primarily on the critical regime and, in most cases, only the part of the critical regime that overlaps with the Purcell regime, as indicated by the red hatched area in Fig. 1.3 (c). In the next section, I will describe how the cavity-emitter-waveguide system in the bad-cavity regime reduces to an effective 'one-dimensional atom'.

1.3.6 Cavity-enhanced one-dimensional atom

In section 1.3.4, I introduced how the cavity-emitter dynamics in the weak-coupling regime can be described by the effective rate R that couples the equation of motion for $\langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle$ and $\langle \hat{a}^{\dagger} \hat{a} \rangle$ (see Eq. (1.32)). We can push this idea one step further by eliminating the cavity mode from the dynamics entirely. In this case, the two-level emitter effectively couples directly to the collection waveguide. This simplified picture of a two-level system coupled to a waveguide is referred to as a one-dimensional atom [101].

Instead of eliminating the correlation between the emitter and the cavity to get Eq. (1.32), we can directly eliminate the cavity amplitude in Eq. (1.29). By setting $d\langle \hat{a} \rangle/dt = 0$, we obtain $\langle \hat{a} \rangle = -(2ig/\kappa) \langle \hat{\sigma} \rangle$. Similarly, from Eq. (1.32) we can eliminate the cavity population to get $\langle \hat{a}^{\dagger} \hat{a} \rangle = R \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle / (\kappa + R)$. In this case, for $\kappa \gg R$, the cavity-emitter system reduces to an effective two-level system whose evolution can be described by the effective Liouville superoperator

$$\mathcal{L} = -\frac{i}{\hbar}\mathcal{H} + \gamma'\mathcal{D}(\hat{\sigma}) + 2\gamma^{\star}\mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma}), \qquad (1.37)$$

where $\hat{H} \simeq \hbar \omega'_o \hat{\sigma}^{\dagger} \hat{\sigma} + \hbar (\Omega^*(t) \hat{\sigma} + \Omega(t) \hat{\sigma}^{\dagger})/2$ is the effective Hamiltonian, $\gamma' \simeq \gamma + R$ is the Purcellenhanced decay rate, and $\omega'_o \simeq \omega_o - 4g^2 \Delta/(\kappa^2 + 4\Delta^2)$ is the cavity Lamb-shifted resonance of the two-level system. Since this model is effectively identical to the model introduced in section 1.3.1, I will write γ' as γ without specifying the parameters of the cavity that give rise to the decay rate.

A special note must be made for including the semi-classical driving in the Hamiltonian. When performing adiabatic elimination, one will find that the effective decay rate γ' depends on Ω and that Ω reduces γ' . Physically, this occurs when the semi-classical driving dresses the two-level system, effectively decoupling it from the cavity mode and reducing the Purcell factor. Furthermore, a time-dependent Ω will cause γ' to become time dependent. However, any transient behaviour of the cavity-emitter interaction due to a time-dependent Ω is inherently neglected when performing adiabatic elimination. Hence, in addition to requiring the bad-cavity condition $\kappa \gg R$ so that any transients are short compared $1/\gamma'$, this effective two-level model is valid either when (1) $\Omega(t)$ is small compared to g or (2) $\Omega(t)$ is nonzero for a time small compared to $1/\gamma'$. This latter condition is often satisfied for state-of-the-art pulsed single-photon sources operating far into the bad-cavity regime, where the effective two-level model performs well [vi]. I will discuss this topic in more detail at the end of section 3.3.1.

For the one-dimensional atom model, the cavity amplitude \hat{a} becomes effectively proportional to the amplitude of the two-level system $\hat{\sigma}$. Strictly speaking, these two operators cannot be proportional because $\hat{\sigma}$ is a finite dimensional operator whereas \hat{a} is an infinite dimensional operator. However, under the condition that $\langle \hat{a} \rangle$, $\langle \hat{a}^{\dagger} \hat{a} \rangle \ll 1$, which is satisfied in the bad-cavity regime, the cavity mode operator becomes an effective two-level operator itself. Furthermore, this effective proportionality extends to the waveguide mode \hat{b} provided that we only perform measurements on the waveguide that are not sensitive to the vacuum fluctuations of the continuum, such as photon-counting measurements.

1.3.7 Quasi-normal mode master equation

The derivation of the Jaynes-Cummings model and the cavity QED master equation makes a key assumption that the modes of the cavity are far separated in frequency compared to their linewidth κ . That is, we must assume that the spectrum of the cavity is described by a sum of well-separated independent Lorentzian-shaped resonances. This allows for the light-matter interaction to be described by a single quantized mode \hat{a} interacting with the two-level system. However, this assumption is often violated for low-Q devices that incorporate very lossy materials, such as metals or other plasmonic material, that can absorb and/or scatter light at a high rate.

If the spectrum of a cavity cannot be modeled by a summation of independent narrow Lorentzian modes, then it is necessary to revisit the light-matter interaction from first principles. Although there have been many approaches to generalizing the Jaynes-Cummings model for highly dissipative environments [102], one recent approach by Franke *et al.* [103] stands above the rest in that it provides a clear and computationally-friendly [104] link between the classical properties of the electromagnetic field and a Markovian master equation predicting the quantum dynamics. This approach is based on a quasi-normal mode (QNM) expansion of the field where the cavity is described by a summation of complex Lorentzian modes that can interfere if they overlap in space and frequency. These QNMs are modes with complex eigenfrequencies that describe the mode resonance peak as the real part of the eigenfrequency and the spectral width, or damping rate, as the imaginary part of the eigenfrequency.

The QNM master equation has been successfully used to analyze Purcell enhancements of single emitters [103] and study the quantum properties of light emitted by a cavity-emitter system [105]. In this thesis, I will explore some properties predicted by the phenomenological application of the quantum model based on the generalized constraints of its parameters. Hence, I will not go into depth on the subtleties and fundamental considerations of its derivation, which can be found in Ref. [106].

The form of the QNM master equation is similar to a multi-mode Jaynes-Cummings interaction

with the key difference that the mode quantization accounts for possible interference and light-matter coupling renormalization. Let us begin by supposing that our cavity field is described by a set of QNMs \tilde{a}_i [103] with complex resonances $\tilde{\omega}_i = \omega_i - i\kappa_i/2$, where ω_i is the mode resonance and κ_i is the FWHM spectral intensity linewidth. The modes \tilde{a}_i contain all the spatial and spectral structure of the cavity field. However, in general, they do not immediately satisfy the canonical commutation relations for harmonic oscillators because they are not plane-wave modes—they have a finite width due to dissipation. To enforce these commutation relations, the modes can be symmetrized by defining new operators \hat{a}_i such that $\hat{a}_i = \sum_j (\mathbf{S}^{-1/2})_{ij} \tilde{a}_j$ and $\hat{a}_i^{\dagger} = \sum_j (\mathbf{S}^{-1/2})_{ji} \tilde{a}_j^{\dagger}$, where \mathbf{S} is a positive semi-definite Hermitian matrix fully describing the spatial and spectral overlap of the QNMs [103].

After QNM symmetrization, the Liouville superoperator that governs the quantum evolution of an emitter-cavity interaction is given generally as [103]

$$\mathcal{L} = -\frac{i}{\hbar}\mathcal{H} + \gamma \mathcal{D}(\hat{\sigma}, \hat{\sigma}) + \sum_{ij} 2\chi_{ij}^{-} \mathcal{D}(\hat{a}_i, \hat{a}_j)$$
(1.38)

where \mathcal{D} is given by $\mathcal{D}(\hat{a}_i, \hat{a}_j)\hat{\rho} = \hat{a}_i\hat{\rho}\hat{a}_j^{\dagger} - \{\hat{a}_j^{\dagger}\hat{a}_i, \hat{\rho}\}/2$ and the Hamiltonian corresponding to $\mathcal{H}\hat{\rho} = [\hat{H}, \hat{\rho}]$ is

$$\hat{H} = \hbar\omega_o \hat{\sigma}^{\dagger} \hat{\sigma} + \sum_{ij} \hbar\chi_{ij}^+ \hat{a}_i^{\dagger} \hat{a}_j + \sum_i \hbar g_i \hat{\sigma} \hat{a}_i^{\dagger} + \hbar g_i^* \hat{\sigma}^{\dagger} \hat{a}_i.$$
(1.39)

The parameters χ_{ij}^{\pm} describing the QNM interference are given by $\chi_{ij}^{+} = (\chi_{ij} + \chi_{ji}^{*})/2$ and $\chi_{ij}^{-} = i(\chi_{ij} - \chi_{ji}^{*})/2$, where $\chi_{ij} = \sum_{k} (\mathbf{S}^{-1/2})_{ik} \tilde{\omega}_{k} (\mathbf{S}^{1/2})_{kj}$. The complex emitter-cavity coupling rate g_{i} is given in terms of the QNM coupling rates before symmetrization \tilde{g}_{j} by $g_{i} = \sum_{j} (\mathbf{S}^{1/2})_{ji} \tilde{g}_{j}$. The quantized mode \hat{a}_{i} has a bare resonance χ_{ii}^{+} and interacts with mode \hat{a}_{j} through the Hermitian coupling described by χ_{ij}^{+} . It is important to note that the quantized mode operators \hat{a}_{i} describe the quasinormal modes after performing a symmetrization that maintains the required commutative properties. Hence, the resonances χ_{ii}^{+} and the decay rates χ_{ii}^{-} do not correspond to $\tilde{\omega}_{i}$ directly, but are already altered due to the interference of the QNMs. Furthermore, we have a coupling between

the quantized modes. This coupling appears in the Hamiltonian described by parameters χ_{ij}^+ and a coupling in the dissipative part of the master equation described by parameters χ_{ij}^- .

Since the matrix χ^- that is described by the elements χ_{ij}^- is Hermitian, there exists a unitary transformation of modes \hat{a}_i that diagonalizes the dissipative part of the master equation [106]. I will denote these dissipative eigenmodes as $\hat{c}_i = \sum_j v_j \hat{a}_j$ for eigenvector elements v_i of χ^- . The eigenvalues κ'_i of this transformation correspond to the exponential decay of modes \hat{c}_i . This transformation allows for a straightforward derivation of input-output relations for the symmetrized QNM operators [106]. It also provides a familiar Markovian master equation form where all of the mode coupling is described by the Hamiltonian and the dissipative part simply describes the exponential decay of those coupled modes. It is important to emphasize that, although this approach provides a model that is similar in form to a multi-mode Jaynes-Cummings model, the light-matter coupling and mode interference now become directly dependent on dissipation. This can lead to a significant reduction in the coupling magnitude compared to the standard Jaynes-Cummings approach when considering highly dissipative cavities [106].

In this thesis, I will not go into detail about the quantum properties of emission predicted by this QNM master equation, as this is already actively being explored [104–106]. However, I do expect that many of the methods I use in this thesis for photonic state analysis can be applied to this generalized system to great effect. That said, in section 4.2, I will apply the QNM master equation to explore the potential application of plasmonic cavities to mediate high-fidelity local spin-spin interactions.

1.4 Figures of merit

I will now introduce a handful of important quantities that appear multiple times throughout this thesis. First, I will discuss how the Purcell factor is related to the cavity cooperativity, which is a figure of merit used extensively in section 4.2. Then I will introduce three important figures of merit for single-photon sources: the brightness, the single-photon purity, and the indistinguishability.

These quantities arise often in chapter 3 and again in section 4.1. Finally, I will define state fidelity and concurrence. These will be used in section 3.3 to quantify the quality of photonic state entanglement, and again in chapter 4 to discuss spin-spin entanglement.

1.4.1 Cavity cooperativity

The rate R not only describes the Purcell enhancement in the Purcell regime as we saw in section 1.3.4, it is also a useful quantity to characterize the cavity-emitter system. As we found in section 1.3.5, it allows us to easily define many interesting regimes of a dephased cavity-emitter system. However, since R is not a unitless quantity, it is not useful when comparing different physical systems. Instead, we can talk about the rate R normalized by the emitter lifetime γ : the cavity cooperativity.

Let us consider a resonant system ($\Delta = 0$) without dephasing in the regime where $\gamma \ll \kappa$. Then the cavity cooperativity is

$$C = \frac{4g^2}{\kappa\gamma},\tag{1.40}$$

which is a quantity that appears very often in the literature and will arise often in section 4.2. Similar to the Purcell factor, the cavity cooperativity quantifies how quickly the system decays via the cavity mode as opposed to direct decay via the emitter. Note that, if $\gamma \ll \kappa$, C > 1 does not imply that there is a vacuum Rabi splitting. Hence, I do not consider a system with C > 1 as necessarily being in the strong-coupling regime. We can also discuss the effective cavity cooperativity, which I denote C^* , that is inhibited by emitter broadening and detuning

$$C^{\star} = CF_{\text{inh}}.\tag{1.41}$$

The effective cavity cooperativity is related to the effective Purcell factor by $C^* = \eta_r F_P$, where $\eta_r = \gamma_r / \gamma$ is the radiative efficiency of the emitter.



Figure 1.4: Characterization of a pulsed single-photon source. (a) Integrated intensity μ measured by direct detection. By knowing the losses prior to detection, this can be used to estimate the source brightness β , which is $\beta = 1$ in the ideal case. (b) Single-photon purity measured by the integrated intensity correlation $g^{(2)}$ using a balanced beam splitter and normalizing by the intensity μ . In the ideal case where the photonic state contains at most one photon, we have that $g^{(2)} = 0$. (c) Characterization of the purity in the time/frequency domain. Subsequent photonic states are interfered in a Mach-Zehnder interferometer with one arm delayed by the pulse separation $T_{\rm p}$. The normalized intensity correlation after Hong-Ou-Mandel (HOM) interference $g^{(2)}_{\rm HOM}$ should be zero in the ideal case due to perfect photon bunching.

1.4.2 Brightness

For many applications of single-photons, it is necessary that the photons are produced very efficiently. This is usually characterized by the source 'first lens' brightness \mathcal{B} , which is the ratio of the photon rate collected at position x = 0 of the waveguide divided by rate of pulsed excitation of the device. In the case that at most one photon can be produced by a single pulse, the first lens brightness becomes equal to the probability of generating a single photon.

Since it is not easy to place a perfectly efficient detector at the first lens, the measurement of brightness is usually dependent on the characterization of additional losses and inefficiencies of the setup, such as transmission and detection efficiency. It is also affected by the probability that the emitter is in the correct 'bright state' configuration to emit a photon, also called the emitter occupation probability, which I will denote $\eta_{\rm b}$.

In this thesis, I will discuss only the contribution to the first lens brightness due to the dynamics of the light-matter interaction when the emitter is in the correct configuration. This is described by

the average photon number in the waveguide $N(t) = \langle \hat{b}^{\dagger}(t) \hat{b}(t) \rangle$ integrated over time

$$\mu = \int N(t)dt = \kappa \eta_{\rm c} \int \langle \hat{a}^{\dagger}(t)\hat{a}(t)\rangle \, dt, \qquad (1.42)$$

where the integration is taken over all dynamics induced by a single pulsed excitation or state preparation protocol. It is also convenient to introduce the quantity $\overline{\mu} = \mu/\eta_c = \sum_n np_n$, where $p_n = \Pr(n)$ is the probability of having *n* photons leaving the cavity mode. This quantity has important implications for the photon statistics of the photonic state produced by the cavity mode. Since I am dividing out the collection efficiency η_c , some of these photons may be directly absorbed by the cavity material, or leave the cavity but never make it to the waveguide. Using $\overline{\mu}$, we can write the total first-lens brightness, including occupation probability, as $\mathcal{B} = \eta_b \eta_c \overline{\mu}$.

For a given emitter system, it is often enlightening to look at what $\overline{\mu}$ is expected for a perfect state preparation of the emitter. This gives insight into the upper bound on the brightness under perfect single-photon generation conditions. I will denote this idealized definition of brightness as β and refer to it either as the intrinsic cavity efficiency or brightness without further specification. For the cavity-emitter system introduced so far, a perfect state preparation corresponds to $|e\rangle$, which can give at most one photon. In the Purcell regime, the value of β then has an upper bound that is given by the ratio of decay from the emitter at the rate γ and the effective rate R. Using the effective rate R, we can write this upper bound in terms of the effective cavity cooperativity as

$$\beta \le \frac{R}{\gamma + R} = \frac{C^{\star}}{1 + C^{\star}} \le 1. \tag{1.43}$$

For single-photon sources, one ultimate goal is to achieve $p_1 = 1$. From the above definition of \mathcal{B} , we can see that $\mathcal{B} = 1$ does not necessarily imply $p_1 = 1$ since $\overline{\mu} = 1$ can imply an infinite number of sets of p_n values. For example, $p_0 = p_2 = 0.5$ can also give $\overline{\mu} = 1$. Therefore, it is necessary to characterize the quality of a single-photon source using more than just the brightness.

1.4.3 Single-photon purity

An ideal single-photon source must provide at most one photon per pulse. This can be characterized by the antibunching of the photonic state at the output of a beam splitter (see Fig. 1.4 (b)). If the photonic state does not contain more than one photon, the intensity correlation between output modes \hat{b}_3 and \hat{b}_4 measured by the two detectors will vanish. For a balanced beam splitter, the modes \hat{b}_3 and \hat{b}_4 will be in identical states proportional to the state of the input waveguide mode $\hat{b} = \hat{b}_1$, however the detection times may be different. Integrating the two-time intensity correlation of the waveguide mode $G^{(2)}(t_1, t_2) = \langle \hat{b}^{\dagger}(t_1)\hat{b}^{\dagger}(t_2)\hat{b}(t_2)\hat{b}(t_1)\rangle$ over all times and normalizing by the integrated intensity of each mode gives the normalized integrated intensity correlation characterizing the single-photon purity

$$g^{(2)} = \frac{1}{\mu^2} \iint G^{(2)}(t_1, t_2) dt_1 dt_2, \tag{1.44}$$

where the integration is taken over the dynamics as for μ . This quantity is often denoted $g^{(2)}(0)$ to emphasize that it is the intensity correlation around zero delay, which is a relic of continuous wave (CW) source characterization. However, the true equivalent to $g^{(2)}(0)$ for pulsed sources would be given by the time-integrated intensity correlation $g^{(2)}(\tau)$ as a function of detection delay τ given by $g^{(2)}(\tau) = \mu^{-1} \int G^{(2)}(t, t + \tau) dt$, which is not equal to the full time-integrated quantity if $\tau = 0$. Hence, for pulsed sources where subsequently produced photon states are uncorrelated, I find the emphasis on 0 unnecessary when discussing the full time-integrated quantity and so I will use the notation $g^{(2)}$ and not $g^{(2)}(0)$.

Furthermore, it is important to note that the intensity normalized $g^{(2)}$ is independent of photon losses, such as non-unity collection efficiency η_c and transmission efficiency η_t . It is also independent of detector efficiency η_d provided that the beam splitter is balanced so that each detector receives the same average intensity. This means that $g^{(2)}$ measured after significant losses is identical to the value at the device.

If the source experiences irreversible evolution besides that which is caused by the collected

emission, it is not straightforward to compute two-time correlations such as $\langle \hat{b}^{\dagger}(t_1)\hat{b}^{\dagger}(t_2)\hat{b}(t_2)\hat{b}(t_1)\rangle$ in the Schrödinger picture when $t_2 < t_1$. However, due to the time symmetry of the correlation function and normalization intensity, we can restrict the integration to a time ordering such that $t_2 \ge t_1$ to compute $g^{(2)}/2$ and multiply by the factor of 2 to obtain the full time-integrated value. As a consequence, I will often switch between the form given in Eq. (1.44) and the equivalent expression

$$g^{(2)} = \frac{2}{\mu^2} \int_{-\infty}^{\infty} \int_{t_1}^{\infty} G^{(2)}(t_1, t_2) dt_2 dt_1, \qquad (1.45)$$

depending on which is more convenient. I will discuss more about time ordering conventions in chapter 2.

1.4.4 Indistinguishability

Some applications of single-photon sources, such as quantum state teleportation [29], quantum repeaters [74], and optical quantum computing [63], require that the photons are indistinguishable. This is characterized by how well two photons interfere at a beam splitter to give rise to two-photon bunching, which is known as the Hong-Ou-Mandel (HOM) effect [107]. The topic of HOM interference is covered in significant detail in section 3.2. However, in this introductory section, I will give the basics in order to define the quantities of mean wavepacket overlap and indistinguishability.

For perfectly indistinguishable photons at the input of a balanced beam splitter $|\psi\rangle = \hat{b}_1^{\dagger}\hat{b}_2^{\dagger}|0\rangle$, the output after the beam splitter transformation $\hat{b}_1^{\dagger} = (\hat{b}_3^{\dagger} + \hat{b}_4^{\dagger})/\sqrt{2}$ and $\hat{b}_2^{\dagger} = (\hat{b}_4^{\dagger} - \hat{b}_3^{\dagger})/\sqrt{2}$ is

$$|\psi\rangle = \frac{1}{2} \left(\hat{b}_3^{\dagger} + \hat{b}_4^{\dagger} \right) \left(\hat{b}_4^{\dagger} - \hat{b}_3^{\dagger} \right) |0\rangle = \frac{1}{2} \left(\hat{b}_4^{\dagger 2} - \hat{b}_3^{\dagger 2} \right) |0\rangle , \qquad (1.46)$$

which implies that the intensity correlation between detectors monitoring modes \hat{b}_3 and \hat{b}_4 vanishes. Here, I am using \hat{b} to represent a plane-wave mode of the waveguide(s) to demonstrate the basic nature of the HOM effect. In practice, finite wavepacket length plays an important role in determining the output state of the beam splitter.

To quantify the indistinguishability of photons emitted by the same source, a Mach-Zehnder interferometer is often used where one arm is delayed by the separation between pulses $T_{\rm p}$ (see Fig. 1.4 (c)). By following the methods similar to Ref. [92], we can take into account the time dynamics of the identical input photonic states. The intensity correlation at the output of a beam splitter is given as in the previous section $G_{\text{HOM}}^{(2)}(t_1, t_2) = \langle \hat{b}_3^{\dagger}(t_1) \hat{b}_4^{\dagger}(t_2) \hat{b}_4(t_2) \hat{b}_3^{\dagger}(t_1) \rangle$. Applying the inverse balanced beam splitter relations $\hat{b}_3^{\dagger} = (\hat{b}_1^{\dagger} - \hat{b}_2^{\dagger})/\sqrt{2}$ and $\hat{b}_4^{\dagger} = (\hat{b}_1^{\dagger} + \hat{b}_2^{\dagger})/\sqrt{2}$ we can expand this intensity correlation in terms of the input modes 1 and 2. By making the assumption that these input modes are not entangled or classically correlated, the correlations such as $\langle \hat{b}_1^{\dagger} \hat{b}_2^{\dagger} \hat{b}_2 \hat{b}_1 \rangle$ can be separated into $\langle \hat{b}_1^{\dagger} \hat{b}_1 \rangle \langle \hat{b}_2^{\dagger} \hat{b}_2 \rangle$. Then, assuming that the states of each mode are identical, we can take $\hat{b}_1(t) = \hat{b}(t)$ and $\hat{b}_2(t) = \hat{b}(t)$. Finally, it is often the case that the input states carry no oneand two-photon coherence so that $\langle \hat{b}(t) \rangle = \langle \hat{b}(t) \hat{b}(t) \rangle = 0$. This is true when the photonic state leaving the cavity gives $p_1 \simeq 1$, because any photon losses from collection or transmission will not generate photon number coherence. This is also true when the Mach-Zehnder interferometer used to measure indistinguishability has a highly unstable phase on the timescale of the measurement. In this case, any effect of nonzero $\langle \hat{b}(t) \rangle$ or $\langle \hat{b}(t) \hat{b}(t) \rangle$ will average to zero. Under either condition, we obtain

$$G_{\text{HOM}}^{(2)}(t_1, t_2) = \frac{1}{2} \left[N(t_1)N(t_2) + G^{(2)}(t_1, t_2) - \left| G^{(1)}(t_1, t_2) \right|^2 \right],$$
(1.47)

where $G^{(1)}(t_1, t_2) = \langle \hat{b}^{\dagger}(t_2)\hat{b}(t_1) \rangle$ is the two-time first order amplitude correlation function of the mode \hat{b} .

By integrating the intensity correlation over all arrival times for a single pulse or statepreparation protocol, and normalizing by the total integrated intensity, we obtain

$$g_{\text{HOM}}^{(2)} = \frac{1}{\mu^2} \iint G_{\text{HOM}}^{(2)}(t_1, t_2) dt_1 dt_2 = \frac{1}{2} \left(1 + g^{(2)} - M \right), \tag{1.48}$$

where I have defined the mean wavepacket overlap as

$$M = \frac{1}{\mu^2} \iint \left| G^{(1)}(t_1, t_2) \right|^2 dt_1 d_2.$$
(1.49)

It is also common to talk about the HOM visibility V_{HOM} , which is further defined to be

$$V_{\rm HOM} = 1 - 2g_{\rm HOM}^{(2)} = M - g^{(2)}.$$
 (1.50)

Like $g^{(2)}$, all of these quantities are normalized by intensity and so they are independent of photon losses. Unlike $g^{(2)}$, good HOM interference requires that the intensity of each input is equal at the beam splitter.

In practice, the HOM visibility is normalized with respect to the total uncorrelated coincident counts of the interferometer [92][vi][xiii]. This is because the uncorrelated counts provide a very good estimate for the squared intensity μ^2 when the measured photonic state contains little single-photon coherence. If this is not the case, the uncorrelated coincident counts can underestimate μ^2 [108], leading to an over-estimation of V_{HOM} as defined above. This subtlety is not an issue for good single-photon sources where $\langle \hat{b}(t) \rangle \simeq 0$, but it is very important to consider when using HOM visibility to characterize more general photonic states. I will discuss this more in section 3.2.3.

The quantities M and V_{HOM} are both often referred to as the indistinguishability of photons. It is also common to use $1 - g_{\text{HOM}}^{(2)}$ [108], which for a single photon is the HOM bunching probability. Indeed, all these quantities characterize the quality of interference between two photonic states. However, the term indistinguishability refers to a property of two particles. Hence, to discuss indistinguishability, we should be clear about which two particles we are considering. In the context of single-photon source characterization, we are concerned with measuring the indistinguishability of two successive uncorrelated photons emitted by the same source. That is, the two particles are in identical states that may not be pure.

Recently, there has also been discussion of another quantity: the trace purity of the single-
photon component of a photonic density matrix [109, 110]. Let $\hat{\varrho}$ be the photonic state where $\hat{\varrho}_1$ is the unnormalized state after being projected onto the one-photon subspace with probability p_1 , then the single-photon trace purity is $\mathcal{P}_1 \equiv \text{Tr}(\hat{\varrho}_1^2)/p_1^2$. In my opinion, $0 \leq \mathcal{P}_1 \leq 1$ is the most appropriate quantity to call indistinguishability, as it is by definition the purity of two photons described by the same mixed photonic state and gives their maximum HOM bunching probability by $(1 + \mathcal{P}_1)/2$.

The mean wavepacket overlap M, the HOM visibility V_{HOM} , and $g_{\text{HOM}}^{(2)}$ are all dependent on the *entire* photonic state measured at the interferometer, including multi-photon components, but are independent of photon losses prior to measurement. On the other hand, \mathcal{P}_1 is given by the single-photon component of a photonic state and hence is generally dependent on losses. For example, a two-photon component will contribute to a measurement of \mathcal{P}_1 if one photon was lost prior to the measurement. In the limit of large losses where only the single-photon component remains, \mathcal{P}_1 of the photonic state *at the interferometer* is equal to $M = V_{\text{HOM}} + g^{(2)}$. This follows from the fact that $G^{(1)}$ in this limit is proportional to temporal density function of the single-photon component. Temporal density functions will be the subject of section 2.4. If $g^{(2)} = 0$, then \mathcal{P}_1 is independent of losses and so $\mathcal{P}_1 = M = V_{\text{HOM}}$. However, in general, it is not possible to exactly obtain \mathcal{P}_1 of the photonic state *at the source* by performing HOM measurements after losses [110].

The single-photon trace purity \mathcal{P}_1 of the photonic state at the source is an important quantity that characterizes the quantum purity of the source. It can also be used to make predictions about the behaviour of the device in various other applications, hence it is useful to try to estimate it. For these reasons, I will define the *single-photon* indistinguishability I as the value of $\mathcal{P}_1 = M = V_{\text{HOM}}$ when $g^{(2)} \rightarrow 0$. In this case, I serves to quantify the upper bound on the interference visibility for a particular device, similar to β for brightness. In section 3.1, I will compute I and β in the critical regime. The subtleties of the relationship between M, V_{HOM} , $g^{(2)}$, \mathcal{P}_1 , and I, was explored in Ref. [xiii] and is a main topic of sections 3.2 and 3.3. In section 4.1, the mean wavepacket overlap becomes the focus as photons from two different sources are interfered after significant losses.

1.4.5 Fidelity

When discussing the general quality of a particular quantum state, it is often convenient to use the measure of state fidelity. However, this requires knowledge about the desired state, which may not be available. If the model is well-understood, then it may be the case that a desired state is apparent. The state fidelity then allows for a convenient way to quantify how close the measured state is to the desired state.

The fidelity between the desired state $\hat{\rho}$ and the measured state $\hat{\varrho}$ is defined generally by [111]

$$\mathcal{F} = \mathrm{Tr}^2 \left(\sqrt{\sqrt{\hat{\rho}} \hat{\varrho} \sqrt{\hat{\rho}}} \right). \tag{1.51}$$

Note that for $\hat{\rho} = \hat{\varrho}$ the fidelity is $\mathcal{F} = 1$, even if $\hat{\rho}$ is not a pure state $\operatorname{Tr}(\hat{\rho}^2) < 1$. Very often, the desired state is a pure state $\hat{\rho} = |\psi\rangle\langle\psi|$. In this case, the fidelity reduces to the expectation value of the density operator with respect to the pure state $\mathcal{F} = \langle\psi|\hat{\varrho}|\psi\rangle$. If both states are pure, the fidelity reduces to $\mathcal{F} = |\langle\phi|\psi\rangle|^2$ where $\hat{\varrho} = |\phi\rangle\langle\phi|$ and $\hat{\rho} = |\psi\rangle\langle\psi|$.

In some protocols, such as those studied in section 4.1, it is possible to post-select different (unnormalized) measured states $\hat{\varrho}_m$ based on the measurement result m in the space of measurements \mathbb{M} such that $\hat{\varrho} = \sum_{m \in \mathbb{M}} \hat{\varrho}_m$ where $\Pr(m) = \operatorname{Tr}(\hat{\varrho}_m)$ is the measurement probability. In this case, I define the protocol fidelity as a weighted average of all the post-selected measurements m in the set of accepted measurements \mathbb{A} so that $\mathcal{F} = \sum_{m \in \mathbb{A}} \Pr(m) \mathcal{F}_m / \eta$ where $\eta = \sum_{m \in \mathbb{A}} \Pr(m)$ is the total protocol efficiency, $\mathcal{F}_m = \operatorname{Tr}^2\left(\sqrt{\sqrt{\hat{\rho}_m}\hat{\varrho}_m}\sqrt{\hat{\rho}_m}\right) / \Pr(m)$, and $\hat{\rho}_m$ is the desired state for measurement m. In this way, the individual measurement probabilities $\Pr(m)$ cancel and the weight is carried by the unnormalized density operator $\hat{\varrho}_m$.

1.4.6 Entanglement and concurrence

Although the fidelity is a good measure of the distance between two states, it is quite influenced by the classical statistics of the states and it requires knowledge about a particular desired state. This

makes it difficult to use the fidelity to determine, for example, how entangled a state is. Formally, a quantum state $|\psi\rangle$ of two or more subsystems is said to be entangled if it cannot be written as a product of individual subsystem states. That is, if $|\psi\rangle \neq |\psi_1\rangle_1 \otimes |\psi_2\rangle_2 \otimes \cdots \otimes |\psi_N\rangle_N$ for any set of states $|\psi_i\rangle$ of N subsystems. Entangled states give rise to quantum correlations that can be very useful in quantum information processing and fundamental experiments such as Bell inequality violations. However, it is generally a difficult task to quantify "how much" entanglement a state has if there are more than a few quantum systems involved.

In this thesis, I only explore the simplest case: entanglement between qubits of two quantum systems. Suppose we have two systems each with states $|0\rangle$ and $|1\rangle$. A very useful set of entangled states that spans the total system of two qubits are the Bell states. These are defined by $|\psi^{\pm}\rangle = (|10\rangle \pm |01\rangle)/\sqrt{2}$ and $|\phi^{\pm}\rangle = (|00\rangle \pm |11\rangle)/\sqrt{2}$. These Bell states give maximum possible quantum correlations and so they are said to be maximally entangled. I will discuss Bell states more in sections 3.3 and 4.1.

Entanglement is straightforward to quantify for two-qubit systems, and there are multiple ways to do so. For this purpose, I will sometimes compute the fidelity with respect to a maximally entangled state. But occasionally, I will also compute the Wootters concurrence [112], which is independent of a reference state. If the subsystems have more than two states, I will compute the concurrence after projecting the total system onto the two-qubit subsystem. This measure of entanglement is defined by

$$\mathcal{C} = \max\left(0, \sqrt{\alpha_1} - \sqrt{\alpha_2} - \sqrt{\alpha_3} - \sqrt{\alpha_4}\right),\tag{1.52}$$

where $\alpha_i \geq \alpha_{i+1}$ is the i^{th} eigenvalue of $(\mathcal{I}_{s}\hat{\varrho})(\mathcal{Y}_{s}\hat{\varrho}^*)$. Here, $\mathcal{I}_{s}\hat{\rho} = (\hat{I}_{s}\otimes\hat{I}_{s})\hat{\rho}(\hat{I}_{s}\otimes\hat{I}_{s})$ is the density operator projected onto the two-qubit subsystem and $\hat{I}_{s} = \hat{\sigma}^{\dagger}\hat{\sigma} + \hat{\sigma}\hat{\sigma}^{\dagger}$ is the identity operator of the two-qubit subsystem, where $\hat{\sigma}$ is the spin qubit lowering operator; and $\mathcal{Y}_{s}\hat{\rho}^* = (\hat{\sigma}_{y}\otimes\hat{\sigma}_{y})\hat{\rho}^*(\hat{\sigma}_{y}\otimes\hat{\sigma}_{y})$, where $\hat{\sigma}_{y} = i(\hat{\sigma} - \hat{\sigma}^{\dagger})$ is the Pauli y operator.

A state with a concurrence greater than 0 is an entangled state. A maximally entangled state of

two qubits, such as a Bell state, has a concurrence of 1. The concurrence is also related to other measures of entanglement [113]. For example, a state with a concurrence bigger than $1/\sqrt{2}$ is guaranteed to violate the Clauser-Horne-Shimony-Holt (CHSH) Bell inequality [114]. However, this does not mean that a concurrence less than $1/\sqrt{2}$ will not violate the CHSH inequality.

1.5 Analytic tools

In this section, I will summarize some important analytic methods used throughout this thesis. Each subsection describes a 'tool' that I use at least once, and some more often than others. The first tool is the Fock-Liouville space, which is a formalism that I use in all of my numerical simulations. So, although it will not appear again directly in the lines of this thesis, nearly all of the figures are taking advantage of it in some way. The second tool is the quantum regression theorem, which is used to compute some analytic results shown in section 3.1 and the details of its application are in appendix B. The third tool is the non-Hermitian Hamiltonian approximation, which I apply to derive results presented in section 4.2. It is also closely related to material in chapter 2. The fourth tool, adiabatic elimination, has already been used in section 1.3.4, and will appear again in section 4.2. Lastly, I will discuss time-dependent perturbation theory in the context of superoperators, which is the foundation for material in chapter 2.

1.5.1 The Fock-Liouville space

To solve the dynamics of a Markovian master equation, it is very convenient to use of the Fock-Liouville space representation [81]. This transforms the square matrix form of the density operator $\hat{\rho}$ into a vector representation notated by $|\rho\rangle\rangle$ and linear superoperators of the form $S\hat{\rho} = \sum_i \hat{A}_i \hat{\rho} \hat{B}_i$ into a square matrix notated by \tilde{S} acting on the density state vector. In this representation, the Markovian master equation is $d |\rho\rangle\rangle/dt = \tilde{\mathcal{L}} |\rho\rangle\rangle$, where $\tilde{\mathcal{L}}$ is the matrix representation of the linear superoperator \mathcal{L} . Given that $\hat{\rho} = \sum_{i,j} \rho_{i,j} |\phi_i\rangle \langle \phi_j|$, the vector form of $\hat{\rho}$ becomes $|\rho\rangle\rangle = \sum_{i,j} \rho_{i,j} |\phi_i\rangle \otimes |\phi_j\rangle$. This can be viewed as taking each column of the matrix $\hat{\rho}$ and stacking them into a single column vector. Likewise, for a superoperator of the form $S\hat{\rho} = \sum_i \hat{A}_i \hat{\rho} \hat{B}_i$, where \hat{A}_i and \hat{B}_i are operators acting on the Hilbert space, the matrix representation is given by the relation $\tilde{S} = \sum_i \hat{A}_i \otimes \hat{B}_i^T$, where \hat{B}_i^T is the transpose of \hat{B}_i . For a finite Hilbert space with dimension N, the Fock-Liouville space has dimension N^2 .

The Fock-Liouville representation can be used to solve the evolution of the master equation using either time-dependent or time-independent techniques. If \mathcal{L} is time dependent, the state evolution can be solved using one of many numerical algorithms, such as Runge-Kutta. If \mathcal{L} is time independent, then the propagation superoperator \mathcal{U} can be solved by standard matrix exponentiation in the Fock-Liouville space: $\tilde{\mathcal{U}}(t, t_0) = e^{(t-t_0)\tilde{\mathcal{L}}}$. This propagator then acts on $|\rho\rangle$ to solve for $|\rho(t)\rangle\rangle$ given $|\rho(t_0)\rangle\rangle$. The original density operator $\hat{\rho}(t)$ can then be recovered by applying the reverse transformation back into its matrix form.

In some cases, $\tilde{\mathcal{L}}$ can be analytically diagonalized, providing an analytic form for the propagation superoperator. This often allows for analytic derivations of useful figures of merit for a quantum device, such as efficiency, fidelity, or emission indistinguishability. Otherwise, it is also often possible to numerically diagonalize $\tilde{\mathcal{L}}$ for a fixed set of parameters resulting in $\tilde{\mathcal{U}}(t_f, t_0)$ that is still analytic with respect to time. This approach can drastically decrease the time needed to simulate the time dynamics and two-time correlations.

1.5.2 Quantum regression theorem

For Markovian systems, there is a useful theorem that sometimes allows for a simplified computation of multi-time correlations. The so-called quantum regression theorem [75] implies that the statistics of multi-time correlation functions are determined by the statistics of the one-time expectation values along with a proper application of the dynamical map. That is, if we know the set of operator expectations $\langle \hat{A}_i(t)\hat{B}_j(t)\rangle$, and the equations of motion for $\langle \hat{A}_i(\tau)\rangle$, we can obtain $\langle \hat{A}_i(t+\tau)\hat{B}_j(t)\rangle$. Formally, the theorem states that if

$$\frac{d}{d\tau} \left\langle \hat{A}_i(\tau) \right\rangle = \sum_j X_{i,j} \left\langle \hat{A}_j(\tau) \right\rangle \tag{1.53}$$

then

$$\frac{\partial}{\partial \tau} \left\langle \hat{A}_i(t+\tau) \hat{A}_k(t) \right\rangle = \sum_j X_{i,j} \left\langle \hat{A}_j(t+\tau) \hat{A}_k(t) \right\rangle, \qquad (1.54)$$

where $X_{i,j}$ are the coefficients of the coupled set of first-order linear differential equations.

Suppose we know

$$\frac{d}{dt} \left\langle \hat{A}_i(t) \hat{A}_j(t) \right\rangle = \sum_{k,l} Y_{i,k}^{j,l} \left\langle \hat{A}_k(t) \hat{A}_l(t) \right\rangle.$$
(1.55)

Let $\mathbf{x}(t)$ be the vector of expectation values $\langle \hat{A}_i(t) \rangle$ and let $\mathbf{y}(t)$ be the vector of expectation values $\langle \hat{A}_i(t) \hat{A}_j(t) \rangle$. Then we can write $\dot{\mathbf{x}} = X\mathbf{x}$ and $\dot{\mathbf{y}} = Y\mathbf{y}$, where I use the dot to represent the time derivative and the matrices X and Y have coefficients $X_{i,j}$ and $Y_{i,k}^{j,l}$ in some ordering. If the matrix X has a dimension N, then Y has the dimension N^2 . For time-independent parameters, we can now write the solution \mathbf{z}_k , which is the N dimensional vector of two-time correlation functions $\langle \hat{A}_i(t+\tau)\hat{A}_k(t) \rangle$, as

$$\mathbf{z}_{k}(t,\tau) = e^{\tau X} \left[e^{tY} \mathbf{y}(0) \right]_{k}, \qquad (1.56)$$

where the subscript k takes the k^{th} N-vector corresponding to the set of $\langle \hat{A}_i(t) \hat{A}_k(t) \rangle$.

The quantum regression theorem approach is particularly useful to solve time integrals of twotime correlation functions for systems where the equations of motion for the first and second order correlation functions are uncoupled, such as the optical Bloch equations described in section 1.3.4. This is because it allows for the separability of functions dependent on t and τ and hence the integrated correlations can be solved by computing two one-dimensional integrals rather than one two-dimensional integral.

The quantum regression theorem is valid for Markovian systems. However, if the Markovian master equation accurately predicts the evolution of the reduced system state, this does not nec-

essarily imply that the quantum regression theorem will also accurately predict the multi-time correlation functions of the same system. That is, the validation of the quantum regression theorem is stronger evidence for Markovianity than the validation of the Markovian master equation. In this sense, it has been suggested that a good test (or perhaps definition) of Markovian behaviour of a system is its satisfaction of the quantum regression theorem [115].

1.5.3 Non-Hermitian Hamiltonians

Moving from the full Liouville-von Neumann system evolution description to a Markovian master equation simplifies a model significantly. In some specific cases, it is possible to further simplify the system dynamics back to an effective Hamiltonian that governs the reduced state evolution via the Schrödinger equation.

To see how this is possible, first consider the cavity-emitter dissipative master equation of the form

$$\frac{d}{dt}\hat{\rho} = -\frac{i}{\hbar}[\hat{H},\hat{\rho}] + \gamma \mathcal{D}(\hat{\sigma})\hat{\rho} + \kappa \mathcal{D}(\hat{a})\hat{\rho}$$
(1.57)

This can be rewritten as [116]

$$\frac{d}{dt}\hat{\rho} = -\frac{i}{\hbar} \left[\hat{H}, \hat{\rho}\right] - \frac{1}{2} \{\gamma \hat{\sigma}^{\dagger} \hat{\sigma} + \kappa \hat{a}^{\dagger} \hat{a}, \hat{\rho}\} + \gamma \hat{\sigma} \hat{\rho} \hat{\sigma}^{\dagger} + \kappa \hat{a} \hat{\rho} \hat{a}^{\dagger}
= -\frac{i}{\hbar} \left(\tilde{H}_{\text{eff}} \hat{\rho} - \hat{\rho} \tilde{H}_{\text{eff}}^{\dagger}\right) + \gamma \hat{\sigma} \hat{\rho} \hat{\sigma}^{\dagger} + \kappa \hat{a} \hat{\rho} \hat{a}^{\dagger},$$
(1.58)

where

$$\tilde{H}_{\text{eff}} = \hat{H} - \frac{i\hbar}{2} \left(\gamma \hat{\sigma}^{\dagger} \hat{\sigma} + \kappa \hat{a}^{\dagger} \hat{a} \right)$$
(1.59)

is an effective non-Hermitian Hamiltonian that captures the amplitude decay of $\hat{\sigma}$ and \hat{a} . The remaining terms that cannot be brought into the effective Hamiltonian correspond to the jump operators $\mathcal{J}(\hat{\sigma})$ and $\mathcal{J}(\hat{a})$. These terms cause the system state to jump down by a quantum of energy perhaps after the spontaneous emission of a photon. Hence, they serve to preserve the trace of $\hat{\rho}$ by recycling the population into the ground state(s) after decay. In fact, this rearrangement is the starting point for the quantum trajectories formalism, where the stochastic action of the jump operators are simulated via Monte Carlo type simulations to reproduce the dynamics of the full density operator $\hat{\rho}$.

Due to the absence of the jump operators, the effective Schrödinger equation defined by the effective non-Hermitian Hamiltonian \tilde{H}_{eff} describes the evolution of a quantum trajectory $|\phi(t)\rangle$ during which no photon is emitted [116]. As a consequence, this evolution does not preserve the normalization condition. That is, we have $p(t) = \langle \phi(t) | \phi(t) \rangle = \text{Tr}(|\phi(t)\rangle \langle \phi(t)|) \leq 1$, which is the time-dependent probability of finding the system in the pure state trajectory $|\phi(t)\rangle$. I will define $\hat{\rho}_{\gamma\kappa}$ where $\text{Tr}(\hat{\rho}_{\gamma\kappa}) = 1$ as the state of the system if it is not in the state $|\phi(t)\rangle$. Then, the total master equation solution can be written $\hat{\rho}(t) = |\phi(t)\rangle \langle \phi(t)| + (1 - p(t))\hat{\rho}_{\gamma\kappa}(t)$, where $\hat{\rho}_{\gamma\kappa}(t)$ is the state of the system at time t given that at least one jump occurred.

The state $|\phi(t)\rangle$ is usually much easier to compute than $\hat{\rho}(t)$. If p(t) is small enough, then $|\phi(t)\rangle$ approximates the state of the system with high fidelity. In turn, it can be used to estimate the fidelity compared to a desired state $|\psi(t_f)\rangle$ at some final time t_f . In this case, the fidelity is

$$\mathcal{F} = p\mathcal{F}_0 + (1-p)\mathcal{F}_{\gamma\kappa} \tag{1.60}$$

where $\mathcal{F}_0 = (1/p) |\langle \phi(t_f) | \psi(t_f) \rangle|^2$ is the fidelity provided that no decay occurred, $p = p(t_f)$, and $\mathcal{F}_{\gamma\kappa} = \langle \psi(t_f) | \hat{\rho}_{\gamma\kappa}(t_f) | \psi(t_f) \rangle$ is the potentially non-zero fidelity after a decay event.

When computing the fidelity by solving only the effective non-Hermitian Hamiltonian part of the master equation, we are making the approximation that $\mathcal{F} \simeq p\mathcal{F}_0$. This approximation is accurate when $p \simeq 1$ or $\mathcal{F}_{\gamma\kappa} \simeq 0$. The precision of this approximation depends on \mathcal{F}_0 and $\mathcal{F}_{\gamma\kappa}$ for a given implementation. If a protocol is interrupted by a decay event, then $\mathcal{F}_{\gamma\kappa} < \mathcal{F}_0$ and so \mathcal{F}_0 can properly capture the scaling of \mathcal{F} with respect to parameters γ and κ in the high-fidelity regime. This approach is used in section 4.2.

1.5.4 Adiabatic elimination

In section 1.3.4, I touched briefly on one application of adiabatic elimination to correlation functions. The same approach can be applied to amplitudes of states or elements of the density matrix, which can be another powerful tool to reduce complicated system dynamics and obtain analytically tractable and simple solutions that can be easily applied in different situations. In this section, I would like to formalise adiabatic elimination in the general sense. To do this, I will follow the succinct definition of adiabatic elimination given in Ref. [117].

Suppose we have a system of first-order linear differential equations $\dot{z} = Zz$. The vector z can be a quantum state $|\psi\rangle$ corresponding to $Z = -i\hat{H}/\hbar$, it can be the density vector $|\rho\rangle\rangle$ corresponding to $Z = \tilde{\mathcal{L}}$, or it can be a vector of correlation functions as in sections 1.3.4 and 1.5.2. Now, suppose we can partition the vector space into two subspaces X and Y, where X contains the initial state. Let Π_x and $\Pi_y = 1 - \Pi_x$ be complementary projection matrices onto these two subspaces, respectively. Then we can write our original set of differential equations as two coupled sets of first-order differential equations

$$\dot{\mathbf{x}} = \Pi_x Z \Pi_x \mathbf{x} + \Pi_x Z \Pi_y \mathbf{y}$$

$$\dot{\mathbf{y}} = \Pi_y Z \Pi_x \mathbf{x} + \Pi_y Z \Pi_y \mathbf{y}$$
(1.61)

where $\Pi_x \mathbf{z} = \mathbf{x} \in \mathbb{X}$ and $\Pi_y \mathbf{z} = \mathbf{y} \in \mathbb{Y}$. We now make the assumption that the eigenvalues of $\Pi_y Z \Pi_y$ are far separated from those of $\Pi_x Z \Pi_x$ so that the coupling between subspaces is weak. Then let $\dot{\mathbf{y}} \simeq 0$ so that $\Pi_y Z \Pi_x \mathbf{x} \simeq -\Pi_y Z \Pi_y \mathbf{y}$. Substituting this into the equation for $\dot{\mathbf{x}}$ gives the effective equation of motion for the vector \mathbf{x} in the subspace \mathbb{X}

$$\dot{\mathbf{x}} = \left[\Pi_x Z \Pi_x - \Pi_x Z \Pi_y \left(\Pi_y Z \Pi_y\right)^{-1} \Pi_y Z \Pi_x\right] \mathbf{x} = \tilde{Z} \mathbf{x},$$
(1.62)

provided that $\Pi_y Z \Pi_y$ is invertible. Hence, \tilde{Z} is the effective Hamiltonian, Liouville superoperator, or generator of the correlation dynamics, of the subspace X undergoing adiabatic evolution with

respect to the subspace \mathbb{Y} .

The validity of adiabatic elimination is not always clear and usually requires justification based on the norm of the correction $\tilde{Z} - \prod_x Z \prod_x$ to the homogeneous evolution of x. In this thesis, I will provide comparisons to the full solutions computed numerically to validate analytic solutions obtained using adiabatic elimination.

1.5.5 Perturbation theory

Suppose we can separate the Liouville superoperator \mathcal{L} into a part that is easily diagonalized \mathcal{L}_0 and a perturbation $\epsilon \mathcal{L}_{\epsilon}$ where the eigenvalues of $\epsilon \mathcal{L}_{\epsilon}$ are much smaller than those of \mathcal{L}_0 . Then, we can write the master equation as

$$\frac{d}{dt}\hat{\rho}(t) = \mathcal{L}_0(t)\hat{\rho}(t) + \epsilon \mathcal{L}_\epsilon(t)\hat{\rho}(t).$$
(1.63)

From here, we can write the integral form of the density operator solution (see appendix A.1) as

$$\hat{\rho}(t) = \mathcal{U}_0(t, t_0)\hat{\rho}(t_0) + \epsilon \int_{t_0}^t \mathcal{U}_0(t, t')\mathcal{L}_\epsilon(t')\hat{\rho}(t')dt'$$
(1.64)

for initial condition $\hat{\rho}(t_0)$ and where \mathcal{U}_0 is the propagation superoperator corresponding to the generator \mathcal{L}_0 . If ϵ is small, we can use the zero-order 'homogeneous' solution to approximate the integrand of the first-order correction: $\hat{\rho}(t') \simeq \mathcal{U}_0(t', t_0)\hat{\rho}(t_0)$. This gives a solution $\hat{\rho}(t) \simeq (\mathcal{U}_0(t, t_0) + \mathcal{U}_1(t, t_0))\hat{\rho}(t_0)$ to first order in ϵ where

$$\mathcal{U}_1(t,t_0) = \epsilon \int_{t_0}^t \mathcal{U}_0(t,t') \mathcal{L}_\epsilon(t') \mathcal{U}_0(t',t_0) dt'.$$
(1.65)

To obtain higher-order solutions, we can repeatedly substitute Eq. (1.64) into itself in a Picard iterative fashion to obtain the series solution for \mathcal{U} , provided that it converges:

$$\mathcal{U}(t,t_0) = \sum_{n=0}^{\infty} \mathcal{U}_n(t,t_0), \qquad (1.66)$$

where for n > 0

$$\mathcal{U}_n(t,t_0) = \epsilon \int_{t_0}^t \mathcal{U}_0(t,t') \mathcal{L}_\epsilon(t') \mathcal{U}_{n-1}(t',t_0) dt'.$$
(1.67)

Then, any order of solution can be computed by truncating the summation:

$$\hat{\rho}(t) = \sum_{n=0}^{N} \mathcal{U}_n(t, t_0) \hat{\rho}(t_0) + \mathcal{O}\left(\epsilon^{N+1}\right).$$
(1.68)

This approach can also be applied to a Schrödinger equation, where it is called the Dyson series, and to any first-order linear differential equation, such as the optical Bloch equations.

Physically, the perturbation order n considers all possible times between t_0 and t that the superoperator \mathcal{L}_{ϵ} can be applied to the system state n times. Moving from order n - 1 to order n requires the integration of one additional operator \mathcal{L}_{ϵ} applied sometime between t_0 and t. In the following chapter, I will discuss the case where \mathcal{L}_{ϵ} is a jump superoperator so that the order n corresponds to the number of jumps that the system experienced during an interval of time.

Chapter 2

Photon number decomposition

One goal of quantum optics theory is to understand and compute the complicated photon statistics of non-classical light, such as photon counting distributions and detection waiting times. This led to the development of the quantum trajectories approach for solving quantum optical master equations [91, 116, 118]. In the last chapter, I introduced the Markovian master equation, non-Hermitian Hamiltonians, and perturbation theory. In this chapter, I will outline how these concepts are connected to the quantum trajectories formalism and show that they provide a useful approach for analyzing optically active quantum systems that may experience excess decoherence.

A quantum trajectory is a particular pure-state evolution of a quantum system that occurs with a certain probability [116]. The total mixed state evolution of the reduced quantum system is then seen to be a composition of all possible quantum trajectories, in the same sense as the definition of the density operator (see section 1.2). In section 1.5.3, I introduced the effective non-Hermitian Hamiltonian that describes the reduced state evolution of the quantum trajectory corresponding to no photon emission. This was used to estimate the system state for the case where the chance for emission is very small. I also highlighted that the terms neglected in the master equation to obtain the effective Hamiltonian correspond to the jump superoperators that would otherwise cause the system to jump to the ground state after the potential emission of a photon. Instead of neglecting these terms, we can treat them perturbatively using the approach described in section 1.5.5. This leads to a series expansion for the density operator in the number of jumps that occurred during the system evolution.

The perturbative expansion of a master equation described above is one example of a master equation *unravelling* [75, 91]. There are infinitely many possible unravellings that describe the master equation dynamics [91], each corresponding to a different choice of perturbation. In this thesis, I will refer to a master equation unravelling in the number of jumps as a photon number decomposition, since it can provide information on the state of the emitter or cavity-emitter system conditioned on the number of photons emitted or detected.

Master equation unravellings are the foundation for the quantum trajectories method [91, 116, 119] and are usually applied to reduce the computational complexity of a master equation. This is done by solving an effective Schrödinger equation (as in section 1.5.3) or by numerically solving a stochastic equation of the open quantum system [118, 120–125]. However, the underlying concept of quantum trajectories is not numerical in nature and can be analytically useful in certain cases even when it does not reduce the complexity of the computation.

Recently, the concept of a photon number decomposition that I will describe in this chapter has been used in part to compute photon statistics for complicated emitter dynamics [124, 126]. It is also connected to scattering matrix theory [127] and the solution for a quantum emitter coupled to a waveguide using the technique of coarse graining time [128]. Furthermore, the decomposition is intrinsically related to continuous measurement and quantum feedback theories [119, 129], which can describe active or coherent feedback schemes [125, 130]. As I will show in this thesis, the photon number decomposition approach is also a useful tool to study the temporal properties of the waveguide photonic state (section 3.3) and analyze photon counting post-selection schemes (section 4.1) under the effects of excess emitter decoherence.

The results in this chapter, such as conditional correlation functions, may be formally derived beginning from the total light-matter Hamiltonian as in [109, 128]. However, in section 2.1, I will instead present the decomposition beginning from a Markovian master equation using the principles

introduced in the previous chapter. This will hopefully give a concise and intuitive picture of the methods I employ in this thesis. In section 2.2, I will then use the decomposition to describe an imperfect photon counting measurement of the quantum system, which is applied in section 4.1. In sections 2.3 and 2.4, I demonstrate that the photon number decomposition can allow for a partial reconstruction of the photonic temporal density function of the collection waveguide that accounts for excess decoherence of the emitter. These results are then applied in section 3.3. Finally, in section 2.5, I show that the photon number decomposition of the reduced system dynamics directly satisfies the expected photon statistics relations for time integrated intensity correlation functions of the waveguide mode.

Sections 2.1 and 2.2 are published in Ref. [ix], although I have expanded some of the descriptions. The remaining material in this chapter is original research that I have not published.

2.1 Conditional propagation superoperators

We begin with the general Markovian master equation of the form

$$\frac{d}{dt}\hat{\rho}(t) = \mathcal{L}\hat{\rho}(t), \qquad (2.1)$$

where \mathcal{L} is the Liouville superoperator that contains all the reversible and irreversible dynamics of the open system, which may contain one or more emitters or emitter-cavity systems. To decompose the master equation dynamics into evolution conditioned on single-photon emission or detection, we can rearrange the master equation in a way similar to the derivation of the non-Hermitian Hamiltonian in section 1.5.3:

$$\frac{d}{dt}\hat{\rho}(t) = \mathcal{L}_0\hat{\rho}(t) + \sum_{i=1}^N \mathcal{J}_i\hat{\rho}(t), \qquad (2.2)$$

where $\mathcal{L}_0 = \mathcal{L} - \sum_i^N \mathcal{J}_i$ is the Liouville superoperator of the state evolution that lacks jumps accounted for by all \mathcal{J}_i . Each jump superoperator $\mathcal{J}_i \hat{\rho} = \hat{C}_i \hat{\rho} \hat{C}_i^{\dagger}$ is given by the collapse operator \hat{C}_i describing the effect of a photon emission or detection event on the system [91]. These collapse operators are in units of square root rate.

For a single reduced system in the Markovian regime described by the standard form of the master equation (see section 1.2.4), the collapse operators are given by the source field [92, 120, 123]. For a two-level emitter, the source field is $\hat{C} = \sqrt{\gamma \eta_r} \hat{\sigma}$, where the coefficient $\gamma \eta_r$ is the emitter Einstein A coefficient [91]. For a cavity coupled to a one-dimensional waveguide, we have $\hat{C} = \sqrt{\kappa \eta_c} \hat{a}$. Note that if there is non-radiative decay $\eta_r < 1$ or imperfect waveguide collection efficiency $\eta_c < 1$, then \mathcal{L}_0 still contains jump dynamics and cannot, in general, be reduced to an effective non-Hermitian Hamiltonian as in section 1.5.3. This illustrates the fact that an emitted photon always implies that the system jumped but a jump does not necessarily imply a photon was emitted. In the language of photon counting measurements, which is the topic of the next section, $\eta_c < 1$ simply implies that not all cavity photons were detected. Hence, the absence of a photon does not mean that the system followed a pure state trajectory.

By choosing a collapse operator $\hat{C} = \sqrt{\kappa \eta_c} \hat{a}$, we would be decomposing the dynamics of the reduced system in terms of the jump statistics that give rise to the emitted photonic state at position x = 0 of the waveguide. This is illustrated by the input-output relation $\hat{b} - \hat{b}_0 = \sqrt{\kappa \eta_c} \hat{a}$. In the quantum-optical regime where we assume that the input mode \hat{b}_0 is in the vacuum state, a photon detected at position x = 0 implies that the system jumped by the collapse operator \hat{C} . Hence, the input-output relations give us a natural way to choose our decomposition to represent the number of jumps giving rise to the photonic state at different points along the waveguide, which can capture additional photon losses or beam splitter operations applied to the source fields from multiple emitters. The input-output relations also imply that the photon number decomposition is related to a particular indirect measurement of the reduced system performed by measuring the emitted photonic state.

In general, I will refer to a reduced system composed of one or more emitters or emitter-cavity

systems as the source. For a source being monitored by N single-photon detectors, we can write $\mathcal{J}_i \hat{\rho} = \hat{d}_i \hat{\rho} \hat{d}_i^{\dagger}$ where \hat{d}_i is the source field at the *i*th single-photon detector. If the detected photonic state is fully described by the source, meaning that we are not performing quadrature measurements using an auxiliary laser, this field operator is given by operators of the source. Interestingly, quadrature measurements can be modeled using master equation unravellings and proportionality relations provided that the quantum dynamics of the local oscillator are explicitly included in the master equation [91].

From Eq. (2.2), we can now expand the solution $\hat{\rho}(t)$ perturbatively around jump superoperators \mathcal{J}_i [91, 131]. Using the method described in section 1.5.5 (see also appendix A.1), the density operator solution $\hat{\rho}(t)$ can be decomposed into a set of conditional states dependent on the cumulative photon count n_i in the *i*th mode of N modes using the Liouville-Neumann series

$$\hat{\rho}(t) = \sum_{i=1}^{N} \sum_{n_i=0}^{\infty} \hat{\rho}_{(n_1, n_2, \cdots, n_N)}(t) = \sum_{\mathbf{n} \in \mathbb{N}_0^N} \hat{\rho}_{\mathbf{n}}(t),$$
(2.3)

where $\mathbf{n} = \sum_{i}^{N} n_i \mathbf{e}_i$, and \mathbf{e}_i is the *i*th natural basis vector in the *N*-dimensional space of nonnegative integers \mathbb{N}_0^N that labels the number of photons at each detector. The conditional state $\hat{\rho}_{\mathbf{n}}(t)$ is the unnormalized density matrix $\hat{\rho}_{\mathbf{n}}(t) = \mathcal{U}_{\mathbf{n}}(t, t_0)\hat{\rho}(t_0)$, where $\mathcal{U}_{\mathbf{n}}$ is the conditional propagation superoperator given recursively by

$$\mathcal{U}_{\mathbf{n}}(t,t_0) = \sum_{i=1}^N \int_{t_0}^t \mathcal{U}_0(t,t') \mathcal{J}_i \mathcal{U}_{\mathbf{n}-\mathbf{e}_i}(t',t_0) dt', \qquad (2.4)$$

and \mathcal{U}_0 is the propagation superoperator of the equation $d\hat{\rho}(t)/dt = \mathcal{L}_0\hat{\rho}(t)$. For convenience, we define $\mathcal{U}_n = 0$ if $n \notin \mathbb{N}_0^N$. This recursive relation is the superoperator equivalent to the stochastic description given by Eq. (6.36) in Ref. [75].

The conditional state $\hat{\rho}_{\mathbf{n}}(t)$ occurs with the probability $\Pr(\mathbf{n}) = \operatorname{Tr}[\hat{\rho}_{\mathbf{n}}(t)]$, where $\sum_{\mathbf{n}} \Pr(\mathbf{n}) = 1$. It is dependent on both the initial state and the evolution time. The conditional state corresponding to \mathbf{n} can be interpreted as the mixed-state analog of a quantum trajectory, which has been referred to as a quantum corridor [132], as it can be seen as a collection of quantum trajectories that each satisfies a particular property. In this case, that property is the number of photons n_i in each mode corresponding to \hat{d}_i .

The photon number decomposition implemented in this way is an exact description of the original master equation dynamics because the total propagation superoperator of Eq. (2.1) is given by $\mathcal{U}(t, t_0) = \sum_{n} \mathcal{U}_n(t, t_0)$. As a consequence, this decomposition provides access to the state of the source after post selecting based on the number of emitted or detected photons (depending on \mathcal{J}). It is also very important to recognize that this post selection does not provide a measurement back-action on the source dynamics. It only selects out a particular quantum corridor through which the source evolved, corresponding to the fact that the underlying master equation is unchanged.

Since the total propagation superoperator \mathcal{U} has the semi-group property that $\mathcal{U}(t, t_0) = \mathcal{U}(t, t')\mathcal{U}(t', t_0)$, we can also discuss conditional states of the source for detection during a window of time (t', t) with duration T = t' - t between an initial time t_0 and a final time t_f . The conditional propagator for this window is given by

$$\mathcal{W}_{\mathbf{n}}(t_{\mathbf{f}}, t', t, t_0) = \mathcal{U}(t_{\mathbf{f}}, t')\mathcal{U}_{\mathbf{n}}(t', t)\mathcal{U}(t, t_0), \qquad (2.5)$$

where $t_0 \leq t \leq t' \leq t_f$ and $\mathcal{U} = \sum_n \mathcal{W}_n$. The conditional state given by \mathcal{W}_n does not distinguish between trajectories that give rise to different numbers of photons outside of the window of time. In the next section, I discuss how these conditional propagators are related to a measurement of the source and how they can be used to define an imperfect gated photon counting measurement where the detectors may falsely indicate the number of detected photons due to classical noise.

2.2 Photon counting measurements

For a gated detector operating at a distance L_d from the source, the measurement depends on the state of the source at the retarded time $r(t) = t - L_d/c$, where c is the transmission speed of light. The gated detector begins at time t_d where $r(t_d) \ge t_0$ and remains open for duration $T_d = t'_d - t_d = r(t'_d) - r(t_d)$. The conditional state $\hat{\rho}_n$ at time t_f after a retarded detection window is

$$\hat{\rho}_{\mathbf{n}}(t_{\rm f}) = \mathcal{W}_{\mathbf{n}}(t_{\rm f}, r(t_{\rm d}'), r(t_{\rm d}), t_0)\hat{\rho}(t_0), \tag{2.6}$$

where $t_{\rm f} - t_0 = T_{\rm d} + 2L_{\rm d}/c$ would be the minimum protocol time after a two-way classical communication. The purpose of describing this measurement delay is to place a limit on the protocol time, which becomes important in section 4.1. It is not used to describe active feedback, measurement back-action, or relativistic consequences. These are not considered in this thesis.

Suppose that we use number-resolving single-photon detectors that are ideal so that they always project the waveguide mode at the detector onto the photon-number basis. In this idealized scenario, the conditional state $\hat{\rho}_n$ is the state of the source after measuring photon numbers n. That is, the photon number decomposition is equivalent to a selective indirect measurement of the source whereby the photonic waveguide is an ideal quantum probe that provides no measurement backaction [75], analogous to a PVM applied to the waveguide (recall section 1.2.1). In this context, the conditional propagation superoperator \mathcal{U}_n is known as the operation of a generalized measurement scheme [75].

To account for classical detector imperfections such as inefficiency, dark counts, and number resolving limitations, we can remix [75] the conditional states to form a non-selective measurement. That is, we can write the imperfect measured state as a linear combination of perfect measured states weighted by conditional probabilities. This is the analogous case to applying a POVM to the waveguide (recall section 1.2.1). Although detector inefficiency could be applied during this remixing step, it is much more convenient to model detector inefficiency using a beam-splitter loss model, whereby the waveguide losses are artificially increased just before each detector. To do this, we simply modify the source fields by $\hat{d}_i \rightarrow \sqrt{\eta_{d_i}} \hat{d}_i$. Note that this does not yet reduce the selective measurement to a non-selective measurement because of our choice of measurement basis. For the remaining two imperfections, we can perform a remixing. Let \mathbb{M} be the space of all measurement outcomes from the set of N single-photon detectors. Then, the source state after measuring outcome $\mathbf{m} \in \mathbb{M}$ is given by the classical remixing of the conditional states

$$\hat{\varrho}_{\mathbf{m}}(t_{\mathbf{f}}) = \sum_{\mathbf{n} \in \mathbb{N}_{0}^{N}} \Pr(\mathbf{m}|\mathbf{n}) \,\hat{\rho}_{\mathbf{n}}(t_{\mathbf{f}}), \tag{2.7}$$

where $\Pr(\mathbf{m}|\mathbf{n})$ is the probability for measurement outcome \mathbf{m} given the state $\hat{\rho}_{\mathbf{n}}$. Note that we distinguish between the conditional state of the source $\hat{\rho}_{\mathbf{n}}$, where \mathbf{n} denotes the true photon distribution (after losses), and the state after the measurement $\hat{\varrho}_{\mathbf{m}}$, where \mathbf{m} includes dark counts and/or number resolution limitations. Naturally, we also require that $\sum_{\mathbf{m}\in\mathbb{M}} \Pr(\mathbf{m}|\mathbf{n}) = 1$ for all $\mathbf{n} \in \mathbb{N}_0^N$.

By assuming that the detectors are identical and independent, we can simplify the conditional probability to $Pr(\mathbf{m}|\mathbf{n}) \equiv \prod_{i}^{N} P_{d}(m_{i}|n_{i})$. For fast photon-number-resolving detectors (PNRDs) that can count all photons arriving during the gate duration T_{d} , we have $\mathbb{M} = \mathbb{N}_{0}^{N}$. Then $P_{d}(m|n)$ is given by all possible combinations of dark counts such that n can appear to be m:

$$P_{\text{PNRD}}(m|n) = \sum_{k=0}^{\infty} \delta_{m,k+n} p_{\mathrm{d},k}(T_{\mathrm{d}},\gamma_{\mathrm{d}}), \qquad (2.8)$$

where δ is the Kronecker delta and $p_{d,k}$ characterizes the dark count distribution. For Poissonian noise, we have that $p_{d,k}(T_d, \gamma_d) = \gamma_d^k T_d^k e^{-\gamma_d T_d} / k!$, where γ_d is the detector dark count rate.

For a bin detector (BD) that simply indicates the presence of one or more photons arriving during the gate duration, then $\mathbb{M} = \Sigma^N$ is the set of binary vectors of length N and $P_d(m|n)$ is given by

$$P_{\rm BD}(m|n) = \sum_{q=0}^{\infty} \delta_{m, \rm sgn(q)} P_{\rm PNRD}(q|n)$$

$$= \delta_{m, \rm sgn(n)} p_{\rm d,0}(T_{\rm d}, \gamma_{\rm d}) + \delta_{m,1} (1 - p_{\rm d,0}(T_{\rm d}, \gamma_{\rm d})) , \qquad (2.9)$$

where sgn : $\mathbb{N}_0 \to \Sigma$ is the signum function.

The PNRD and BD models are appropriate for many different detector types [133] (see also Ref. [i]). For example, single-photon avalanche photodiodes (APDs) and superconducting nanowire

single-photon detectors (SNSPDs) [134] can be modeled by BDs while transition edge sensors (TESs) [135, 136] or pixel arrays are ideally considered to be PNRDs.

2.3 Conditional correlation functions

Using the conditional propagation superoperators \mathcal{U}_n , we can also perform a decomposition for time-ordered correlation functions. I will demonstrate this by giving an explicit example for the two-time correlation function $G^{(1)}$ for a decomposition of a single cavity mode \hat{a} of a source. I will also give the decomposition for $G^{(2)}$. However, the idea can applied to a two-level emitter, be extended to multi-mode decompositions, and other multi-time correlations.

In section 2.1, I introduced the decomposition in terms of jump operators \mathcal{J} corresponding to the source field. For a single mode decomposition at x = 0, we have $\hat{C} = \sqrt{\kappa \eta_c} \hat{a}$ so that $\mathcal{J}\hat{\rho} = \hat{C}\hat{\rho}\hat{C}^{\dagger}$. Let S and \mathcal{R} be the superoperators corresponding to $S\hat{\rho} = \hat{a}\hat{\rho}$ and $\mathcal{R}\hat{\rho} = \hat{\rho}\hat{a}^{\dagger}$, respectively. Then we have that $S\mathcal{R} = \mathcal{R}S$ and $\mathcal{J} = \kappa \eta_c S\mathcal{R}$. From section (1.2.5), we have

$$G^{(1)}(t',t) = \langle \hat{a}^{\dagger}(t')\hat{a}(t) \rangle = \operatorname{Tr}(\mathcal{RU}(t',t)\mathcal{SU}(t,t_0)\hat{\rho}(t_0))$$
(2.10)

where $t' \ge t \ge t_0$. By substituting the photon number decomposition for \mathcal{U} , we have

$$\langle \hat{a}^{\dagger}(t')\hat{a}(t)\rangle = \sum_{l,m} \operatorname{Tr}(\mathcal{R}\mathcal{U}_{l}(t',t)\mathcal{S}\mathcal{U}_{m}(t,t_{0})\hat{\rho}(t_{0})).$$
(2.11)

In order to obtain a correlation function that probes the corridor corresponding to a specific outcome n, we need to consider the decomposition of evolution after t' as well. To do this, we can note that $\mathcal{U} = \sum_k \mathcal{U}_k$ is trace preserving because it is the propagation superoperator corresponding to a master equation. Hence, we can substitute it after \mathcal{R} and obtain

$$\langle \hat{a}^{\dagger}(t')\hat{a}(t)\rangle = \sum_{k,l,m} \operatorname{Tr}(\mathcal{U}_k(t_{\mathrm{f}},t')\mathcal{R}\mathcal{U}_l(t',t)\mathcal{S}\mathcal{U}_m(t',t)\hat{\rho}(t_0)), \qquad (2.12)$$

which also requires that $t_f \ge t'$. We can now count how many n total jumps occur between time t_0 and t_f . To do so, we add the condition from each propagation superoperator plus the combined action of \mathcal{R} and \mathcal{S} to get n = k + l + m + 1.

By rearranging the summation so that k + l + m + 1 = n, and taking $t_f \to \infty$, we have our final result

$$\langle \hat{a}^{\dagger}(t')\hat{a}(t)\rangle = \sum_{n} \langle \hat{a}^{\dagger}(t')\hat{a}(t)\rangle_{n}$$
(2.13)

where

$$\left\langle \hat{a}^{\dagger}(t')\hat{a}(t)\right\rangle_{n} = \sum_{k,l,m} \lim_{t_{\mathrm{f}}\to\infty} \mathrm{Tr}(\mathcal{U}_{k}(t_{\mathrm{f}},t')\mathcal{R}\mathcal{U}_{l}(t',t)\mathcal{S}\mathcal{U}_{m}(t,t_{0})\hat{\rho}(t_{0}))\,\delta_{n,k+l+m+1}$$
(2.14)

is the correlation function conditioned on n jumps \mathcal{J} between an initial state $\hat{\rho}(t_0)$ until time $t_f \to \infty$. Note that this conditional correlation is only well-defined if the master equation satisfies $\lim_{t\to\infty} \mathcal{J}\hat{\rho}(t) = 0$ so that the source eventually reaches a dark state. Otherwise, the correlation could be defined for a finite time t_f . The necessity for adding 1 to k + l + m to account for the combined action of \mathcal{R} and \mathcal{S} becomes apparent when noting that we require that $\langle \hat{a}^{\dagger}(t')\hat{a}(t)\rangle_0 = 0$ but that the case for k + l + m = 0 can give a nonzero result in general. Hence, k = l = m = 0 should correspond to the first non-zero term of the expansion, which is the correlation over the single-jump corridor $\langle \hat{a}^{\dagger}(t')\hat{a}(t)\rangle_1$. If this argument is not convincing, we could take n to be an arbitrary definition and leave its interpretation as 'number of jumps' to be justified in the following section. At this point, the validity of Eq. (2.14) does not depend on our interpretation of n.

This decomposition can also be used to compute any time-ordered correlation function. For example, the second-order correlation $G^{(2)}$ is

$$\langle \hat{a}^{\dagger}(t)\hat{a}^{\dagger}(t')\hat{a}(t')\hat{a}(t)\rangle = \operatorname{Tr}(\mathcal{RSU}(t',t)\mathcal{RSU}(t,t_0)\hat{\rho}(t_0))$$

$$= \sum_{n} \langle \hat{a}^{\dagger}(t)\hat{a}^{\dagger}(t')\hat{a}(t')\hat{a}(t)\rangle_{n}$$

$$(2.15)$$

where

$$\left\langle \hat{a}^{\dagger}(t)\hat{a}^{\dagger}(t')\hat{a}(t')\hat{a}(t)\right\rangle_{n} = \sum_{k,l,m} \lim_{t_{f} \to \infty} \operatorname{Tr}(\mathcal{U}_{k}(t_{f},t')\mathcal{RSU}_{l}(t',t)\mathcal{RSU}_{m}(t,t_{0})\hat{\rho}(t_{0}))\,\delta_{k+l+m+2,n}.$$
(2.16)

Similar to above, I am adding 2 to k + l + m to account for the additional two jumps \mathcal{RS} within the dynamics probed by $\langle \rangle_n$. As before, the association of n = k + l + m + 2 to the number of jumps becomes more apparent in the following section where I will show that this additional counting is necessary so that $\langle \rangle_n$ defined in Eq. (2.16) is consistent with that of Eq. (2.14) when reconstructing the state of the waveguide within photon number subspaces.

2.4 Temporal density functions

Suppose that \mathcal{L} satisfies $\lim_{t\to\infty} \mathcal{J}\hat{\rho}(t) = 0$ so that the source reaches a dark state. Suppose also that the total system operates in the quantum optical regime where the Fourier transform of plane wave states allows for a photonic state description of the waveguide in the continuous temporal basis of modes $\hat{b}(t)$ [128]. Then, if the waveguide state is pure, it can be written as $|\Psi\rangle = \sum_{n} |\psi_n\rangle$ where

$$|\psi_n\rangle = \sqrt{p_n} \int f_n(\mathbf{t}) \prod_{k=1}^n \hat{b}^{\dagger}(t_k) |0\rangle \, \mathbf{dt}$$
(2.17)

for $n \ge 1$ and $|\psi_0\rangle = \sqrt{p_0} |0\rangle$. The values $0 \le p_n \le 1$ are the photon-number probabilities of the waveguide photonic state where $\sum_n p_n = 1$. The complex temporal wavefunctions $f_n(\mathbf{t})$ describing the state of n photons are normalized so that $\int |f_n(\mathbf{t})|^2 d\mathbf{t} = 1$, where $\mathbf{t} = (t_1, \dots, t_n)$. I will assume the time-ordered convention where the integration is taken so that $t_i \le t_{i+1}$, or equivalently, over the entire real space \mathbb{R}^n provided that we restrict $f_n(\mathbf{t}) = 0$ if \mathbf{t} is not time ordered. An alternative convention would be to take f_n to be symmetric upon the exchange of two times, which differs from the time-ordered convention by a normalization factor of n!.

In general, if the source suffers from excess decoherence, the waveguide may not be in a pure

state. By extending Eq. (2.17) to the mixed-state picture, we have the waveguide density operator $\hat{\varrho} = \sum_{n,m} \hat{\varrho}_{n,m}$ where

$$\hat{\varrho}_{n,m} = \sqrt{p_n p_m} \iint \xi_{n,m}(\mathbf{t}, \mathbf{t}') \left[\prod_{k=1}^n \hat{b}^{\dagger}(t_k) \right] |0\rangle \langle 0| \left[\prod_{l=1}^m \hat{b}(t_l') \right] \mathbf{dt} \mathbf{dt'},$$
(2.18)

for $n, m \geq 1$ and

$$\hat{\varrho}_{0,0} = p_0 |0\rangle \langle 0|
\hat{\varrho}_{1,0} = \sqrt{p_0 p_1} \int \xi_{1,0}(t_1) \hat{b}^{\dagger}(t_1) |0\rangle \langle 0| dt_1$$

$$\hat{\varrho}_{n,m} = \hat{\varrho}_{m,n}^{\dagger},$$
(2.19)

where the temporal density functions $\xi_n = \xi_{n,n}$ are normalized such that $\operatorname{Tr} \left[\hat{\varrho}_{n,n}\right] = p_n$ and the coherence functions $\zeta_{n,m} = \xi_{n,m}$ $(n \neq m)$ are bounded by $|\zeta_{n,m}(\mathbf{t}, \mathbf{t}')|^2 \leq \xi_n(\mathbf{t}, \mathbf{t})\xi_m(\mathbf{t}', \mathbf{t}')$. By the Hermitian property, we have that $\xi_{n,m}^*(\mathbf{t}, \mathbf{t}') = \xi_{n,m}(\mathbf{t}', \mathbf{t})$. In addition, we assume the same time ordering convention as for the pure state description where the integration is taken over $t_i \leq t_{i+1}$ and $t'_i \leq t'_{i+1}$. We can again extend the integration over the entire \mathbb{R}^{n+m} space by restricting $\xi_{n,m}(\mathbf{t}, \mathbf{t}') = f_n(\mathbf{t})f_m^*(\mathbf{t}')$ for $n, m \geq 1$ and $\zeta_{1,0}(t) = \zeta_{0,1}^*(t) = f_1(t)$.

As a consequence of the input-output relation $\hat{b} = \sqrt{\kappa \eta_c} \hat{a} + \hat{b}_o$, we have that normally-ordered, time-ordered correlations of the waveguide mode can be written in terms of the conditional correlations of the source

$$\langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle = \kappa \eta_{\rm c} \sum_{n} \left\langle \hat{a}^{\dagger}(t')\hat{a}(t) \right\rangle_{n}, \qquad (2.20)$$

where $t' \ge t$. This naturally defines a corresponding decomposition of the waveguide correlation functions by association $\langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle_n = \kappa \eta_c \langle \hat{a}^{\dagger}(t')\hat{a}(t)\rangle_n$. If *n* indeed corresponds to the number of jumps giving rise to *n* photons in the waveguide, then we expect that $\langle \rangle_n$ is the expectation value of the waveguide substate $\hat{\varrho}_{n,n}$. A proof of this claim would require a derivation from the total system Hamiltonian [128]. However, we can show that this association is perfectly consistent with all the properties of the photonic state and other known results in the literature. To this end, let us assume that $\langle \rangle_n$ is the expectation value for the state $\hat{\varrho}$ after being projected onto the *n*-photon subspace. This assumption should rely on all the same assumptions used to arrive at the Markovian master equation and input-output relation for a vacuum input.

Using this notion, we have a clear definition of the emitted photon number probabilities in terms of the source dynamics:

$$p_{n} = \operatorname{Tr}(\hat{\varrho}_{n,n}) = \langle \hat{I} \rangle_{n}$$

$$= \lim_{t_{f} \to \infty} \operatorname{Tr}(\mathcal{U}_{n}(t_{f}, t_{0})\hat{\rho}(t_{0})), \qquad (2.21)$$

which is equivalent to the Mandel counting formula [109]. This derivation can be seen as an analytic application of quantum trajectories simulations. In this case, all of the quantum trajectories (or quantum corridors in the mixed-state picture) are buried inside the conditional propagator U_n and $t_f \rightarrow \infty$ implies we simulate for much longer than the source lifetime.

The waveguide field correlations in the photon number subspace are directly related to the temporal density functions:

$$\begin{aligned} \langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle_{0} &= 0\\ \langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle_{1} &= p_{1}\xi_{1}(t,t')\\ \langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle_{2} &= p_{2}\int \left[\xi_{2}(t,t'',t',t'') + \xi_{2}(t,t'',t'',t'') + \xi_{2}(t'',t,t'',t'')\right]dt''\\ &\vdots \end{aligned}$$
(2.22)

Where I have applied the time-ordering convention so that the term $\xi_2(t'', t, t', t'')$ does not appear. From these relations, we can then compute the single-photon temporal density function $\xi_1(t, t')$ from the master equation dynamics by

$$\xi_{1}(t,t') = \frac{\kappa \eta_{c}}{p_{1}} \left\langle \hat{a}^{\dagger}(t')\hat{a}(t) \right\rangle_{1}$$

$$= \frac{\kappa \eta_{c}}{p_{1}} \operatorname{Tr}(\mathcal{U}_{0}(\infty,t')\mathcal{R}_{0}(t',t)\mathcal{S}_{0}(t,t_{0})\hat{\rho}(t_{0}))$$
(2.23)

for $t' \ge t$ and $\xi_1(t',t) = \xi_1^*(t,t')$. In this expression, I have introduced the shorthand $\mathcal{U}_0(\infty,t) =$

 $\lim_{t_f\to\infty} \mathcal{U}_0(t_f, t)$ and the Heisenberg-like notations $\mathcal{R}_n = \mathcal{R}\mathcal{U}_n$ and $\mathcal{S}_n = \mathcal{S}\mathcal{U}_n$. The result for the single-photon temporal density function in Eq. (2.23) is also given in Ref. [109] (up to a difference in normalization convention) as derived from the full light-matter interaction for an emitter coupled to a waveguide. This justifies defining the index of the decomposition in the previous section such that n = 1 corresponds to a single jump resulting in a photon in the waveguide. We can also show that the extension to n = 2 is self-consistent with the two-photon temporal density function.

The correlation function $\langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle_2$ computed using Eq. (2.14) gives three components for when $t' \geq t$. These components correspond to the cases where conditional evolution \mathcal{U}_1 occurs before t, in between t and t', and after t', respectively. Expanding \mathcal{U}_1 in terms of \mathcal{U}_0 and $\mathcal{J} = \kappa \eta_c S \mathcal{R}$ gives

$$\langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle_{2} = \kappa\eta_{c}\int_{t'}^{\infty} \operatorname{Tr}(\mathcal{U}_{0}(\infty,t'')\mathcal{J}_{0}(t'',t')\mathcal{R}_{0}(t',t)\mathcal{S}_{0}(t,t_{0})\hat{\rho}(t_{0})) dt'' + \kappa\eta_{c}\int_{t}^{t'} \operatorname{Tr}(\mathcal{U}_{0}(\infty,t')\mathcal{R}_{0}(t',t'')\mathcal{J}_{0}(t'',t)\mathcal{S}_{0}(t,t_{0})\hat{\rho}(t_{0})) dt'' + \kappa\eta_{c}\int_{t_{0}}^{t} \operatorname{Tr}(\mathcal{U}_{0}(\infty,t')\mathcal{R}_{0}(t',t)\mathcal{S}_{0}(t,t'')\mathcal{J}_{0}(t'',t_{0})\hat{\rho}(t_{0})) dt'',$$

$$(2.24)$$

where $\mathcal{J}_0(t',t) = \mathcal{J}\mathcal{U}_0(t',t)$. In this form, we can identify the similarity between this expression written in terms of the source dynamics and the expression in terms of temporal density functions in Eq. (2.22). When restricting $t' \ge t$, we can notice that indeed $\xi_2(t, t'', t', t'')$ is nonzero only for $t' \le t''$, $\xi_2(t, t'', t'', t')$ is nonzero only for $t \le t'' \le t'$, and $\xi_2(t'', t, t'', t')$ is nonzero only for $t'' \le t$. These three components have clear physical interpretations. Given that two photons were emitted between t_0 and ∞ , the first term describes the temporal coherence within the first photon emitted, the second term describes the temporal entanglement between the two photons, and the last term describes the temporal coherence within the second photon. The integrands of Eq. (2.24) hint at the form to solve for the full two-photon temporal density function ξ_2 in terms of the source dynamics. We can solve for ξ_2 using the second-order conditional correlation for two photons and then show that it is indeed consistent with Eq. (2.24). The second-order four-time amplitude correlation for the two-photon component is

$$\langle \hat{b}^{\dagger}(t_1')\hat{b}^{\dagger}(t_2')\hat{b}(t_2)\hat{b}(t_1)\rangle_2 = p_2\xi_2(t_1, t_2, t_1', t_2'), \qquad (2.25)$$

where I have applied the time-ordered convention where $\xi_2(\mathbf{t}, \mathbf{t}') = 0$ if $\mathbf{t} = (t_1, t_2)$ or $\mathbf{t}' = (t'_1, t'_2)$ is not time ordered. In terms of the source dynamics, we can expand the total two-photon temporal density function into

$$\xi_{2}(t_{1}, t_{2}, t_{1}', t_{2}') = \frac{\kappa^{2} \eta_{c}^{2}}{p_{2}} \begin{cases} \operatorname{Tr}(\mathcal{U}_{0}(\infty, t_{2}')\mathcal{R}_{0}(t_{2}', t_{2})\mathcal{S}_{0}(t_{2}, t_{1}')\mathcal{R}_{0}(t_{1}', t_{1})\mathcal{S}_{0}(t_{1}, t_{0})\hat{\rho}(t_{0})) & t_{1} \leq t_{1} \leq t_{2} \leq t_{2}' \\ \operatorname{Tr}(\mathcal{U}_{0}(\infty, t_{2}')\mathcal{R}_{0}(t_{2}', t_{2})\mathcal{S}_{0}(t_{2}, t_{1})\mathcal{S}_{0}(t_{1}, t_{1}')\mathcal{R}_{0}(t_{1}', t_{0})\hat{\rho}(t_{0})) & t_{1}' \leq t_{1} \leq t_{2} \leq t_{2}' \\ \operatorname{Tr}(\mathcal{U}_{0}(\infty, t_{2}')\mathcal{R}_{0}(t_{2}', t_{1}')\mathcal{R}_{0}(t_{1}', t_{2})\mathcal{S}_{0}(t_{2}, t_{1})\mathcal{S}_{0}(t_{1}, t_{0})\hat{\rho}(t_{0})) & t_{1} \leq t_{2} \leq t_{1}' \leq t_{2}' \end{cases}$$

$$(2.26)$$

and $\xi_2(t'_1, t'_2, t_1, t_2) = \xi_2^*(t_1, t_2, t'_1, t'_2)$. In its general form, we can now see that the expression for the case $t_1 \leq t'_1 \leq t_2 \leq t'_2$ applied to $\langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle_2$ in Eq. (2.22) indeed recovers Eq. (2.24). This consistency then justifies defining n = 2 as the lowest-order nonzero term for correlations of quantities involving two jumps.

It may be possible to extend this method to compute the number coherence functions $\zeta_{n,m}$ by looking at correlations such as $\langle \hat{b}(t) \rangle_n \propto \zeta_{n+1,n}(t)$ and $\langle \hat{b}(t_2) \hat{b}(t_1) \rangle_n \propto \zeta_{n+2,n}(t_1, t_2)$ under the same decomposition. However, the association is not as intuitive because of our choice to decompose the master equation in terms of the photon number in the waveguide. To my knowledge, unlike ξ_1 , there are no results in the literature with which to compare such an approach when including excess decoherence, such as pure dephasing. Alternatively, we could unravel the master equation in the context of a physical phase measurement setup, such as a homodyne measurement [91]. As I explore in section 3.3, it is also possible to obtain some information about number coherence using a self-homodyne setup, where two copies of the same photonic state are interfered in exactly the same spirit as the standard indistinguishability measurements discussed in section 1.4.4.

2.5 Relation to photon statistics

Using the photon number decomposition, we can derive the well-known photon statistics relations for brightness μ and $g^{(2)}$ from the perspective of the source rather than the field. In addition to providing these relations for use later in this thesis, it will also help show that the photon number decomposition is consistent with the properties expected of the emitted photonic state.

To derive these relations, we will need to use an important property of the conditional propagation superoperators. That is, that the propagator does not depend on the order of jumps because all jumps are identical:

$$\int_{t_0}^{t_f} \mathcal{U}_k(t_f, t) \mathcal{J} \mathcal{U}_l(t, t_0) dt = \mathcal{U}_{k+l+1}(t_f, t_0).$$
(2.27)

I provide a detailed proof of this property in appendix A.3 for the sake of interest, because I was not able to find it in the literature, although I imagine this is a known result in perturbation or measurement theory. It is also a very physically intuitive result because we would expect that the dynamics of the reduced system is unchanged when exchanging the order of any two jumps.

The integrated brightness of a pulsed source is defined in section 1.4.2 as $\mu = \int N(t)dt$ where $N(t) = \langle \hat{b}^{\dagger}(t)\hat{b}(t)\rangle$. Using the invariance of order property and the input-output relations for a vacuum input, we can see that

$$\mu = \kappa \eta_{c} \lim_{t_{f} \to \infty} \int_{t_{0}}^{t_{f}} \langle \hat{a}^{\dagger}(t) \hat{a}(t) \rangle dt$$

$$= \sum_{n,k,l} \lim_{t_{f} \to \infty} \operatorname{Tr} \left(\int_{t_{0}}^{t_{f}} \mathcal{U}_{k}(t_{f},t) \mathcal{J} \mathcal{U}_{l}(t,t_{0}) dt \hat{\rho}(t_{0}) \right) \delta_{k+l+1,n}$$

$$= \sum_{n} n \lim_{t_{f} \to \infty} \operatorname{Tr} (\hat{\rho}_{n}(t_{f})) .$$
(2.28)

Finally, knowing that $p_n = \lim_{t_f \to \infty} \text{Tr}(\hat{\rho}_n(t_f))$, we recover the expected result for the average photon number of the pulsed photonic state: $\mu = \sum_n np_n$.

The pulsed $g^{(2)}$ can be computed in a similar fashion:

$$\frac{\mu^2}{2}g^{(2)} = \kappa^2 \eta_c^2 \lim_{t_f \to \infty} \int_{t_0}^{t_f} \int_t^{t_f} \langle \hat{a}^{\dagger}(t) \hat{a}^{\dagger}(t') \hat{a}(t') \hat{a}(t) \rangle dt' dt$$

$$= \sum_{k,l,m,n} \lim_{t_f \to \infty} \operatorname{Tr} \left(\int_{t_0}^{t_f} \left(\int_t^{t_f} \mathcal{U}_k(t_f,t') \mathcal{J} \mathcal{U}_l(t',t) dt' \right) \mathcal{J} \mathcal{U}_m(t,t_0) dt \hat{\rho}(t_0) \right) \delta_{n,k+l+m+2}$$

$$= \sum_n \frac{n(n-1)}{2} \lim_{t_f \to \infty} \operatorname{Tr} (\hat{\rho}_n(t_0)) . \qquad (2.29)$$

Hence, we obtain the well-known [128] expression for the pulsed integrated intensity correlation

$$g^{(2)} = \frac{\sum_{n} n(n-1)p_n}{\left(\sum_{n} np_n\right)^2}$$
(2.30)

as expected. These results confirm that the conditional expectation $\langle \rangle_n$ correctly preserves the total photon number and that it is capable of reconstructing the photonic state within photon number subspaces that are consistent with both the photon statistics and correlations predicted by the master equation. Although I will not demonstrate this, we could generalize this derivation for $g^{(m)}$ where $m \geq 2$ to confirm that

$$g^{(m)} = \frac{m!}{\mu^m} \sum_n \binom{n}{m} p_n.$$
(2.31)

To summarize, the results presented in this chapter are direct consequences of assuming that the source is modeled by a Markovian master equation that satisfies an effective proportionality $\hat{b} \propto \hat{a}$ with the waveguide mode. Physically, we can imagine that these two assumptions together imply that the state of the waveguide is a chronicle of the state of the source over time. By treating the waveguide as a quantum probe for the source state (section 2.2), we can find the state of the source conditioned on the number of emitted or detected photons. Conversely, by analyzing the conditional correlations of the source using this same indirect measurement (section 2.3), we can also reconstruct the corresponding state of the waveguide (section 2.4) and describe the resulting photon counting statistics in terms of the source dynamics (section 2.5).

Chapter 3

Single photons

Single photons are one of the prime physical systems used to demonstrate quantum phenomena theoretically, experimentally, and pedagogically. Their presence is practically ubiquitous in quantum physics. Understanding their properties and behaviour is key to developing fundamental and technological advancements. Single photons can be used to demonstrate particle anti-bunching, single-particle interference, and the Hong-Ou-Mandel (HOM) effect [107]. As discussed in section 1.1.2, they are key ingredients for quantum communication and quantum sensing. Many proposals for universal quantum computers also rely on photonic states built from single photons [137, 138].

In this chapter, I explore three main topics related to single photons. In section 3.1, I discuss the feasibility of developing a deterministic solid-state source of indistinguishable single photons that can operate at room temperature. Section 3.2 then expands on section 1.4.4 by going into more detail regarding HOM interference, which is the standard method for quantifying single-photon indistinguishability. Finally, in section 3.3, I demonstrate how the photon number decomposition can be used to explore the time dynamics of pulsed photonic states emitted by a purely-dephased two-level system.

The content of section 3.1, with the exception of subsection 3.1.5, is published in Ref. [iii]. This content is presented in its published form with only minor changes made for clarification and consistency. Because this paper was published three years before this thesis, I provide a brief

preface after this chapter outline to give an updated context.

Section 3.2 is based on projects that were completed in collaboration with the group of Prof. Pascale Senellart. The analysis of the effect of multi-photon events on indistinguishability is published in the experimental paper given by Ref. [xiii]. This project also spurred the unpublished content of subsection 3.2.2, which explores the same problem for a specific emitter model. Subsection 3.2.3 is also unpublished material that was inspired by Refs. [xiii] and [xiv]. Some of these results may be published in combination with an ongoing project related to photon number coherence with the Senellart group.

The final section 3.3 of this chapter deals more explicitly with the time dynamics and properties of an emitted photonic state from a pulsed two-level system that may experience pure dephasing. Although I focus on pure dephasing, the methods can be applied to other Markovian master equation models with excess decoherence, at least numerically if not analytically. The majority of the content in this section is not published, although it is related to Refs. [xiii] and [xiv]. Some of the results in subsection 3.3.3 do appear in the supplementary of Ref. [xiv].

Preface for Ref. [iii].—The primary goal of the project presented in section 3.1 was to motivate the plasmonics community to focus on developing coherent light-matter interactions for use in quantum information processing applications. However, although this paper has seen success in terms of citations since its publication in 2018 and the topic continues to be relevant [139], the model used in this paper is outdated and neglects some important effects. I will now outline these issues in the following three paragraphs.

First, the QNM master equation, developed by Ref. [103] and discussed in section 1.3.7, supersedes the simpler dissipative Jaynes-Cummings model and multi-pole Markovian quenching model [140] that is used in section 3.1. The latter model is only valid for specific cavity geometries. Instead, the QNM approach generally describes a dissipative cavity in terms of modes that each account for both radiative and non-radiative components, giving rise to inter-dependent coupling strengths and dissipative rates.

Second, section 3.1 only explores limitations due to photon emission, and hence makes a crucial assumption that the emitter is excited on a timescale faster than the Purcell-enhanced lifetime, which itself must be sufficiently short at room temperature to overcome dephasing. Otherwise, detrimental re-excitation of the emitter can degrade the emission single-photon purity and HOM visibility. For QDs, this amounts to requiring pulses on the femtosecond timescale [105], which can then cause additional problems such as reducing the excitation pulse filtering efficiency [xiii] or incidentally exciting QD higher-energy biexciton states. However, as we mention in Ref. [iii], a QD is unlikely to ever be successful as a room temperature single-photon source even using fast pulses due to having a very broad emission zero-phonon line and phonon sideband at room temperature, which fundamentally limits indistinguishability even with a Purcell enhancement [141, 142].

Third, for solid-state emitters, excitation pulses can cause additional power-dependent dephasing due to phonon interactions [85], which is not captured by the constant phenomenological pure dephasing model used in this thesis. In addition, non-Markovian behaviour becomes important for pulses on the order of the phonon bath memory timescale. This effect has been well-studied for quantum dots [67], and the lower bound is around 5-10 ps at low temperature. However, to my knowledge, this effect has not been thoroughly investigated for atomic-scale defects such as rare-earth ions and color centers in diamond. These latter systems generally have longer lifetimes and more spectrally-separated phonon sidebands than quantum dots. Hence, some of these systems could remain in a Markovian regime even under ultrafast optical excitation. As a motivating case, ultrafast coherent optical control using pulses as short as 1 ps has been demonstrated with negatively-charged silicon vacancy centers in diamond [143], albeit at low temperature. Moreover, ultrafast excitation that is faster than the bath memory time can decouple the phonon bath, allowing for efficient excitation that should be quite robust even at room temperature [144, 145].

Given its shortcomings, I still believe that Ref. [iii] is an insightful study that contains valid conclusions. First, it provides a good analytic approach for estimating the emission-limited indistinguishability for a cavity-emitter system in the critical regime that includes a correction due to unfiltered phonon sideband emission. Second, we find that room temperature solid-state indistinguishable single photon sources should still be a feasible goal for the plasmonics community, but that accomplishing this goal will necessarily require emitters more narrow than QDs or NV centers. Third, plasmonic antennas or cavities with low Q factors cannot filter the large phonon sidebands of solid-state emitters at room temperature. Therefore, a mid-Q hybrid plasmonic-dielectric cavity combined with a relatively narrow emitter and ultrafast excitation is the most promising route.

3.1 Single-photon sources at room temperature

While there is substantial excitement about quantum technology and quantum information processing (QIP), many practical applications are still held back by the fact that critical components are restricted to operating at cryogenic temperatures. Trying to overcome the thermal restrictions of quantum devices also tests fundamental questions about the physical regimes in which quantum processes can exist and be manipulated.

Indistinguishable single-photon sources (SPSs) are basic components of numerous different optical QIP implementations. As discussed earlier in this thesis, they are required for tasks such as linear-optical quantum computing [138] and boson sampling [146, 147]. In addition, an efficient indistinguishable photon source can be used to construct quantum repeaters and would assist in the development of a quantum internet [21, 31, 32, 148].

The most common way to generate indistinguishable photons at room temperature is by heralded spontaneous parametric down-conversion (SPDC) [16]. This technique has seen pioneering success in quantum research, yet its probabilistic nature limits its range of applications, although this limitation could in principle be addressed by multiplexing many SPDC sources [149, 150].

In contrast, individual quantum emitters not only promise near-deterministic single-photon emission at room temperature, but moreover the emitted photon could be entangled with coherent solid-state spins [70]. This would allow many QIP applications to be performed at room temperature, including optically-mediated entanglement of distant spins in solids [151].

Low-temperature indistinguishable SPSs have been achieved [152] and are becoming more efficient [68, 153]. Ref. [86] showed that relatively inefficient indistinguishable SPSs can be realized beyond the low-temperature regime using weakly-coupled narrow-bandwidth micro-cavities. These results were applied to extend quantum dot indistinguishable SPS operation up to 20 K [142]. Distinguishable SPSs have also been demonstrated at room temperature using solid-state quantum emitters [154]. In particular, recently, plasmonic cavities have been proposed to enhance emission rates for distinguishable SPSs in spite of high losses [155, 156].

Achieving indistinguishable photon emission from a solid-state emitter is a difficult task, especially at higher temperatures. The main problem is that optical transitions in solid-state materials experience rapid phonon-induced dephasing that homogeneously broadens the zero-phonon line (ZPL) at room temperature [157, 158]. This dephasing reduces the degree of indistinguishability between emitted photons [86, 92]. In addition, phonon-assisted optical transitions can produce a phonon sideband (PSB) that reduces indistinguishability and consequently must be filtered, which sacrifices efficiency. If the PSB spectrum overlaps with the desired ZPL emission, it cannot be entirely filtered and hence fundamentally limits indistinguishability [72, 141]. Furthermore, when using plasmonic materials to enhance emission, plasmon-induced quenching poses another detrimental effect that affects both efficiency and indistinguishability. A very recent analysis on a single spherical metallic nanoparticle with the goal of producing an on-chip room temperature single-photon source [140] found that simultaneous high efficiency and high indistinguishability are difficult to achieve with integrated plasmonics.

Here we show that achieving high efficiency and high indistinguishability simultaneously at room temperature should be possible by using ultrasmall mode volume cavities and by operating in the critical regime (see section 1.3.5). We argue that such systems are within reach of current technology, and we provide theoretical guidelines for designing successful plasmonic cavities for this purpose.

A common approach to improving indistinguishability and efficiency simultaneously is to place a quantum emitter inside a cavity. In addition to suppressing off-resonant PSB emission, the cavity reduces the emitter lifetime through the Purcell effect [95] and allows the photon to be emitted before the emitter coherence is destroyed by interactions with the phonon bath. At room temperature, this fast emitter dephasing is very difficult to overcome, requiring Purcell factors exceeding 10^4 for most emitters [86]. Obtaining a large Purcell factor can be accomplished by either increasing the cavity quality factor or by decreasing the effective mode volume. However, for highly dissipative emitters, increasing the quality factor too high prolongs the interaction between the cavity photon and the emitter, which causes emitter dephasing. As a result, it is necessary to use cavities with mode volumes far below the diffraction limit.

Such ultrasmall mode volume cavities have seen significant development over the last decade and, in particular, the last few years [159–169]. Many of these cavities utilize plasmonic materials to concentrate the electromagnetic field along a material interface in the form of plasmon-polaritons [160–163, 166–169]. There are also interesting proposals for pure-dielectric ultrasmall cavities [159, 164, 165] and plasmonic-Fabry-Pérot hybrid cavities [166, 169]. However, the application of ultrasmall mode volume cavities to quantum information processing is still relatively unexplored.

The interaction between a quantum emitter and the phonon bath will broaden the emitter's ZPL and could also create a PSB. Recently, the effect of the PSB on the indistinguishability of single-photon sources was studied [72, 141]. It was found that the indistinguishability is limited by the fraction of PSB not filtered by the cavity. This limitation can be highly detrimental at room temperature where the ZPL is very broad, which necessitates a broad cavity that might not filter the PSB. Therefore, an ideal emitter for an efficient room-temperature indistinguishable SPS should have a small PSB that is spectrally well-separated from its ZPL. For such an emitter, pure dephasing and possible quenching effects will be the primary limitation. Consequently, we first consider a Markovian system that neglects non-Markovian PSB effects in order to describe the most ideal parameter regime for a good (Markovian) emitter. Then we estimate the correction induced by a non-zero PSB using the results of Iles-Smith *et al.* [141].

3.1.1 Indistinguishability in the critical regime

We now describe our proposed approach in detail. We begin with the interaction Hamiltonian for a driven two-level system coupled to a resonant cavity $\hat{H} = \hbar g(\hat{\sigma} \hat{a}^{\dagger} + \hat{\sigma}^{\dagger} \hat{a}) + (\hbar \Omega/2)(\hat{\sigma} + \hat{\sigma}^{\dagger})$. The driving term of the Hamiltonian significantly complicates the derivation for indistinguishability. However, the single-photon purity of emission from a two-level system depends on how the system is excited. Slow excitation allows for multi-photon emission, which increases $g^{(2)}$. For a high single-photon purity, excitation of the emitter will require ultrafast optical control, which has been demonstrated for defects in diamond [143, 170]. The common practice, which we also adopt, is to assume $\Omega \gg g$ during excitation so that the system is effectively instantaneously prepared in the excited state [86, 140], implying $g^{(2)} \simeq 0$. We then explore the dominant emission dynamics as governed by the system when $\Omega = 0$. This allows us to make use of the single-excitation approximation and decouple the optical Bloch equations (see section 1.3.4). However, for large Purcell factors, the excitation timescale must still be shorter than the Purcell-enhanced lifetime.

We describe the dissipative dynamics of the system using the Markovian master equation governed by the Liouville superoperator $\mathcal{L} = -(i/\hbar)\mathcal{H} + \kappa \mathcal{D}(\hat{a}) + \gamma^* \mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma}) + \gamma \mathcal{D}(\hat{\sigma})$. Please note that the definition of γ^* differs by a factor of 2 in this entire section 3.1 and appendix B compared to the rest of this thesis. This choice follows the convention in [86] so that $\Gamma = \gamma + \gamma^*$ is the FWHM of the ZPL rather than $\gamma + 2\gamma^*$. We take the cavity linewidth $\kappa = \kappa_r + \kappa_{nr}$ to have a radiative part κ_r and non-radiative part κ_{nr} . The cavity quality factor is $Q = \omega/\kappa$ and we define the cavity quantum efficiency as $\eta_c = \kappa_r/\kappa$. Similarly, the natural decay rate $\gamma = \gamma_r + \gamma_{nr}$ also has a radiative part γ_r and a non-radiative part γ_{nr} .

We can now state the requirements for high efficiency and indistinguishability. As discussed in section 1.3.5, the rate of population transfer from the emitter to the cavity must exceed the total emitter decoherence rate $R > \Gamma$. In addition, the photon must escape the cavity before it is dephased by the emitter: $\kappa > \Gamma$. In this section, we define $R = 4g^2/\kappa$, which is consistent with the more general definition given in section 1.3.4 only in the critical regime where $\Gamma \ll \kappa$ and when the cavity and emitter are resonant so that $\Delta = 0$. In this way, R is directly related to the uninhibited Purcell factor by $F_{\rm p} = R/\gamma_{\rm r}$.

To quantify the efficiency of the system, we use the cavity efficiency $\mu = \eta_c \beta$ where $\beta = \kappa \int_0^\infty \langle \hat{a}^{\dagger}(t) \hat{a}(t) \rangle dt$ [86]. The derivation is given in appendix **B** by inverting the optical bloch equations, and the result is $\beta = R\kappa/(R(\gamma + \kappa) + \gamma(\kappa + \Gamma))$. In our analysis, we only explicitly compute β —the efficiency of population dissipated through the cavity mode—hereafter referred to as the intrinsic cavity efficiency. The total collected radiative quantum efficiency is given by $\mu = \beta \eta_c$. The efficiency of an SPS in the presence of plasmonic materials has been studied [155, 156]. However, for many quantum information applications, the efficiency is not the only relevant quantity, but the indistinguishability of the emitted photons is also essential.

The quantity that we use for indistinguishability is derived from the probability that two photons emitted from the same source interfere and bunch at a beamsplitter [86, 92] (see also section 1.4.4). An ideal indistinguishable SPS should have near-unity indistinguishability I and efficiency β . To this end, we focus on maximizing the indistinguishability-efficiency product $I\beta$.

Using the quantum regression theorem [171], we derived an expression for indistinguishability valid in the critical regime for arbitrary γ . As we will show, this allows the expression to capture possible effects of plasmon quenching. The details of the derivation and the full solution valid for arbitrary γ are given in appendix **B**. Here we only write the expression in the case that $\gamma < \gamma^* < \kappa$, which arises when quenching is weak:

$$I = \frac{R^2 \kappa^2 \left(1 + I_1\right)}{(R+\gamma)(\kappa+\gamma)(R+\Gamma)(\kappa+\Gamma)\beta^2},\tag{3.1}$$

where $I_1 = (\gamma^*/\kappa)(6\kappa - R)/(3\kappa + 4R)$. This expression is accurate in the critical regime to first order in γ^*/κ .

When there are no plasmon quenching effects and when the system is far within the critical regime boundaries, we have that $\gamma \simeq 0$ and $I_1 \simeq 0$; hence, $I = R\kappa(R + \gamma^*)^{-1}(\kappa + \gamma^*)^{-1}$ and $\beta = 1$ for perfect excitation. From this, it can be seen that $I\beta$ is maximized when $R = \kappa$ for a given


Figure 3.1: Product of indistinguishability I and intrinsic cavity efficiency β for an emitter-cavity system in the critical regime with cavity coupling g, emitter dephasing rate $\gamma^* \simeq \Gamma$, and cavity linewidth κ with non-radiative portion $\kappa_{nr} = (1 - \eta_c)\kappa$ where η_c is the bare cavity radiative efficiency. (a) The case without quenching (γ_q =0). $I\beta$ plotted along with the critical regime boundaries (blue solid lines) and the boundary between the strong-coupling and bad-cavity regimes, $R = \kappa$ (black dotted line). (b) The case with quenching ($\gamma_q \neq 0$). $I\beta$ for a simple one-mode example, $\gamma_q = g_2^2 \kappa_{nr}/(\Delta_2^2 + (\kappa_{nr}/2)^2)$, to illustrate the different regimes and boundaries. Here we use $g_2 = g/2$, $\Delta_2 = 30\gamma^*$, and $\eta_c = 0.5$, corresponding to $\Delta_q = 60\gamma^*$. The maximum of $I\beta = 0.92$ occurs at the intersection between $\gamma^* = 4\gamma_q$ and $R = \kappa$ (white star). The regime above the $\kappa = \gamma_q$ line is dominated by quenching. The vertical dashed line divides the mode-detuned case to the left and the nearly resonant case to the right.

cavity coupling rate g (see Fig. 3.1 (a)). This implies $2g = \kappa$, which is also the strong-coupling boundary in the limit that $\kappa \gg \Gamma$.

3.1.2 A Markovian quenching model

For a plasmonic cavity, g is the coupling rate to the dominant radiating mode; however, the emitter will also couple to higher-order modes [140, 156, 172]. These higher-order modes contribute to the enhanced decay rate of the emitter but are predominantly non-radiative and hence quench the emission.

We treat the coupling of the emitter to higher-order modes using the Markovian approximation [140]. In this case, the emitter decay rate γ is increased by the quenching rate γ_q so that γ becomes $\gamma = \gamma_r + \gamma_{nr} + \gamma_q$. The quenching contribution γ_q can be described by the plasmon spectral density,

which is approximated by a sum of Lorentzian functions in the quasistatic limit [172]:

$$\gamma_{\mathbf{q}} = \sum_{l=2}^{\infty} \frac{g_l^2 \kappa_{\mathbf{n}\mathbf{r}}}{\Delta_l^2 + (\kappa_{\mathbf{n}\mathbf{r}}/2)^2},\tag{3.2}$$

where $\Delta_l > 0$ is the detuning of the respective mode from the emitter, $g_l = k_l g$ is the coupling rate where k_l is approximately independent of g, and we assume that each mode has the same non-radiative rate $\kappa_{nr} = \kappa(1 - \eta_c)$ as the dominant mode [140]. This Markovian approximation is justified when the coupling rate to each individual higher-order mode is not too strong—when $g_l^2/(\Delta_l^2 + (\kappa_{nr}/2)^2) < 1$. Since we only require the system to achieve the optimal relation $R = \kappa$ in the critical regime, the system is only on the brink of strong coupling with the resonant dominant mode. Hence, we expect that the detuned higher-order modes should not display any significant strong coupling. In the critical regime, the quenching rate dominates $\gamma_q > \gamma_r + \gamma_{nr}$ so that $\gamma \simeq \gamma_q$.

To attain large $I\beta$, the dissipation through the cavity must be faster than the quenching rate, implying: $4g^2/\kappa > \gamma_q$ and $\kappa > \gamma_q$. There are two upper-bound cases to consider for γ_q . In the limit that the higher-order modes are near-resonant with the emitter $(4\Delta_l^2 < \kappa_{nr}^2)$, we have $\gamma_q \simeq (4g^2/\kappa_{nr}) \sum_{l=2}^{\infty} k_l^2$. Then $4g^2/\kappa > \gamma_q$ implies $\sum_{l=2}^{\infty} k_l^2 < (1 - \eta_c)$. This condition opposes high radiative efficiency and it is very difficult to satisfy when there are many modes, making it unsuitable. On the other hand, if the modes are detuned from the emitter $(4\Delta_l^2 > \kappa_{nr}^2)$, we have $\gamma_q \simeq g^2 \kappa_{nr}/\Delta_q^2$ where we define $1/\Delta_q^2 = \sum_{l=2}^{\infty} k_l^2/\Delta_l^2$ for simplicity. In this case, $4g^2/\kappa, \kappa > \gamma_q$ implies that we require $\kappa, 2g < 2\Delta_q(1 - \eta_c)^{-1/2}$ to achieve large $I\beta$. Applying the condition $\kappa > \gamma^*$ that is required to reach the critical regime, we find the main condition: $\gamma^*(1 - \eta_c)^{1/2} < 2\Delta_q$.

The quantity Δ_q can be seen as an effective detuning parameter that describes the severity of quenching for a plasmonic-emitter system. The value of Δ_q depends on the geometry of the cavity and the position of the emitter relative to the cavity. For example, Δ_q for a single spherical metallic nanoparticle is dependent on the ratio d/r of the distance between the emitter and the particle surface d and particle radius r [172]. For a silver sphere, Δ_q can range from $2\pi \times 6$ THz for d/r = 0.05 to $2\pi \times 117$ THz for d/r = 2. In this example, the limit $\gamma_q \simeq g^2 \kappa_{nr}/\Delta_q^2$ is also a good approximation when Q > 5.

In the absence of quenching, $I\beta$ can be increased arbitrarily by increasing both g and κ while following $R = \kappa$ to maximize the Purcell enhancement (see Fig. 3.1 (a)). However, in the presence of quenching, $I\beta$ decreases when κ , $2g > 2\Delta_q(1 - \eta_c)^{-1/2}$. This restricts $I\beta$ to a maximum value for a given Δ_q and γ^* . We analytically maximized $I\beta$ in the mode-detuned case for small γ^*/Δ_q . The values of g and κ that maximize $I\beta$ were found to be $\kappa_{max} \simeq 2g_{max} \simeq [\Delta_q^2 \gamma^*/(1 - \eta_c)]^{1/3}$. From this solution, we see that decreasing γ^*/Δ_q or increasing η_c will increase the maximum possible $I\beta$. We also notice that, at the $I\beta$ maximum, $\gamma_q \simeq \gamma^*/4$ is independent of Δ_q and η_c . Hence, the maximum occurs roughly at the intersection between $2g = \kappa$ and $4\gamma_q = \gamma^*$, see Fig. 3.1 (b).

3.1.3 Phonon sideband corrections

As we have shown, decreasing γ^*/Δ_q can increase the maximum attainable $I\beta$. This heavily favors narrow linewidth emitters. An emitter with a smaller γ^*/γ_r ratio is also advantageous because the Purcell factor required to reach the maximum (at $2g = \kappa$ and $\gamma_q = 4\gamma^*$) is not as large. Furthermore, a good emitter for this application must have a small PSB that has a small overlap with the cavity spectrum [72, 141]. This is necessary because any photons emitted through the PSB into the cavity will decrease the indistinguishability and any photons emitted from the PSB directly will decrease the intrinsic cavity efficiency.

The reduction of indistinguishability and intrinsic cavity efficiency due to PSB non-Markovian effects can be approximated in the cavity-emitter weak-coupling regime by [141]:

$$I = I_0 \left[\frac{B^2}{B^2 + F(1 - B^2)} \right]^2$$
(3.3)

and

$$\beta = \beta_0 \frac{B^2 + F(1 - B^2)}{1 - \beta_0 (1 - F)(1 - B^2)},$$
(3.4)

where I_0 and β_0 are the values computed in the Markovian approximation; *B* is the Franck-Condon factor, and *F* is the fraction of the PSB not filtered by the cavity, which depends on the cavity

linewidth κ and the exact PSB spectrum for an emitter. In writing Eq. (3.3), we also assume that the additional interaction between the PSB and ZPL via the cavity does not significantly alter the phonon-induced pure-dephasing rate γ^* , which is a reasonable assumption in the regime of interest and at room temperature where γ^* is already quite large (see appendix B for a more detailed discussion).

In general, F increases with cavity linewidth. In the broad-cavity limit where no PSB is filtered $(F \rightarrow 1)$, the $I\beta$ product is limited to $I\beta = I_0\beta_0B^4$, where B^4 can be interpreted as the probability that both of the photons being interfered at a beamsplitter were emitted from the ZPL.

3.1.4 Proposal for silicon vacancy centers

The above restrictions on ZPL and PSB do not favor popular emitters such as quantum dots (QDs) and nitrogen-vacancy (NV⁻) centers in diamond, both of which have significant PSBs at room temperature and generally broad ZPLs. However, a promising candidate is the negatively-charged silicon-vacancy (SiV⁻) center in diamond. A nanodiamond SiV⁻ center can have γ^* as small as $2\pi \times 380$ GHz [173] and $1/\gamma_r \simeq 8.3$ ns $(1/\gamma = 0.58$ ns with $\gamma_r/\gamma = 0.07$) [174]. They can also have a very small PSB, with a ZPL emission proportion (Debye-Waller factor) up to $B^2 = DW = 0.88$ [173]. The SiV⁻ PSB is also spectrally well-separated from the ZPL, allowing most of the PSB emission to be filtered by the cavity.

For a nanodiamond SiV⁻ center with DW = 0.88, the PSB corrections from the previous section implies $I\beta \simeq 0.77I_0\beta_0$ whereas an NV⁻ center with DW $\simeq 0.03$ [175] implies $I\beta \simeq 0.0009I_0\beta_0$. A cavity can improve this by allowing F < 1, provided that the PSB is spectrally well-separated from the cavity resonance.

A promising plasmonic cavity design for room-temperature applications is the plasmonic bowtie antenna, which was used to demonstrate vacuum Rabi splitting with single QDs at room temperature [167]. Unfortunately, the close proximity of the emitter to the bowtie structure makes it difficult to achieve a large Δ_q and η_c , causing the system to be dominated by quenching. Moreover, a bowtie antenna has a very broad resonance ($Q \simeq 7$), which is a poor filter for any PSB emission. These problems can be solved by placing the bowtie inside a detuned Fabry-Pérot cavity [166]. This hybrid approach promises to alleviate quenching effects, improve cavity quantum efficiency, and increase the cavity quality factor. A simulated Fabry-Pérot-bowtie hybrid cavity shows a Purcell factor (as determined from local density of states (LDOS) enhancement) up to $R/\gamma_r = 2.7 \times 10^5$ with a Q as high as 10^3 and near-unity radiative efficiency ($\beta \eta_c \simeq 0.95$). In addition, high collection efficiencies should be possible with these systems, e.g. 81%[166]. Furthermore, by tuning the parameters of a hybrid cavity, it should be possible to optimize the system to produce a large $I\beta$ using the guidelines derived above.

To estimate the $I_0\beta_0\eta_c$ achievable using a nanodiamond SiV⁻ center inside a hybrid cavity, we used the spectrum of sample 5 from Neu *et al.* [173] along with the predictions for cavity properties from Gurlek *et al.* [166]. For the emitter, we used resonance frequency $\omega = 2\pi \times 405$ THz, $1/\gamma_r = 8.3$ ns [174], and $\gamma^* = 2\pi \times 500$ GHz [173]. For the cavity, we used $R/\gamma_r = 2.7 \times 10^5$, Q = 60, and $\beta_0\eta_c = 0.95$. With these parameters, the Markovian approximation gives $I_0\beta_0\eta_c =$ 0.86 ($I_0 = 0.90$, $\beta_0 = 0.98$) for $\Delta_q = 2\pi \times 5$ THz.

To approximate the effect of the PSB, we estimated the PSB spectrum using a sum of Lorentzian functions to match the measured spectrum and Debye-Waller factor $B^2 = DW = 0.88$ of sample 5 [173] (see appendix B for the expression). For a cavity quality factor of Q = 60, the fraction of PSB not filtered by the cavity is F = 0.15. This leads to a correction of $I \simeq 0.96I_0$ and $\beta \simeq 0.997\beta_0$. Hence the estimation becomes $I\beta\eta_c = 0.83$ (I = 0.87, $\beta = 0.97$). This correction is accurate in the weak-coupling regime $2g < \kappa$ [141]. For the parameters used in this estimation we have $2g/\kappa \simeq 0.8$. See Fig. 3.2 for estimations using other Q and Δ_q values.

For applications in quantum information processing, it is necessary to have near-unity indistinguishability for high fidelity of quantum operations. This can be accomplished by spectrally filtering the broad emission from the hybrid cavity, which sacrifices efficiency. For an approximately-Markovian source, the $I\beta$ product after spectral filtering cannot exceed the $I\beta$ product from the source [86]; hence, at best it remains constant. By assuming an outcoupling of 0.81 [166] and



Figure 3.2: Indistinguishability and efficiency estimations for a nanodiamond negatively-charged silicon vacancy (SiV⁻) center enhanced by an ultrasmall mode volume cavity when taking into account the effects of quenching and a non-zero phonon sideband (PSB). The effect of the SiV⁻ PSB on indistinguishability I and intrinsic cavity efficiency β is estimated using an expression valid in the weak-coupling regime [141]. This small correction could be inaccurate in the strong-coupling regime (blue shaded region in (a)) where the zero-phonon line (ZPL) begins to display a vacuum Rabi splitting. Parameters used for the SiV⁻ center are as follows: resonance frequency $\omega = 2\pi \times 405$ THz, radiative lifetime $1/\gamma_r = 8.3$ ns [174], phonon-induced pure-dephasing rate $\gamma^{\star} = 2\pi \times 500$ GHz [173], and Debye-Waller factor DW = 0.88 [173]. See appendix B for the PSB spectrum used to calculate the correction. (a) $I\beta$ plotted in the critical regime and in the mode-detuned case with $\Delta_q(1 - \eta_c)^{-1/2} = 2\pi \times 30$ THz, where Δ_q is the effective detuning parameter for higher-order non-radiative plasmon modes and η_c is the bare cavity quantum efficiency. Here $R = 4g^2/\kappa$ where g is the cavity coupling rate and κ is the bare cavity linewidth. The blue dots represent the plasmonic bowtie [167] and a plasmonic-Fabry-Pérot hybrid cavity [166]. For the bowtie, $R/\gamma_r = 1.7 \times 10^6$ is determined from g = 60 meV, Q = 7.3 [167], and $1/\gamma_r = 20$ ns [176]. The dashed line marks the $R/\gamma_{\rm r} = 2.7 \times 10^5$ expected for the hybrid cavity (at Q = 986) from the enhancement of the local density of states (LDOS) [166]. The white star shows the proposed single-photon source. (b) Cross-section for the dashed line in (a). The indistinguishability reaches the value of $I = DW^2 \simeq 0.77$ in the limit $Q \rightarrow 0$ where the PSB is not filtered. (c) Quality factor Q_{max} required to reach the maximum $I\beta$ for a given Δ_q and $R/\gamma_{\rm r} = 2.7 \times 10^5$. (d) I β expected at $Q_{\rm max}$. I is limited by both ZPL broadening and PSB emission in the limit that quenching is negligible.

an initial value of $I\beta\eta_c = 0.83$, this source could be capable of providing near-unity indistinguishability with a single-photon emission efficiency as high as 0.67. This would be comparable to state-of-the-art semi-conductor sources that operate at low temperature [68, 153].

Although we only discussed the SiV⁻ center as a candidate emitter, there are other emitters that have potential. A few other diamond defects might have narrow homogeneous linewidths at room temperature, such as the N3, H2, H1b, and H1c defects [177]. There is also evidence that the infrared transition of the NV⁻ center is weakly coupled to phonons [178–180] and could have a very narrow homogeneous linewidth with a relatively small spectrally-separated PSB [178]. In addition, the neutral silicon vacancy (SiV⁰) center shows a promising combination of optical and spin properties [181, 182]. Its symmetry and electronic configuration allow it to exhibit stable optical properties and a long spin coherence time, combining the best aspects of the SiV⁻ and NV⁻ centers, respectively. Moreover, it has a reported Debye-Waller factor of DW > 0.9—a single-defect measurement limited only by the noise floor [182]. If the spin properties of the SiV⁰ center can be made comparable to those of the NV⁻ center, it could function (in combination with a hybrid plasmonic cavity) as a spin-photon interface for a room-temperature solid-state quantum network.

An interesting alternative approach could be to use proposed pure-dielectric ultrasmall cavities [159, 164, 165]. These cavities should maintain high quality factors ($Q \simeq 10^6$) with mode volumes as small as $7 \times 10^{-5}\lambda^3$ providing unprecedented Purcell factors without being affected by plasmon quenching [165]. These high-Q ultrasmall cavities would need to be used with very narrow emitters, since the cavity must decay faster than the pure-dephasing rate ($\kappa > \gamma^*$). In particular, single rareearth ions are known for having very narrow but dim lines [183]. These quantum emitters can contain very phonon-resistant transitions (such as the ${}^5D_0 \rightarrow {}^7F_0$ transition in the europium(III) ion [184]) that might remain quite narrow at room temperature and exhibit a very small phonon sideband. Such a pure-dielectric ultrasmall cavity could also be useful with broader emitters if its cavity quality factor could be lowered to $\sim 10^2 - 10^3$ without sacrificing an increase in the effective mode volume. We have shown that highly efficient solid-state room-temperature indistinguishable SPSs should be within reach using ultrasmall mode volume cavities, and we have described the most promising regime of operation. For cavities containing plasmonic materials, this regime exists only for emitters that are narrow and bright enough to allow a large effective Purcell enhancement without being dominated by plasmon quenching. In addition, it exists only for emitters with a small PSB that is spectrally well-separated from its ZPL. In particular, a nanodiamond SiV⁻ defect combined with a hybrid plasmonic-Fabry-Pérot cavity appears to be exceptionally promising. Room-temperature indistinguishable SPSs would be a significant advance for quantum technology, while also helping to answer fundamental questions about the physical regimes in which quantum phenomena can be observed [185, 186].

3.1.5 Generalized Purcell factor

The above four sections were published in Ref. [iii]. Here, I would like to make a few brief comments to connect some of the results in [iii] to the content presented in section 1.3.4. First, the expression for intrinsic cavity efficiency, or brightness, presented in section 3.1.1 can be written in terms of the more general definition of cavity-emitter population transfer rate R given in section 1.3.4. This is

$$\beta = \frac{R\kappa}{\gamma\kappa + R(\gamma + \kappa)},\tag{3.5}$$

which holds for all regimes of the purely-dephased Jaynes-Cummings model. This result was also derived using a retarded Green's function approach in Ref. [86]. Second, the brightness for a twolevel system with a Purcell factor F_p is also defined for all parameter regimes as $\beta = F_p/(F_p + \eta_r^{-1})$, where $\eta_r = \gamma_r/\gamma$ is the radiative efficiency of the emitter. By equating these results, we can write a generalized Purcell factor that holds in all the previously discussed regimes, including the strongcoupling regime:

$$F_{\rm p} = \frac{R\kappa}{\gamma_{\rm r}(\kappa + R)}.\tag{3.6}$$

This can be written in terms of cavity quality factor Q and V as in Eq. (1.35) but with a slightly modified inhibition factor F_{inh} that now exactly takes into account the Purcell saturation in the strong coupling regime:

$$F_{\rm inh} = \frac{R\kappa}{\gamma C(\kappa + R)},\tag{3.7}$$

where $C = 4g^2/\kappa\gamma$ is the uninhibited cavity cooperativity and

$$R = \frac{4g^2(\kappa + \Gamma)}{(\kappa + \Gamma)^2 + 4\Delta^2}$$
(3.8)

depends on Δ and g, as well as γ^* through Γ . Taking into account both the saturation and pure dephasing inhibition of the Purcell factor in the critical regime, we can now see that the timescale for excitation must be smaller than $(\gamma + F_p\gamma_r)^{-1} \simeq (F_p\gamma_r)^{-1}$ to avoid significant re-excitation. For the scenario described in Fig. 3.1 for the SiV⁻ center, we have $(F_p\gamma_r)^{-1} \simeq 60$ fs, which is almost two orders of magnitude faster than what has been demonstrated for solid-state systems [187][xiii] and may only be achievable for multi-level systems using chirped pulse engineering [145, 188].

Third, because this generalized Purcell factor holds in all regimes, we can use it to estimate the indistinguishability of photons in the critical regime. Knowing that, for a purely dephased emitter without a cavity, the indistinguishability is $I = \gamma/\Gamma$, we can apply the generalized Purcell factor to get

$$I \simeq \frac{F_p \gamma_r + \gamma}{F_p \gamma_r + \Gamma} = \frac{\gamma(\kappa + R) + R\kappa}{\Gamma(\kappa + R) + R\kappa},$$
(3.9)

which turns out to be a decent estimation. However, it underestimates the indistinguishability near the critical point $R = \kappa = \Gamma$ when moving towards the cavity funneling regime and along the edges of the critical regime compared to the more accurate first-order expression derived in section 3.1.2.

3.2 Hong-Ou-Mandel interference

In section 1.4.4, I introduced the quantities of mean wavepacket overlap and indistinguishability for pulsed systems using Hong-Ou-Mandel (HOM) interference of single-photon states. In this section, I will discuss the relationship between indistinguishability, mean wavepacket overlap, and the integrated intensity correlation for two specific scenarios. The first scenario occurs when one photon of the two-photon component causing a non-zero $g^{(2)}$ is approximately unentangled with the desired single-photon signal. This 'separable noise model' is highly applicable to state-of-theart sources, as demonstrated in the experimental papers of Refs. [xiii] and [vi]. For the second scenario, I will further analyze this relationship for states produced by a pulsed purely-dephased two-level emitter. I will also introduce the notion that a HOM indistinguishability measurement is actually just a special case of a self-homodyne measurement. The broader picture of self-homodyne measurements expands the photonic state properties that can be probed using an identical optical setup to an indistinguishability measurement. This includes quantifying the quantum properties of photonic states that carry photon number coherence or have two-mode entanglement, which is a topic explored in section 3.3 related to Ref. [xiv].

3.2.1 Separable noise model

We now develop a theoretical model to describe the visibility of HOM interference for a specific type of imperfect single photon. This imperfect single photon source is modeled by adding a small amount of 'noise' to an ideal 'signal' single photon using a beam splitter interaction, as shown in Fig. 3.3 (a). Here, we assume that the noise is separable and exhibits no entanglement with the single photon. For simplicity, we also model the noise by another single photon, which should be valid in the limit of a weak noise field or, equivalently, when $g^{(2)}$ is small. The approach and results in this section are derived from the material presented in the supplementary of [xiii].

The initial state of the signal (s) and noise (n) is given by $\hat{\varrho} = \hat{\varrho}_{s} \otimes \hat{\varrho}_{n}$ where $\hat{\varrho}_{i} = p_{i,0} |0\rangle \langle 0| + p_{i,1}\hat{\varrho}_{i,1}$ and $\hat{\varrho}_{i,1} = \int \int \xi_{i}(t,t')\hat{b}_{i}^{\dagger}(t) |0\rangle \langle 0| \hat{b}_{i}(t')dtdt'$, where $\xi_{i}(t,t')$ is the normalized single-photon



Figure 3.3: Two specific models for an imperfect single-photon source. (a) The separable noise model where the nonzero two-photon component is composed of a perfect single photon 'signal' combined with a 'noise' single photon. The state is constructed using a beam-splitter interaction with a variable ratio controlled by the parameter θ . In addition, both the signal and the noise may not be perfectly efficient $(p_{n,1}, p_{s,1} \leq 1)$. (b) A single-photon produced by the pulsed excitation of a purely-dephased two-level emitter. The pulse length t_p compared to the system lifetime $1/\gamma$ determines the amount of multi-photon contribution to the desired single-photon state.

temporal density wavefunction for $i \in \{s, n\}$ (recall section 2.4). The imperfect photon in the output transmission mode of the beam splitter \hat{b} is obtained by tracing out the reflected loss mode after applying the beam splitter relation, so that $\hat{\varrho} = \text{Tr}_{\text{loss}}(\hat{\varrho}_{s} \otimes \hat{\varrho}_{n})$, where

$$\begin{pmatrix} \hat{b}(t)\\ \hat{b}_{loss}(t) \end{pmatrix} = \begin{pmatrix} \cos\theta & -\sin\theta\\ \sin\theta & \cos\theta \end{pmatrix} \begin{pmatrix} \hat{b}_{s}(t)\\ \hat{b}_{n}(t) \end{pmatrix}.$$
(3.10)

The total state of the imperfect single photon can then be written as $\hat{\rho} = p_0 |0\rangle \langle 0| + p_1 \hat{\rho}_1 + p_2 \hat{\rho}_2$. From this, we can write $g^{(2)}$ and μ using the photon number probability relations $\mu = p_1 + 2p_2$ and $g^{(2)} = 2p_2/\mu^2$ derived in section 2.5. For the average photon number, we have $\mu = p_{s,1} \cos^2 \theta + p_{n,1} \sin^2 \theta$ and $g^{(2)}$ is given by $\mu^2 g^{(2)} = 2p_{s,1}p_{n,1}(1 + M_{sn}) \cos^2 \theta \sin^2 \theta$, where

$$M_{\rm sn} = \iint \xi_{\rm s}(t,t')\xi_{\rm n}^*(t,t')dtdt'$$
(3.11)

is the mean wavepacket overlap of the single photon and noise, and θ along with $p_{n,1}$ quantifies the amount of noise. Note that M_{sn} is a real value as a consequence of the Hermitian property of ξ_s and ξ_n . However, one could restrict $t \leq t'$ and compute the integral over $2\text{Re}(\xi_s(t, t')\xi_n^*(t, t'))$ to obtain the same result. This case is useful when evaluating ξ using the time-ordered convention described in section 2.4.

The total mean wavepacket overlap of the imperfect photon with itself is given by

$$\mu^2 M = \iint \left| \langle \hat{b}^{\dagger}(t) \hat{b}(t') \rangle \right|^2 dt' dt$$

= $p_{s,1}^2 I \cos^4 \theta + p_{n,1}^2 M_n \sin^4 \theta + 2p_{s,1} p_{n,1} M_{sn} \cos^2 \theta \sin^2 \theta,$ (3.12)

where $M_{\rm s} = \iint |\xi_{\rm s}(t,t')|^2 dt dt' = \operatorname{Tr} \left[\hat{\varrho}_{{\rm s},1}^2\right] = I$ quantifies the intrinsic single-photon indistinguishability or single-photon trace purity [109, 110] and $M_{\rm n} = \iint |\xi_{\rm n}(t,t')|^2 dt dt'$ is the indistinguishability of the noise photon with itself. It is worth emphasizing that Eq. (3.12) holds only when the states of modes $\hat{b}_{\rm s}$ and $\hat{b}_{\rm n}$ are uncorrelated and carry no number coherence, which is the case for ϱ as defined above.

We can now reparametrize the expressions for μ , $g^{(2)}$, and M by defining an effective noise parameter ϑ so that $\cos^2 \vartheta = (p_{s,1} \cos^2 \theta)/\mu$ and $\sin^2 \vartheta = (p_{n,1} \sin^2 \theta)/\mu$. The fact that this reparametrization exists stems from the independence of M and $g^{(2)}$ from photon loss. It also implies that the fundamental quantity affecting the photon statistics of this imperfect single photon model is ϑ , which depends on both the beam splitter angle θ and the relative input intensities through $p_{s,1}$ and $p_{n,1}$. Knowing that $V_{\text{HOM}} = M - g^{(2)}$ (see section 1.4.4), the visibility and $g^{(2)}$ in terms of the noise parameter ϑ are

$$V_{\text{HOM}}(\vartheta) = I \cos^4 \vartheta + M_{\text{n}} \sin^4 \vartheta - 2 \cos^2 \vartheta \sin^2 \vartheta$$

$$g^{(2)}(\vartheta) = 2(1 + M_{\text{sn}}) \cos^2 \vartheta \sin^2 \vartheta.$$
 (3.13)

From an experimental point of view, we are interested in the relationship between V_{HOM} and $g^{(2)}$ in the limit that $g^{(2)}$ is small. To this end, let us analyze the slope and intercept of the parametric curve formed by $\{g^{(2)}(\vartheta), V_{\text{HOM}}(\vartheta)\}$. The solution for the intercept is clear since $g^{(2)}(\vartheta) = 0$ implies $\vartheta = 0$ and $V_{\text{HOM}}(0) = I$. To solve for the slope at small ϑ , we have

$$\lim_{\vartheta \to 0} \frac{dV_{\text{HOM}}(\vartheta)}{dg^{(2)}(\vartheta)} = -\frac{1+I}{1+M_{\text{sn}}}.$$
(3.14)

Hence, the HOM visibility for small $g^{(2)}$ is given by

$$V_{\text{HOM}} = I - \left(\frac{1+I}{1+M_{\text{sn}}}\right) g^{(2)}.$$
 (3.15)

This implies that sources with identical noise where $M_{\rm sn} = I$ (i.e. some heralded sources from SPDC) should show a higher HOM visibility compared to sources with distinguishable noise $(M_{\rm sn} = 0)$ for the same $g^{(2)}$ and single-photon indistinguishability I. Note that for identical noise $(M_{\rm sn} = I)$, we recover the usual estimate for single-photon indistinguishability $I = V_{\rm HOM} + g^{(2)} = M$ from the measured values. For the distinguishable noise case $(M_{\rm sn} = 0)$ we obtain the correction $I = (V_{\rm HOM} + g^{(2)})/(1 - g^{(2)}) = M/(1 - g^{(2)})$, which is larger by a factor of $(1 - g^{(2)})^{-1}$ compared to the identical noise case.

The relationship between V_{HOM} and $g^{(2)}$ derived in this section has been confirmed experimentally in a collaboration with the group of Prof. Pascale Senellart for the cases $M_{\text{sn}} = I$ and $M_{\text{sn}} = 0$ [xiii]. In that work, we also found that pulsed single-photon sources based on semi-conductor QDs, in both trion and exciton configurations, follow very closely to the $M_{\text{sn}} = 0$ case when $g^{(2)}$ is below 0.1. Furthermore, in Ref. [vi], we applied Eq. (3.15) using $M_{\text{sn}} = 0$ to 15 different QD sources with varying $g^{(2)}$ and V_{HOM} values. The resulting 15 estimates for I showed a smaller deviation than that of $M = V_{\text{HOM}} - g^{(2)}$, suggesting that the estimated value of I is indeed less dependent on $g^{(2)}$ than M is. These experimental results are particularly interesting because 'noise' from re-excitation processes should *not*, in general, be separable from the single photon signal because of entanglement due to emission time jitter. Hence, one might expect these sources to violate the primary assumption made to derive Eq. (3.15). In the next section, I will look deeper into this relationship for a pulsed two-level system to discuss under which regimes the separable noise model is actually valid for single-photon sources where the nonzero $g^{(2)}$ arises from re-excitation.

3.2.2 Pulsed two-level emitter

In this section, I consider the relationship between visibility and $g^{(2)}$ for the specific case where the nonzero $g^{(2)}$ is caused only by re-excitation of the source (see Fig. 3.3 (b)). I model the emission using a driven two-level system that may experience pure dephasing. Then, I compute the HOM visibility and $g^{(2)}$ using a Markovian master equation. The results shown in this section hold in the far-field of the dipole radiation from a two-level emitter where the proportionality relationship $\hat{b} \propto \hat{\sigma}$ is valid [92]. If the two-level system is inside a cavity, as described in section 1.3.4, we must be sufficiently far into the bad-cavity regime where adiabatic elimination of the cavity mode is valid (recall section 1.3.6). Otherwise, this effective two-level model neglects important emittercavity non-Markovian effects such as Rabi oscillations and the drive-dependent Purcell factor [108]. For this two-level model, the correlation functions of the waveguide can be computed using the proportionality relation $\hat{b} = \sqrt{\eta_r \gamma} \hat{\sigma}$ where $\eta_r = \eta_c F_p/(1 + F_p)$ now depends on the Purcell factor of the cavity. However, the important quantities computed in this section are loss-independent and so I will consider $\eta_r = 1$ so that $\hat{b} = \sqrt{\gamma} \hat{\sigma}$ for simplicity.

I begin by writing the Liouville superoperator for a purely-dephased classically-driven two-level system:

$$\mathcal{L} = -\frac{i}{\hbar} \mathcal{H}(t) + \gamma \mathcal{D}(\hat{\sigma}) + 2\gamma^* \mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma}), \qquad (3.16)$$

where $\hat{H}(t) = (\hbar \Omega(t)/2)(\hat{\sigma} + \hat{\sigma}^{\dagger})$ and $\mathcal{H}\hat{\rho} = [\hat{H}, \hat{\rho}]$. It is then possible to evaluate the total intensity μ , mean wavepacket overlap M, and intensity correlation $g^{(2)}$ for any pulse shape $\Omega(t)$ using time-dependent integration methods. However, the quantities M and $g^{(2)}$ require the evaluation of two-time correlation functions, which can be very time-consuming to compute when exploring a very large parameter set. Alternatively, we can assume that the pulse is square in shape and divide the Hamiltonian into two separate time-independent parts. For a square pulse with temporal width

 $t_{\rm p}$ and Rabi frequency Ω beginning at t=0, we can write the propagation superoperator as

$$\mathcal{U}(t',t) = \begin{cases} \mathcal{U}_{p}(t',t) & t' \leq t_{p} \\ \mathcal{U}_{d}(t',t_{p})\mathcal{U}_{p}(t_{p},t) & t \leq t_{p} \leq t' \\ \mathcal{U}_{d}(t',t) & t > t_{p} \end{cases}$$
(3.17)

where $\mathcal{U}_{p}(t',t) = e^{(t'-t)\mathcal{L}_{p}}$ is the propagation superoperator when the pulse is on, corresponding to $\mathcal{L}_{p} = \mathcal{L}(\Omega(t) = \Omega)$, and $\mathcal{U}_{d}(t',t) = e^{(t'-t)\mathcal{L}_{d}}$ is the propagation superoperator when only decay is occurring, corresponding to $\mathcal{L}_{d} = \mathcal{L}(\Omega(t) = 0)$.

For the piecewise evolution, the two-time integrals for M and $g^{(2)}$ can be divided into three parts: (1) $0 \le t \le t_p$ and $t \le t' \le t_p$, (2) $0 \le t \le t_p$ and $t_p < t' < \infty$, and (3) $t_p < t < \infty$ and $t \le t' < \infty$. For example, M can be evaluated by

$$\frac{\mu^2}{2\gamma^2}M = \int_0^{t_{\rm p}} \int_t^{t_{\rm p}} \left| \langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle_{\rm p} \right|^2 dt' dt + \int_0^{t_{\rm p}} \int_{t_{\rm p}}^{\infty} \left| \langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle_{\rm dp} \right|^2 dt' dt
+ \int_{t_{\rm p}}^{\infty} \int_t^{\infty} \left| \langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle_{\rm d} \right|^2 dt' dt,$$
(3.18)

where using the superoperators $S\hat{\rho} = \hat{\sigma}\hat{\rho}$ and $\mathcal{R}\hat{\rho} = \hat{\rho}\hat{\sigma}^{\dagger}$ we have

$$\langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle_{\rm p} = \operatorname{Tr}(\mathcal{R}\mathcal{U}_{\rm p}(t',t)\mathcal{S}\mathcal{U}_{\rm p}(t,0)\hat{\rho}(0)) \langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle_{\rm dp} = \operatorname{Tr}(\mathcal{R}\mathcal{U}_{\rm d}(t',t_{\rm p})\mathcal{U}_{\rm p}(t_{\rm p},t)\mathcal{S}\mathcal{U}_{\rm p}(t,0)\hat{\rho}(0)) \langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle_{\rm d} = \operatorname{Tr}(\mathcal{R}\mathcal{U}_{\rm d}(t',t)\mathcal{S}\mathcal{U}_{\rm d}(t,t_{\rm p})\mathcal{U}_{\rm p}(t_{\rm p},0)\hat{\rho}(0)) .$$

$$(3.19)$$

By evaluating the dynamics in this piecewise time-independent way, it is possible to evaluate exact analytic expressions for μ and $g^{(2)}$, although the expressions are quite large and do not provide much physical insight. It may also be possible to evaluate an exact expression for M but the squared integrand complicates the solution. Regardless, we are still able to evaluate exact analytic expressions for the two-time correlation functions used to compute M, which can drastically reduce the computation time required to solve M numerically. Fig. 3.5 (a) shows μ , $g^{(2)}$, and M for a



Figure 3.4: Single-photon figures of merit for a purely-dephased two-level emitter driven by a square π pulse. (a) Single-photon quality quantified by the mean photon number μ , second-order correlation $g^{(2)}$, and mean wavepacket overlap M computed using the proportionality relation $\hat{b} = \sqrt{\gamma}\hat{\sigma}$. (b) The Hong-Ou-Mandel visibility $V = V_{\text{HOM}} = M - g^{(2)}$ plotted against $g^{(2)}$ for different pure dephasing values γ^* . The dashed lines indicate the slope in the limit of small $g^{(2)}$ and correspond to the $V_{\text{HOM}} = I - (1+I)g^{(2)}$ result from the previous section.

two-level system with a pure dephasing rate of $\gamma^* = 0.04\gamma$ driven by a π pulse where $\Theta = t_p \Omega = \pi$. With the piecewise evaluation approach, we can also efficiently analyze the shape of the $\{g^{(2)}, V_{\text{HOM}}\}$ curve when varying different pulse parameters. For the π pulse case in the limit that $t_p \to 0$, the slope of the $\{g^{(2)}, V_{\text{HOM}}\}$ curve can indeed be seen to approach -(1 + I) where $I = \gamma/\Gamma$ is the value of M at $g^{(2)} = 0$ (see Fig. 3.4). This corresponds to the distinguishable noise scenario in the previous section, as was found experimentally. However, if Θ is not fixed, the shape and slope of the parameteric curve $\{g^{(2)}, V_{\text{HOM}}\}$ actually depends on which parameter is being varied, Ω or t_p . Interestingly, regardless of the parameters Ω and t_p , we can find that the value of V_{HOM} is always bounded between the straight lines $V_{\text{HOM}} = I - (1 + I)g^{(2)}$ and $V_{\text{HOM}} = I - g^{(2)}$ (see Fig. 3.5), just as we found from the separable noise model. Thus we can postulate that

$$V_{\text{HOM}} + g^{(2)} \le I \le \frac{V_{\text{HOM}} + g^{(2)}}{1 - g^{(2)}}$$
 (3.20)

is true for any pulse shape applied to a driven emitter following a two-level model. This result is quite useful since it allows one to estimate both a lower bound and an upper bound on I by measuring V_{HOM} and $g^{(2)}$ only once.



Figure 3.5: The relationship between Hong-Ou-Mandel visibility and the integrated intensity correlation for a purely-dephased two-level emitter. Numerically exact computed positive values of $V = V_{\text{HOM}} = M - g^{(2)}$ and $g^{(2)}$ reachable using a square pulse with width t_p and Rabi frequency Ω for a) no pure dephasing and b) pure dephasing giving $M_s = I = 0.8$ corresponding to $\gamma^* = \gamma/8$. For small $g^{(2)}$, pulses near an area of π approach the lower bound of V_{HOM} while pulses near an area of 0 approach the upper bound. Points corresponding to smaller pulse areas are plotted on top of those corresponding to larger pulse areas if there is overlap.

Numerically computing the $\{g^{(2)}, V_{HOM}\}$ curve gives a good prediction and a useful relation, but it doesn't really give much physical insight into the nature of the emitted wavepacket. Ultimately, it would be nice to understand how the properties of the re-excited photon affects the HOM visibility. Let us begin that endeavour now.

Recall from section 2.4 that the emitted photonic state density matrix $\hat{\rho}$ can be decomposed into the photon number subspaces by $\hat{\rho} = \sum_{n,m} \hat{\rho}_{n,m}$ where $\hat{\rho}_n = \hat{\rho}_{n,n}$ is the unnormalized state given that *n* photons were emitted and $\hat{\rho}_{n,m}$ quantifies possible coherence between photon number states. The probability of emitting *n* photons after one pulse is given by $p_n = \text{Tr} [\hat{\rho}_n]$. In this notation, the $g^{(2)}$ and μ can be written in terms of p_n using the standard photon statistics relations (see again section 2.5). Similarly, we can decompose the mean wavepacket overlap using conditional correlation functions of the waveguide (see sections 2.3 and 2.4):

$$M = \frac{1}{\mu^2} \sum_{j,k} \iint \langle \hat{b}^{\dagger}(t')\hat{b}(t) \rangle_j \langle \hat{b}^{\dagger}(t')\hat{b}(t) \rangle_k^* dt' dt$$

$$= \frac{1}{\mu^2} \sum_{j,k} \mu_j \mu_k M_{j,k},$$
 (3.21)

where $\mu_n = np_n$ and

$$M_{j,k} = \frac{1}{\mu_j \mu_k} \iint \operatorname{Re}\left[\langle \hat{b}^{\dagger}(t') \hat{b}(t) \rangle_j \langle \hat{b}^{\dagger}(t') \hat{b}(t) \rangle_k^* \right] dt' dt, \qquad (3.22)$$

is the normalized mean wavepacket overlap between density operators $\hat{\varrho}_k$ and $\hat{\varrho}_j$ in different photonnumber subspaces. Here, I have used the property $M_{j,k} = M_{k,j}^*$ to define the intensity normalized overlap $M_{j,k}$ as a real valued quantity.

We can now analyze the overlap in terms of the one- and two-photon temporal density functions introduced in section 2.4. Using this formalism, we can compute relevant $M_{j,k}$ using the timeordered normalization convention by

$$M_{1,1} = I = \iint |\xi_1(t, t')|^2 dt dt'$$

$$M_{1,2} = \frac{1}{2} (\Lambda_1 + \Lambda_2 + \Lambda_{12})$$
(3.23)

and

$$\Lambda_{1} = \iiint \operatorname{Re} \left[\xi_{1}(t, t')\xi_{2}^{*}(t, t'', t', t'')\right] dt dt' dt''$$

$$\Lambda_{2} = \iiint \operatorname{Re} \left[\xi_{1}(t, t')\xi_{2}^{*}(t'', t, t'', t')\right] dt dt' dt''$$

$$\Lambda_{12} = \iiint \operatorname{Re} \left[\xi_{1}(t, t')\xi_{2}^{*}(t, t'', t'', t')\right] dt dt' dt'',$$
(3.24)

where Λ_1 quantifies the overlap of the single photon and the first photon of the two-photon component, Λ_2 quantifies the overlap between the single photon and the second photon of the two-photon component, and Λ_{12} is related to how much entanglement exists between the noise and the desired single photon. We can compute these integrals exactly for the square pulse case by applying a photon number decomposition to \mathcal{U}_p and \mathcal{U}_d for $\mathcal{J} = \gamma \mathcal{RS}$ (see section 2.1) and then applying Eqs. (2.23) and (2.26) using the two-level emitter proportionality relation $\hat{b} = \sqrt{\gamma}\hat{\sigma}$ instead of the cavity-waveguide relation $\hat{b} = \sqrt{\kappa \eta_c}\hat{a}$. In Fig. 3.6, we can see that Λ_1 and Λ_{12} approach zero as t_p decreases, regardless of Θ . In addition Λ_2 approaches 1 when $\gamma^* = 0$. I will discuss the one- and two-photon density functions in much more detail in section 3.3. For now, let us use this numerical



Figure 3.6: Constituent mean wavepacket overlap integrals illustrating the effect of re-excitation on the single-photon indistinguishability from a purely-dephased emitter. Components of $M_{1,2} = (\Lambda_1 + \Lambda_2 + \Lambda_{12})/2$ computed from the temporal density functions when $\gamma^* = 0$ at (a) π -pulse excitation and (b) in the weak-driving limit. Regardless of $t_p\Omega$, $M_{1,2}$ converges to 1/2 from below in the limit that $t_p\gamma \to 0$. The curves in this panel correspond to those in panel (a) and follow the same ordering of $\Lambda_{12} \leq \Lambda_1 \leq M_{1,2} \leq \Lambda_2$.

result to motivate a useful approximation.

The result in Fig. 3.6 is intuitive in the limit $t_p \gamma \rightarrow 0$ because the first photon of the two-photon component ξ_2 must have been emitted during the pulse if the emitter was re-excited. Hence, if the pulse is short, the first photon must have a very small temporal overlap with the single-photon component, which is emitted primarily after the pulse (hence $\Lambda_1 \rightarrow 0$). The second photon of the two-photon component must also have very little temporal entanglement with the first photon when the pulse is short (hence $\Lambda_{12} \rightarrow 0$), simply because they have very different timescales. Finally, the second photon emitted in the two-photon component must have a very good overlap with the single-photon component, since both are emitted primarily after the pulse and have similar intensity profiles (hence $\Lambda_2 \rightarrow 1$).

We can formalize this separability approximation in the fast-pulse limit as $\xi_2(t_1, t_2, t'_1, t'_2) \simeq \xi_n(t_1, t'_1)\xi_1(t_2, t'_2)$, where n symbolizes the noise photon (the first photon emitted from ξ_2). Then, applying this to Eq. (3.24), we find that $\Lambda_1 \simeq 0$, $\Lambda_2 \simeq I$, and $\Lambda_{12} \simeq 0$, which implies $M_{1,2} = I/2$.

Applying this approximation to M gives

$$M \simeq \frac{\mu_1^2 I + \mu_1 \mu_2 I + \mu_2^2 M_{2,2}}{\mu^2}$$

$$\simeq I \left(1 - p_1 g^{(2)} \right).$$
(3.25)

Finally, in the fast-pulse limit of a driven two-level system, we know that $p_1 \simeq \sin^2(\Theta/2)$. Thus, for a driven two-level system in the fast-pulse limit and when $g^{(2)}$ is small, we have

$$V_{\text{HOM}} = I - \left(1 + I\sin^2\left(\Theta/2\right)\right) g^{(2)}.$$
(3.26)

For a π -pulse, this becomes $V_{\text{HOM}} = I - (1+I)g^{(2)}$ and for very low power we have $V_{\text{HOM}} = I - g^{(2)}$. These limits agree with the numerical simulations in Fig. 3.5 and, in particular, when $g^{(2)}$ is small. However, this expression is not accurate for Θ around $2n\pi$ for integers $n \ge 1$, because $g^{(2)}$ will not be small for these pulse areas [189]. This high- $g^{(2)}$ regime is within the gray area in Fig. 3.5.

The results presented in this section are only valid for HOM interference measurements that are normalized by the total squared intensity using an interferometer that is completely phase averaged during the timescale of the measurement. This is because, as I will show in the next section, a photonic state with photon number coherence can alter the HOM visibility if it is normalized by the uncorrelated peaks or if the interferometer has a stable phase on the timescale of the measurement. This latter situation allows a HOM setup to perform a self-homodyne measurement of one- and two-photon coherence.

3.2.3 Self-homodyne measurements

Hong-Ou-Mandel interference of single photons produced by one emitter can be seen as a special case of a self-homodyne measurement, where two copies of the same photonic state are interfered. Instead of using a local oscillator (LO) as a phase reference, as is the case for classical homodyne detection, we can instead define one of the two interfering states as the reference state. The phase

of this reference state is primarily defined by the LO used to excite the emitter(s). Hence, it can serve in place of the LO in the phase measurement. However, in addition to phase fluctuations of the LO, the phase of the self-homodyne reference state is affected by phase fluctuations due to the source environment, such as pure dephasing and spectral diffusion. This makes self-homodyne measurements useful for characterizing the purity of a photonic state, such as indistinguishability [68] and photon-number purity [190].

When performing HOM interference in this way, the individual detector rates can be affected by the phase of the interferometer in a way analogous to classical homodyne detection, provided that the input photonic states carry coherence in the number basis. For a balanced beam splitter with input modes \hat{b}_1 and \hat{b}_2 , the output modes \hat{b}_3 and \hat{b}_4 can be described by the relation

$$\begin{pmatrix} \hat{b}_3(t)\\ \hat{b}_4(t) \end{pmatrix} = \frac{1}{\sqrt{2}} \begin{pmatrix} 1 & -e^{-i\phi}\\ e^{i\phi} & 1 \end{pmatrix} \begin{pmatrix} \hat{b}_1(t)\\ \hat{b}_2(t) \end{pmatrix}.$$
(3.27)

Here, we take ϕ to be the relative phase difference imparted on the two otherwise identical photonic input states. The intensities received at each detector, denoted $\mu^+ = \int N_3(t)dt$ and $\mu^- = \int N_4(t)dt$, will fluctuate in opposition by $\mu^{\pm} = \mu[1 \pm \lambda^{(1)}\cos(\phi)]$ where $\lambda^{(1)} = \mu^{-1} \int |\langle \hat{b}(t) \rangle|^2 dt$ is the integrated squared magnitude of the single-photon coherence and $\mu = \mu^+ + \mu^-$. Thus, we can define the pulsed self-homodyne signal to be

$$V_{\rm SH} = \frac{\mu^+ - \mu^-}{\mu^+ + \mu^-} = \lambda^{(1)} \cos(\phi).$$
(3.28)

The coincident events of the detectors at the output are determined from the two-time intensity correlation between the output fields $G_{\text{HOM}}^{(2)}(t,t') = \langle \hat{b}_3^{\dagger}(t) \hat{b}_4^{\dagger}(t') \hat{b}_4(t') \hat{b}_3(t) \rangle$. Using the same approach as in section 1.4.4, but keeping terms arising from coherence, we obtain

$$2G_{\text{HOM}}^{(2)}(t,t') = N(t)N(t') + G^{(2)}(t,t') - \left|G^{(1)}(t,t')\right|^{2} + 2\Lambda_{-}^{(1,2)}(t,t')\cos(\phi) - \left|\Lambda^{(2)}(t,t')\right|^{2}\cos(2\phi),$$
(3.29)

where the new term $\Lambda^{(2)}(t,t') = \langle \hat{b}(t')\hat{b}(t) \rangle$ captures the two-photon coherence and $\Lambda^{(1,2)}_{-}(t,t') =$ Re $\left[\langle \hat{b}(t') \rangle \langle \hat{b}^{\dagger}(t) \hat{b}^{\dagger}(t') \hat{b}(t) \rangle - \langle \hat{b}(t) \rangle \langle \hat{b}^{\dagger}(t) \hat{b}^{\dagger}(t') \hat{b}(t') \rangle \right]$ is a term that is nonzero only if there is onephoton coherence and specifically coherence between the 1 and 2 photon subspaces. For emission from a two-level system or a cavity-emitter device operating in the Purcell regime, the term $\Lambda^{(1,2)}_{-}$ is nonzero only during the coherent excitation of the system. Because of this, it could serve as a way to measure the time dynamics of the re-excitation processes, although I do not explore that idea in this thesis.

After integrating $G_{\text{HOM}}^{(2)}$ over the photonic state and normalizing by the total intensity $\mu = \int N(t)dt$, we get

$$V_{\text{HOM}} = M - g^{(2)} - 2\lambda_{-}^{(1,2)}\cos(\phi) + \lambda^{(2)}\cos(2\phi), \qquad (3.30)$$

where we now have two new quantities affecting the HOM visibility: the integrated two-photon coherence $\lambda^{(2)} = \mu^{-2} \int \int |\Lambda^{(2)}(t,t')|^2 dt' dt$ and the integrated asymmetry component $\lambda_{-}^{(1,2)} = \mu^{-2} \int \int |\Lambda_{-}^{(1,2)}(t,t')|^2 dt' dt$. I call this last term the asymmetry component because, when looking at the parameteric curve $\{V_{\text{SH}}(\phi), V_{\text{HOM}}(\phi)\}$, we would expect a quadratic relationship centered about $V_{\text{SH}} = 1/2$ when $\lambda_{-}^{(1,2)} = 0$ and $\lambda^{(2)} \neq 0$ [xiv]. The presence of a very small $\lambda_{-}^{(1,2)}$ component will add a linear dependence of V_{HOM} on V_{SH} , skewing this quadratic relationship and making the parametric curve asymmetric about $V_{\text{SH}} = 1/2$. By measuring V_{SH} and V_{HOM} simultaneously while varying ϕ , one can extract a lot of information about an input photonic state [190][xiv].

It is very important to note that I have defined all of the above HOM quantities normalized with respect to μ^2 and not the total uncorrelated coincidence counts, even though the latter normalization is standard for single-photon source characterization [xiii]. The purpose for choosing to normalize by the intensity is to avoid conflating important quantities that we are attempting to extract from the measurement. If the measured photonic state carries one-photon coherence, then the uncorrelated coincidence counts will depend on $\langle \hat{b}(t) \rangle$ [108] and hence a calculated HOM visibility normalized by the uncorrelated coincidence counts will depend on V_{SH} . Then, it would be necessary to correct for nonzero $\lambda^{(1)}$ in order to obtain M, $\lambda^{(2)}$, and $\lambda^{(1,2)}_{-}$.

To see this normalization problem explicitly, consider $G_{\text{HOM}}^{(2)}(t, t' + mT_{\text{p}})$ where T_{p} is the delay between successive photonic states. Here we must choose $m \ge 2$ so that we do not probe the first sideband peak, which can still contain correlations due to the Mach-Zehnder setup (see the supplementary of [xiii]), but we must also not choose m so large that potential emitter blinking introduces correlations [191]. For this uncorrelated peak, we have that $G_{\text{HOM}}^{(2)}(t, t' + mT_{\text{p}}) =$ $\langle \hat{b}_{3}^{\dagger}(t)\hat{b}_{3}(t)\rangle \langle \hat{b}_{4}^{\dagger}(t')\hat{b}_{4}(t')\rangle = N_{3}(t)N_{4}(t')$, where I have applied the periodic condition of our source to eliminate T_{p} . By integrating over the entire uncorrelated peak, we obtain $g_{\text{unc.}}^{(2)} = \mu^{+}\mu^{-} =$ $\mu^{2}(1 - V_{\text{SH}}^{2})$. Hence, the uncorrelated peak underestimates μ^{2} depending on the amount of singlephoton coherence and the interferometer phase. It is critical to note that, even if the interferometer phase ϕ is averaged during data collection, so long as the phase is stable on the timescale of mT_{p} then $g_{\text{unc.}}^{(2)} = \mu^{2}(1 - (\lambda^{(1)})^{2}/2)$ still underestimates μ^{2} . If $g_{\text{HOM}}^{(2)}$ is normalized with respect to $g_{\text{unc.}}^{(2)}$, then the important quantities such as M will be over-estimated by the factor $(1 - V_{\text{SH}}^{2})^{-1}$.

If the two states being measured are produced by the same emitter, the phase coherence probed by the self-homodyne measurement is susceptible to all phase fluctuations occurring on the timescale of the delay T_p between the reference state and the measured state. This case arises when measuring indistinguishability, as was discussed in the previous section. If the two states are produced by independent (but identical) emitters, then the self-homodyne measurement is susceptible to all emitter dephasing mechanisms on all timescales. This case arises when performing spin-spin entanglement via which-path erasure, which will be discussed in section 4.1.

3.3 Photonic state from a purely-dephased emitter

In this section, I will further explore the one- and two-photon temporal density functions that were discussed in the previous section to justify the separability approximation. I will then extend the idea of a photon number decomposition to analyze self-homodyne measurements of photonic states to gain insight into the amount of coherence between photon number subspaces. Finally, I will discuss and analyze how a two-level emitter can be manipulated to generate photon-number Bell

states encoded in discrete time bins.

3.3.1 Temporal coherence and photon statistics

Let us again consider a two-level system experiencing some pure dephasing that is excited by a square pulse in the semi-classical approximation. The Liouville superoperator for this system is given by Eq. (3.16). We can now perform a photon number decomposition using the jump superoperator $\mathcal{J}\hat{\rho} = \eta_r \gamma \hat{\sigma} \hat{\rho} \hat{\sigma}^{\dagger}$ to obtain the conditional propagation superoperators \mathcal{U}_n that are of the order η_r^n . Furthermore, we can break the evolution into piecewise time-independent parts using Eq. (3.17). Using the conditional propagation superoperators along with the expressions given in section 2.4, it is possible to compute the photon number probabilities p_n and temporal density functions of photon number subspaces ξ_n by correctly choosing the integration limits for each \mathcal{U}_n , similar to Eq. (3.18). With this method, we can obtain analytically exact expressions by evaluating the conditional propagation superoperators in the Fock-Liouville space (see section 1.5.1). When $\gamma^* \neq 0$, analytic solutions become very difficult to simplify, but the problem can still be numerically computed in an efficient way. If we choose $\gamma^* = 0$ and $\eta_r = 1$, then for n = 0 we can reproduce the known result [128] for the vacuum probability for a pulsed two-level system

$$p_0 = \frac{1}{4\tilde{\Omega}^2} \left[2\tilde{\Omega} \cos\left(\frac{t_{\rm p}\tilde{\Omega}}{2}\right) + \gamma \sin\left(\frac{t_{\rm p}\tilde{\Omega}}{2}\right) \right]^2 e^{t_{\rm p}\gamma/2},\tag{3.31}$$

where $2\tilde{\Omega} = \sqrt{4\Omega^2 - \gamma^2}$. For $n \ge 1$ the exact expressions are complicated, but in the approximation that $t_p \ll 1/\gamma$, we can find that the photon number decomposition of the master equation gives the solutions

$$p_{0} \simeq \cos^{2} \left(\frac{\Omega}{2} t_{\rm p}\right) e^{-\gamma t_{\rm p}/2}$$

$$p_{1} \simeq \sin^{2} \left(\frac{\Omega}{2} t_{\rm p}\right) e^{-\gamma t_{\rm p}/2}$$

$$p_{2} \simeq \frac{\gamma t_{\rm p}}{8} \left(2 + \cos(\Omega t_{\rm p})\right) e^{-\gamma t_{\rm p}/2}.$$
(3.32)



Figure 3.7: Photon emission statistics for pulsed excitation of a two-level system for square pulse area $\Theta = \pi$, 2π , and 3π from left to right, respectively. Chosen parameters are $\Delta = 0$, $\eta_r = 1$, and $\gamma^* = 0$. The curve labeled $p_{n\geq 3}$ (red curve) is computed by $1 - p_0 - p_1 - p_2$.

See Fig. 3.7 for plots of the exact $p_{n<3}$ for an emitter driven at $\Theta = \pi$, 2π , and 3π when $\gamma^* = 0$. The above expressions for p_n are not novel, and can also be derived from a Hamiltonian emitter-waveguide system [128], which (for the case of pure states) reduces to the photon number decomposition I have presented. This photon decomposition approach has also been used in Ref. [126] to compute photon number probabilities for quantum dot sources.

The full analytic expressions for ξ_1 and ξ_2 that include γ^* are quite complicated and do not have a form that is physically insightful. However, if $\gamma^* = 0$, we find that the solutions for ξ_1 and ξ_2 using Eqs. (2.23) and (2.26) can be factored into amplitudes $\xi_1(t, t') = f_1(t)f_1^*(t')$ and $\xi_2(t_1, t_2, t'_1, t'_2) = f_2(t_1, t_2)f_2^*(t'_1, t'_2)$. The solutions for the amplitudes f_1 and f_2 obtained from ξ_1 and ξ_2 , respectively, written in the frame rotating at the emitter frequency are

$$f_{1}(t) = \frac{\sqrt{\gamma}e^{i\phi_{1}}}{\sqrt{p_{1}}} \begin{cases} \frac{\Omega}{2\tilde{\Omega}^{2}}\sin\left(\frac{\tilde{\Omega}}{2}t\right) \left[2\tilde{\Omega}\cos\left(\frac{\tilde{\Omega}}{2}(t_{\rm p}-t)\right) + \gamma\sin\left(\frac{\tilde{\Omega}}{2}(t_{\rm p}-t)\right)\right] e^{-\gamma t_{\rm p}/4} & t < t_{\rm p} \\ \frac{\Omega}{\tilde{\Omega}}\sin\left(\frac{\tilde{\Omega}}{2}t_{\rm p}\right) e^{-\gamma(2t-t_{\rm p})/4} & t \ge t_{\rm p} \end{cases}$$
(2.20)

and

$$f_{2}(t_{1},t_{2}) = \frac{\sqrt{p_{1}\gamma}e^{i\phi_{2}}}{\sqrt{p_{2}}} \begin{cases} \frac{\Omega}{\tilde{\Omega}}\sin\left(\frac{\tilde{\Omega}}{2}t_{1}\right)\csc\left(\frac{\tilde{\Omega}}{2}t_{2}\right)\sin\left(\frac{\tilde{\Omega}}{2}(t_{2}-t_{1})\right)f_{1}(t_{2}) & t_{1} \leq t_{2} \leq t_{p} \\ \frac{\Omega}{\tilde{\Omega}}\sin\left(\frac{\tilde{\Omega}}{2}t_{1}\right)\csc\left(\frac{\tilde{\Omega}}{2}t_{p}\right)\sin\left(\frac{\tilde{\Omega}}{2}(t_{p}-t_{1})\right)f_{1}(t_{2}) & t_{1} \leq t_{p} < t_{2} \\ 0 & \text{otherwise} \end{cases}$$
(3.34)

where I have restricted $t_1 \leq t_2$ by our time ordering convention. This factorization is not the same as the separability approximation discussed in section 3.2.2. In that case, we were separating the two-photon density function ξ_2 into two one-photon density functions ξ_n and ξ_1 . In this case, we are factoring the two-photon density function ξ_2 into two two-photon amplitudes f_2 and f_2^* . The former case is valid when the two photons are not entangled, but they may be temporally impure. The latter case is valid when the total two-photon component is temporally pure, but there may still be entanglement. As I will discuss below, there can be situations where both of these cases apply so that the two photons described by ξ_2 are both unentangled and temporally pure.

Before moving on, I would like to note that the above solutions for the one and two-photon temporal wavefunctions are also derived in section 4.1 of Ref. [128] by explicit consideration of the full emitter-waveguide system undergoing reversible evolution. However, the equivalent solutions (up to a difference in normalization convention) presented here are lacking information about the relative phase between f_1 and f_2 . This is because we cannot obtain ϕ_1 and ϕ_2 when deriving the amplitudes by factoring ξ_1 and ξ_2 . To obtain this phase information, it is necessary to compute $\zeta_{n,m}$. However, often only ξ_n are relevant for a particular analysis, and the derivation using conditional correlations to compute density functions ξ_n instead of wavefunctions f_n can account for excess emitter decoherence and can be applied to a wide variety of emitter systems in a straightforward way.

I have chosen to write f_2 in terms of f_1 to illustrate some important points about f_2 . First, we can see that f_2 is implicitly dependent on t_2 through $f_1(t_2)$ when $t_1 \leq t_p \leq t_2$. Hence, it can be separated into $f_2(t_1, t_2) = f_n(t_1)f_1(t_2)$, which is the pure-state analog to the separability assumption made in section 3.2.2 for ξ_2 . This separability is not true when $t_1 \le t_2 \le t_p$, because f_2 is dependent on t_2 explicitly. This implies that the two photons are entangled in time. Second, when t_p is small compared to $1/\gamma$, the part of f_2 corresponding to $t_1 \le t_2 \le t_p$ has a negligible contribution. Therefore, when $t_p \ll 1/\gamma$ we have $f_2(t_1, t_2) \simeq f_n(t_1)f_1(t_2)$ and $\tilde{\Omega} \simeq \Omega$, where the 'noise' photon due to re-excitation has a temporal amplitude of

$$f_{\rm n}(t) = \frac{\sqrt{p_1 \gamma} e^{i\phi_2}}{\sqrt{p_2}} \sin\left(\frac{\Omega}{2}t\right) \csc\left(\frac{\Omega}{2}t_{\rm p}\right) \sin\left(\frac{\Omega}{2}(t_{\rm p}-t)\right)$$
(3.35)

for $t \le t_p$ and $f_n(t) = 0$ otherwise. This separability of 'noise' from 'signal' agrees with the observations made in section 3.2.2 and the assumptions of the derivation in section 3.2.1.

The result that ξ_1 and ξ_2 correspond to temporally pure quantum states when $\gamma^* = 0$ may seem physically obvious, but it is not mathematically obvious in the context of the photon number decomposition. When solving the evolution from the total emitter-waveguide Hamiltonian as in Ref. [128], the temporal amplitudes f_1 and f_2 are solved naturally by applying a unitary transformation to the total system state. Once the emitter has decayed to the ground state, the waveguide mode is left in a pure quantum state. However, using the photon number decomposition, the entire derivation was performed in the mixed state picture of the reduced system evolution. I find it interesting that the dynamics of just the emitter, which experiences decoherence during emission (recall section 1.3.1), can still allow us to compute the temporally-pure state of the waveguide after emission.

Let us now turn to the case with a nonzero pure dephasing rate. By evaluating ξ_1 , we can see that the temporal coherence, $\xi_1(t_1, t_2)$ when $t_1 \neq t_2$, is suppressed by pure dephasing (see Fig. 3.8). However, the intensity $\xi_1(t, t)$ remains mostly unchanged for $\gamma^* < \Omega$. Any small change in the intensity is due to the damping of the coherent driving.

The photon number decomposition approach is not only useful for analyzing single-photon generation, it is also a powerful tool if one needs to access and analyze a particular photon-number



Figure 3.8: Single-photon temporal density function $\xi_1(t, t')$ for a two-level system coherently driven with a square π pulse (a) without pure dephasing and (b) with pure dephasing. The dashed blue line shows the single-photon intensity $\xi_1(t, t) = N(t)$ corresponding to the diagonal of the density function. Chosen parameters are $\Omega/\gamma = 5$, $\eta_r = 1$, and $\gamma^* = \gamma/2$ if not zero.

subspace of the full photonic state, even if there are potentially many photons emitted. As an interesting example, consider a 7π pulse for $\Omega = 3\gamma$. In this case, we expect significant multiphoton processes, yet we can easily extract the temporal density function of just the single-photon subspace. In Fig. 3.9, we can see quite a large effect of pure dephasing on both the temporal coherence and the intensity profile of the single photon component for a 7π pulse. The shape of this temporal density function suggests that the single-photon subspace is in an entangled state of four well-defined time-bin modes—taking a form similar to $(|1000\rangle + |0100\rangle + |0010\rangle + |0001\rangle)/2$. Adding pure dephasing significantly reduces the temporal coherence of this state, with the most degradation affecting the coherence between time bins that are far-separated in time (coherence further from the diagonal). It is also important to note that I have shown only the result when neglecting losses so that $\eta_r = 1$. If there are multiphoton processes, the single-photon subspace. With this method, it is also very straightforward to include losses by decreasing η_r .

Unlike the single-photon density function, the two-photon density function ξ_2 cannot be easily visualized. This is because it is a four-dimensional function of time. It is more enlightening to



Figure 3.9: Single-photon temporal density function $\xi_1(t, t')$ for a two-level system coherently driven by a square 7π pulse (a) without pure dephasing and (b) with pure dephasing. The dashed blue line shows the single-photon intensity $\xi_1(t, t) = N(t)$ corresponding to the diagonal of the density function. Chosen parameters are $\Omega/\gamma = 3$, $\eta_r = 1$, and $\gamma^* = \gamma/2$ for panel (b).



Figure 3.10: Two-photon amplitude correlation components and intensity correlation for a purelydephased two-level emitter coherently driven by a square π pulse. (a) The amplitude correlation of the first photon emitted of the two-photon component. (b) The amplitude correlation of the second photon emitted of the two-photon component. (c) The amplitude correlation between the first and second photons emitted of the two-photon component. (d) The intensity correlation function $G^{(2)}(t_1, t_2) = \xi_2(t_1, t_2, t_1, t_2)$ of the two photon component. Parameters chosen are $\gamma^* = \gamma/2$, $\eta_r = 1$, and $\Omega = 5\gamma$.



Figure 3.11: A comparison of measures for the indistinguishability of photons emitted from a pulsed two-level emitter experiencing emitter pure dephasing. (a) The intensity-normalized HOM visibility V_{HOM} , mean wavepacket overlap M, single-photon purity $1 - g^{(2)}$, and single-photon trace purity \mathcal{P}_1 plotted for different π -pulse widths t_p relative to the spontaneous emission rate γ . The quantity $M/(1 - g^{(2)})$ is also shown. This was derived in section 3.2 as an upper-bound approximation to I and \mathcal{P}_1 in the limit that $\gamma t_p \rightarrow 0$. (b) The same quantities as in panel (a) but for a nonzero emitter pure dephasing rate $\gamma^* = 0.1\gamma$. Note that the vertical axis scale is different in panel (b) than in panel (a). In both panels, the black dotted line illustrates V_{HOM} as computed from the full pulsed cavity-emitter model for $\hbar \gamma_r = 0.5 \mu \text{eV}$, $\hbar g = 17 \mu \text{eV}$, and $\hbar \kappa = 400 \mu \text{eV}$ corresponding to QD devices used in Refs. [vi] and [xiii] that fall well within the bad-cavity regime giving $R/\kappa \simeq 0.007$ for $\gamma = \gamma' = \gamma_r (1 + F_p) \simeq 3.4 \mu \text{eV}$. Note that an effective γ^* for these QD devices at 7K is usually estimated to be $\gamma^* \simeq 0.04\gamma'$, which falls between the cases of panels (a) and (b).

analyze the two photon intensity correlation $\xi_2(t_1, t_2, t_1, t_2)$ and the three components composing $\langle \hat{b}^{\dagger}(t')\hat{b}(t)\rangle_2$ (see Eq. (2.22)). Fig. 3.10 shows these four quantities for the two-photon component produced by a finite-width π -pulse. From this, we can clearly see that the first photon emitted is very temporally separated from the second. Also, we can see that the second photon has a shape quite similar to the single-photon component illustrated in Fig. 3.8.

Using ξ_1 , we are finally in a good position to check the validity of the estimation for I derived using the separable noise model and also compare it to other quantities describing HOM interference. Fig. 3.11 shows that, regardless of dephasing or pulse width, both the indistinguishability I and the trace purity of the single-photon component $\mathcal{P}_1 = \int \int |\xi_1(t,t')|^2 dt dt'$ at the source are bounded between the mean wavepacket overlap M and the curve $M/(1 - g^{(2)})$. In addition, the curve $M/(1 - g^{(2)})$ is a much more accurate (upper-bound) estimate of both of these quantities when $\gamma t_p < 1$. Furthermore, we can see that the indistinguishability as computed by $I = \gamma/\Gamma$ and



Figure 3.12: A comparison of the HOM visibility from a driven two-level emitter and from a driven cavity-emitter source in the bad-cavity regime. The HOM visibility V_{HOM} is plotted for a square π pulse of width $\gamma' t_p = 0.1$ where $\gamma' = (1 + F_p)\gamma$ and F_p is given by Eq. (3.6) that captures the Purcell saturation in the critical regime. To allow for a comparison of the approximation accuracy for different Purcell factors when dephasing is nonzero, I have scaled the dephasing rate γ^* proportional to the Purcell-enhanced decay rate. For a given R/κ and F_p of either 10 (solid curves) or 1000 (dashed curves), the values of g and κ are determined relative to γ , which would then fix the timescale if specified. The cavity-emitter non-Markovian behaviour during pulsed excitation suppresses $g^{(2)}$ when the system is not very far into the bad-cavity regime where $R/\kappa \ll 1$. This brings V_{HOM} (red curves) from the bad-cavity limit value (red dot-dashed horizontal lines) nearly up to the emission-limited indistinguishability I (black curves) computed using Eq. (3.1). Although I do not show the mean wavepacket overlap M, it is necessarily bounded between I and V_{HOM} . In addition, $(V_{\text{HOM}} + g^{(2)})/(1 - g^{(2)})$ would very closely approximate I in this scenario.

the trace purity of the single-photon component are only truly equal in the fast-pulse regime where $g^{(2)} \rightarrow 0.$

In Fig. 3.11, I also show an example of the HOM visibility computed using the cavity-emitter model described in section 1.3.4 compared to the cavity-enhanced 1-dimensional atom model described in section 1.3.6. For a system operating very far into the bad-cavity regime as in Refs. [vi] and [xiii], the two-level approximation used extensively in this section is quite accurate. However, adiabatic elimination of the cavity mode cannot capture the brief suppression of the Purcell factor during excitation [108]. This transient non-Markovian behaviour can increase V_{HOM} by suppressing re-excitation for pulses shorter than the Purcell-enhanced lifetime even when $R/\kappa < 1$. To illustrate this point more clearly, in Fig. 3.12 I have plotted V_{HOM} as a function of the bad-cavity parameter R/κ for two fixed Purcell factors $F_{\rm p} = 10$ and $F_{\rm p} = 1000$ where $F_{\rm p}$ is given by Eq. (3.6). For this plot, I have chosen to fix the excitation pulse so that $\gamma' t_{\rm p} = 0.1$, where $\gamma' = \gamma (1 + F_{\rm p})$ is the Purcell-enhanced decay rate. Thus, $0.1/\gamma'$ determines the pulse timescale, which may or may not be realistic for a given implementation. To relate the results in this section to the results of section 3.1 that neglect non-zero $g^{(2)}$, I have also plotted the upper-bound value of I given by Eq. (3.1) that was derived by considering only the emission dynamics of a cavity-emitter system. Note that R and γ^* are both defined slightly differently in Eq. (3.1), which comes from Ref. [iii], compared to Eq. (3.6), which uses the standard convention in this thesis. From this result, we can see that the two-level emitter model is accurate in the bad-cavity regime when $R/\kappa < 0.01$ (roughly implying $2g/\kappa < 0.1$) and that the accuracy is nearly independent of the inhibited Purcell factor for a given R/κ . In addition, we can see that the HOM visibility is truly dominated by the emission dynamics in the critical regime where $R/\kappa = 1$ due to the suppression of $g^{(2)}$ if the excitation pulse is much faster than $1/\gamma'$.

All of the values for $g^{(2)}$ computed in this thesis assume that the collected emission is perfectly filtered from the resonant excitation pulse. For example, this can be accomplished using a crosspolarization setup. However, often the excitation pulse is applied to the emitter through another mode of the same cavity that is used to collect the emission. If the polarization of excitation is not ideally matched to the correct cavity mode, the cavity can induce a birefringence effect [vi] where some of the excitation pulse is collected along with the emission from the emitter, leading to an increased $g^{(2)}$. This effect becomes worse with increasing excitation power, leading to an increased $g^{(2)}$ for short pulses [xiii] that may partially cancel any gain due to potential non-Markovian suppression of the Purcell effect. Cross-polarization also inherently limits the efficiency of the source due to necessarily filtering approximately half of the desired emission. Therefore, excitation techniques that allow for near-perfect filtering of the excitation pulse without sacrificing collection efficiency are highly desired. This can be accomplished by using two-photon excitation [126, 192], phonon-assisted off-resonant excitation at low temperature [x][84, 192–194], or cavities with highly non-degenerate orthogonally-polarized modes [195].

3.3.2 Number coherence

In the previous section, I discussed how we can access the temporal coherence within photon number subspaces of a photonic state produced by a driven emitter using the photon number decomposition. In this section, I will explore one approach to gaining information about the number coherence functions $\zeta_{n,m}$ of the photonic state.

Master equation unravellings have already been used to study classical homodyne measurements [91], which can give access to the homodyne photon statistics and phase information about a photonic state. However, in this section, I will demonstrate that applying the photon number decomposition to a pulsed self-homodyne measurement (see section 3.2.3) can also give us access to some information about the number coherence between distinct photon number subspaces. In particular, the coherence function $\zeta_{1,0}$. This technique corresponds to a physically realistic setup, and this setup was recently used to quantify the magnitude of the coherence produced by a coherently driven artificial atom [190]. The work of Ref. [190] inspired the material in this section and motivated a related project on photon number entanglement [xiv], which I will discuss in the sections following this one.

I will begin by studying a slightly simplified case compared to the previous section by first assuming that the emitter is prepared in a pure superposition state so that we are only analyzing the emission dynamics for when $g^{(2)} = 0$. In this case, our Liouville superoperator is $\mathcal{L}_s = -(i/\hbar)\mathcal{H} + \gamma \mathcal{D}(\hat{\sigma}) + 2\gamma^* \mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma})$ and $\hat{H} = \hbar \omega \hat{\sigma}^{\dagger}\hat{\sigma}$. Let us now consider the initial state $\hat{\rho}(0) = |\psi(0)\rangle \langle \psi(0)|$, where $|\psi(0)\rangle = \cos(\vartheta) |g\rangle + \sin(\vartheta) |e\rangle$ and $\vartheta = \Theta/2$ is half the pulse area of the perfect state preparation pulse. This allows our emitter to be prepared with some amount of coherence that will then be transferred to our photonic state.

For perfect state preparation, we can have at most one photon emitted from the system and so the photonic state $\hat{\varrho}$ is truncated at $n \leq 1$. As a consequence, any potential photon number coherence must be contained between the 0- and 1-photon subspaces. Thus, from the input-output relation $\hat{b} - \hat{b}_0 = \sqrt{\gamma \eta} \hat{\sigma}$, we must have $\sqrt{p_0 p_1} \zeta_{1,0}(t) = \sqrt{\gamma \eta} \langle \hat{\sigma}(t) \rangle$ since all higher-order coherence functions vanish: $\zeta_{n+1,n}(t) = 0$ for n > 0. This simplified scenario allows us to obtain $\zeta_{1,0}(t)$ using a standard approach, which will then allow us to verify that the self-homodyne photon number decomposition approach reproduces this known solution.

For the simplified scenario described above, it is straightforward to solve for $\zeta_{1,0}(t)$ without the photon number decomposition. To do so, we can first recognize that $p_n = 0$ for n > 1 and so we must have $p_1 = \eta \sin^2(\vartheta)$ and $p_0 = 1 - p_1$. We can then apply the full propagation superoperator $\mathcal{U}(t,0)$ to compute $\langle \hat{\sigma}(t) \rangle = \text{Tr} \left(S\mathcal{U}(t,0)\hat{\rho}(0) \right)$ and obtain

$$\zeta_{1,0}(t) = \sqrt{\gamma F_{\eta}} e^{-\Gamma t/2 - i\omega t}, \qquad (3.36)$$

where $F_{\eta} = \cos^2(\vartheta)/(1 - \eta \sin^2(\vartheta))$ is a factor that captures decoherence due to photon losses. This factor is important in the context of quantum repeaters [196] and it arises again in section 4.1.

Using $\langle \hat{\sigma}(t) \rangle$, we can also compute the self-homodyne signal V_{SH} from Eq. (3.28):

$$V_{\rm SH} = \lambda^{(1)} \cos(\phi) = I \cos^2(\vartheta) \cos(\phi), \qquad (3.37)$$

where $I = \gamma/\Gamma$ and the self-homodyne signal amplitude $\lambda^{(1)} = I \cos^2(\vartheta)$ is the total integrated single-photon coherence. We can see that the magnitude of the number coherence is limited by pure dephasing in the same way as the HOM visibility. To see this explicitly, we can compute the one-photon temporal density function for this model in a similar way to $\zeta_{1,0}$ using $p_1\xi_1(t, t') = \eta\gamma \langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle$, which gives

$$\xi_1(t,t') = \gamma e^{-\gamma(t+t')/2 - i\omega(t-t') - \gamma^* |t-t'|}.$$
(3.38)

From this, we derive the indistinguishability $\iint |\xi(t,t')|^2 dt dt' = I$, which in this scenario is equivalent to the HOM visibility. Knowing $\zeta_{1,0}$, ξ_1 , $p_1 = \eta \sin^2 \vartheta$, and $p_0 = 1 - p_1$, we have now reconstructed the source field photonic density operator $\hat{\rho}$ of the waveguide arising from the initial state $\hat{\rho}(0)$ of the emitter. If more than one photon can be emitted, then the amplitude $\langle \hat{\sigma}(t) \rangle$ and correlation function $\langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle$ will have contributions from multiple photon number subspaces. Hence, the approach illustrated above will not allow us to reconstruct the coherence between specific photon-number subspaces. In section 2.3, I illustrated how to decompose $\langle \hat{\sigma}^{\dagger}(t')\hat{\sigma}(t) \rangle$ into different photon number subspaces. Let us now consider a self-homodyne measurement to decompose $\langle \hat{\sigma}(t) \rangle$ in a similar way.

Consider the system describing the emission of two identical photonic states. One photonic state, which I will denote using the subscript r, provides the reference frame for our phase measurement. The other, which I will denote using s, is the photonic state that we would like to measure. These two states could be produced by two identical emitters, or by a single emitter sequentially excited, so long as the two photonic states are initially uncorrelated. For conceptual simplicity, let us describe the situation using two identical emitters. The reduced system evolution of two identical and independent emitters can be described by the Liouville superoperator $\mathcal{L} = \mathcal{L}_r \otimes \mathcal{I} + \mathcal{I} \otimes \mathcal{L}_s$ where $\mathcal{L}_r = \mathcal{L}_s$ is given as above.

I will now outline the general idea of the self-homodyne decomposition. Suppose that the waveguide modes collecting emission from these two emitters are interfered at a balanced beam splitter with the relative phase ϕ . Then, the measurement of a single photon at either detector will induce a joint measurement of the emitters described by the jump superoperators $\mathcal{J}_{\pm}\hat{\rho} = (\eta\gamma/2)(\hat{\sigma}_{\rm r} \pm \hat{\sigma}_{\rm s}e^{i\phi})\hat{\rho}(\hat{\sigma}_{\rm r}^{\dagger} \pm \hat{\sigma}_{\rm s}^{\dagger}e^{-i\phi})$ [91], where $\sigma_{\rm r} = \hat{\sigma} \otimes \hat{I}$ and $\hat{\sigma}_{\rm s} = \hat{I} \otimes \hat{\sigma}$. Here, I assume that both inputs of the beam splitter have the same intensity and both detectors have the same detection efficiency $\eta_{\rm d}$ so that $\eta = \eta_{\rm r}\eta_{\rm t}\eta_{\rm d}$, where $\eta_{\rm t}$ the transmission efficiency. Using these jump superoperators, we have that $\mathcal{L}_0 = \mathcal{L} - \mathcal{J}_+ - \mathcal{J}_-$ describes the reduced system evolution conditioned on no detection at either detector of our self-homodyne setup. We can then decompose the reduced system dynamics conditioned on the individual photon numbers at each detector.

Since we have at most two photons emitted in our idealized scenario (at most one from each emitter), the full propagation superoperator is decomposed into 7 conditional propagators. The vacuum case $\mathcal{U}_0(t, t_0)$ is computed from \mathcal{L}_0 in the usual way. Then, the two one-photon propagators

are given by

$$\mathcal{U}_{\pm}(t,t_0) = \int_{t_0}^t \mathcal{U}_{\mathbf{0}}(t,t') \mathcal{J}_{\pm} \mathcal{U}_{\mathbf{0}}(t',t_0) dt', \qquad (3.39)$$

and the four two-photon cases are

$$\mathcal{U}_{\pm\pm}(t,t_0) = \int_{t_0}^t \mathcal{U}_{\mathbf{0}}(t,t') \mathcal{J}_{\pm} \mathcal{U}_{\pm}(t',t_0) dt'$$

$$\mathcal{U}_{\pm\mp}(t,t_0) = \int_{t_0}^t \mathcal{U}_{\mathbf{0}}(t,t') \mathcal{J}_{\mp} \mathcal{U}_{\pm}(t',t_0) dt'.$$
(3.40)

Here, I am distinguishing between the detection order, but this may not always be the case in practice.

Before moving on to solve $\zeta_{1,0}$ with this approach, let us take a look at the detection probabilities of the self-homodyne measurement. These are given by

$$Pr(0) = \lim_{t_{f} \to \infty} Tr(\mathcal{U}_{0}(t_{f}, t_{0})\hat{\rho}(t_{0})) = (1 - \eta \sin^{2}(\vartheta))^{2} = p_{0}^{2}$$

$$Pr(\pm) = \lim_{t_{f} \to \infty} Tr(\mathcal{U}_{\pm}(t_{f}, t_{0})\hat{\rho}(t_{0})) = p_{1} \left(p_{0} \pm I \cos^{2}(\vartheta) \cos(\phi)\right)$$

$$Pr(\pm\pm) = \lim_{t_{f} \to \infty} Tr(\mathcal{U}_{\pm\pm}(t_{f}, t_{0})\hat{\rho}(t_{0})) = \frac{1}{4}(1 + I)\eta^{2} \sin^{4}(\vartheta) = \frac{1}{4}(1 + I)p_{1}^{2}$$

$$Pr(\pm\mp) = \lim_{t_{f} \to \infty} Tr(\mathcal{U}_{\pm\mp}(t_{f}, t_{0})\hat{\rho}(t_{0})) = \frac{1}{4}(1 - I)\eta^{2} \sin^{4}(\vartheta) = \frac{1}{4}(1 - I)p_{1}^{2}.$$
(3.41)

The detection probabilities sum to 1, as expected, which verifies the completeness of the decomposition. It is also clear from these probabilities that the self-homodyne decomposition approach captures the degradation of the HOM visibility due to dephasing, since the indistinguishability $0 \le I \le 1$ dictates the photon bunching probability.

Using the above detection probabilities, we can compute the normalized self-homodyne signal within the single-photon subspace of the measured joint photonic state

$$V_{\rm SH,1} = \frac{\Pr(+) - \Pr(-)}{\Pr(+) + \Pr(-)} = I\cos(\phi)F_{\eta} \ge V_{\rm SH}.$$
(3.42)

It is important to note that $V_{SH,1}$ is normalized by the probability of measuring the joint photonic
state in the single-photon subspace whereas the full signal $V_{\rm SH}$ is normalized by the total intensity. Hence, even though the only nonzero signal arises from the single-photon subspace, $V_{\rm SH,1}$ is not necessarily equal to the total signal $V_{\rm SH}$. This result illustrates two important points about self-homodyne measurements and photon-number post-selection. In general, post-selecting on photon number resolved detection produces loss-dependent quantities. However, post-selection can increase the self-homodyne signal and hence purify the coherence, which is evident from $V_{\rm SH,1} \ge V_{\rm SH}$. This inequality saturates in the limit that $\eta \sin^2(\vartheta) \ll 1$, which is valid when losses are high or driving is weak so that multi-photon events at the detectors are rare.

Let us now turn back to the original goal of finding $\zeta_{1,0}$. In the same spirit as section 2.4, we can compare the general form of the photonic state density operator of the joint waveguide system to the correlations of our two-emitter system. Doing this, we can see that

$$\langle \mathcal{V}_{\rm SH}(t) \rangle_1 = p_{1,0} \operatorname{Re} \left(\zeta_{\rm s,1,0}(t) \zeta^*_{\rm r,1,0}(t) \right) \equiv p_{0,1} \left(\zeta_{\rm r,1,0}(t) \right)^2 \cos(\phi),$$
 (3.43)

where $\mathcal{V}_{\text{SH}} = \mathcal{J}_+ - \mathcal{J}_-$ is the self-homodyne signal superoperator, $p_{1,0} = \Pr(+) + \Pr(-) = 2p_1p_0$ is the probability of detecting one photon, and $\langle \rangle_1$ is the expectation over the single-photon subspace of the joint photonic state. The coherence function $\zeta_{r,1,0}$ is *defined* to be real because we choose it to be the phase reference for the self-homodyne measurement. Also, since we have two identical photonic states, $\zeta_{s,1,0}(t) = \zeta_{r,1,0}(t)e^{i\phi}$. Eq. (3.43) is analogous to classical homodyne detection where, instead of $\zeta_{r,1,0}$, we would have a classical coherent state amplitude. Computing this expression for our simple scenario gives

$$\langle \mathcal{V}_{\rm SH}(t) \rangle_1 = \operatorname{Tr}(\mathcal{U}_0(\infty, t) \mathcal{V}_{\rm SH} \mathcal{U}_0(t, t_0) \hat{\rho}(t_0)) = \frac{\gamma}{2} e^{-\Gamma t} \cos(\phi) \sin^2(\Theta).$$
 (3.44)

Finally, combining Eqs. (3.43) and (3.44), we find

$$\zeta_{\rm s,1,0}(t) = \sqrt{\gamma F_{\eta}} e^{-\Gamma t/2} e^{i\phi} \tag{3.45}$$



Figure 3.13: The self-homodyne signal amplitude for a purely-dephased emitter driven by coherent pulses. (a) The total self-homodyne signal amplitude $\lambda^{(1)}$ (orange) and the signal amplitude in the single-photon subspace $\lambda_1^{(1)}$ (red for $\eta = 1$ and blue for $\eta = 0.5$) as a function of Rabi frequency Ω for a fixed pulse area of $\Theta = \pi/2$. (b) The self-homodyne signal amplitudes for different pulse areas while fixing $t_p \gamma = \pi/20$ and varying Ω . The dotted black sinusoidal line indicates the curve $\lambda^{(1)} = I \cos^2(\Theta/2)$ that is expected without re-excitation processes. The vertical gray lines indicate the values consistent in both plots. For both plots, I have fixed $\gamma^* = 0.1\gamma$ corresponding to I = 0.833, which is indicated by the horizontal black dashed line.

in the rotating frame of the reference state. By adding back the rotating frame $e^{-i\omega t}$ and taking $\phi = 0$, we recover the solution given in Eq. (3.36).

Of course, the power of this self-homodyne photon number decomposition is not fully illustrated with this simple example since we are able to derive the same result in a much simpler way at the beginning of the section. However, the agreement between these two approaches verifies that the self-homodyne measurement post-selected in the single-photon subspace extracts the coherence between the vacuum and the single-photon component. It also allows us to access the coherence functions between individual photon-number subspaces even when the emitter dynamics includes both multi-photon processes and excess decoherence for a wide variety of emitter models. Furthermore, it corresponds to a physically realistic optical setup that is already commonly used to characterize the indistinguishability of single photons.

As an illustrative example of the self-homodyne photon number decomposition approach, I have computed the first-order integrated coherence of the self-homodyne signal amplitude post-

selected in the single-photon subspace $\lambda_1^{(1)} = V_{\text{SH},1}/\cos(\phi)$ for a purely-dephased two-level emitter driven with a finite square pulse (see Fig. 3.13). I have plotted this value alongside the total selfhomodyne signal amplitude $\lambda^{(1)}$ for two different measurement efficiency values. Notably, there is a drastic suppression of the amplitude $\lambda^{(1)}$ at $\Theta = 2\pi$ which is not captured when neglecting re-excitation processes. This suppression is alleviated when post-selecting in the single-photon subspace provided that loss is not significant. From this result, we can see that the amount of coherence in the lossy regime is small unless the excitation power is weak (either $\Omega \ll \gamma$ or $\Theta \ll \pi$). This is a well-known conclusion and the reason why single-photon repeater protocols require weak excitation [197]. I will discuss more about this consequence in section 4.1 in the context of heralded spin-spin entanglement generation. Before getting to that discussion, I would like to briefly touch on how self-homodyne measurements can also be used to characterize photon number Bell states encoded in time.

3.3.3 Single-photon entanglement

It is well-known that a single photon can give rise to multi-partite entanglement [198]. The simplest example is achieved by separating a single photon into multiple spatial paths using beam splitters. A balanced beam splitter takes a single-photon state $|1\rangle$ and divides it into two equal spatial modes, where it becomes a single-particle Bell state of the form $|\psi^+\rangle = (|10\rangle + |01\rangle)/\sqrt{2}$. This operation can be repeated using multiple beam splitters, giving rise to a single excitation encoded in many spatial modes. The multi-partite single-particle entanglement carried by a single photon also becomes evident when mapped onto an ensemble of atoms using a frequency comb atomic memory [199], where it generates a Dicke state encoded in frequency.

As we have seen so far, the time dynamics of a single-photon wavepacket emitted by a twolevel system into a waveguide can be described using a temporal wavefunction $f_1(t)$. In the more general case, the photonic state can be described by a temporal density function $\xi_1(t, t')$ where $\xi_1(t, t') = f_1(t)f_1^*(t')$ for a pure state. Having access to the time-dependent dynamics of the single-photon wavepacket allows us to probe deeper quantum properties of a single photon, such as the amount of single particle entanglement encoded in time.

Consider the ideal single-photon state in the waveguide

$$|1\rangle = \int f_1(t)\hat{b}^{\dagger}(t) |0\rangle dt.$$
(3.46)

Suppose now that we have a detector that can resolve the time dynamics of this system. We can then divide our single photon into two time bins: an early photon which arrives before time T since the state preparation, and a late photon that arrives after time T. I will refer to T as the time bin threshold. Let us denote the early photon arriving during the first interval $\Delta S = (-\infty, T)$ using the subscript s and the late photon arriving during the interval $\Delta T = (T, \infty)$ using the subscript t. If our detector is perfectly capable of resolving these two time bins, we can write our single photon as a bipartite state

$$|1\rangle = \sqrt{\mu_{\rm s}}\hat{s}^{\dagger}|0\rangle + \sqrt{\mu_{\rm t}}\hat{t}^{\dagger}|0\rangle. \qquad (3.47)$$

The corresponding second-quantized time-bin modes are given by

$$\hat{s}^{\dagger} = \frac{1}{\sqrt{\mu_{\rm s}}} \int_{\Delta S} f_1(t) \hat{b}^{\dagger}(t) dt$$

$$\hat{t}^{\dagger} = \frac{1}{\sqrt{\mu_{\rm t}}} \int_{\Delta T} f_1(t) \hat{b}^{\dagger}(t) dt,$$
(3.48)

where $\mu_{\rm s} = \int_{\Delta S} |f_1(t)|^2 dt$ and $\mu_{\rm t} = \int_{\Delta T} |f_1(t)|^2 dt$.

Suppose now that we have an imperfect photonic state $\hat{\varrho}$. We can then compute the density matrix elements in the second-quantized picture $\varrho_{nmkl} = \langle nm | \hat{\varrho} | kl \rangle$, $|nm \rangle = (\hat{s}^{\dagger})^n (\hat{t}^{\dagger})^m | 0 \rangle / \sqrt{n!m!}$ for our ideal time bin modes. For simplicity, I use the abbreviated notation $\varrho_{1010} = \varrho_{ss}$, $\varrho_{0101} = \varrho_{tt}$, and $\varrho_{1001} = \varrho_{st}$. Using this second quantized picture, it is then straightforward to compute the entanglement fidelity and concurrence (see sections 1.4.5 and 1.4.6) of the imperfect single photon in the two-qubit photon-number subspace of $\{|00\rangle, |10\rangle, |01\rangle, |11\rangle\}$.

The entanglement carried by a single photon can be seen to be rooted in the process of

spontaneous emission. Recall from section 1.3.1 that, during spontaneous emission, the emitter briefly becomes entangled with its environment. This manifests as a dip in the purity of the emitter state. For example, consider a two level system in its excited state $|e\rangle$. As time passes, it decays to its ground state by spontaneously emitting a photon into the waveguide through a *coherent* interaction with the continuum of modes. At time T after state preparation, the joint emitter-waveguide system is in the state $\alpha |e\rangle |0\rangle + \beta |g\rangle |1\rangle$, where $|0\rangle$ and $|1\rangle$ represent the state of the time-bin mode \hat{s} . After complete emission, the state becomes $|g\rangle (\beta |10\rangle + \alpha |01\rangle) = |g\rangle (\beta \hat{s}^{\dagger} + \alpha \hat{t}^{\dagger}) |0\rangle$, where the emitter is now separable leaving a single-photon entangled state in time.

Following the example above, let us consider the impure single photon produced by a perfectly excited purely-dephased two-level emitter, as described by ξ_1 in Eq. (3.38). If the emitter has no pure dephasing, we expect that $f_1(t) = \sqrt{\gamma}e^{-\gamma t/2 - i\omega t}$. Using ξ_1 and f_1 , we can then compute the entanglement figures of merit for ρ . For this scenario, Fig. 3.14 shows the density matrix elements in the bipartite time bin basis along with the computed Bell-state fidelity and entanglement concurrence as a function of time bin threshold T. The fidelity compared to the ideal Bell state $|\psi^+\rangle = (\hat{s}^{\dagger} + \hat{t}^{\dagger}) |0\rangle /\sqrt{2}$ and the concurrence are maximized near when the time bin threshold is chosen to be at the half-life of the emitter $T = \ln 2/\gamma$. However, the presence of pure dephasing shifts this ideal choice of threshold to smaller values.

Let us now discuss the properties of photons within each time bin for an impure single-photon state. From ξ_1 , we can compute the total indistinguishability $I = \int \int |\xi_1(t, t')|^2 dt = \gamma/\Gamma$ as before. The indistinguishability I can then be divided into its bipartite components for modes \hat{s} and \hat{t} by:

$$I = \frac{\mu_{\rm s}^2 I_{\rm ss} + 2\mu_{\rm s}\mu_{\rm t} I_{\rm st} + \mu_{\rm t}^2 I_{\rm tt}}{\mu^2}$$
(3.49)

where $\mu = \mu_s + \mu_t$ is the total intensity. The constituent parts of indistinguishability for postselected intervals ΔS and ΔT are also explicitly related to the total indistinguishability I of the



Figure 3.14: Bipartite time-bin decomposition of a single photon: density matrix elements, Bell-state fidelity \mathcal{F}_{ψ^+} , and concurrence \mathcal{C}_{ψ^+} . (a) The density matrix elements along with Bell-state fidelity and concurrence in the second-quantized time-bin basis defined by the threshold T. For this panel, I have used $\gamma^* = 0.1\gamma$. The vertical line shows the half-life $\gamma T = \ln(2)$. (b) The concurrence as a function of time bin threshold T and pure dephasing rate γ^* .

single-photon:

$$I_{\rm ss} = \frac{1}{\mu_{\rm s}^2} \int_{\Delta S} \int_{\Delta S} |\xi_1(t,t')|^2 dt dt' = \frac{I}{\beta^4} \left[1 - \frac{\alpha^4 - 2I\alpha^{2/I}}{1 - 2I} \right]$$

$$I_{\rm st} = \frac{1}{\mu_{\rm s}\mu_{\rm t}} \int_{\Delta S} \int_{\Delta T} |\xi_1(t,t')|^2 dt dt' = \frac{I^2}{\alpha^2 \beta^2} \left[\frac{\alpha^4 - \alpha^{2/I}}{1 - 2I} \right]$$

$$I_{\rm tt} = \frac{1}{\mu_{\rm t}^2} \int_{\Delta T} \int_{\Delta T} |\xi_1(t,t')|^2 dt dt' = I,$$

(3.50)

where $\mu_{\rm s} = \int_{\Delta S} \xi_1(t,t) dt = \beta^2 = 1 - e^{-\gamma T}$ and $\mu_{\rm t} = \int_{\Delta T} \xi_1(t,t) dt = \alpha^2 = 1 - \beta^2$. Fig. 3.15 shows this time bin decomposition of indistinguishability in panel (a) and the weights of their contribution to the total indistinguishability in panel (b).

The bipartite time bin decomposition illustrates how temporal post-selection of photons that are emitted early can increase the indistinguishability I_{ss} at the cost of a reduced efficiency $\mu_s < 1$. This fast temporal post-selection eliminates the tail of the photon's exponential profile, creating a sharper single-photon pulse in time. The temporal truncation can alternatively be seen as a spectral broadening of the post-selected photon so that it begins to overcome the incoherent broadening



Figure 3.15: **Bipartite time-bin decomposition of the single-photon indistinguishability.** (a) The three components of the bipartite indistinguishability I_{ss} , I_{tt} and I_{st} composing the total indistinguishability I = 0.8. (b) The weights of contribution for each of the three components of indistinguishability in panel (a) normalized to the squared intensity μ^2 . The vertical line corresponds to the half-life of the emitter decay.

induced by dephasing. However, the indistinguishability I_{tt} for post-selected photons that are emitted late remains constant regardless of T. This is because rejecting early photons does not alter the temporal shape of late photons. That is, the tail always looks like the tail no matter how much we remove from the beginning of the profile. Hence, in this case, the spectrum is not broadened and the indistinguishability is not increased.

The value of I_{st} quantifies the amount of two-mode entanglement that the single-particle state carries. It is also much easier to measure than ρ_{st} in practice because ρ_{st} necessarily requires knowledge about the ideal mode amplitude f_1 . It is then natural to ask how closely the post-selected indistinguishability measurements can estimate the density matrix elements in the second-quantized picture of the ideal time bin modes. To this end, we can find a good estimate by noting that the postselected values of I are very similar to the overlaps needed to obtain the density matrix elements, except that they account for decoherence twice compared to the elements of ρ and are normalized differently with respect to intensity. From this, we know that a lower bound on the density matrix elements must be $I_{ab}\sqrt{\mu_a\mu_b} \leq \rho_{ab}$. However, we can do better than this if we attempt to correct for this double-counting of decoherence by taking a square root of the weighted component of I. That is, we can also guess that $\rho_{ab}\sqrt{\mu_a\mu_bI_{ab}}$ for $\{a, b\} \in \{t, s\}$. Fig. 3.16 shows a comparison



Figure 3.16: A comparison of the bipartite time bin density matrix elements and their estimates from indistinguishability measurements. (a) The density matrix elements in the second-quantized basis (solid black curves) compared to the upper and lower estimates using bipartite indistinguishability measurements (colored regions) for $\gamma T = \ln(2)$. (b) The same comparison but for fidelity \mathcal{F}_{ψ^+} and concurrence \mathcal{C}_{ψ^+} computed using the values in panel (a).

between these two estimated elements using I_{ab} compared to the true values ρ_{ab} along with the implied fidelity and concurrence values. From this, we can see that the latter approximation will overestimate the values of the density matrix elements, but it is a much more reasonable approximation for highly-indistinguishable photons ($\gamma^* < \gamma$) than the lower-bound approximation.

The single-particle photon number Bell state $|\psi^+\rangle$ and its analysis can be seen as a timeresolved perspective on indistinguishability measurements. However, the underlying concept of using spontaneous emission as a resource to generate entanglement can be extended to more novel situations. I will explore this extension in the following section.

3.3.4 Two-photon coherence

If we can create the $|\psi^+\rangle$ Bell state by exploiting the entangling nature of spontaneous emission, it is natural to ask whether the other three Bell states can also be generated in this way. Interestingly, the answer seems to be yes! The state $|\psi^-\rangle$ can be produced by performing a rapid relative phase flip between state $|e\rangle$ and $|g\rangle$ at the half-life of emission, essentially flipping the phase of the dipole halfway through emitting the photon. The states $|\phi^{\pm}\rangle$ can be generated by applying a second excitation pulse at time T after the first pulse to flip the population of the two-level system. This latter case can be explained by noting that a perfect π pulse takes $\alpha |e\rangle |0\rangle + \beta |g\rangle |1\rangle$ to $\alpha |g\rangle |0\rangle + \beta |e\rangle |1\rangle$. This leaves the photonic modes in the state $\alpha |00\rangle + \beta |11\rangle = (\alpha + \beta \hat{s}^{\dagger} \hat{t}^{\dagger}) |0\rangle$.

The photon number $|\psi^{\pm}\rangle$ Bell states are very different from the $|\phi^{\pm}\rangle$ Bell states. Although both carry the same average energy and have the same intensity profiles, they have very different quantum properties. The $|\psi^{\pm}\rangle$ states are single-photon states that carry significant temporal coherence whereas the $|\phi^{\pm}\rangle$ states have intensity correlations and carry two-photon coherence. Because of this, they require very different techniques to characterize. We can no longer rely on indistinguishability measurements to understand $|\phi^{\pm}\rangle$. In this section, I will analyze the state $|\phi^{+}\rangle$ generated by an imperfect emitter and discuss how to quantify the amount of entanglement using a self-homodyne setup.

To observe the intensity correlations caused by the state $|11\rangle$ within $|\phi^+\rangle$, we can use timeresolved $g^{(2)}$ measurements to look for coincidence counts between time bin modes. In addition, as I alluded to in section 3.2.3, by simultaneously monitoring the self-homodyne signal and the HOM interference visibility as a function of the relative phase, it is possible to observe and quantify the effect of two-photon coherence between the vacuum state $|00\rangle$ and the two-photon state $|11\rangle$. This technique was employed in the experimental paper of Ref. [xiv] to characterize $|\phi^+\rangle$ produced by a coherently driven artificial atom using a self-homodyne setup.

Following the approach in the previous section, let us compute the photonic state $\hat{\varrho}$ corresponding to $|\phi^+\rangle$ generated from a purely-dephased two-level system. Let us again assume that we are using perfect excitation pulses so that we can neglect re-excitation. In this case, the photonic state is truncated at $n \leq 2$ photons and so the density matrix elements that we are interested in can be directly related to correlations of the reduced system operators without considering a photonnumber decomposition. Let S, \mathcal{R} , $\mathcal{J} = S\mathcal{R}$ and \mathcal{X} be the superoperators defined by $S\hat{\rho} = \hat{\sigma}\hat{\rho}$, $\mathcal{R}\hat{\rho} = \hat{\rho}\hat{\sigma}^{\dagger}$, and $\mathcal{X}\hat{\rho} = \hat{\sigma}_x\hat{\rho}\hat{\sigma}_x$. The superoperator \mathcal{X} is the action of an ideal π -pulse, which we take to arrive at time τ after the initial state preparation. In addition, let us assume we start in the excited state $\hat{\rho}(0) = |e\rangle\langle e|$. Then, we can compute all the necessary time-dependent expectation values and correlation functions of our system in the standard way (see section 1.2.5).

First, we can verify that the first-order coherence $\Lambda^{(1)}(t) = \langle \hat{\sigma}(t) \rangle$ vanishes for all time because we are only ever using perfect π -pulses, leaving no coherence:

$$\Lambda^{(1)}(t) = \begin{cases} \sqrt{\gamma} \operatorname{Tr}(\mathcal{S}\hat{\rho}(t)) = 0 & 0 \le t \le \tau \\ \sqrt{\gamma} \operatorname{Tr}(\mathcal{SU}(t,\tau)\mathcal{X}\hat{\rho}(\tau)) = 0 & t > \tau \end{cases}$$
(3.51)

Second, we can show that the system produces no single-photon temporal coherence between time bins, but that each time bin itself can contain a single photon:

$$G^{(1)}(t,t') = \begin{cases} \gamma \operatorname{Tr}(\mathcal{R}\mathcal{U}(t',t)\mathcal{S}\hat{\rho}(t)) = \xi_1(t,t') & 0 \le t \le t' \le \tau \\ \gamma \operatorname{Tr}(\mathcal{R}\mathcal{U}(t',\tau)\mathcal{X}\mathcal{U}(\tau,t)\mathcal{S}\hat{\rho}(t)) = 0 & 0 \le t \le \tau < t' \\ \gamma \operatorname{Tr}(\mathcal{R}\mathcal{U}(t',t)\mathcal{S}\mathcal{U}(t,\tau)\mathcal{X}\hat{\rho}(\tau)) = (\beta^2/\alpha^2)\xi_1(t,t') & 0 \le \tau < t \le t' \end{cases}$$
(3.52)

where now $\alpha^2 = e^{-\gamma\tau}$ and $\beta^2 = 1 - \alpha^2$ depend on the pulse separation τ . Third, we can show that there are intensity correlations $G^{(2)}(t, t')$ between the two time bins, but not within each time bin:

$$G^{(2)}(t,t') = \begin{cases} \gamma^2 \operatorname{Tr}(\mathcal{JU}(t',t)\mathcal{J}\hat{\rho}(t)) = 0 & 0 \le t \le t' \le \tau \\ \gamma^2 \operatorname{Tr}(\mathcal{JU}(t,\tau)\mathcal{XU}(\tau,t)\mathcal{J}\hat{\rho}(t)) = \gamma^2 e^{-(t+t'-\tau)\gamma} & 0 \le t \le \tau < t' \\ \gamma^2 \operatorname{Tr}(\mathcal{JU}(t',t)\mathcal{JU}(t,\tau)\mathcal{X}\hat{\rho}(\tau)) = 0 & 0 \le \tau < t \le t' \end{cases}$$
(3.53)

Finally, we can show that there is nonzero two-photon coherence $\Lambda^{(2)}(t, t') = \langle \hat{b}(t)\hat{b}(t') \rangle$ between different time bins, and the two-photon coherence is equal to the temporal density function of a single photon:

$$\Lambda^{(2)}(t,t') = \begin{cases} \gamma \operatorname{Tr}(\mathcal{SU}(t',t)\mathcal{S}\hat{\rho}(t)) = 0 & 0 \le t \le t' \le \tau \\ \gamma \operatorname{Tr}(\mathcal{SU}(t,\tau)\mathcal{XU}(\tau,t)\mathcal{S}\hat{\rho}(t)) = \xi_1(t,t') & 0 \le t \le \tau < t' \\ \gamma \operatorname{Tr}(\mathcal{SU}(t',t)\mathcal{SU}(t,\tau)\mathcal{X}\hat{\rho}(\tau)) = 0 & 0 \le \tau < t \le t' \end{cases}$$
(3.54)

Using these correlation functions, we can solve for μ and $g^{(2)}$ by

$$\mu = \int_0^\infty G^{(1)}(t,t)dt = \int_0^\tau \xi_1(t,t)dt + \frac{\beta^2}{\alpha^2} \int_\tau^\infty \xi_1(t,t)dt = 2\beta^2$$
(3.55)

and

$$g^{(2)} = \frac{2}{\mu^2} \int_0^\infty \int_t^\infty G^{(2)}(t,t') dt' dt = \frac{\gamma^2}{2\beta^4} \int_0^\tau \int_\tau^\infty e^{-(t+t'-\tau)\gamma} dt' dt = \frac{1}{2\beta^2}.$$
 (3.56)

Using the photon statistics relations (see section 2.5), and knowing $p_{n\geq 3} = 0$, we have $\mu = p_1 + 2p_2 = 2\beta^2$ and $g^{(2)} = 2p_2/\mu^2 = \mu^{-1}$. This implies $p_1 = 0$ and $p_2 = \beta^2$, leaving $p_0 = 1 - \beta^2 = \alpha^2$. Hence, the single-photon component vanishes for all pulse separations, leaving only the two-photon component and the vacuum component.

The last thing we need is the two-photon density function $\xi_2(t_1, t_2, t'_1, t'_2)$. This can be solved using the form of Eq. (2.26), although we can use \mathcal{U} in place of \mathcal{U}_0 since we know that all components for $n \ge 3$ vanish. With this approach, we can verify that the only non-vanishing component corresponds to $t_1, t'_1 \le \tau \le t_2, t'_2$ and is given by the separable function

$$\xi_{2}(t_{1}, t_{2}, t_{1}', t_{2}') = \frac{1}{\alpha^{2}\beta^{2}} \begin{cases} \xi_{1}(t_{1}, t_{1}')\xi_{1}(t_{2}, t_{2}') & t_{1}, t_{1}' \leq \tau \leq t_{2}, t_{2}' \\ 0 & \text{otherwise} \end{cases}$$
(3.57)

Note that the division by $\alpha^2 \beta^2 = p_0 p_2$ is necessary so that ξ_2 is normalized. We now have enough information to write down the photonic density operator of our non-ideal $|\phi^+\rangle$ state:

$$\hat{\varrho} = p_0 |0\rangle \langle 0| + \frac{p_2}{\alpha^2 \beta^2} \int_0^\tau \int_0^\tau \xi_1(t_1, t_1') \int_\tau^\infty \int_\tau^\infty \xi_1(t_2, t_2') \hat{b}^{\dagger}(t_2) \hat{b}^{\dagger}(t_1) |0\rangle \langle 0| \hat{b}(t_1') \hat{b}(t_2') dt_1 dt_1' dt_2 dt_2' + \int_0^\tau \int_\tau^\infty \xi_1(t_1, t_2) \hat{b}^{\dagger}(t_2) \hat{b}^{\dagger}(t_1) |0\rangle \langle 0| dt_1 dt_2 + \text{h.c.}$$
(3.58)

Up until this point, I have not discussed the definition for our time bin measurement. We should expect that the best option to maximize entanglement is $T = \tau$, as this is the natural choice



Figure 3.17: The concurrence of a $|\phi^+\rangle$ photonic Bell state generated by a pulsed two-level emitter experiencing pure dephasing. (a) The concurrence \mathcal{C}_{ϕ^+} as a function of time bin threshold T and pulse separation τ for $\gamma^* = 0$. The white lines show the $\gamma \tau = \gamma T = \ln(2)$ half-life condition. (b) The concurrence along the $T = \tau$ condition (diagonal dashed blue line in panel (a) as a function of pure dephasing γ^* .

to capture all of the two-photon coherence. By computing the density matrix elements in the second-quantized basis, we can compute the concurrence in the $\{|00\rangle, |10\rangle, |01\rangle, |01\rangle, |11\rangle\}$ subspace as before. Now, our solution is a function of both the time bin threshold T and pulse separation τ (see Fig. 3.17 (a)). We can see that the concurrence for a given pulse separation τ is maximized by choosing $T = \tau$, as expected. Along this condition, we can also look at how dephasing degrades the concurrence to find that it is very similar to the single-particle $|\psi^+\rangle$ case (see Fig. 3.17 (b)).

When choosing $T = \tau$, we can show that the density matrix elements of the $\rho_{1111} = \langle 11|\hat{\rho}|11\rangle$ and $\rho_{0011} = \langle 00|\hat{\rho}|11\rangle$ are related to ρ_{ss} , ρ_{tt} , and ρ_{st} from the previous section. By performing the overlaps using time bin modes Eq. (3.48), we find that $\rho_{1111} = p_2 \rho_{ss} \rho_{tt} / (\alpha^2 \beta^2)$ and $\rho_{0011} = \rho_{st}$. Also, we have $\rho_{0000} = \alpha^2 = p_0$.

As before, it would be useful to have an alternative, simpler, way to estimate the density matrix elements. This can be done with the help of a time-resolved self-homodyne measurement (recall section 3.2.3 for self-homodyne measurements), which gives us access to M_{ab} and $\lambda_{ab}^{(2)}$ that can be post-selected into our two time bins $a, b \in \{t, s\}$ by an appropriate choice of integration limits. For simplicity, let us still focus on the case where $T = \tau$. Using $G^{(1)}$, we can find that $M_{st} = 0$,



Figure 3.18: A comparison of the bipartite time bin density matrix elements for a $|\phi^+\rangle$ photon number Bell state and their estimates from self-homodyne measurements. (a) The density matrix elements in the second-quantized basis (solid black curves) compared to the estimated values using bipartite indistinguishability measurements (blue region, top panel) and bipartite two-photon coherence measurements (purple region, bottom panel) for $\gamma T = \ln(2)$. The third relevant density matrix element $\rho_{0000} = p_0$ is constant for any given T and $p_0 = 1/2$ for $T = \ln(2)$. (b) The fidelity and concurrence computed using the values in panel (a). The fidelity approaches 1/4 for large γ^* due to the constant contribution from p_0 .

 $M_{\rm ss} = I_{\rm ss}$ and $M_{\rm tt} = I_{\rm tt}$. Furthermore, using $\Lambda^{(2)}$, we can find that $\lambda_{\rm ss}^{(2)} = \lambda_{\rm tt}^{(2)} = 0$ and $\lambda_{\rm st}^{(2)} = I_{\rm st}$. The values for $I_{\rm ss}$, $I_{\rm st}$, and $I_{\rm tt}$ are given in Eq. (3.50). From this, and the fact that $\rho_{0011} = \rho_{\rm st}$, it is clear that applying a second pulse at $\tau = T$ transfers the temporal coherence between $|10\rangle$ and $|01\rangle$ given by $I_{\rm st}$ to a two-photon coherence between $|00\rangle$ and $|11\rangle$, as expected. It also implies that, for the condition $\tau = T$, we can estimate the density matrix elements by measuring the integrated two-photon coherence to get $\lambda_{\rm st}^{(2)}\sqrt{\mu_{\rm s}\mu_{\rm t}} \leq \rho_{0011} \simeq \sqrt{\mu_{\rm s}\mu_{\rm t}\lambda_{\rm st}^{(2)}}$ and the mean wavepacket overlap to get $M_{\rm ss}M_{\rm tt}\mu_{\rm s}\mu_{\rm t}/p_2 \leq \rho_{1111} \simeq \mu_{\rm s}\mu_{\rm t}\sqrt{M_{\rm ss}M_{\rm tt}}/p_2$ using the time-resolved self-homodyne setup. By comparing these estimates to the full overlaps computed using f_1 , we can see that they are very accurate (see Fig. 3.18). In addition, the concurrence is very similar to the case for $|\psi^+\rangle$ given in Fig. 3.18, even though the density elements are different. This is a particularity associated with the choice $\tau = T = \ln(2)/\gamma$, which maximizes the entanglement in both cases and when γ^* is small.

The ability to deterministically produce entangled states using only two-level single-photon sources could be very beneficial for quantum information applications. This is because single-photon sources based on two-level system models are well developed [68, 153]. The scheme

presented in this section to produce and measure photon-number Bell states can also be extended by applying additional pulses before the emitter has fully decayed. This extension can allow for the deterministic generation of multi-partite W-class entangled states [xiv], which may have applications in all-optical quantum repeaters or other quantum communication schemes.

Deterministic generation of entangled photonic states using optically-active defects can also be implemented if the defect contains additional level structure allowing for a quantum memory [200], such as a ground-state spin qubit. This additional level structure expands the class of entangled states that can be produced. The basic principle takes advantage of the persistent spin-photon entanglement generated between a photonic state produced by the defect and the spin state of the emitter. In the following chapter, I will analyze how this spin-photon entanglement generated by an emitter containing a spin qubit can be used to generate spin-spin entanglement between two remote defects.

Chapter 4

Spin-spin entanglement

Photon-mediated entanglement generation between spins in different quantum defects is important for implementing quantum repeaters [27, 196, 197, 201, 202] and distributed quantum computing protocols [203, 204] for quantum networks. Such a quantum network could form the basis for a global quantum internet [31–33]. The many benefits and applications of quantum networks are detailed in section 1.1.2.

The nodes of a quantum network are composed of clusters of stationary qubits (spins). These qubits serve as a quantum memory to store quantum information. They must also interact locally, enabling quantum logic gates for quantum information processing. To carry quantum information over long distances, nodes of a quantum network must couple to flying qubits (photons). This coupling is mediated by a quantum interface between matter and light that operates in a quantum coherent way to preserve the quantum information [57, 205]. Photonic qubits connect nodes together and distribute quantum information over the network. They can propagate using direct optical links over short distances (up to \sim 500 km). For longer distances (up to \sim 2000 km), a quantum repeater could be used to overcome transmission loss [27]. For a truly global network, satellites in low-earth or geostationary orbits might be used [32].

For distances up to about 500 km, photons of telecommunication wavelength (~ 1550 nm) can be faithfully transmitted directly through readily-available fiber-optic infrastructure [22, 206]. By performing a joint Bell-state measurement [i] on photons emitted from two different nodes, the spin qubits within separated nodes can be projected onto a maximally entangled state [123].

Unfortunately, losses of about 0.2 dB/km in optical fibers plague long-distance single-photon transmission. In classical fiber optics communication, this problem is solved by a repeater component that amplifies the signal. However, because of the no-cloning theorem in quantum physics [207, 208], it is not possible to faithfully duplicate the quantum state of a single photon. Therefore, to extend single-photon transmission distances, one can use a quantum repeater protocol [27].

The basic principle of a quantum repeater is to first generate entanglement between intermediate nodes over shorter (<500 km) distances through direct fiber or free-space transmission. These short-range entanglement connections can then be extended over a long distance (<5000 km) by performing a local exchange of quantum information between stationary qubits in each intermediate node of repeater, known as an entanglement swapping. Although there are many ensemble-based approaches to quantum repeater memory nodes [202], I will focus on using single spin qubits. In this case, entanglement swapping can be accomplished deterministically by performing a combination of local gates and spin-state measurements [iv].

In this chapter, I explore two main types of photon-mediated spin-spin entanglement related to quantum networks for single defects: (1) between two remote optically-active defects that cannot interact locally and (2) between two local defects that can interact via coupling to the same optical cavity mode. These two types of spin-spin entangling processes are directly applicable to the entanglement generation and deterministic entanglement swapping steps, respectively, of quantum repeater schemes using single defects [iv][vii].

To achieve remote entanglement between systems that emit visible or near-infrared photons, it is convenient to use pulsed schemes that herald entanglement by the detection of single photons [123, 209–211]. Such schemes have already been implemented using atomic ensembles [212, 213], single trapped atoms [214] or ions [215, 216], quantum dots [217, 218], and defects in diamond [10, 151]. This is usually done by first generating spin-photon entanglement between the spin state

of the individual defect and their emission. Then, by performing a probabilistic joint entangling measurement on the emission from each remote system, the joint spin state of the systems is projected onto an entangled state.

For local interactions, if the cavity-emitter system is in the strong-coupling regime, then the vacuum Rabi oscillations can be used to engineer a strong emitter-emitter interaction before dissipation occurs. However, since the strong-coupling regime is not ideal for remote entanglement generation that relies on the Purcell effect to increase fidelity and efficiency, it is attractive to consider deterministic local entanglement schemes that can operate under the same bad-cavity conditions needed for high-fidelity remote entanglement generation. This is possible by engineering an adiabatic cavity-mediated interaction between two defects whereby a photon is never actually created in the cavity mode [vii], and hence decoherence due to the cavity dissipation is suppressed.

As discussed in section 1.1.3, solid-state systems are particularly attractive as a quantum technology platform for their scalability, ease of manufacturing, and potential to integrate with classical information processing hardware [56, 57]. However, solid-state systems suffer from decoherence (see section 1.1.4) that limits the initial amount of generated entanglement between systems as well as their longevity as a quantum memory [204]. Spin decoherence can be caused by the interaction of the spin qubit with a surrounding bath of nuclear spins [219] or lattice phonons [220]. These interactions can randomly flip the spin state of the qubit during or after entanglement generation, or cause a pure-dephasing of the spin coherence. Phonon interactions can also cause homogeneous broadening of the zero-phonon line (ZPL) for solid-state optical transitions [158, 221], which degrades the indistinguishability of photons emitted from the quantum system [86] and can degrade the generation of local entanglement that uses superpositions involving the emitter excited state. All these decoherence processes limit the amount of final spin-spin entanglement that can be generated.

The two most critical figures of merit for entanglement generation are efficiency and fidelity (section 1.4.5). The efficiency impacts the overall rate of quantum information transfer. For example, the quantum key distribution rate for a repeater protocol is proportional to the entanglement

generation efficiency. The fidelity quantifies the quality of entanglement in addition to our knowledge about the state of the system, and can be related to the entanglement concurrence (section 1.4.6). High-fidelity entanglement is necessary for many quantum information applications. In addition, purification and error correction protocols require minimum fidelity thresholds to be satisfied [74].

In section 4.1, I will present the work of Ref. [ix] where we analyze the efficiency and fidelity for three different common remote entanglement generation protocols while taking into account many realistic imperfections including decoherence. These three protocols are respectively implemented via (1) spin-photon number entanglement with a single pulse [209], (2) spin-time bin entanglement with two sequential π -pulses [123], and (3) spin-polarization entanglement using an excited Λ system [210, 211]. Each of these protocols requires fast resonant pulsed excitation of the quantum systems. For each protocol, we derive the spin-spin conditional states for photon counting measurements using the methods introduced in sections 2.1 and 2.2 to compute expressions for the entanglement figures of merit. We also discuss their relationship to the properties of single-photon emission from the individual quantum systems, such as brightness and mean wavepacket overlap. The results of this section have already been applied to analyze quantum repeaters [viii][xi][xii] and have been included in software for simulating quantum networks [222].

In section 4.2, I will present the analysis of local spin-spin entanglement for one of the three adiabatic photon-mediated approaches discussed in Ref. [vii]. The approach I study considers a controlled phase gate operation between two emitters in a cavity using a virtual photon interaction initiated by exciting one of the two emitters. In my analysis, I will explicitly include the effect of emitter pure dephasing that was not considered in Ref. [vii]. In addition, I will explore the possibility of using a cavity with an optical Fano resonance to greatly enhance the fidelity of this adiabatic phase gate, which perhaps could be implemented using the same Fabry-Pérot-plasmonic hybrid cavities discussed in section 3.1.

4.1 Heralded entanglement generation

To analyze heralded entanglement generation, we apply a photon number decomposition to compute the entanglement generation efficiency and fidelity of the final spin-spin entanglement conditioned on the number of detected photons. This approach uses a Liouville-Neumann series [91, 131, 223] to decompose the master equation dynamics into a set of propagation superoperators that describe the spin state evolution conditioned on the cumulative detector photon count during a window of time (see chapter 2).

In section 4.1.1, I will introduce the system model describing two remote optically active defects that contain a spin qubit. This will include a summary of all the imperfections included in the entanglement generation model. Sections 4.1.2 through 4.1.4 present the analytical and numerical results for each of the three analyzed protocols. Finally, section 4.1.5 presents a comparison of the protocols as a function of loss and distance between the remote spin qubits.

4.1.1 System

We consider five main imperfections in the entanglement generation process: (1) decoherence, (2) spectral diffusion, (3) photon loss, (4) phase errors, and (5) dark counts. Solid-state systems may suffer from mechanisms that degrade the spin coherence and the coherence of emitted photons. These mechanisms are usually strongly dependent on temperature [142, 158, 221]. These systems can also experience spectral diffusion, which can inhibit the indistinguishability of emitted photons [224–226]. In addition, photon losses due to non-radiative pathways or collection/transmission inefficiency can affect the protocol figures of merit; and in some cases, protocols can moreover be susceptible to initialization and propagation phase errors. Finally, the detectors may have a non-negligible dark count rate [133].

Decoherence.—For each protocol, we consider that a transition with decay rate γ is subject to a pure dephasing rate γ^* that degrades the coherence of photons emitted by the system [86]. This



Figure 4.1: Diagram of the system model used to analyze remote spin-spin entanglement. The three level system shown here illustrates the model used for both defects k = 1 and k = 2 that emit photons to be interfered at the beam splitter. These two defects may have different parameters, indicated by the subscripts k, and a spectral detuning Δ . The defects are subject to decoherence processes such as pure dephasing and spectral diffusion. The spin qubits of the two defects are projected onto an entangled state by performing an imperfect joint measurement of the photons emitted by each defect. The measurement imperfections include transmission loss, detector inefficiency, detector dark counts, and number resolving limitations.

dephasing can be seen as a fluctuation of the transition energy on a timescale much faster than its decay rate, and it affects the indistinguishability between photons emitted by the systems. We separate the decay rate of the transition into a radiative component γ_r and a non-radiative component γ_{nr} so that $\gamma = \gamma_r + \gamma_{nr}$. In addition, we consider that the spin qubits experience incoherent spin flip excitation (decay) at the rate $\gamma_s^+ = 1/T_1^+$ ($\gamma_s^- = 1/T_1^-$) and a pure dephasing at the rate $\gamma_s^* = 1/T_2^*$ for a total spin decoherence rate of $1/T_2 = 1/T_2^* + 1/2T_1^+ + 1/2T_1^-$.

Spectral diffusion.—In contrast to pure dephasing, spectral diffusion is a fluctuation of the transition energy on a timescale much slower than its decay rate. In many solid-state systems, this fluctuation can shift the emitted photon frequency over time by more than its linewidth [225, 226]. This degrades the mean wavepacket overlap of photons emitted by the same source at different times [224–226]. Hence, this fluctuation also significantly degrades interference between fields from different sources. We account for spectral diffusion by averaging entanglement figures of merit over a Gaussian distribution $h(\omega_k - \overline{\omega}_k, \delta_k) = (\delta_k \sqrt{2\pi})^{-1} e^{-(\omega_k - \overline{\omega}_k)^2/2\delta_k^2}$ for each emitter frequency ω_k with an average value of $\overline{\omega}_k$ and a spectral diffusion standard deviation δ_k . For example, for two systems, the entanglement fidelity \mathcal{F} becomes $\iint h(\omega_1 - \overline{\omega}_1, \delta_1)h(\omega_2 - \overline{\omega}_2, \delta_2)\mathcal{F}d\omega_1d\omega_2$.

Photon loss.—Losses can occur due to non-radiative transitions at a rate γ_{nr} . We also quantify the imperfect collection fraction η_c of emission and the fraction of photons transmitted to the detectors by η_t . In addition, we consider that each detector has a probability η_d of detecting an incident photon. This detector inefficiency can be applied during the measurement step. However, the beam-splitter loss model used to describe detector inefficiency can be mapped to the identical model for transmission loss [i]. Thus, for convenience, we choose to simulate detector inefficiency as part of the conditional dynamics rather than the measurement itself. This allows us to use the total efficiency parameter $\eta = \eta_c \eta_t \eta_d \gamma_r / \gamma$.

Phase errors.—Phase errors can arise when the individual quantum systems are locally initialized and read out using pulses from a source that does not maintain phase stability over the duration of the protocol. We account for this by considering an initial phase φ when a quantum system is initialized in a superposition state. Phase errors can also arise when photons from each source do not accumulate the same propagation phase ϕ before interference. If these phases are unstable or left uncorrected, then they may degrade the entanglement fidelity. We account for phase errors by assuming that the phase fluctuates between entanglement generation attempts and then average the fidelity over a random phase with a Gaussian distribution.

Dark counts.—A realistic detector may falsely indicate the arrival of a photon or detect a photon that did not originate from a desired emitter [133]. In our study, we assume that each detector is gated for an interval T_d that begins at time t_d after the start of the protocol and ends at time $t'_d = t_d + T_d$. We also assume that the dark counts are classical noise described by a Poisson distribution with a rate γ_d . Then for a given detector, the probability that n dark counts have occurred during the gate duration T_d is given by $p_{d,n}(T_d, \gamma_d) = \gamma_d^n T_d^n e^{-\gamma_d T_d}/n!$.

To capture the defect-specific imperfections, we use a Markovian master equation for two independent three-level systems (see Fig. 4.1). The total Liouville superoperator of the two

optically-active defects is $\mathcal{L} = \mathcal{L}_1 \otimes \mathcal{I} + \mathcal{I} \otimes \mathcal{L}_2$ where

$$\mathcal{L}_{k} = -\frac{i}{\hbar}\mathcal{H}_{k} + \sum_{j}\gamma_{j_{k}}^{-}\mathcal{D}(\hat{\sigma}_{j}) + \gamma_{j_{k}}^{+}\mathcal{D}(\hat{\sigma}_{j}^{\dagger}) + 2\gamma_{k}^{\star}\mathcal{D}(\hat{\sigma}_{\uparrow}^{\dagger}\hat{\sigma}_{\uparrow}) + \frac{\gamma_{\mathbf{s}_{k}}^{\star}}{2}\mathcal{D}(\hat{\sigma}_{z}), \qquad (4.1)$$

for $k \in \{1, 2\}$. The system operators are defined in their respective Hilbert spaces as $\hat{\sigma}_{\uparrow} = |\uparrow\rangle\langle e|$, $\hat{\sigma}_{\downarrow} = |\downarrow\rangle\langle e|$, $\hat{\sigma}_{s} = |\uparrow\rangle\langle\downarrow|$, and $\hat{\sigma}_{z} = |\uparrow\rangle\langle\uparrow| - |\downarrow\rangle\langle\downarrow|$. The rate $\gamma_{j_{k}}^{-}(\gamma_{j_{k}}^{+})$ is the total incoherent decay (excitation) rate across the transition associated with $\hat{\sigma}_{j}$ where $j \in \{\uparrow, \downarrow, s\}$, γ_{k}^{\star} is the emitter pure dephasing rate, and $\gamma_{s_{k}}^{\star}$ is the spin pure dephasing rate. The three-level system Hamiltonian is $\hat{H}_{k} = \hbar\omega_{\uparrow k}\hat{\sigma}_{\uparrow}^{\dagger}\hat{\sigma}_{\uparrow} + \hbar\omega_{s_{k}}\hat{\sigma}_{s}^{\dagger}\hat{\sigma}_{s}$ where $\omega_{\uparrow k}$ is the separation between $|\uparrow\rangle$ and $|e\rangle$, $\omega_{s_{k}}$ is the separation between $|\uparrow\rangle$ and $|\downarrow\rangle$.

The model used in this section is the three-level extension of the model used in sections 3.2 and 3.3. Therefore, if the defects are placed inside a cavity, it is necessary that the system is sufficiently far into the bad-cavity regime such that adiabatic elimination is valid (recall section 1.3.6). Furthermore, in this section we assume that a coherent state preparation is achieved on a timescale much faster than the total system decay rate. Thus, the results in this section neglect the degrading effects of nonzero $g^{(2)}$. That said, all of the methods used in this section can be directly applied to the full cavity-emitter source model described in section 1.3.4 to capture cavity-emitter non-Markovian effects and realistic state preparation pulses.

4.1.2 Spin-photon number entanglement

Consider the scheme where two spatially separated L-type systems are entangled by heralding a single-photon emission after erasing the which-path information using a beam splitter (see Fig. 4.2 (a)). This scheme is similar to the scheme used in the DLCZ repeater protocol to generate entanglement between spatially separated quantum memories [197]. However, by using fast resonant pulses, the quantum system requires only one optical transition. The scheme generates spin-spin entanglement by using spin-photon number entanglement [196]. For this reason, we will denote it as protocol N. For this protocol, we assume that the excited state $|e\rangle$ can only decay to



Figure 4.2: **Diagram illustrating remote entanglement generation between pulsed L-type systems.** (a) An L-type system as used in protocols N and T; a ground state doublet $|\uparrow\rangle$ and $|\downarrow\rangle$ with one optically excited state $|e\rangle$ that decays back to its initial ground state $|\uparrow\rangle$. (b) A diagram illustrating the fields from each quantum system interfering at a central beam splitter (BS) that has its output ports monitored by single-photon detectors D₁ and D₂.

spin state $|\uparrow\rangle$. That is, we assume that $\gamma_{\downarrow_k}^- \ll \gamma_{\uparrow_k}^- = \gamma_k$, where $k \in \{1, 2\}$ indexes the system.

Protocol description.—Each system is first prepared in the state $|\downarrow\rangle$. Then a microwave pulse resonant with the $|\downarrow\rangle \longleftrightarrow |\uparrow\rangle$ transition with a pulse area of $\Theta = 2\vartheta$ and phase φ_k brings the spin qubit to the state $\cos(\vartheta) |\downarrow\rangle + \sin(\vartheta)e^{i\varphi_k} |\uparrow\rangle$. After an optical π -pulse is applied to excite $|\uparrow\rangle$, each system is left in a superposition of ground and excited states. The excited state then decays back to $|\uparrow\rangle$ and the system emits a photon with a probability $\sin^2(\vartheta)$. By perfectly interfering the fields from two quantum systems (see Fig. 4.2 (b)), the which-path information is erased and a single detection event will herald one of the Bell states $|\psi^{\pm}\rangle = (|\uparrow\downarrow\rangle \pm |\downarrow\uparrow\rangle)/\sqrt{2}$.

To show protocol N in detail, consider the simpler case where $\vartheta = \pi/4$ and $\varphi_k = 0$. Then the total state of the quantum systems before decay is

$$|\psi(t_0)\rangle = \frac{1}{2}(|\downarrow\rangle + |\mathbf{e}\rangle)_1 \otimes (|\downarrow\rangle + |\mathbf{e}\rangle)_2.$$
(4.2)

After decay, $|e\rangle \rightarrow |\uparrow\rangle$, each system is in a spin-photon number entangled state $|\psi\rangle_k = (|\downarrow\rangle |0\rangle + |\uparrow\rangle |1\rangle)_k/\sqrt{2}$, where $|0\rangle$ is the vacuum state and $|1\rangle$ is the single-photon state of emission mode.

After interfering photons $|1\rangle_1$ and $|1\rangle_2$ at a beam splitter, the state before detection is

$$\frac{1}{2} \left(\left| \downarrow \downarrow \right\rangle \left| 00 \right\rangle + \left| \psi^{+} \right\rangle \left| 01 \right\rangle + \left| \psi^{-} \right\rangle \left| 10 \right\rangle + \left| \uparrow \uparrow \right\rangle \left| \psi^{-}_{2002} \right\rangle \right)_{s,p}, \tag{4.3}$$

where $|n_1n_2\rangle_p$ is the state with n_1 (n_2) photons in the mode of detector D_1 (D_2), $|\psi^{\pm}\rangle_s = (|\uparrow\downarrow\rangle \pm |\downarrow\uparrow\rangle)/\sqrt{2}$ are spin Bell states, and $|\psi_{2002}^-\rangle_p = (|20\rangle - |02\rangle)/\sqrt{2}$ is a two-photon NOON state. Hence, a single photon at either detector heralds a maximally entangled spin state with a phase determined by which detector received the photon.

The maximum efficiency of the above scheme is 50%, which is the Bell analyzer efficiency of a single beam splitter [227][i]. However, any amount of photon loss will cause errors due to states $|20\rangle$ and $|02\rangle$ contributing to single-photon measurement outcomes. If ϑ is small enough, then the probability for both quantum systems to emit photons becomes much less than the probability that only one system emits a photon. Thus, to combat errors due to multi-photon events, the parameter ϑ can be reduced to improve fidelity at the cost of efficiency [196]. This trade-off also improves the protocol fidelity for BD type detectors (see section 2.2 for the definition of BD and PNRD detector models).

Conditional states.—In the far field limit or for a cavity-emitter system in the Purcell regime, the source field component collected from a quantum emitter dipole by the waveguide is described by the proportionality relation $\hat{b}_k = \hat{\sigma}_{\uparrow_k} \sqrt{\eta_{c_k} \gamma_{r_k}}$ [91–93]. After transmission losses and a propagation phase we have $\hat{b}_k \rightarrow \hat{b}_k \sqrt{\eta_{t_k}} e^{-i\phi_k}$ where $\phi_k = L_k \omega_{\uparrow_k} / v_p$, L_k is the propagation distance, and v_p is the phase velocity. Then, the collected fields are interfered at a beam splitter so that the fields \hat{d}_1 and \hat{d}_2 at detectors D₁ and D₂, respectively, are

$$\begin{pmatrix} \hat{d}_1 \\ \hat{d}_2 \end{pmatrix} = \hat{R}(\theta) \begin{pmatrix} \hat{b}_1 \sqrt{\eta_{t_1}} e^{-i\phi_1} \\ \hat{b}_2 \sqrt{\eta_{t_2}} e^{-i\phi_2} \end{pmatrix},$$
(4.4)

where $\hat{R}(\theta)$ is the 2×2 rotation unitary matrix. Using a beam-splitter model for detector inefficiency, the effective detected field is $\hat{d}_i \rightarrow \hat{d}_i \sqrt{\eta_{d_i}}$. In this case, the jump superoperator corresponding to the effective detected field at the *i*th detector is $\mathcal{J}_i \hat{\rho} = \eta_{d_i} \hat{d}_i \hat{\rho} \hat{d}_i^{\dagger}$, where to simplify calculations we will take $\eta_{d_i} = \eta_d$ for each detector.

If we assume that the pulse Rabi frequency is much faster than the rates of dissipation and decoherence, and that the excitation pulses are resonant with the emitters, then we can consider the initial state to be approximated by $\hat{\rho}(t_0) = |\psi(t_0)\rangle \langle \psi(t_0)|_1 \otimes |\psi(t_0)\rangle \langle \psi(t_0)|_2$ where $|\psi(t_0)\rangle_k = \cos(\vartheta) |\downarrow\rangle + \sin(\vartheta)e^{i\varphi_k} |e\rangle$. Under these conditions, the set of conditional states can also be truncated to those n such that $n_1 + n_2 \leq 2$ as a consequence of each emitter only being able to emit up to a single photon.

The successful conditional states are associated with the single-photon detection conditions $\mathbf{n} = (1,0)$, and (0,1), which are given by their corresponding conditional propagators $\mathcal{W}_{\mathbf{n}}$ (see section 2.1). For convenience, we notate these vectors by 10, and 01, respectively. For example, the outcome associated with $|\psi^+\rangle$ is $\hat{\rho}_{01}(t_f) = \mathcal{W}_{01}(t_f, t'_d, t_d, t_0)\hat{\rho}(t_0)$ where $\mathcal{W}_{01}(t_f, t'_d, t_d, t_0) = \mathcal{U}(t_f, t'_d)\mathcal{U}_{01}(t'_d, t_d)\mathcal{U}(t_d, t_0)$ and

$$\mathcal{U}_{01}(t'_{\mathrm{d}}, t_{\mathrm{d}}) = \int_{t_{\mathrm{d}}}^{t'_{\mathrm{d}}} \mathcal{U}_{\mathbf{0}}(t'_{\mathrm{d}}, t) \mathcal{J}_{2} \mathcal{U}_{\mathbf{0}}(t, t_{\mathrm{d}}) dt.$$

$$(4.5)$$

Likewise, the outcome $\hat{\rho}_{10}$ associated with $|\psi^-\rangle$ is given by \mathcal{W}_{10} , which is determined using Eq. (4.5) but with \mathcal{J}_1 in place of \mathcal{J}_2 . The states corresponding to the remaining relevant conditions $\mathbf{n} = (2,0), (1,1), \text{ and } (0,2)$ are similarly obtained from Eqs. (2.4) and (2.5).

Measurement duration.—The spin entanglement is generated at the moment a single photon from one of the emitters is detected by one of the two detectors. However, a subsequent detection of another photon will destroy this entanglement [228]. As a consequence, unless the probability for two-photon events is very small, the detection duration must be long enough to ensure that only one photon was emitted. Hence, the fidelity can be very low for small T_d . On the other hand, for a detection window much longer than the lifetime, the fidelity becomes limited by spin decoherence processes. Fig. 4.3 shows the entanglement generation fidelity and efficiency as the detection window duration is increased, illustrating the peak in fidelity when the duration is on the order of the lifetime. To show this qualitative behaviour, we have chosen parameters in the regime $\gamma > \gamma^* \gg \gamma^*_s$ to represent a solid-state system that could potentially serve as a quantum communication node. The two-photon probabilities for photon bunching (p_2) and coincident counts (p_{11}) are also illustrated. Note that the coincident counts are nonzero due to the emitter pure dephasing that degrades the HOM interference [107] between photons from different sources.

Under the condition where the systems experience negligible spin decoherence on the timescale of the lifetime of the emitter, we can analytically solve the conditional states as a function of detection window duration. This can then be used to estimate simple figures of merit for the quality of entanglement based only on the optical properties of the emitters. These limits are illustrated by the asymptotes of the dashed lines in Fig. 4.3. The analytic solutions can also then be used to estimate the fidelity under the effects of additional imperfections such as noisy detectors and spectral diffusion using the methods outlined in sections 2.2 and 4.1.1, respectively.

Suppose that the detection window begins at t_d such that $r(t_d) = t_0$ and ends at t'_d such that $r(t'_d) = T_d + t_0$. Using the appropriate conditional propagators W_n , we compute the final (unnormalized) spin-spin conditional states in the rotating frame of the spin qubits after time $t_f \gg 1/\gamma_k$. In this limit of time, neither quantum system remains in $|e\rangle$.

The conditional spin state of the quantum systems given that both detector modes do not contain a photon from an emitter is

$$\hat{\rho}_{0} = \frac{1}{4} \sin^{2}(2\vartheta) \left((1 - \overline{\beta}_{1}) |\uparrow\downarrow\rangle \langle\uparrow\downarrow| + (1 - \overline{\beta}_{2}) |\downarrow\uparrow\rangle \langle\downarrow\uparrow| \right) + (1 - \overline{\beta}_{1}) (1 - \overline{\beta}_{2}) \sin^{4}(\vartheta) |\uparrow\uparrow\rangle \langle\uparrow\uparrow| + \cos^{4}(\vartheta) |\downarrow\downarrow\rangle \langle\downarrow\downarrow| ,$$
(4.6)

where $\overline{\beta}_k = \eta_d \beta_k / \gamma_k = \eta_k \left(1 - e^{-T_d \gamma_k} \right)$ is the rate-normalized brightness and $\eta_k = \eta_d \eta_{t_k} \eta_{c_k} \gamma_{r_k} / \gamma_k$ is the total single-photon efficiency of each quantum system.

The single-photon conditioned states of the quantum system are

$$\hat{\rho}_{\rm s}^{\pm} = \rho_{\uparrow\uparrow\uparrow}^{\pm} |\uparrow\uparrow\rangle\langle\uparrow\uparrow| + \rho_{\uparrow\downarrow}^{\pm} |\uparrow\downarrow\rangle\langle\uparrow\downarrow| + \rho_{\downarrow\uparrow}^{\pm} |\downarrow\uparrow\rangle\langle\downarrow\uparrow| + \left(\rho_{\rm c}^{\pm} |\downarrow\uparrow\rangle\langle\uparrow\downarrow| + \rm h.c.\right), \tag{4.7}$$



Figure 4.3: The time dynamics of remote entanglement generation via spin-photon number entanglement: protocol N. Entanglement generation fidelity \mathcal{F}_{gen} and efficiency η_{gen} as a function of protocol time $t_{\rm f}$ for an initial state $\hat{\rho}(t_0) = |\psi(t_0)\rangle\langle\psi(t_0)|$ where $|\psi(t_0)\rangle = (1/2)(|\downarrow\rangle + |e\rangle)^{\otimes 2}$ and where there is no loss and noiseless local photon-number resolving detectors. The asymptotic dashed lines indicate the limits on fidelity $\mathcal{F}_{\rm op}$ and efficiency $\eta_{\rm op}$ reached when spin decoherence is neglected and when the detection window encompasses the entire photon lifetime. The detection window is set to be equal to the entire protocol duration: $t_{\rm d} = t_0 = 0$ and $t'_{\rm d} = t_{\rm f} = T_{\rm d}$. The gray lines show the probabilities for no photon emission from either system $p_0 = \text{Tr}[\hat{\rho}_0]$ and for photon bunching $p_2 = p_{20} + p_{02} = \text{Tr}[\hat{\rho}_{20} + \hat{\rho}_{02}]$. The red line indicates coincident counts $p_{11} = \text{Tr}[\hat{\rho}_{11}]$ caused by imperfect HOM interference. Parameters chosen: $\gamma_k = \gamma, \gamma_k^* = 0.1\gamma$, and $\gamma_{\rm s_k}^{\pm} = \gamma_{\rm s_k}^* = 0.001\gamma$ for $k \in \{1, 2\}$.

where

$$\begin{split} \rho_{\uparrow\uparrow}^{\pm} &= \frac{1}{2} \left(\overline{\beta}_1 + \overline{\beta}_2 - 2\overline{\beta}_1 \overline{\beta}_2 \pm (\overline{\beta}_1 - \overline{\beta}_2) \cos(2\theta) \right) \sin^4(\vartheta) \\ \rho_{\downarrow\uparrow\uparrow}^{\pm} &= \frac{\overline{\beta}_1}{8} (1 \pm \cos(2\theta)) \sin^2(2\vartheta) \\ \rho_{\uparrow\downarrow}^{\pm} &= \frac{\overline{\beta}_2}{8} (1 \mp \cos(2\theta)) \sin^2(2\vartheta) \\ \rho_{c}^{\pm} &= \pm \frac{\tilde{C}(T_d)}{8} \sqrt{\eta_1 \eta_2} \sin(2\theta) \sin^2(2\vartheta) \\ \tilde{C}(T_d) &= \frac{2\sqrt{\gamma_1 \gamma_2}}{\Gamma_1 + \Gamma_2 + 2i\Delta} \left(1 - e^{-\frac{1}{2}T_d(\Gamma_1 + \Gamma_2 + 2i\Delta)} \right) e^{i(\varphi + \phi)}, \end{split}$$
(4.8)

and where $\Gamma_k = \gamma_k + 2\gamma_k^*$ is the FWHM of the emission ZPL for system k, $\Delta = \omega_{\uparrow 1} - \omega_{\uparrow 2}$ is the spectral detuning, $\varphi = \varphi_1 - \varphi_2$ is the relative initialization phase, and $\phi = \phi_1 - \phi_2$ is the relative propagation phase. The sign of $\hat{\rho}_s^{\pm}$ is given by which detector received the photon: $\hat{\rho}_{10} = \hat{\rho}_s^-$ and $\hat{\rho}_{01} = \hat{\rho}_s^+$. The quantity \tilde{C} is a complex value related to the spin-spin coherence and entanglement concurrence, not to be confused with the cavity cooperativity C used in other sections of this thesis. Note that $\rho_{\downarrow\downarrow} = 0$ because after either system emits one photon, the system is guaranteed to not be

in $|\downarrow\downarrow\rangle$ due to our assumptions about the allowed emitter transitions and under the approximation of a negligible spin flip rate.

The individual two-photon conditioned states $\hat{\rho}_{20}$, $\hat{\rho}_{11}$, and $\hat{\rho}_{02}$ are all proportional to $|\uparrow\uparrow\rangle\langle\uparrow\uparrow|$, as expected. However, their trace has a complicated dependency on T_d due to the HOM effect. Regardless, their sum can be easily simplified to the intuitive result

$$\hat{\rho}_{20} + \hat{\rho}_{11} + \hat{\rho}_{02} = \overline{\beta}_1 \overline{\beta}_2 \sin^4(\vartheta) \left|\uparrow\uparrow\rangle \left\langle\uparrow\uparrow\right|.$$
(4.9)

Using all these conditional states, we can also verify that

$$\hat{\rho}_{0} + \hat{\rho}_{s}^{+} + \hat{\rho}_{s}^{-} + \hat{\rho}_{20} + \hat{\rho}_{11} + \hat{\rho}_{02} = \sin^{4}(\vartheta) |\uparrow\uparrow\rangle \langle\uparrow\uparrow| + \cos^{4}(\vartheta) |\downarrow\downarrow\rangle \langle\downarrow\downarrow| + \frac{1}{4} \sin^{2}(2\vartheta) (|\uparrow\downarrow\rangle \langle\uparrow\downarrow| + |\downarrow\uparrow\rangle \langle\downarrow\uparrow|)$$
(4.10)

is indeed the solution $\hat{\rho}(t)$ of the total master equation in the limit $t \gg 1/\gamma_k$ and when spin decoherence is neglected. This confirms the completeness of the photon-number decomposition.

The states $\hat{\rho}_s^{\pm}$ correspond to the expected Bell states $|\psi^{\pm}\rangle$ and so the average entanglement fidelity (recall section 1.4.5) is

$$\mathcal{F}_{\text{gen}} = \frac{1}{\eta_{\text{gen}}} \left(\langle \psi^+ | \, \hat{\varrho}_s^+ \, | \psi^+ \rangle + \langle \psi^- | \, \hat{\varrho}_s^- \, | \psi^- \rangle \right), \tag{4.11}$$

where $\eta_{\text{gen}} = \text{Tr} \left[\hat{\varrho}_{\text{s}}^{+} \right] + \text{Tr} \left[\hat{\varrho}_{\text{s}}^{-} \right]$ is the entanglement generation efficiency and $\hat{\varrho}_{\text{s}}^{\pm}$ is the state after measurement computed from the conditional state $\hat{\rho}_{\text{s}}^{\pm}$.

Optical limits.—Suppose that the interference is balanced so that $\theta = \pi/4$ and $\eta_1 = \eta_2 = \eta$. Also, suppose that the protocol is phase corrected so that $\varphi + \phi = 0$ (see section 4.1.5 for a discussion on phase errors). Then in the limit that $T_d \gg 1/\gamma_k$ we have $\overline{\beta}_1 \to \eta$, $\overline{\beta}_2 \to \eta$, and

$$\tilde{C} \to \frac{2\sqrt{\gamma_1 \gamma_2}}{\Gamma_1 + \Gamma_2 + 2i\Delta}.$$
(4.12)

If we also assume that the measurement is performed by ideal noiseless PNRDs, then $\hat{\varrho}_s^{\pm} = \hat{\rho}_s^{\pm}$ and under these conditions—which we refer to as the optical limit—the corresponding entanglement generation fidelity \mathcal{F}_{op} gives an estimate of the fidelity determined only by the optical properties of the emitters. In principle, this estimate could be exceeded using spectral or temporal post selection of photons, consequently sacrificing efficiency.

In the optical limit, the fidelity for protocol N becomes

$$\mathcal{F}_{\rm op} = \frac{1}{2} \left(1 + \operatorname{Re}(\tilde{C}) \right) F_{\eta}(\vartheta) \tag{4.13}$$

with concurrence $C_{op} = |\tilde{C}|F_{\eta}(\vartheta)$, where the loss compensation factor is $F_{\eta}(\vartheta) = \cos^2(\vartheta)/(1 - \eta \sin^2(\vartheta))$ that arose also in the context of self-homodyne measurements of number coherence (recall section 3.3.2). The efficiency becomes $\eta_{op} = (\eta/2) \sin^2(2\vartheta)/F_{\eta}(\vartheta)$ and the two-photon conditioned states reduce to

$$\hat{\rho}_{20} = \hat{\rho}_{20} = \frac{1}{4} \left(1 + M_{12} \right) \eta^2 \sin^4(\vartheta) \left|\uparrow\uparrow\rangle\rangle\langle\uparrow\uparrow|$$

$$\hat{\rho}_{11} = \frac{1}{2} \left(1 - M_{12} \right) \eta^2 \sin^4(\vartheta) \left|\uparrow\uparrow\rangle\rangle\langle\uparrow\uparrow| ,$$
(4.14)

where

$$M_{12} = M_{\gamma} \frac{(\Gamma_1 + \Gamma_2)(\gamma_1 + \gamma_2)}{(\Gamma_1 + \Gamma_2)^2 + 4\Delta^2} \le \sqrt{M_1 M_2},$$
(4.15)

is the mean wavepacket overlap between photons from defects k = 1 and k = 2, $M_k = \gamma_k/\Gamma_k$ is the individual system indistinguishability from Eq. (1.49), and $M_{\gamma} = 4\gamma_1\gamma_2/(\gamma_1 + \gamma_2)^2 \ge M_{12}$ quantifies the temporal profile mismatch. We emphasize that Eq. (4.14) and M_{12} in Eq. (4.15) were solved using the methods of chapter 2 and not by computing the mean wavepacket overlap of two different input photons as in Eq. (3.11). However, we have verified that solving the mean wavepacket overlap indeed gives the same result as Eq. (4.15), which confirms that the photon statistics of the HOM interference are independent of whether the calculation is performed from the perspective of the source or the field. For a given $F_{\eta}(\vartheta)$, the fidelity and concurrence are limited by the spectral and temporal properties of the individual emitters. In particular, we can identify that $C_{op}^2 \leq M_{12}F_{\eta}^2(\vartheta)$. Hence for protocol N, the square root of the mean wavepacket overlap M_{12} gives an upper bound on the entanglement generation concurrence, which itself can be used to determine an upper bound on the entanglement generation fidelity by $\mathcal{F}_{op} \leq (1 + \sqrt{M_{12}}) F_{\eta}(\vartheta)$. On the other hand, we have that $\operatorname{Re}(\tilde{C}) = M_{12}/\sqrt{M_{\gamma}} \geq M_{12}$. Hence the optical limit of fidelity \mathcal{F}_{op} for protocol N is bounded by

$$\frac{1}{2}(1+M_{12}) \le \frac{\mathcal{F}_{\text{op}}}{F_{\eta}(\vartheta)} \le \frac{1}{2}\left(1+\sqrt{M_{12}}\right).$$
(4.16)

Detector noise and number resolution.—The entangled spin-spin state after a single-photon measurement by a PNRD with non-negligible noise is $\hat{\varrho}_s^{\pm} = p_{d,1}^2 \hat{\rho}_s^{\pm} + p_{d,0} p_{d,1} \hat{\rho}_0$, where $p_{d,n}(T_d, \gamma_d)$ is the probability to have *n* dark counts within the detection window T_d . For $T_d \gg 1/\gamma_k$, we can write

$$\mathcal{F}_{gen} = \frac{1}{\eta_{gen}} \left(p_{d,0}^2 \mathcal{F}_{op} \eta_{op} + \frac{1}{2} p_{d,0} p_{d,1} (1-\eta) \sin^2(2\vartheta) \right), \tag{4.17}$$

where

$$\eta_{\text{gen}} = p_{d,0}^2 \eta_{\text{op}} + 2p_{d,0} p_{d,1} \left(1 - \eta \sin^2(\vartheta)\right)^2$$
(4.18)

is the total efficiency. For a measurement by a BD, the state after heralding is given by

$$\hat{\varrho}_{s}^{-} = p_{d,0}(\hat{\rho}_{s}^{-} + \hat{\rho}_{20}) + p_{d,0}(1 - p_{d,0})\hat{\rho}_{0}$$

$$\hat{\varrho}_{s}^{+} = p_{d,0}(\hat{\rho}_{s}^{+} + \hat{\rho}_{02}) + p_{d,0}(1 - p_{d,0})\hat{\rho}_{0},$$
(4.19)

which can be used to compute the fidelity and efficiency in the same way as for the PNRD case.

In the absence of detector noise and for a given η , \mathcal{F}_{op} can be maximized by increasing F_{η} arbitrarily close to 1 by taking $\vartheta \to 0$ and sacrificing efficiency. However, detector noise places an additional constraint on the fidelity due to the presence of a finite noise floor. This gives rise to an optimal $\vartheta \neq 0$ that maximizes fidelity. In the regime where $1 - p_{d,0} \simeq p_{d,1} \ll \eta$, we find that Eq. (4.17) for the PNRD case is maximized when $\vartheta \simeq [p_{d,1}/(\eta(1-\eta))]^{1/4}$. Note that

this estimate is also only accurate for $\eta < p_{d,0} \simeq 1$ as evidently $\vartheta = \pi/4$ is the optimal choice for $\eta = 1$ when using a PNRD. As for the BD case, the optimal ϑ depends on M_{12} due to the contribution from two-photon events. For $M_{12} \simeq 1$ we use the conditional states in Eq. (4.19) to find that $\vartheta \simeq [2p_{d,1}/(\eta(2-\eta))]^{1/4}$ maximizes the fidelity. When $\eta = 1$ this optimal choice becomes $\vartheta \simeq (2p_{d,1})^{1/4}$. In the regime of quantum communication where $\eta \ll 1$, two-photon detections are suppressed due to losses and so the PNRD and BD models give equivalent results.

4.1.3 Spin-time bin entanglement

For the second protocol (denoted by T), which uses spin-time bin entanglement, we focus on the extension of protocol N where two successive photons herald entanglement between L-type systems (see Fig. 4.2). This protocol is also referred to as the Barrett-Kok scheme [123], which was utilized to demonstrate the first loophole-free Bell inequality violation [10].

Protocol description.—Each system is first prepared in the maximal superposition state $(|\uparrow\rangle + |\downarrow\rangle)/\sqrt{2}$. Then a resonant π pulse excites the $|\uparrow\rangle$ states at t_0 , giving Eq. (4.2). Following protocol N, we could obtain the entangled state $|\psi^{\pm}\rangle$ by post-selecting on a single photon. However, to eliminate the errors caused by both systems emitting photons after the first pulse, we can flip the spin state of both systems and re-excite $|\uparrow\rangle$ some time $t_x - t_0$ after the first pulse. If the quantum systems emit only one photon either before or after the second pulse, then they are each in a spin-time bin entangled state $|\psi\rangle_k = (|\downarrow\rangle |\text{early}\rangle + |\uparrow\rangle |\text{late}\rangle)/\sqrt{2}$, where $|\text{early}\rangle$ and $|\text{late}\rangle$ represent the presence of a photon in the early and late time bin modes, respectively. The joint state $|\psi\rangle_1 \otimes |\psi\rangle_2$ can be written in the Bell basis of the spin and photon states

$$\frac{1}{2} \left(\left| \psi^{+} \right\rangle \left| \psi^{+} \right\rangle - \left| \psi^{-} \right\rangle \left| \psi^{-} \right\rangle + \left| \phi^{+} \right\rangle \left| \phi^{+} \right\rangle - \left| \phi^{-} \right\rangle \left| \phi^{-} \right\rangle \right|_{s,p},$$
(4.20)

where $|\psi^{\pm}\rangle_{s}$ is as before and $|\phi^{\pm}\rangle_{s} = \frac{1}{\sqrt{2}}(|\uparrow\uparrow\rangle\pm|\downarrow\downarrow\rangle)$. The Bell states $|\psi^{\pm}\rangle_{p}$ and $|\phi^{\pm}\rangle_{p}$ are similarly defined using time bin states $|\text{early}\rangle$ and $|\text{late}\rangle$. Interfering the joint state at a beam splitter performs a partial Bell-state measurement (BSM), allowing the identification of $|\psi^{+}\rangle_{p}$ and $|\psi^{-}\rangle_{p}$ from $|\phi^{\pm}\rangle_{p}$.

This projects the spin state onto either $|\psi^+\rangle_s$ or $|\psi^-\rangle_s$ with a 50% total probability.

In the absence of spin-flipping decoherence, and with perfect spin-flipping operations, neither quantum system can emit a photon in both the early and late time bins. Thus, when neglecting detector dark counts, the entanglement fidelity is independent of photon losses and the protocol does not suffer from an inherent efficiency-fidelity trade-off. The ramification is that a photon from each emitter must be transmitted to the beam splitter, which reduces the overall protocol efficiency.

Conditional states.— Let $\mathcal{X}(t'_{x}, t_{x})$ be the superoperator propagator that performs a spin-flip and re-excitation of both systems beginning at time t_{x} and concluding at time t'_{x} . The conditional state $\hat{\rho}_{\mathbf{n}_{l},\mathbf{n}_{e}}$ given photon counts \mathbf{n}_{e} in the early detection window and \mathbf{n}_{l} in the late detection window is $\hat{\rho}_{\mathbf{n}_{l},\mathbf{n}_{e}} = \mathcal{W}_{\mathbf{n}_{l}}\mathcal{X}\mathcal{W}_{\mathbf{n}_{e}}\hat{\rho}$, where $\hat{\rho}$ is the state after the first excitation and $\mathcal{W}_{\mathbf{n}_{l}}$ ($\mathcal{W}_{\mathbf{n}_{e}}$) is the conditional propagator for the late (early) time bin detection window dependent on the time ordering $\mathcal{W}_{\mathbf{n}_{l}}(t_{f}, r(t'_{l}), r(t_{l}), t'_{x})$ ($\mathcal{W}_{\mathbf{n}_{e}}(t_{x}, r(t'_{e}), r(t_{e}), t_{0}$)). The detector window duration for the early and late bins are $T_{e} = t'_{e} - t_{e}$ and $T_{l} = t'_{l} - t_{l}$, respectively.

There are four measurement outcomes that may indicate successful entanglement: { \mathbf{n}_l , \mathbf{n}_e } = {(1,0), (1,0)}, {(1,0), (0,1)}, {(0,1), (1,0)}, and {(0,1), (0,1)}. For notation convenience, we concatenate the sets of vectors. For example, $\hat{\rho}_{(1,0),(0,1)} = \hat{\rho}_{1001}$. The conditional states are then given by the appropriate conditional propagators $\mathcal{W}_{\mathbf{n}}$. Using the case 1001 as an example, we have $\hat{\rho}_{1001} = \mathcal{W}_{10}\mathcal{X}\mathcal{W}_{01}\hat{\rho}$, where \mathcal{W}_{10} and \mathcal{W}_{01} are the same as in protocol N. The remaining conditional states can be similarly expressed in terms of \mathcal{U}_0 , \mathcal{J} , and \mathcal{X} superoperators, although for brevity we do not display them.

Measurement duration.—For protocol T, there are two detection windows beginning at t_e and t_1 with duration T_e and T_1 , respectively. Suppose that the detection window is continuous between the pulses. Then we have $t_e = t_0$, $t'_e = t_x$. Also, if the spin-flip and re-excitation is much faster than other system dynamics so that $t'_x - t_x \simeq 0$, then we have $t'_e = t_1 = t_x$. To simplify the problem, we also make the time bins equal in duration so that $T_e = T_1 = T_d$.

If the detection windows do not encompass the entire photon lifetime, then a high fidelity can



Figure 4.4: The time dynamics of remote entanglement generation via spin-photon number entanglement: protocol T. Entanglement generation fidelity \mathcal{F}_{gen} and efficiency η_{gen} as a function of protocol time t_f for an initial state $\hat{\rho}(t_0) = |\psi(t_0)\rangle \langle \psi(t_0)|$ where $|\psi(t_0)\rangle = (1/2)(|\downarrow\rangle + |e\rangle)^{\otimes 2}$ and where there is no loss and noiseless local photon-number resolving detectors. The asymptotic dashed lines indicate the limits of fidelity \mathcal{F}_{op} and efficiency η_{op} reached when spin decoherence is neglected. The detection window for each time bin is set to be equal to half the protocol duration, which begins after the first system excitation: $t_d = t_0 = 0, t'_d = t_x = T_d$, and $t_f = 2T_d$. The gray lines show the probabilities for no photon emission from either system p_0 , for single-photon detection events p_1 , and for photon bunching events p_2 . The red solid line shows coincident counts p_{11} caused by imperfect HOM interference and the red dashed line indicates the probability $p_{n\geq 3}$ for 3 or more photons to be emitted as a consequence of spin relaxation between the pulses. Parameters chosen: $\gamma_k = \gamma$, $\gamma_k^* = 0.1\gamma$, and $\gamma_{s_k}^{\pm} = \gamma_{s_k}^* = 0.001\gamma$ for $k \in \{1, 2\}$.

be attained because after heralding by two-photon events, both systems will be in the ground state with a high probability. When this is the case, the detection window post selects photons that were emitted early compared to the total lifetime (see Fig. 4.4). This again demonstrates how fast detector gate times can potentially purify photon indistinguishability and increase the overall spin-spin entanglement fidelity (recall section 3.3.3). Consequently, the efficiency in this regime is very low. Note that this type of temporal post selection can also be applied to protocol N provided that ϑ is very small.

When the time bin duration is on the order of the emission lifetime, the fidelity briefly plateaus at the optical limit where the non-zero coincidence count probability p_{11} indicates imperfect interference. In this regime, the efficiency approaches the ideal Bell-analyzer efficiency of 50%. However, if the duration is much longer than the optical lifetime, then spin flips occurring between the excitation pulses increase the probability to have three or more photons emitted during the protocol, which reduces the efficiency and fidelity to their thermal limits of 0.25.

For brevity we do not show the full conditional state solutions for protocol T. However, due to the symmetry of this protocol and its close relationship with protocol N, the fidelity for $\theta = \pi/4$ when neglecting spin decoherence and detector noise takes the simple form

$$\mathcal{F}_{\text{gen}} = \frac{1}{2} \left(1 + \frac{\eta_1 \eta_2}{2\eta_{\text{gen}}} |\tilde{C}(T_{\text{d}})|^2 \right), \tag{4.21}$$

where $\eta_{\text{gen}} = \overline{\beta}_1 \overline{\beta}_2 / 2$ is the efficiency, $\tilde{C}(T_d)$ is given by Eq. (4.8), and $\overline{\beta}_k$ is the same as in protocol N. This expression accounts for emitter pure dephasing through $\tilde{C}(T_d)$ and can also be averaged for a fluctuating detuning Δ to capture spectral diffusion.

Optical limits.—In the limit that $T_d \gg 1/\gamma_k$ we have $\overline{\beta}_k \to \eta_k$ and \tilde{C} again reduces to Eq. (4.12). Then the optical limits of efficiency and fidelity are $\eta_{op} = \eta^2/2$ and $\mathcal{F}_{op} = (1+|\tilde{C}|^2)/2$, respectively, for $\eta_k = \eta$. The corresponding concurrence is simply $\mathcal{C}_{op} = |\tilde{C}|^2$.

As with protocol N, the fidelity and concurrence can be related to the mean wavepacket overlap M_{12} by noting

$$\mathcal{C}_{\rm op} = M_{12} \left(\frac{\gamma_1 + \gamma_2}{\Gamma_1 + \Gamma_2} \right) \le M_{12}. \tag{4.22}$$

On the other hand, it can be shown that $\mathcal{F}_{op} \geq M_{12}$. Hence the optical limit of fidelity for T is bounded by

$$M_{12} \le \mathcal{F}_{op} \le \frac{1}{2} \left(1 + M_{12} \right).$$
 (4.23)

We note that the upper bound result presented here has also been derived in the supplementary of Ref. [151] using arguments from interference visibility. However, since pure dephasing degrades both the interference visibility and the coherence of the emitter states, the final spin-spin fidelity must be less than or equal the value computed from interference visibility alone.

Detector noise and number resolution.—Because of detector dark counts, it is possible that zero or single-photon conditioned states appear to give successful measurements. After taking detector

noise into consideration with PNRDs as described in subsection 2.2 we have, for example,

$$\hat{\varrho}_{1001} = p_{d,0}^4 \hat{\rho}_{1001} + p_{d,0}^3 p_{d,1} \left(\hat{\rho}_{1000} + \hat{\rho}_{0001} \right) + p_{d,0}^2 p_{d,1}^2 \hat{\rho}_{\mathbf{0}}.$$
(4.24)

In the absence of detector noise, only conditional states corresponding to three or more total detected photons will cause errors when using BDs with protocol T. This only occurs if the probability for a spin flip in between the pulses is non-negligible and photon loss is not too low. Conditional states where two photons arrive at one detector can combine with a single dark count at another detector to cause errors. However, for reasonably high photon losses or a reasonably low spin flip probability, both of these contributions to errors are negligible compared to other sources. Hence, Eq. (4.24) also well-approximates the measured state for BDs in this regime. This illustrates the robustness of protocol T against losses.

4.1.4 Spin-polarization entanglement

We now look at the third protocol (denoted by P), which is based on spin-spin entanglement generation via spin-polarization entanglement. For this scheme, we analyze a Λ -type system where a single excited state $|e\rangle$ can decay to either $|\uparrow\rangle$ or $|\downarrow\rangle$, emitting photons of orthogonal polarization depending on the transition (see Fig. 4.5).

Protocol description.— Initially, we prepare each of the quantum systems in one of the two ground states. Then, using a short π -pulse, each system is brought to the excited state. This excited state will then decay to one of the ground states while emitting a photon.

To illustrate this more clearly, suppose the probability is equal to decay to either ground state. Then the state of the qubit and the emitted photon for each system is $|\psi\rangle_k = \frac{1}{\sqrt{2}}(|\uparrow\rangle |L\rangle + |\downarrow\rangle |R\rangle)$, where $|L\rangle$ and $|R\rangle$ denote the left and right circular polarization modes of the photon. The joint state of both systems can then be written in the Bell basis for the spin and photon as Eq. (4.20), where the polarization modes replace the time bin modes of protocol T.



Figure 4.5: Diagram illustrating remote entanglement generation between pulsed Λ -type systems. (a) A Λ -type system as used in protocol P; an excited state $|e\rangle$ can decay to either ground state $|\uparrow\rangle$ or $|\downarrow\rangle$. (b) Entanglement generation via a polarization Bell-state measurement. Placing $\lambda/2$ and $\lambda/4$ waveplates before each polarizing beam splitter (PBS) can control the measurement basis.

To perform a BSM, we require a beam splitter (BS) and a polarizing beam splitter (PBS) at each output port of the BS. Then, we place detectors L_1 and R_1 (L_2 and R_2) on the left (right) output port of the BS, as shown in Fig. 4.5. For $|\psi^+\rangle_p (|\psi^-\rangle_p)$ photon bunching (anti-bunching) happens on the BS. Therefore, considering a perfect interference of the fields, a coincidence in detectors (L_1 , R_1) or (L_2 , R_2) will project the photon state onto the entangled state $|\psi^+\rangle_p$ and a coincidence in detectors (L_1 , R_2) or (L_2 , R_1) results in the entangled state $|\psi^-\rangle_p$ [229]. This projects the state of the qubits onto the corresponding spin Bell state. As in protocol T, this setup is not able to distinguish $|\phi^+\rangle_p$ and $|\phi^-\rangle_p$ since photon bunching happens for both of these cases. However, with the addition of a source of local auxiliary polarization-entangled photon states, the Bell analyzer success rate could be increased to 75% [230][i].

Conditional states.—We can describe the source field collected from each transition by $\hat{b}_{\uparrow k} = \hat{\sigma}_{\uparrow k} \sqrt{\eta_{c_k} \gamma_{r\uparrow k}}$ and $\hat{b}_{\downarrow k} = \hat{\sigma}_{\downarrow k} \sqrt{\eta_{c_k} \gamma_{r\downarrow k}}$ where k denotes the quantum system 1 and 2 and r indicates the radiative decay rate. Considering the transmission loss and the beam splitter, we can compute the L-polarized fields (\hat{d}_1, \hat{d}_2) at detectors L₁ and L₂ and the R-polarized fields (\hat{d}_3, \hat{d}_4) at detectors R₁ and R₂ using Eq. (4.4). The associated jump superoperators are then $\mathcal{J}_i \hat{\rho} = \eta_d \hat{d}_i \hat{\rho} \hat{d}_i^{\dagger}$.
Similar to protocol T, the conditions for a successful protocol are $\mathbf{n} = (1, 0, 1, 0), (1, 0, 0, 1), (0, 1, 1, 0), and (0, 1, 0, 1)$ where the vectors notate the photon count at the detectors in the order $(\mathbf{L}_1, \mathbf{L}_2, \mathbf{R}_1, \mathbf{R}_2)$. Like with the previous protocol, we simplify the notation by concatenating the vector elements. In contrast to protocol T, the conditional states for protocol P are true two-photon events rather than sequential one-photon events. These two-photon conditioned states are computed from their corresponding two-photon conditioned propagators. For example, $\hat{\rho}_{1001}(t_f) = \mathcal{W}_{1001}(t_f, t'_d, t_d, t_0)\hat{\rho}(t_0)$ where $\mathcal{W}_{1001} = \mathcal{U}(t_f, t'_d)\mathcal{U}_{1001}(t'_d, t_d)\mathcal{U}(t_d, t_0)$ is computed using

$$\mathcal{U}_{1001}(t'_{d}, t_{d}) = \int_{t_{d}}^{t'_{d}} \int_{t_{d}}^{t''} \mathcal{U}_{0}(t'_{d}, t'') \mathcal{J}_{1} \mathcal{U}_{0}(t'', t') \mathcal{J}_{4} \mathcal{U}_{0}(t', t_{d}) dt' dt'' + \int_{t_{d}}^{t'_{d}} \int_{t_{d}}^{t''} \mathcal{U}_{0}(t'_{d}, t'') \mathcal{J}_{4} \mathcal{U}_{0}(t'', t') \mathcal{J}_{1} \mathcal{U}_{0}(t', t_{d}) dt' dt''.$$
(4.25)

Note that by how we defined a photon counting measurement in this thesis, we are not tracking the arrival time within the detection window. Thus U_{1001} does not discriminate between the cases where L₁ clicks before R₂ and cases where R₂ clicks before L₁. This is illustrated in Eq. (4.25) as a consequence of the summation in Eq. (2.4). Such a restriction can be lifted if the detectors have sufficient time resolution capabilities.

Measurement duration.— The time dynamics of protocol P shows features in common with both protocols N and T. Like T, it is a two-photon scheme and so the fidelity is high for small T_d compared to the system optical lifetimes $1/\gamma_k$. However, like N, P only requires a single excitation of each system. Thus the efficiency is unaffected by spin flip processes when T_d is much larger than the lifetime (see Fig. 4.6).

Although it is possible to derive analytic expressions for protocol P for arbitrary measurement duration when neglecting spin decoherence, they do not provide new physical insight. For brevity, we only show analytic results in the optical limit to compare with protocols N and T.

Optical limits.— Consider the case where spin decoherence is negligible, the measurement window encompasses the lifetime $T_d \gg 1/\gamma_{j_k}$, and the interference is balanced so that $\theta = \pi/4$ and $\eta_{j_k} = \eta$ for $j \in \{\uparrow,\downarrow\}$ and $k \in \{1,2\}$. Then the entanglement generation efficiency is given



Figure 4.6: The time dynamics of remote entanglement generation via spin-polarization entanglement: protocol P. Entanglement generation fidelity \mathcal{F}_{gen} and efficiency η_{gen} as a function of protocol time t_f for an initial state $\hat{\rho}(t_0) = |ee\rangle\langle ee|$ when there is no loss and noiseless local photon-number resolving detectors. The asymptotic dashed lines indicate the limits of fidelity \mathcal{F}_{op} and efficiency η_{op} reached when spin decoherence is neglected and when the detection window encompasses the entire photon lifetime. The detection window is set to be equal to the entire protocol duration: $t_d = t_0 = 0$ and $t'_d = t_f = T_d$. The gray lines show the probabilities for no photon emission from either system p_0 , for single-photon events p_1 where only one photon is detected, and for photon bunching where two photons arrive at one detector p_2 . The red line indicates the probability for coincident counts p_{11} caused by imperfect HOM interference. Parameters chosen: $\gamma_{j_k} = \gamma/2$ for $j \in \{\uparrow, \downarrow\}$ and $k \in \{1, 2\}$ so that $\gamma_k = \gamma_{\uparrow_k} + \gamma_{\downarrow_k} = \gamma$, $\gamma_k^* = 0.1\gamma$; and $\gamma_{s_k}^{\pm} = \gamma_{s_k}^* = 0.001\gamma$.

by $\eta_{\rm op} = \eta^2/2$ and the fidelity becomes

$$\mathcal{F}_{\rm op} = \frac{1}{2} \left(1 + \operatorname{Re}\left(\frac{\tilde{C}^*_{\uparrow} \tilde{C}_{\downarrow}}{\tilde{M}_{\gamma^*}}\right) \right), \qquad (4.26)$$

where

$$\tilde{C}_{j} = \frac{2\sqrt{\gamma_{1}\gamma_{2}}}{\Gamma_{1} + \Gamma_{2} + 2i\Delta_{j}}$$

$$\tilde{M}_{\gamma^{\star}} = \frac{\gamma_{1} + \gamma_{2} - i(\Delta_{\uparrow} - \Delta_{\downarrow})}{\Gamma_{1} + \Gamma_{2} - i(\Delta_{\uparrow} - \Delta_{\downarrow})},$$
(4.27)

and where $\gamma_k = \gamma_{\uparrow k} + \gamma_{\downarrow k}$ is the total decay rate of the k^{th} system, $\Gamma_k = \gamma_k + 2\gamma_k^*$ is the total optical decoherence rate, Δ_{\uparrow} and Δ_{\downarrow} are the optical detunings between the left and right circularly polarized transitions (respectively) of the systems. Similar to protocols N and T, the factor \tilde{C}_j quantifies the coherence for the which-path erasure of photons from the j transitions at the beam splitter, which depends only on the total decay rates relative to the detuning and dephasing. We attribute the factor $1/\tilde{M}_{\gamma^*}$ to the gain in fidelity due to the systems being initialized in the excited

state, compared to protocols N and T where the systems are initialized in a superposition state and are directly affected by emitter pure dephasing.

The fidelity is bounded from above by the mean wavepacket overlaps $M_{12\uparrow}$ and $M_{12\downarrow}$ of photons from each transition: $\mathcal{F}_{op} \leq (1 + \sqrt{M_{12\uparrow}M_{12\downarrow}})/2$. Interestingly, this inequality can be saturated if $\Delta_{\uparrow} - \Delta_{\downarrow}$ is much smaller than $\gamma_1 + \gamma_2$, implying that the systems have nearly identical spin splittings compared to the system decay rate γ_k . Then we have $\tilde{C}_{\uparrow} \simeq \tilde{C}_{\downarrow}$ and $\tilde{M}_{\gamma^*} \simeq (\gamma_1 + \gamma_2)/(\Gamma_1 + \Gamma_2)$. In this case, the fidelity becomes $\mathcal{F}_{op} = (1 + M_{12})/2$ where $\Delta = \Delta_{\uparrow} = \Delta_{\downarrow}$.

We note that the fidelity of protocol P does not depend on the ratio of the decay rates to each ground state. Rather, it only depends on the total decay rate γ_k of each system, which dictates the photon temporal profile. However, it is still necessary to balance the input intensity at the beam splitter, which may require artificially reducing η_{j_k} if some transitions are brighter than others, consequently reducing η and the overall efficiency.

Detector noise and number resolution.— Because protocol P is a two-photon heralded scheme, it behaves almost identically to protocol T in terms of robustness against photon loss and detector noise. This means that the final measured states $\hat{\varrho}_n$ can be determined using the form of Eq. (4.24). However, unlike protocol T, protocol P is quite robust against non-number resolving detectors even when spin flips occur on the order of the emission timescale. This is because spin flips cannot directly affect the photon statistics of protocol P and so the chance to have more than 1 photon arriving at a given detector remains very small.

4.1.5 Protocol comparison

In this section, we compile the results for each protocol and compare their optical limits of fidelity with respect to each other and the mean wavepacket overlap. We also show the performance of each protocol when including photon loss, detector noise, and spin decoherence.

Optical limits.—In section 4.1.2, the maximum fidelity achievable for the spin-photon number entanglement scheme (protocol N) in the limit that $T_d \gg 1/\gamma_k$, $F_{\eta}(\vartheta) \rightarrow 1$, and $\varphi + \phi = 0$ was



Figure 4.7: A comparison of the optical limits to fidelity for remote entanglement generation protocols. Fidelity in the optical limit \mathcal{F}_{op} for each protocol as affected by emitter temporal profile mismatch γ_2/γ_1 , pure dephasing γ^* , spectral detuning Δ , and spectral diffusion standard deviation δ . These limits are attained when the detection window encompasses the entire photon lifetime and when spin decoherence and loss are negligible. The long dashed blue line represents the fidelity limit for protocol N when $F_{\eta} \rightarrow 1$, the short dashed orange line represents protocol T, and the solid red line represents protocol P. The thin red lines and shaded region represent values bounded by the mean wavepacket overlap M_{12} of photons from each source: $M_{12} \leq \mathcal{F}_{op} \leq (1 + \sqrt{M_{12}})/2$. The labeling and order of all lines are the same across all four panels. The thin vertical gray lines in each plot show the fixed values used for each of the other plots. Parameters used unless otherwise stated: $\gamma_2 = 0.85\gamma_1$, $\gamma_1^* = \gamma_2^* = 0.002\gamma_1$, $\Delta = 0$, and $\delta = 0.02\gamma_1$. For protocol P, we also assume that $\Delta_{\uparrow} = \Delta_{\downarrow} = \Delta$.

found to be $\mathcal{F}_{N} = (1 + \operatorname{Re}(\tilde{C}))/2$ corresponding to a concurrence $\mathcal{C}_{N} = |\tilde{C}|$. The fidelity in section 4.1.3 for the scheme using time-bin entanglement (protocol T) was found to be $\mathcal{F}_{T} = (1 + |\tilde{C}|^{2})/2$ corresponding to a concurrence $\mathcal{C}_{T} = \mathcal{C}_{N}^{2}$. In section 4.1.4, we found the fidelity for the spin-polarization entanglement generation scheme (protocol P) to be $\mathcal{F}_{P} = (1 + M_{12})/2$ when $\Delta_{\uparrow} \simeq \Delta_{\downarrow}$. The corresponding concurrence is $\mathcal{C}_{P} = M_{12}$, where M_{12} is the mean wavepacket overlap of photons from each source.

Knowing that $|\tilde{C}|^2 \leq M_{12}$ (see section 4.1.3) but also $|\tilde{C}| \geq \operatorname{Re}(\tilde{C}) \geq M_{12}$ (see section 4.1.2), we find that $\mathcal{C}_{\mathsf{T}} \leq \mathcal{C}_{\mathsf{P}} \leq \mathcal{C}_{\mathsf{N}}$. In addition, we have that the order is the same for the fidelity as well: $\mathcal{F}_{\mathsf{T}} \leq \mathcal{F}_{\mathsf{P}} \leq \mathcal{F}_{\mathsf{N}}$. Furthermore, since $\mathcal{F}_{\mathsf{N}} \leq (1 + \sqrt{M_{12}})/2$ and $\mathcal{F}_{\mathsf{T}} \geq M_{12}$, the optical limits of fidelity for all three protocols are bounded by

$$M_{12} \leq \mathfrak{F}_{\mathsf{T}} \leq \mathfrak{F}_{\mathsf{P}} \leq \mathfrak{F}_{\mathsf{N}} \leq \frac{1}{2} \left(1 + \sqrt{M_{12}} \right).$$
 (4.28)

From Fig. 4.7, we can see that protocols N and T have parallel behaviour in terms of dephasing

and temporal overlap due to the fact that T can be seen as two applications of N. However, protocol P is implemented with a single pulse on each system like N but it is still a two-photon scheme like T. Hence, it matches the fidelity of N or T in different scenarios.

The dominance of protocol N in the ideal case is expected because the two-photon schemes can naively be seen as two single-photon schemes applied back-to-back, which would compound the error. Because of this, it is tempting to believe that protocol N would then also be less susceptible to spectral diffusion. However, protocols T and P have a symmetry advantage that protocol N does not have. In protocol T, the fact that the second photon must come from the opposite side of the beam splitter causes an opposing phase rotation on the entangled spin state. These two phases cancel, leaving only a reduction in the magnitude of the coherence due to nonzero Δ rather than both a reduction and a phase rotation as seen in protocol N. This is illustrated in Eq. (4.21) where the fidelity depends on $|\tilde{C}|^2$ rather than $\text{Re}(\tilde{C})^2$. A similar symmetry occurs for protocol P, however, the detuning phase is only fully eliminated if $\Delta_{\uparrow} = \Delta_{\downarrow}$. Because of these symmetry advantages, a sufficient amount of spectral detuning or spectral diffusion eliminates the fidelity advantage that the single-photon scheme had over the two-photon protocols (see Fig. 4.7).

Phase errors.—Let us now discuss the impact of the relative initialization phase $\varphi = \varphi_1 - \varphi_2$ and propagation phase $\phi = \phi_1 - \phi_2$ errors. The initialization phase φ_k for each quantum system can independently fluctuate over time causing significant phase errors if the two quantum systems do not share a phase reference. In addition, it may be necessary to stabilize or correct the propagation phase ϕ by monitoring the phase fluctuations of the communication channel [231, 232]. Since the propagation phase depends on distance, this propagation phase fluctuation can become severe for large entanglement generation distances.

As discussed in the previous section, protocols T and P have a symmetry advantage over protocol N for the spectral detuning phase. This advantage extends to propagation phase errors and other possible local phase errors such as initialization phase and the relative precession of the two spin qubits. On the other hand, the upper bound on fidelity for protocol N can be severely degraded by any phase error ϕ becoming $(1/2)(1 + \operatorname{Re}(\tilde{C}e^{i\phi}))$. If this phase fluctuates in a Gaussian



Figure 4.8: A comparison of phase errors for remote entanglement generation protocols. The reduction in entanglement generation fidelity of protocol N as compared to phase-robust protocols T and P for a phase error ϕ fluctuating in a Gaussian distribution around $\phi = 0$ with a standard deviation of σ_{ϕ} . For this comparison, we are neglecting spin decoherence, spectral diffusion, detector dark counts, and detector number resolving limitations. Other parameters used: $\gamma_2 = 0.85\gamma_1$, $\gamma_1^* = \gamma_2^* = 0.002\gamma_1$, and $\Delta = 0.02\gamma_1$. We also assume that $F_{\eta} \rightarrow 1$ for protocol N and that $\Delta_{\uparrow} = \Delta_{\downarrow}$ for protocol P.

distribution [231] centered around $\phi = 0$ with a standard deviation of σ_{ϕ} , then the fidelity reduces to $\mathcal{F}_{N} = (1/2)(1 + \operatorname{Re}(\tilde{C})e^{-\sigma_{\phi}^{2}/2})$ where, as in the previous section, we have assumed $F_{\eta} \to 1$.

For a large enough phase fluctuation σ_{ϕ} , protocol N loses its fidelity advantage over the other two protocols (see Fig. 4.8). We find that the value for the variance σ_{ϕ}^2 where $\mathcal{F}_{N} \leq \mathcal{F}_{T}$ is $\sigma_{\phi}^2 \geq \ln((\Gamma_1 + \Gamma_2)^2/(4\gamma_1\gamma_2))$. Likewise, for $\mathcal{F}_{N} \leq \mathcal{F}_{P}$ we would need $\sigma_{\phi}^2 \geq \ln((\gamma_1 + \gamma_2)^2/(4\gamma_1\gamma_2))$.

Although protocols T and P are very robust against phase errors, they can still be affected under some conditions. If the phase fluctuation occurs on a timescale faster than the separation between pulses for protocol T, then \mathcal{F}_T can be degraded. This could be accounted for in our method by adding different phases for the second detection window when computing the conditional propagators. In addition, significant birefringence in protocol P, quantified by ω_{s_k} , can cause a small degradation of \mathcal{F}_P due to propagation phase errors. However, since $\omega_{s_k} \ll \omega_k$, this effect is orders of magnitude smaller than the degradation experienced by protocol N.

Loss and distance.—Let us now compare the fidelity and efficiency of all three protocols while taking into account all imperfections aside from spectral diffusion and phase errors, which were discussed above.

When including losses and detector noise, the two-photon protocols distinguish themselves

significantly from the single-photon protocol. Although less flexible, T and P are more robust in terms of fidelity than N (see Fig. 4.9 (a)). However, using PNRDs, N can exceed T and P in terms of fidelity in the regime of distributed quantum computing (DQC) where errors may be dominated by optical imperfections. It can also exceed T and P in terms of efficiency in the loss regime of quantum communication (QComm). This latter advantage can come at a significant cost to fidelity if there is significant detector noise, even after optimizing ϑ to minimize the errors caused by both systems emitting a photon.

To simulate each protocol's performance over distance, it is necessary take the classical communication time into account as shown in Eq. (2.6), such that the measurement takes place at the retarded time r(t) = t - L/2c, where $L = 2L_d$ is the total distance between the systems. The final protocol time also cannot be less than $t_f = N_w T_d + L/c$, where N_w is the number of detection windows; $N_w = 1$ for N and P, $N_w = 2$ for T. This delay caused by the classical communication time can cause a degradation of the entanglement generation fidelity due to spin decoherence.

To compare the protocols, we have selected a set of parameters that best illustrate their differences while also remaining relevant to realistic systems. We have chosen an optical lifetime of 10 ns, with a spin T_1^{\pm} time of 20 ms and a spin T_2^{\star} of 10 ms typical of a nitrogen-vacancy center in diamond [151, 158]. However, we have chosen an optimistic pure dephasing rate of 0.2 MHz corresponding to nearly Fourier-transform limited lines, which for many systems would likely require some cavity enhancement or spectral filtering to achieve. In Fig. 4.9 (b), we set $\eta_0 = 0.999$ for L = 0 to illustrate the distance-limited values. In practice, η_0 is much lower due to other inefficiencies such as collection losses. This may include filtering losses as a consequence of suppressing the excitation laser or phonon sideband emission.

Some differences in fidelity between the protocols seen in Fig. 4.9 (a) are washed out by spin decoherence when the distance approaches or exceeds the fibre attenuation length $L_{\text{att}} = 22$ km. However, the differences in efficiency scaling remain apparent, with N having the potential to exceed the efficiency of T and P by a couple orders of magnitude for long-distance entanglement generation, although with a modest fidelity for our chosen parameter set.



Figure 4.9: A comparison of loss and distance for remote entanglement generation protocols. (a) A split-scale plot of the fidelity \mathcal{F}_{gen} and efficiency η_{gen} as a function of total single-photon efficiency η for entanglement generation via spin-photon number entanglement (N), spin-time bin entanglement (T), and spin-polarization entanglement (P) using local detectors with a finite detector dark count probability of $1 - p_{d,0} = 10^{-5} \simeq p_{d,1}$ per detection window of duration T_d . (b) The effect of distance L between the quantum systems on the fidelity and efficiency for each protocol in panel (a) when taking into account spin decoherence in addition to photon loss and detector noise. An increased distance affects both the total protocol duration $t_{\rm f} = N_{\rm w}T_{\rm d} + L/c$ and the single-photon efficiency $\eta = \eta_0 10^{-L/2L_{att}}$, where $c = 2 \times 10^8$ m/s is the speed of transmission in the fibre, $N_{\rm w} = 1$ for N and P, $N_{\rm w} = 2$ for T, and $L_{\rm att} = 22$ km is the fibre attenuation length. (a, b) The dotted blue line represents the protocol N using non-number resolving detectors (BD) and with a fixed $\vartheta = \pi/8$, corresponding to a probability of 1/4 for a source to emit a photon. The longdashed blue line indicated by N' illustrates the noise-limited maximum possible fidelity for N using photon-number resolving detectors (PNRD). The top panel shows the corresponding numerically optimized ϑ and the solid light blue line indicates the analytic approximation for the PNRD model: $\vartheta^4 = (1 - p_{d,0})/(\eta(1 - \eta))$. The short-dashed orange line and solid red line corresponding to T and P, respectively, are visually unaffected when accounting for non-number resolving detectors. Other parameters used: $T_{\rm d} = 5/\gamma_1, \gamma_2 = 0.85\gamma_1, \gamma_k^{\star} = 0.002\gamma_1, \Delta = 0.02\gamma_1, \delta_k = 0, \gamma_{s_k}^{\pm} = 0.5 \times 10^{-6}\gamma_1, \delta_k = 0.02\gamma_1, \delta_k = 0.02\gamma_1, \delta_k = 0.002\gamma_1, \delta_k$ and $\gamma_{s_k}^{\star} = 10^{-6} \gamma_1$ for $k \in \{1, 2\}$. For P, we also assume that $\Delta_{\uparrow} = \Delta_{\downarrow} = \Delta$. For panel (b) we set $\eta_0 = 0.999$ for continuity with panel (a) and choose $\gamma_1 = 10^8$ Hz.

Conclusions.—In this section, we have demonstrated a powerful and intuitive approach based on conditional propagator superoperators to analytically and numerically compute figures of merit for single-photon heralded entanglement generation protocols subject to dephasing. Our method relies on concepts from quantum trajectories and is apt given its resurgence in related techniques for analyzing emitted field states [109, 126, 128]. Our approach includes a multitude of realistic imperfections that must be considered when developing a platform for quantum information processing based on solid-state emitters.

We have provided simple relations to estimate the fidelity and efficiency for three popular entanglement generation protocols. These results are directly useful for developing future proposals for system-specific applications and may also help guide the experimental development of solidstate emitters for quantum information processing. Furthermore, we have used our results to compare these three protocols in order to reveal their strengths and weaknesses in detail.

Although the analysis in this section focused on a simplified three-level model for the quantum systems, our approach can be applied to more complicated systems, such as those in the critical cavity coupling regime [iii], spin-optomechanical hybrid systems [v], or perhaps emitters in unconventional hybrid cavities [103, 166]. It may also prove to be a powerful tool to analyze other photon counting applications when exposed to decoherence process such as novel single-photon interference phenomena [190] or deterministic entanglement generation using feedback [228]. Moreover, by extending the decomposition and measurements to include detector temporal resolution, the methods presented in this paper may provide a foundation to analyze the effects of decoherence on photon time-tagging heralded measurements.

4.2 Deterministic entanglement generation

In addition to remote entanglement generation, quantum networks and related information processing applications also require local gates between stationary qubits. For quantum repeaters, this is necessary to perform entanglement swapping between two quantum memories within a repeater node [202]. The entanglement swapping step can, in principle, be performed by applying a local version of one of the remote entanglement protocols described in the previous section. However, the efficiency of that approach would be limited to at most 50% due to the reliance on a passive linear-optical Bell state measurement. Active feedback [228] or auxiliary photons [i] could be used to increase this efficiency above 50%. Alternatively, if the two stationary qubits can interact directly and in a quantum coherent way, it is possible to perform deterministic two-qubit gates that do not rely on post-selection or active feedback. This can significantly increase the entanglement distribution rates for quantum repeater protocols [iv].

Different types of interactions have been proposed to mediate local two-qubit gates between solid-state defects. For example, one could utilize phonon-mediated couplings [233] or electric [234][iv] or magnetic [235] dipole-dipole interactions. However, in many cases, the single defects may already require an optical cavity to enhance the emission rate in order to improve the entanglement generation efficiency and fidelity. Hence, it is natural to explore the option of using the same optical cavity to mediate the local interaction needed for entanglement swapping.

The idea of using photon-mediated local interactions to build two-qubit gates has been proposed in many forms [236–239]. One major advantage of photon mediated interactions over direct electric or magnetic dipole-dipole interactions is that the two defects do not need to be in close proximity. This makes fabrication and addressability less of a challenge. Optical photon-mediated interactions also have an advantage over phonon-related approaches because the higher-frequency photon modes are less susceptible to environmental noise than lower-frequency phonon modes, especially at noncryogenic temperatures.

Defects coupled to a cavity operating in the strong cavity coupling regime can be used to engineer two-qubit gates, and strong cavity-coupling regime has been observed for some optical systems such as quantum dots [240–242]. However, it is a generally difficult task to achieve strong coupling with single optically-active solid-state defects. It is particularly difficult using defects that have weaker dipole magnitudes than quantum dots, such as rare earth ions and color centers in diamond. In addition, the strong-coupling regime is not the most ideal regime for implementing

remote entanglement generation due to the Purcell saturation and vacuum Rabi splitting in the emission spectrum. Hence, in this section, I take a look at using a cavity system operating in the bad-cavity regime to mediate local interactions between two defects within the same cavity.

The content in this section is related to Ref. [vii] where we analyzed three different approaches to engineering a two-qubit gate between defects coupled to a cavity operating in the bad-cavity regime. The first approach that we studied is a probablistic gate performed by scattering a single photon off of the cavity, as proposed in Refs. [236, 243]. The second approach is an effective adiabatic dipole-dipole interaction induced by a virtual photon exchange between the defects, which has already been explored for microwave QED systems [238, 244]. The required interaction has also been observed for cavity-coupled defects in diamond [245]. The third approach that we explored was a modification of the second approach where weak continuous driving of the defect optical transitions induces a direct spin-spin interaction using a Raman transition. This latter Raman-assisted approach is based on the proposal of Ref. [237], although we modified the implementation to avoid a flaw in the original proposal.

In this section 4.2, I will primarily discuss the second approach analyzed in Ref. [vii] based on a simple virtual photon exchange interaction. However, I will include a more detailed discussion on the effects of pure dephasing in section 4.2.2 that was not included in Ref. [vii]. This discussion about pure dephasing is also related to some results briefly presented in Ref. [viii], which used the results of Ref. [vii] in a repeater analysis. I will comment on this overlap when it arises. In section 4.2.3, I will also extend the analysis of the virtual photon exchange to include a second cavity mode that may interfere with the first, as described by the quasi-normal mode (QNM) model introduced in section 1.3.7. To my knowledge, this analysis provides original results that are not present in the literature.

4.2.1 System

The most general case of system model studied in this section is that of two three-level defects described in section 4.1.1 coupled to a cavity described by a summation of interfering QNMs as described in section 1.3.7. However, I will only use up to two QNMs in my analysis and I will assume that the defect-QNM interactions are independent, which may be violated under certain conditions; for example, if the two dipoles experience a non-negligible electric dipole-dipole interaction. Each defect system is composed of an optically active transition that is coupled to the cavity and a ground state spin qubit. In addition, I assume that the states of each defect can be individually prepared and manipulated using microwave or optical fields.

To capture the defect-specific imperfections, we use the same Markovian master equation for two three-level systems discussed in section 4.1.1, with an additional cavity coupling term $\mathcal{L} = \mathcal{L}_1 \otimes \mathcal{I} + \mathcal{I} \otimes \mathcal{L}_2 + \mathcal{L}_c$. The emitter superoperators \mathcal{L}_k are given by Eq. (4.1) and the cavity-mediated coupling superoperator is given in section 1.3.7 as

$$\mathcal{L}_{c} = -\frac{i}{\hbar} \mathcal{H}_{c} + \sum_{ij} 2\chi_{ij}^{-} \mathcal{D}(\hat{a}_{i}, \hat{a}_{j}), \qquad (4.29)$$

where

$$\hat{H}_c = \sum_{ij} \hbar \chi_{ij}^{\dagger} \hat{a}_i^{\dagger} \hat{a}_j + \sum_{ik} \hbar g_{ik} \hat{\sigma}_{\uparrow k} \hat{a}_i^{\dagger} + \hbar g_{ik}^* \hat{\sigma}_{\uparrow k}^{\dagger} \hat{a}_i.$$
(4.30)

This general system model takes into account the following processes: (1) spin pure dephasing, (2) incoherent spin decay and excitation, (3) emitter pure dephasing, (4) incoherent optical decay and excitation, (5) independent emitter-QNM dipole coupling, (6) QNM interference, (7) QNM dissipation, and (8) independent semi-classical driving of defect optical and spin transitions. However, following the trend in this thesis, I assume that all state preparation and retrieval pulses are perfect in order to obtain upper-bound estimates on the quality of the protocol. In practice, the finite speed and non-unity fidelity of pulsed manipulation of the system will degrade the final protocol fidelity.

To be transparent, this generalized model is composed under some strong assumptions in

addition to those made to reach a Markovian master equation. Most importantly, as I mentioned before, the emitter-QNM dipole coupling is assumed to be independent for each defect, which is likely violated under a proper derivation of the light-matter coupling for more than one dipole from first principles. In addition, the semi-classical driving of the defects may incidentally excite the surrounding cavity environment, causing complicated dynamics not captured by this model. Therefore, the analysis presented in this section should be seen as an idealized scenario and the results for fidelity must be taken as an upper bound on what may be possible with a realistic implementation of this model. I will briefly discuss possible implementations in section 5.2.

4.2.2 Simple virtual photon exchange

The basic principle of the simple virtual photon exchange interaction is to use the cavity to mediate an adiabatic interaction between the defect excited states. This type of interaction may best be described as a long-range dipole-dipole interaction that is enhanced by the confinement of the electromagnetic field, which increases the local density of states between the two dipoles. In the bad-cavity regime, however, the Purcell effect rapidly dissipates any photons that would otherwise mediate the interaction between the two dipoles. This issue can be solved by recognizing that the Purcell effect falls off as $1/\Delta^2$ (see section 1.3.4) while the mediated interaction only falls off as $1/\Delta$ [vii], where Δ is the detuning between the cavity resonance and the two dipole resonances. Therefore, by detuning the cavity away from the resonant dipoles, one can suppress the Purcell effect to the point where the long-range interaction between the dipoles dominates the dynamics. That is, the rate of population exchange R between the defects and the cavity can be made much smaller than the dipole-dipole interaction rate $|\lambda|$. This implies that, if the cavity is initially in the vacuum state, it will remain approximately in the vacuum state during the entire protocol. Since this long-range interaction causes an efficient coherent exchange of energy between the two defects, yet no photon is ever produced in the cavity, it is said that the dipoles exchange a virtual, interaction mediating, photon.

To put this into a more mathematical context, consider the simplified case where there is no dissipation or decoherence. Then, suppose we prepare our system by exciting one defect so that the initial state includes some non-zero amplitude of the state $|e\uparrow 0\rangle_{12c}$. If we assume that the transitions $|\uparrow\rangle_k \longleftrightarrow |e\rangle_k$ are each coupled to the cavity by the rate g, then the state $|\uparrow\uparrow1\rangle$ will couple to both our initial state and the state $|\uparrow e 0\rangle_{12c}$, which is resonant with our initial state. The Hamiltonian for such a simplified case is given by the two-emitter independent Jaynes-Cummings interaction $\hat{H} = \hbar \Delta \hat{a}^{\dagger} \hat{a} + \sum_{k} \hbar g (\hat{\sigma}_{\uparrow k} \hat{a}^{\dagger} + \hat{\sigma}_{\uparrow k}^{\dagger} \hat{a})$ in the frame rotating with the resonant dipoles. If $|\uparrow\uparrow\uparrow1\rangle$ is far-detuned so that $\Delta \gg g$ we can apply adiabatic elimination (see section 1.5.4) to eliminate the mediating cavity state $|1\rangle_c$ from the dynamics. This results in the effective interaction Hamiltonian $\hat{H}_{\text{eff}} = \hbar (g^2 / \Delta) (\hat{\sigma}_{\uparrow 1} \hat{\sigma}_{\uparrow 2}^{\dagger} + \hat{\sigma}_{\uparrow 1}^{\dagger} \hat{\sigma}_{\uparrow 2}) \text{ in the subspace } \{ |e \uparrow 0 \rangle, |\uparrow e 0 \rangle \}.$ By letting the initial state evolve for time $T_{\text{gate}} = \pi \Delta/g^2$, the system will return back to the initial state but with an additional relative π phase compared to the other product states of the two defects, which have not experienced the exchange interaction. Hence, the virtual photon exchange allows one to implement a deterministic entangling controlled phase gate between the two defects. By de-exciting the initially-excited defect after time T_{gate} , the interaction is halted and the state is retrieved from the interacting subspace, transferring the entanglement to the spin qubits.

Let us now discuss how decoherence can affect this two-qubit gate. As mentioned before, the interaction is successful only if the effective dipole-dipole coupling rate g^2/Δ is larger than R. However, we must also ensure that g^2/Δ is larger than the bare decay rates of the individual defects γ . This is because Δ cannot be increased too much or the defect excited states will decay before the gate is complete. Furthermore, the phase gate relies on the maintenance of coherence between the two excited states $|e\uparrow 0\rangle$ and $|\uparrow e 0\rangle$ and also between these excited states and the ground state spin qubits. Hence, any emitter dephasing γ^* will degrade the entanglement fidelity, in addition to causing a damping effect on the effective interaction rate g^2/Δ . Finally, any decoherence processes affecting the spin qubits, represented by rates γ_s^* and γ_s^{\pm} , must occur on a timescale slower than T_{gate} . To summarize, the condition for a successful gate is $g^2/\Delta \gg R$, γ , γ^* , γ_s^{\pm} , γ_s^* . In the remainder of the analysis, I will assume that spin decoherence, represented by parameters γ_s^{\pm} and γ_s^* , are

negligible on the timescale of the gate.

Before discussing more details about how emitter pure dephasing degrades the gate, let us consider just the optical dissipation. In Ref. [vii], we analyzed the simple virtual photon exchange mediated by a single cavity mode. This model can be recovered from the general QNM master equation described in the previous section by taking $\chi_{11}^+ = \omega_c$, $\chi_{11}^- = \kappa/2$, $g_{11} = g_{12} = g$, and $\chi_{ij}^\pm = g_{jk} = 0$ otherwise. Then, I will assume the optical transitions are resonant $\omega_{\uparrow_1} = \omega_{\uparrow_2}$ so that $\Delta = \omega_{\uparrow_k} - \omega_c$ is the cavity-emitter detuning for both emitters. In Ref. [vii], we considered the case where the optical transitions may be slightly detuned and may also have different cavity coupling rates, but I will not discuss those subtleties here.

To obtain an analytic expression for the entanglement fidelity that includes dissipative phenomena, we first apply a non-Hermitian approximation to the Liouville superoperator as described in section 1.5.3. This gives a non-Hermitian Hamiltonian that describes the evolution of the quantum trajectory conditioned on no photon emission from either defect or the cavity mode. Then, making the adiabatic approximation for $\Delta \gg g$, we eliminate the cavity mode to obtain an effective non-Hermitian Hamiltonian using the methods described in section 1.5.4. The adiabatic non-Hermitian Hamiltonian for the single-excited system subspace is

$$\tilde{H}_{\rm AE} = \hbar \lambda (\hat{\sigma}_{\uparrow_1}^{\dagger} \hat{\sigma}_{\uparrow_2} + \hat{\sigma}_{\uparrow_1} \hat{\sigma}_{\uparrow_2}^{\dagger}) - \frac{i\hbar}{2} (\gamma + R) \left(\hat{\sigma}_{\uparrow_1}^{\dagger} \hat{\sigma}_{\uparrow_1} + \hat{\sigma}_{\uparrow_2}^{\dagger} \hat{\sigma}_{\uparrow_2} \right)$$
(4.31)

where $\lambda = -2g^2/(2\Delta - i\kappa)$ is the effective (non-Hermitian) dipole-dipole coupling rate and $R = 4g^2\kappa/(\kappa^2 + 4\Delta^2)$ is the Purcell enhancement. The ideal gate time is given by $T_{\text{gate}} = \pi/|\lambda| = \pi \left(2g^2/\sqrt{\kappa^2 + 4\Delta^2}\right)^{-1}$, which is slightly damped by the cavity dissipation. However, since high fidelity is only obtained when $\Delta \gg \kappa$, it is reasonable to ignore this damping and choose $T_{\text{gate}} = \pi g^2/\Delta$. From this effective non-Hermitian dipole-dipole dynamics, it is then straightforward to solve for the phase gate fidelity that includes dissipative effects.

In this section, I will define the phase gate fidelity \mathcal{F}_{gate} as the fidelity of the final state after applying the gate to the initially unentangled state after perfect excitation of defect 1: $|\psi(0)\rangle =$

 $(1/2)(|e\rangle + |\downarrow\rangle)_1 \otimes (|\uparrow\rangle + |\downarrow\rangle)_2 \otimes |0\rangle_c$. In this case, a perfect phase gate would provide the expected final entangled state $|\psi_f\rangle = (1/2)(|e\downarrow\rangle - |e\uparrow\rangle + |\downarrow\downarrow\rangle + |\downarrow\uparrow\rangle)_{12} \otimes |0\rangle_c$. Using this definition along with Eq. (4.31), we obtain

$$\mathcal{F}_{\text{gate}} = \frac{1}{4} \left[1 + e^{-(\gamma + R)t/2} \left| \sin\left(\frac{t\lambda}{2}\right) \right|^2 \right]^2.$$
(4.32)

The product states $|\downarrow\downarrow\rangle$ and $|\downarrow\uparrow\rangle$ of the initial state $|\psi(0)\rangle$ experience no evolution under Eq. (4.31) and so they contribute 1/4 to the fidelity by default. Only the relative phase and magnitudes of the excited states $|e\downarrow\rangle$ and $|e\uparrow\rangle$ contribute to the degradation of the fidelity, which is captured by the second term within the brackets of Eq. (4.32). Note also that this fidelity expression may not give the expected sin² behaviour in time due to the non-Hermitian coupling term λ , which is complex in general.

By setting the evolution time to the chosen value of $t = T_{\text{gate}} = \pi \Delta/g^2$ we have $t\lambda \simeq \pi$ if $\Delta \gg \kappa$. This latter condition holds in the far-adiabatic regime. In this regime, only the exponential term remains in the fidelity. Thus, the fidelity is limited only by the dissipative rate in the high-fidelity, far-adiabatic, regime. Writing the fidelity in terms of the uninhibited cavity cooperativity $C = 4g^2/\kappa\gamma$, we obtain

$$\mathcal{F}_{\text{gate}} \simeq \frac{1}{4} \left(1 + e^{-2\pi\Delta/C\kappa - \pi\kappa/2\Delta} \right)^2. \tag{4.33}$$

This expression can be maximized to find that $\Delta = (\kappa/2)\sqrt{C}$ is the ideal detuning condition [vii]. This condition is identical to requiring that the bare cavity dissipation is equal to the (detuning inhibited) Purcell rate: $\gamma = R \simeq g^2 \kappa/\Delta^2$, which illustrates that the fidelity is truly limited by decoherence from the rate trade-off between the cavity dissipation and the defect spontaneous emission. However, the general way to maximize fidelity in this far-adiabatic regime is to minimize the dissipation during the gate time: $T_{\text{gate}}(R + \gamma)$. It just happens to be the case that $T_{\text{gate}}R$ and $T_{\text{gate}}\gamma$ are reciprocally related via Δ , and hence the maximum fidelity is obtained when the two rates are equal. In any case, by substituting this solution back into the fidelity, we can reveal the simple maximum fidelity scaling of $\max(\mathcal{F}_{\text{gate}}) = \left(1 + e^{-2\pi/\sqrt{C}}\right)^2/4$, which for large C converges to $1 - 2\pi/\sqrt{C}$ [vii]. This sets a fundamental upper bound on the simple virtual photon exchange control phase gate fidelity for a given cavity cooperativity when only one cavity mode is involved.

Let us now take a look at how pure dephasing degrades the fidelity. In principle, we can perform the non-Hermitian approximation even when pure dephasing is present. But the interpretation in terms of quantum trajectories is not as clear because pure dephasing is an elastic process. That is, it can be seen to cause 'jumps' from a state back to that same state. Consider the emitter pure dephasing term $2\gamma^{\star}\mathcal{D}(\hat{\sigma}^{\dagger}\hat{\sigma})$. Expanding this out, we can identify that the pure-dephasing part of the non-Hermitian Hamiltonian is $-i\gamma^{\star}\hat{\sigma}^{\dagger}\hat{\sigma}$. Hence, it should have an identical effect as $\gamma/2$ resulting in a simple replacement of C with an inhibited cavity cooperativity $C^{\star} = 4g^2/\kappa\Gamma$; but, as we briefly mentioned in Ref. [viii], this is not actually the case. Recall that, in the context of fidelity, the non-Hermitian approximation makes the assumption that the fidelity obtained when not following the ideal quantum trajectory is much smaller than the fidelity obtained when following the ideal quantum trajectory (see section 1.5.3). This is always true for dissipation in this scheme because a single jump will remove the system from the ideal subspace causing immediate and total decoherence. However, pure dephasing can never remove the system from the ideal subspace. Hence, dissipation and dephasing do not have an equivalent effect on the state fidelity as they would, for example, on the temporal coherence of an emitted photonic state. For this reason, replacing Cwith C^{\star} in the above fidelity expression generally leads to an *underestimation* of the true fidelity.

Instead of using the non-Hermitian approximation to obtain a non-Hermitian Hamiltonian, we can make use of the conditional propagator superoperator \mathcal{U}_0 corresponding to no photon emission to keep effects of pure dephasing fully intact. That is, we can consider the dynamics associated with the Liouville superoperator $\mathcal{L}_0 = \mathcal{L} - \gamma \mathcal{J}(\hat{\sigma}_{\uparrow 1}) - \gamma \mathcal{J}(\hat{\sigma}_{\uparrow 2}) - \kappa \mathcal{J}(\hat{a})$, where $\mathcal{J}(\hat{A})\hat{\rho} = \hat{A}\hat{\rho}\hat{A}^{\dagger}$. Although this does not allow us to reduce the dimension of the problem as before—because the pure dephasing term within \mathcal{L}_0 prevents us from writing an effective Hamiltonian—it does still decouple the equations of motion for many of the density matrix elements. By adiabatically eliminating all density matrix elements (coherence and population) corresponding to the cavity mode, we can diagonalize the effective Liouville superoperator and obtain an analytic solution for the fidelity valid in the bad-cavity regime that captures pure dephasing.

The exact expression for fidelity obtained by diagonalizing the Liouville superoperator is too large to write out, but by applying the same far-adiabatic regime assumptions used to get Eq. (4.33), the expression is found to be well-approximated by

$$\mathcal{F}_{\text{gate}} \simeq \frac{1}{4} + \frac{1}{2} e^{-2\pi\Delta/C^{\star}\kappa - \pi\kappa/2\Delta} + \frac{1}{16} \left(1 + 3e^{(2\pi\Delta/\kappa)(1/C - 1/C^{\star})} \right) e^{-4\pi\Delta/C\kappa - \pi\kappa/\Delta}, \tag{4.34}$$

Note that by setting $C^* = C$ (hence $\gamma^* = 0$), we recover Eq. (4.33) by factorization. This solution explicitly shows that pure dephasing does not affect the fidelity in the same way as dissipation. If it did, C would not appear in the fidelity. Interestingly, it turns out that we can estimate an effective cavity cooperativity C_{eff} for this particular scheme. To do so, we can postulate that $C_{\text{eff}} = 4g^2/\kappa(\gamma + c\gamma^*) = C(1 + c(\gamma^*/\gamma))^{-1}$ for some unknown coefficient c. For example, if c = 0we get $C_{\text{eff}} = C$ and if c = 2 we get $C_{\text{eff}} = C^*$. By substituting this in place of C in Eq. (4.33) and minimizing the difference with Eq. (4.34) that includes pure dephasing exactly, we can analytically solve for the best value of c to lowest order for when $C \gg 1$. The result is c = 11/8, which is independent of the relative magnitudes of γ and γ^* . Actually, it turns out that, without making the additional far-adiabatic assumptions to simplify Eqs. (4.33) and (4.34), the coefficient can be found to be c = 21/16, which turns out to be slightly more accurate. In any case, since c < 2, this implies that pure dephasing has less of an impact on fidelity than would be expected from C^* alone. I should also note that in Ref. [viii], we mention a value of c = 0.61, but this was guessed from numerical observations and not analytically derived as was done here.

Fig. 4.10 shows a comparison of the analytic solution obtained by applying adiabatic elimination to the conditional Liouville superoperator and points computed from a numerical simulation of the full master equation. I have also plotted the fidelity from Eq. (4.33) computed using the effective cavity cooperativity with c = 21/16. All these computations assume that the system is in the bad-cavity regime with $\kappa \gg \gamma C$ and C > 1 (hence $\gamma \ll \kappa$). For the numerical simulation, I used $\kappa = 20g$ and $\gamma = 10^{-7}\kappa$ for a cavity cooperativity of $C = 10^5$. It is impressive to note that the



Figure 4.10: The effect of emitter pure dephasing on the simple photon exchange controlled phase gate fidelity for local entanglement generation. (a) The phase gate fidelity as a function of cavity detuning Δ in units of the cavity FWHM linewidth κ . The black points are numerically exact values computed from simulating the full master equation with perfect state preparation and retrieval pulses. The orange and cyan colored lines indicate the analytic solution obtained from applying adiabatic elimination to the conditional Liouville superoperator using no pure dephasing (orange) and pure dephasing of $\gamma^*/\gamma = 10$ (cyan). The black dashed lines indicate the solution Eq. (4.33) using an effective cavity cooperativity C_{eff} . The emitter pure dephasing degrades the fidelity for this scheme by inhibiting the cavity cooperativity by a factor $C_{\text{eff}} = C(1 + c\gamma^*/\gamma)^{-1}$ where c = 21/16. The maximum fidelity occurs at $2\Delta = \kappa \sqrt{C_{\text{eff}}}$ with a value well-approximated by $\max(\mathcal{F}_{\text{gate}}) = (1 + e^{-2\pi/\sqrt{C}})^2/4 \simeq 1 - 2\pi/\sqrt{C_{\text{eff}}}$ when $C_{\text{eff}} \gg 1$. (b) The maximum phase gate fidelity as a function of emitter pure dephasing for three different values of cavity cooperativity. The horizontal gray lines carry over from panel (a).

fidelity computed using the effective cavity cooperativity with c = 21/16 provides a near flawless estimate of the fidelity in this regime. However, both analytic solutions slightly deviate from the numerical values when $\Delta/\kappa < 1$ or $C \simeq 1$, which is expected because this is where adiabatic evolution breaks down.

The main message to take away from this section is that the simple virtual photon exchange interaction using a single cavity mode allows for a maximum fidelity near $1-2\pi/\sqrt{C_{\text{eff}}}$. This means that, for a high fidelity such as >0.99, it is necessary to use a system with an effective cooperativity exceeding 10^5 , which is very demanding for solid-state systems that suffer from dephasing. In the next section, I will outline how a cavity with a Fano resonance could surpass this inverse square-root bound, potentially allowing for high fidelity gates using much smaller cavity cooperativity values.

4.2.3 Subradiant virtual photon exchange

As I described in the previous section, aside from pure dephasing, the two main processes hindering the phase gate fidelity are the bare spontaneous emission and the Purcell enhanced emission. These two processes place opposing constraints on the ideal detuning, giving rise to a maximum fidelity at some intermediate detuning value. By introducing a second cavity mode that can interfere with the first, this trade-off can be modified, potentially allowing for regimes that surpass the single-mode case.

A Fano resonance is a spectral phenomenon that occurs when a broad spectral mode resonance interferes with a detuned narrow mode resonance. This interference gives rise to a characteristic asymmetric spectral feature surrounding the narrow mode resonance frequency. This asymmetry occurs because the phase contribution from a detuned broad mode changes monotonically as the frequency is swept over the narrow mode resonance. The abrupt change in phase at the narrow mode resonance then causes the interference between the two modes to be very different on either side of the narrow mode resonance. Thus, the situation can occur where destructive interference occurs on one side of the narrow resonance, forming a 'Fano dip' where the modes are out of phase and a constructive interference forms a 'Fano peak' where the modes are in phase on the other side.

Fano resonances often arise in the context of energy transfer between two systems, in particular involving plasmonics [246, 247]. Fano resonances have also been proposed to enhance single-photon indistinguishability [248] and suppress quenching for hybrid cavities [166]. However, to my knowledge, the quantum properties of a phase gate interaction taking advantage of an optical Fano resonance have not been studied. The methods detailed in the previous section have now set us up to analyze this question using the quantum model recently presented in Ref. [103], which I introduced in section 1.3.7.

Let us consider two QNMs with complex frequencies $\tilde{\omega}_k = \omega_k - i\kappa_k/2$ coupled to two opticallyactive three-level systems. For simplicity, I will neglect spin decoherence and emitter pure dephasing for now. In this case, we can use the non-Hermitian approximation to obtain an effective non-Hermitian Hamiltonian. To do so, we must first diagonalize the dissipative part of the Liouville superoperator [106] to obtain a superradiant and a subradiant mode. As explained in section 1.3.7, the dissipation for the dual QNM system is given by the 2×2 matrix χ^- where $\chi_{ij}^- = i(\chi_{ij} - \chi_{ij}^*)/2$, $\chi_{ij} = \sum_k (\mathbf{S}^{-1/2})_{ik} \tilde{\omega}_k (\mathbf{S}^{1/2})_{kj}$, and \mathbf{S} is a positive semi-definite overlap matrix describing the amount of interference between the modes. For simplicity, I will fix the diagonal elements of \mathbf{S} to 1 so that the modes are not renormalized by the interference. However, in practice this renormalization depends strongly on the system geometry and dissipation, ranging from being fairly small [103] to quite significant [106]. I will also consider a general case for the off-diagonal element $S_{12} = S_{12}^* = |S_{12}|e^{i\phi_s}$. The assumption that S_{12} is independent of the diagonal elements may not be true when varying parameters of the system geometry or material composition. Hence, these simplifying assumptions are used to explore what possibilities arise from this model and to motivate further investigation.

In addition to the simplifying assumptions about S detailed above, we must also ensure that S cannot allow for a physically impossible amount of interference based on the spectral overlap of the QNMs. This means that we must restrict the eigenvalues of χ^- to be positive. Applying these additional constraints gives

$$|S_{12}| \le 2\sqrt{\frac{\kappa_1 \kappa_2}{4\Delta_c^2 + (\kappa_1 + \kappa_2)^2}},$$
(4.35)

where $\Delta_c = \omega_1 - \omega_2$ is the detuning between the two QNM resonances. For convenience, I define a new parameter $0 \le s \le 1$ where $|S_{12}| = 2s\sqrt{\kappa_1\kappa_2/(4\Delta_c^2 + (\kappa_1 + \kappa_2)^2)}$. Then s = 0 implies no interference and s = 1 implies maximum possible interference. Caution must also be taken when considering the case of s = 0. If the modes still significantly overlap in frequency, then s = 0inherently implies that the modes must not overlap significantly in space. This then complicates the notion of a two-mode cavity. For realistic implementations, the matrix S should be calculated fully from classical electromagnetic simulations [103, 106].

After diagonalizing χ^- so that $\sum_{ij} \chi^-_{ij} \mathcal{D}(\hat{a}_i, \hat{a}_j) = \kappa^+ \mathcal{D}(\hat{c}_+, \hat{c}_+) + \kappa^- \mathcal{D}(\hat{c}_-, \hat{c}_-)$, we have a superradiant cavity mode \hat{c}_+ and a subradiant cavity mode \hat{c}_- corresponding to the exponential

dissipative rates $\kappa^+ > \kappa^-$ that are the eigenvalues of χ^- . On one hand, if s = 1, then we have $\kappa^- = 0$ and $\kappa^+ = \kappa_1 + \kappa_2$, which represents the most extreme case of interference. On the other hand, if s = 0, then we have $\kappa^- = \kappa_2$ and $\kappa^+ = \kappa_1$. In this diagonalized dissipative basis, the non-Hermitian approximation can be easily applied by considering the evolution induced by the Liouville superoperator $\mathcal{L}_0 = \mathcal{L} - \kappa^+ \mathcal{J}(\hat{c}_+) - \kappa^- \mathcal{J}(\hat{c}_-) - \gamma \mathcal{J}(\hat{\sigma}_{\uparrow 1}) - \gamma \mathcal{J}(\hat{\sigma}_{\uparrow 2})$ conditioned on no photon emission (or absorption) from either cavity mode or bare spontaneous emission from either dipole systems. This approximation gives a lower bound on the phase gate fidelity (see section 1.5.3). If no other decoherence processes are captured by \mathcal{L}_0 , such as pure dephasing, we can write an effective non-Hermitian Hamiltonian governing the evolution of the ideal quantum trajectory.

$$\tilde{H}_{\text{eff}} = \hat{H}_1 + \hat{H}_2 + \hat{H}_c - \frac{i\hbar\gamma}{2} \left(\hat{\sigma}^{\dagger}_{\uparrow_1} \hat{\sigma}_{\uparrow_1} + \hat{\sigma}^{\dagger}_{\uparrow_2} \hat{\sigma}_{\uparrow_2} \right) - \frac{i\hbar}{2} \left(\kappa^+ \hat{c}^{\dagger}_+ \hat{c}_+ + \kappa^- \hat{c}^{\dagger}_- \hat{c}_- \right).$$
(4.36)

Note that \hat{H}_c contains the quantum interaction between the symmetrized modes \hat{a}_1 and \hat{a}_2 via the rates χ_{ij}^+ . Since \hat{c}_{\pm} are given as a linear combination of \hat{a}_1 and \hat{a}_2 dictated by the eigenvectors of χ^- , the operators $\hat{c}_{\pm}^{\dagger}\hat{c}_{\pm}$ will alter the QNM interaction in \hat{H}_c . In the symmetrized QNM basis, the non-Hermitian QNM Hamiltonian simply reverts to $\sum_{ij} \hbar \chi_{ij} \hat{a}_i^{\dagger} \hat{a}_j$. Hence, we have

$$\tilde{H}_{\text{eff}} = \sum_{k} \hbar \left(\omega_o - \frac{i}{2} \gamma \right) \hat{\sigma}_{\uparrow_k}^{\dagger} \hat{\sigma}_{\uparrow_k} + \sum_{ij} \hbar \chi_{ij} \hat{a}_i^{\dagger} \hat{a}_j + \sum_{ik} \hbar \left(g_{ik} \hat{\sigma}_{\uparrow_k} \hat{a}_i^{\dagger} + g_{ik}^* \hat{\sigma}_{\uparrow_k}^{\dagger} \hat{a}_i \right).$$
(4.37)

By adiabatically eliminating the cavity modes using the methods in section 1.5.4, we can obtain the effective non-Hermitian dipole-dipole Hamiltonian.

The first thing we can look at is the expected Purcell enhancement on the individual dipole systems. Similar to sections 1.3.4 and 1.3.6, we can estimate the dissipative rate in the bad-cavity regime by adiabatically eliminating the cavity modes and looking at the remaining imaginary part of the non-Hermitian Hamiltonian corresponding to one of the dipole excited states. Let \tilde{H}_{AE} be the effective non-Hermitian Hamiltonian after the cavity excited state amplitudes are adiabatically eliminated (see section 1.5.4). Then the Purcell rate for an individual dipole is given

by $R_k \simeq -2 \operatorname{Im}\left[\langle e | \tilde{H}_{AE} | e \rangle_k\right] - \gamma$, when the other dipole system is in the state $|\downarrow\rangle$ so as to not be coupled to the cavity. The solution can be computed from applying adiabatic elimination to Eq. (4.37) or by expanding the general expression derived for the Purcell rate in Ref. [103]

$$R_{k} = \frac{|\tilde{g}_{1k}|^{2}\kappa_{1}}{|\tilde{\Delta}_{1}|^{2}} + \frac{|\tilde{g}_{2k}|^{2}\kappa_{2}}{|\tilde{\Delta}_{2}|^{2}} + 2\operatorname{Re}\left[\tilde{g}_{1k}\tilde{g}_{2k}^{*}S_{12}\left(\frac{1}{\tilde{\Delta}_{1}^{*}} + \frac{1}{\tilde{\Delta}_{2}}\right)\right],$$
(4.38)

where $\tilde{\Delta}_i = \omega_o - \tilde{\omega}_i$ is the complex detuning between the resonant dipoles and the i^{th} cavity mode resonance. Furthermore, we can compute the effective (non-Hermitian) dipole-dipole coupling rates from $\lambda_{12} = \langle \uparrow e | \tilde{H}_{AE} | e \uparrow \rangle$ and $\lambda_{21} = \langle e \uparrow | \tilde{H}_{AE} | \uparrow e \rangle$, which are

$$\lambda_{21} = -\frac{\tilde{g}_{11}}{\tilde{\Delta}_1} \left(\tilde{g}_{12}^* + S_{12}^* \tilde{g}_{22}^* \right) - \frac{\tilde{g}_{21}}{\tilde{\Delta}_2} \left(\tilde{g}_{22}^* + S_{12} \tilde{g}_{12}^* \right) \lambda_{12} = -\frac{\tilde{g}_{12}}{\tilde{\Delta}_1} \left(\tilde{g}_{11}^* + S_{12}^* \tilde{g}_{21}^* \right) - \frac{\tilde{g}_{22}}{\tilde{\Delta}_2} \left(\tilde{g}_{21}^* + S_{12} \tilde{g}_{11}^* \right).$$

$$(4.39)$$

Note that, due to the complex values S_{12} and $\tilde{\Delta}_k$, we do not have $\lambda_{21} = \lambda_{12}^*$ in general.

Fig. 4.11 shows example spectra for the Purcell enhancement and the effective dipole-dipole coupling with and without interference. It is crucial to note that the characteristics of the spectrum depend strongly on the phases of the dipole-QNM couplings g_{ik} , the cavity mode overlap phase ϕ_s , and magnitude *s*. The values used in Fig. 4.11 and throughout this section are chosen without consideration for a given physical implementation. Rather, my intention is to illustrate the limits on what could be physically possible for a system following this model. To that end, let us uncover the ideal operating parameter regime and derive the upper bound cooperativity scaling.

To achieve a high phase gate fidelity using a Fano resonance, it is necessary to engineer the system parameters so that the effective dipole-dipole coupling rates induced by each cavity mode constructively interfere within the Fano dip. When the dipoles are far-detuned from the two modes that are themselves quite far separated, then we have $\Delta_i = \omega_o - \omega_i \gg \kappa_i$ so that $\tilde{\Delta}_i = \Delta_i$ and $|S_{12}| \ll 1$ even if s = 1. Under these conditions, the effective dipole-dipole coupling rate reduces to $\lambda = \lambda_{12} = \lambda_{21} = \lambda_1 + \lambda_2$ where $\lambda_1 = -\tilde{g}_{11}\tilde{g}_{12}^*/\Delta_1$ is the contribution from the broad mode and



Figure 4.11: Purcell enhancement spectrum and effective dipole-dipole coupling rate for a dual quasinormal mode model. (a) Example spectrum for the Purcell enhancement caused by coupling to two QNMs that are not interfering (s = 0). The orange and black lines show the Purcell decay enhancement R_k given by Eq. (4.38) of each dipole and the green line shows the effective dipole-dipole coupling rate $\lambda = |\lambda_{12}| = |\lambda_{21}|$ given by Eq. (4.39) that is induced by coupling to the cavity modes. (b) The same case as in panel (a) but with maximum interference s = 1 forming a Fano resonance with a dip appearing to the right of the narrow mode. Notably, the effective dipole-dipole coupling rate remains large even though the Purcell enhancement is suppressed, indicating that an adiabatic coupling can be mediated by the subradiant mode of the cavity. Parameters used: $\gamma = \gamma_1 = \gamma_2 = 10^{-3}\kappa_2$, $\kappa_1 = 5\kappa_2$, $\Delta_c = 10\kappa_2$, $\phi_s = 0$, $\tilde{g}_{11} = \tilde{g}_{12} = i\sqrt{C\kappa_1\gamma/4}$, and $\tilde{g}_{22} = \tilde{g}_{21} = \sqrt{C\kappa_2\gamma/4}$ with C = 100.

 $\lambda_2 = -\tilde{g}_{21}\tilde{g}_{22}^*/\Delta_2$ is the contribution from the narrow mode. For constructive quantum interference to occur, it is necessary that we have $\operatorname{Arg}(\lambda_1) = \operatorname{Arg}(\lambda_2)$. We can also identify the ideal gate time in this regime as $T_{\text{gate}} = \pi/|\lambda|$.

For the phase gate to have an enhancement from the Fano dip, it is necessary that both dipoles experience a Fano dip when they are resonant. From the expression for R_k , we can identify that the Fano dip occurs when the third term is negative. Hence, if there is a Fano dip, then for it to occur at the same frequency for both dipoles it is necessary to have $\operatorname{Arg}(\tilde{g}_{11}\tilde{g}_{21}^*) = \operatorname{Arg}(\tilde{g}_{12}\tilde{g}_{22}^*)$. Note that for small $|S_{12}|$, this is also roughly equivalent to the condition $\operatorname{Arg}(g_{11}g_{21}^*) = \operatorname{Arg}(g_{12}g_{22}^*)$. For convenience, let us define the phases of these cavity coupling rates as $\phi_{ik}^g = \operatorname{Arg}(\tilde{g}_{ik})$ so that the condition becomes $\phi_{11}^g - \phi_{21}^g = \phi_{12}^g - \phi_{22}^g$.

Going back to the condition $\operatorname{Arg}(\lambda_1) \simeq \operatorname{Arg}(\lambda_2)$, we have three cases: (1) the dipole resonances are blue-detuned from both cavity resonances implying $\Delta_i > 0$, (2) the dipole resonances are in between the cavity resonances implying $\Delta_1 \Delta_2 < 0$, and (3) the dipole resonances are red-



Figure 4.12: Time dynamics of the virtual photon exchange phase gate fidelity with a Fano resonance enhancement. (a) The top panel shows the Purcell enhancement R_k/γ and effective dipole-dipole coupling $|\lambda|$ for a single cavity mode with FWHM linewidth κ_2 . The lower panel shows the phase gate fidelity $\mathcal{F}_{\text{gate}}$ as a function of cavity-dipole detuning Δ and evolution time t. The dashed line shows the ideal gate time $T_{\text{gate}} = \pi |\Delta|/g^2$ and the star indicates the point of maximum fidelity corresponding to $\Delta = \kappa_2 \sqrt{C}/2$. (b) The top panel shows the characteristic Fano resonance spectrum and effective dipole-dipole coupling for maximum interference s = 1. The bottom panel shows the ideal gate time $T_{\text{gate}} = \pi/|\lambda|$ where, $\lambda \simeq \lambda_1 + \lambda_2$, $\lambda_1 = \tilde{g}_{11}\tilde{g}_{12}^*/(\Delta_c + \Delta_2)$ and $\lambda_2 = \tilde{g}_{21}\tilde{g}_{22}^*/\Delta_2$. Parameters used: $\Delta_c = 20\kappa_2$, $\kappa_1 = 10\kappa_2$, $\gamma = \gamma_1 = \gamma_2 = \kappa_1 10^{-5}$, $\phi_s = \pi/2$, $\tilde{g}_{11} = \tilde{g}_{12}$, and $\tilde{g}_{21} = \tilde{g}_{22}$. For panel (a): C = 1000. For panel (b): $C_{ik} = 4|\tilde{g}_{ik}|^2/\gamma_i \kappa_k = 500$.

detuned from both cavity resonances implying $\Delta_i < 0$. Cases (1) and (3) imply that we need $\phi_{11}^g - \phi_{12}^g = \phi_{21}^g - \phi_{22}^g + \pi$. By applying the condition $\phi_{11}^g - \phi_{21}^g = \phi_{12}^g - \phi_{22}^g$ required for both dipoles to experience a Fano dip while resonant with each other, we can see that case (2) suggests $\pi = 0$, which is a direct contradiction. However, cases (1) and (3) give $\phi_{11}^g + \phi_{22}^g = \phi_{12}^g + \phi_{21}^g$, which can (in principle) be satisfied. Therefore, a high fidelity phase gate is only possible if both dipoles are resonant with each other and either red-detuned from both cavity modes or blue-detuned from both cavity modes, but never in between. Finally, it is still necessary to engineer the system so that a Fano dip actually

occurs in the spectrum. From R_k , we can see that a good Fano dip would require that the Re function picks out the most negative term possible. By expanding the third term, we can see that a good estimate of the phase condition showing the best Fano dip is $\phi_s + \phi_{1k}^g - \phi_{2k}^g \simeq \pm \pi/2 + 2m\pi$ for an integer m. This picks out the term(s) $\pm 2|\tilde{g}_{1k}\tilde{g}_{2k}S_{12}|(\Delta_1/|\tilde{\Delta}_1|^2 - \Delta_2/|\tilde{\Delta}_2|^2)$. Since I have defined $\Delta_1 > \Delta_2$, for a Fano dip to occur for case (1) or (3), we must select the + case whereas the – case puts a Fano dip between the mode resonances. All of these conditions are satisfied in the example given in Fig. 4.11, where $\phi_{11}^g = \phi_{12}^g = \pi/2$, $\phi_{22}^g = \phi_{21}^g = 0$, and $\phi_s = 0$. However, there is no single unique choice. For example, $\phi_{11}^g = \phi_{12}^g = \phi_{22}^g = \phi_{21}^g = 0$ and $\phi_s = \pi/2$ also satisfy the conditions.

Using the effective Hamiltonian \hat{H}_{AE} , it is very numerically efficient to compute the virtual photon exchange phase gate fidelity. Fig. 4.12 shows the phase gate fidelity as a function of detuning Δ and evolution time for a Lorentzian resonance compared to a Fano resonance following the phase conditions given above. From this example, it is clear that the Fano resonance dramatically reduces the detuning required to achieve a high phase gate fidelity. This reduces the gate time and increases the fidelity as a consequence. For a reasonably-optimistic cooperativity of C = 1000, the maximum fidelity for the Lorentzian case is just $\mathcal{F}_{gate} = 0.83$. However, two modes each with C = 500 in this case can allow for up to $\mathcal{F}_{gate} = 0.96$ while simultaneously reducing the gate time by more than a factor of 6.

Let us now explore the maximum fidelity scaling as a function of cavity cooperativity. To that end, I will assume that the dipole-QNM coupling strengths are equal in magnitude $|\tilde{g}_{i1}| = |\tilde{g}_{i2}|$ so that $R = R_1 = R_2$. Then the fidelity is given similar to Eq. (4.32) by

$$\mathcal{F}_{\text{gate}} = \frac{1}{4} \left[1 + e^{-T_{\text{gate}}(R+\gamma)/2} \left| \sin\left(\frac{T_{\text{gate}}\sqrt{\lambda_{12}\lambda_{21}}}{2}\right) \right|^2 \right]^2.$$
(4.40)

In the regime where the modes do not overlap much in frequency $(|S_{12}| \ll 1 \text{ implying } \Delta_c \gg \kappa_i \text{ and/or } \kappa_1 \gg \kappa_2)$ and the dipoles are far detuned relative to each mode $\Delta_i \gg \kappa_i$ we have $\lambda_{12} \simeq \lambda_{21} \simeq \lambda$. Hence, by choosing $T_{\text{gate}} = \pi/|\lambda|$ along with satisfaction of the phase conditions



Figure 4.13: Subradiant virtual photon exchange phase gate fidelity in a perfect Fano dip. (a) The phase gate fidelity for $T_{\text{gate}} = \pi/|\lambda|$ as a function of the dipole detuning Δ_2 from the narrow cavity mode with a FWHM of κ_2 and the splitting between the two cavity modes $\Delta_c = \Delta_2 - \Delta_1$. The vertical dashed white lines indicate the ideal detunings for each individual mode given by the condition $\kappa_i \sqrt{C_i}/2$. The dashed blue line indicates the Fano dip condition $\Delta_1/(\kappa_1\sqrt{C_1}) = \Delta_2/(\kappa_2\sqrt{C_2})$. (b) Cross-section (purple curve) along the horizontal line from panel (a) superimposed with the fidelity expected from each individual mode (red and blue curves), which are given by the extreme cases of Δ_c in panel (a). The horizontal lines indicate the maximum fidelity for each individual mode alone. Parameters used: $\kappa_1 = 100\kappa_2$, $C_1 = 1000$, $C_2 = 2000$, s = 1, $\tilde{g}_{i1} = \tilde{g}_{i2} > 0$, and $\phi_s = \pi/2$.

mentioned previously, we can find a significant Fano dip suppression of R that increases the fidelity compared to the expected enhancement from each individual mode alone (see Fig. 4.13).

Using the same approach as with the single-mode case, we can minimize $T_{\text{gate}}(R + \gamma)$ to solve for the ideal detuning conditions that achieve a good Fano dip enhancement of the fidelity. By making the assumption that $\kappa_2 \ll \kappa_1$ and that $C_i = 4\tilde{g}_i^2/\kappa_i\gamma \gg 1$ in the bad-cavity regime for both modes, we can write this term as

$$T_{\text{gate}}(R+\gamma) = 4\pi \left(1 + \frac{C_1}{1+4r_1^2} + \frac{C_2}{1+4r_2^2} - \frac{4sr_2\sqrt{C_1C_2}}{(1+4r_2^2)\sqrt{1+4r_1^2}} \right) \left(\frac{C_1}{r_1} + \frac{C_2}{r_2}\right)^{-1}, \quad (4.41)$$

where $r_1 = \Delta_1/\kappa_1$ and $r_2 = \Delta_2/\kappa_2$. By observation, we can identify that the Fano dip condition is given when both modes satisfy equal single-mode detuning conditions so that $\Delta_2/(\kappa_2\sqrt{C_2}) = \Delta_1/(\kappa_1\sqrt{C_1})$. Under this condition, which corresponds to the dashed blue line in Fig. 4.13, I minimize Eq. (4.41) for s = 1 to find that $r_1/\sqrt{C_1} = r_2/\sqrt{C_2} \simeq (1/2)(3/C_2)^{1/4}$ maximizes the fidelity. Substituting this solution into the fidelity, under the same ideal conditions, gives our main result

$$\max(\mathcal{F}_{gate}) \simeq \frac{1}{4} \left(1 + e^{-4\pi/\left(3^{3/4} \left(\sqrt{C_1} + \sqrt{C_2}\right) C_2^{1/4}\right)} \right)^2 \simeq 1 - \frac{4\pi}{3^{3/4} \left(\sqrt{C_1} + \sqrt{C_2}\right) C_2^{1/4}}.$$
 (4.42)

Here we can identify that the term $(\sqrt{C_1} + \sqrt{C_2})^{-1}$ in the errors arises from two cavity modes that are split in frequency (two independent single-mode cases). The extra factor $C_2^{-1/4}$ increasing the fidelity is directly due to operating in the Fano dip near the narrow mode.

As a consequence of the combined action of the two modes plus the Fano dip, the ideal gate time to maximize the fidelity is reduced compared to a gate using a single Lorentzian mode. Consider $T_{\text{gate}} = \pi/\lambda$ where $|S_{12}| \ll 1$, $\tilde{g}_i = |\tilde{g}_{i1}| = |\tilde{g}_{i2}|$ and where the ideal phase conditions are met. This gives $T_{\text{gate}} \simeq \pi (\tilde{g}_1^2/\Delta_1 + \tilde{g}_2^2/\Delta_2)^{-1}$. Then, in the far adiabatic regime with the ideal detuning conditions $r_1/\sqrt{C_1} = r_2/\sqrt{C_2} \simeq (1/2)(3/C_2)^{1/4}$, the ideal gate time becomes $T_{\text{gate}} = (2\pi/\gamma)(3/C_2)^{1/4} (\sqrt{C_1} + \sqrt{C_2})^{-1}$. Hence, compared to a single mode where the ideal gate time is $2\pi/\gamma\sqrt{C}$, the Fano enhancement can allow for a gate that is multiple times faster primarily due to the additional $C_2^{-1/4}$ scaling from operating in the Fano dip.

For $C = C_1 = C_2$, the maximum fidelity for the subradiant virtual photon exchange scheme reduces to $1 - 2\pi/(3C)^{3/4}$. This shows that the Fano enhancement gives a significant improvement to the cooperativity scaling compared to the $1 - 2\pi/\sqrt{C}$ scaling for the single mode case. To be fair, we need to recognize that if the two modes were resonant, we would expect the cooperativity of each cavity mode to add giving a scaling of $1 - 2\pi/\sqrt{C_1 + C_2}$. Hence, we should consider the single mode case using C compared to a two-mode case with C/2. Even then, the minimum cavity cooperativity needed to possibly reach a fidelity of 0.9 (0.99) for a single mode is 3400 (390000) and for two modes is just 150 (3600). In addition, we can see that the fidelity for the two-mode case benefits more from an increase in C_2 than from an equivalent increase in C_1 .

To summarize, the ideal conditions to achieve maximum fidelity are (1) constructive quantum



Figure 4.14: Susceptibility to deviations from the ideal phase conditions for the subradiant virtual photon exchange phase gate fidelity. (a) The gate fidelity dependence on the phase and magnitude of the mode overlap parameter S_{12} . The white color is set to indicate the fidelity value for no Fano dip enhancement (s = 0). The dashed white contour indicates the boundary where the fidelity exceeds what is possible using a single mode with twice the cavity cooperativity. In this case, I have set $\phi_{ij}^g = 0$ so that $\phi_s \simeq \pi/2$ indicates the maximum fidelity (horizontal black dashed line). (b) The susceptibility of the fidelity for deviations from the ideal dipole-cavity mode coupling phases. Here, I have set $\phi_s = \pi/2$ and s = 1. Parameters used: $\kappa_1 = 100\kappa_2$, $C = C_1 = C_2 = 2000$, and $2\Delta_i/\kappa_i = (3C)^{1/4}$.

interference of the dipole-dipole interaction $\phi_{11}^g - \phi_{21}^g = \phi_{12}^g - \phi_{22}^g$, (2) Fano dip arising on the exterior of the two modes $\phi_s + \phi_{1k}^g - \phi_{2k}^g \simeq \pi/2$, (3) maximum allowed mode overlap s = 1, and (4) Fano dip detuning condition $2\Delta_i/(\kappa_i\sqrt{C_i}) = (3/C_2)^{1/4}$. Knowing these conditions, we can now explore how the fidelity is degraded when the phase conditions are not perfectly satisfied. Fig. 4.14 shows how the fidelity depends on the exact values of s, ϕ_s , and the dipole-cavity coupling phases. From these plots, it is clear that a Fano resonance allows for a significant improvement over what is possible using a Lorentzian mode without requiring extreme precision for the ideal conditions. For $C = C_1 = C_2 = 2000$, there can be an improvement over a single mode with cooperativity 2C when s > 0.75. For $s \simeq 1$, the phases ϕ_s and $\phi_{i1}^g - \phi_{i2}^g$ can deviate by up to $\sim \pi/4$ from their ideal values before the Fano resonance is no better than a Lorentzian. This implies that an enhancement could still be possible even with some fabrication imperfections, which would be promising for increasing the yield of high-quality devices.



Figure 4.15: The effect of emitter pure dephasing on the subradiant photon exchange controlled phase gate fidelity using a Fano dip. (a) The phase gate fidelity as a function of detuning from the narrow mode resonance Δ_2 with the broad mode detuning set using the Fano dip condition $\Delta_1/(\kappa_1\sqrt{C_1}) = \Delta_2/(\kappa_2\sqrt{C_2})$ corresponding to the dashed blue line in Fig. 4.13 (a). The black points are numerically exact values computed from simulating the full master equation with perfect state preparation and retrieval pulses. The orange and cyan colored curves show the full analytic solution obtained from applying adiabatic elimination to the effective Hamiltonian with no pure dephasing (orange curve) and using the effective cooperativity $C_{\text{eff},i} = C_i(1 + (21/16)\gamma^*/\gamma)^{-1}$ for a pure dephasing of $\gamma^* = 10\gamma$ (cyan curve). The black dashed lines indicate the solution Eq. (4.40) assuming $T_{\text{gate}}\sqrt{\lambda_{12}\lambda_{21}} = \pi$ along with an effective cavity cooperativity to capture pure dephasing. The maximum fidelity occurs at the ideal detuning condition $2\Delta_2 = \kappa_2(3C_2)^{1/4}$ with a value given by Eq. (4.42). (b) The phase gate fidelity at the ideal detuning condition as a function of the emitter pure dephasing ratio γ^*/γ for four different sets of cavity cooperativity. The points, colored curves, and the dashed curves represent the same approaches as in panel (a). Parameters used: s = 1, $\phi_s = \pi/2$, and $\phi_{ij}^g = 0$. For panel (a): $C_1 = C_2/2 = 1000$, corresponding to the top curve in panel (b).

Let us now consider emitter pure dephasing. Recall that both the simple and subradiant virtual photon exchange schemes reduce to being described by the same effective dipole-dipole Hamiltonian in the ideal regime of operation, albeit with different rates. Also, pure dephasing primarily affects this subspace. Thus, so long as dephasing is small enough to not significantly dampen the dipole-cavity coupling, it is reasonable to expect that the same effective cooperativity derived for the simple scheme will apply for the two-mode case as well. To verify this postulate, we can compare Eq. (4.40) to the numerically exact solution from the master equation. Indeed, Fig. 4.15 shows that $C_{\text{eff},i} = C_i (1 + (21/16)(\gamma^*/\gamma))^{-1}$ again provides a very good estimate of the fidelity when there is pure dephasing even if $C_1 \neq C_2$. It also illustrates the asymmetric dependence on the cavity cooperativity of each mode, showing a higher fidelity when $C_2 > C_1$ than $C_1 > C_2$ for the same values of C_1 and C_2 . However, the analytic solution does slightly overestimate the

fidelity when $C_2 \gg C_1$. The good correspondence between the numerical solution and the analytic solution also helps to verify all of the assumptions made in deriving the analytic solution.

Reducing the cavity cooperativity requirement undoubtedly helps any application of this type of phase gate. However, it is particularly interesting to consider how the Fano enhancement could potentially open up entirely new regimes of operation. For example, plasmonic cavities may allow for an (uninhibited) cavity cooperativity nearing $C \simeq 10^6$ [105, 166]. Even with a single mode, this would be enough to reach very high fidelity gates. But, at room temperature where optical defects may experience pure dephasing rates exceeding $\gamma^*/\gamma = 10^4$, the effective phase gate cavity cooperativity will be $C_{\text{eff}} \simeq 800$. For a single mode with $C = 10^6$, this limits the maximum gate fidelity to around 0.78. On the other hand, the Fano dip enhancement with $C = 10^6$ may still allow for a gate fidelity of up to 0.98 even with $\gamma^*/\gamma = 10^4$, which may be sufficient to perform cavity-mediated information processing at or near room temperature.

The results in this section are promising and warrant further investigation for both low and room temperature applications. However, inhibition of the cavity cooperativity due to dephasing is not the only issue that must be overcome for solid-state systems. As discussed in section 3.1, phonon non-Markovian effects often play a role in determining the ultimate fidelity possible for a realistic implementation of a solid-state device. In addition, it is necessary to excite one of the two defects much faster than the gate time. Luckily, since this scheme operates in the Fano dip where dissipation is suppressed, the excitation problem is much less severe as compared to a single-photon source. That said, for large $C = C_1 = C_2$, the ideal gate time $T_{\text{gate}} = 3^{1/4} \pi / \gamma C^{3/4}$ still requires excitation pulses orders of magnitude faster than the bare lifetime $1/\gamma$. When dephasing is included, the ideal gate time is reduced by an additional factor of $(C_{\text{eff}}/C)^{1/4}$. Therefore, when designing an implementation for a specific system, it is important to consider realistic limitations on the pulse used for state preparation and retrieval.

Chapter 5

Conclusion

5.1 Summary

Over the course of this thesis, I have presented the analysis of light-matter interaction models for basic components used to process quantum information with solid-state optical devices. These basic components ranged from single-photon sources to optically-mediated spin-spin entanglement.

In many cases, the analysis in this thesis relies on material from chapter 2, where I described a framework based on quantum trajectories to study the state of a quantum optical device when subject to a variety of imperfections. The approach can be used to analytically or numerically decompose a wide variety of emitter Markovian master equations into a set of propagation superoperators conditioned on the emitted photonic state. These operators can then be applied to reconstruct the quantum photonic state of the emission. It can also be used to compute imperfect measurements of the emitter, or joint measurements of multiple emitters. In addition, it is connected to the method of non-Hermitian Hamiltonians, which I used to analyze local entangling gates.

In section 3.1, I discussed how ultra-small mode volume cavities, such as cavities containing plasmonic material, may allow for indistinguishable single-photon sources that operate at room temperature. In particular, we derived an expression for single-photon indistinguishability in the

critical cavity coupling regime that is valid for any magnitude of emitter decay rate. With this expression, we studied the optimal regime of operation for when the emitter quenching rate is proportional to the squared magnitude of the cavity coupling rate. Motivated by our results, we proposed that a Fabry-Pérot-plasmonic hybrid cavity combined with a negatively-charged silicon-vacancy center is a good candidate system to achieve a room temperature indistinguishable single-photon source.

Section 3.2 presented an analysis of how multi-photon emission affects the indistinguishability of single-photon sources measured using Hong-Ou-Mandel interference. The main conclusion from this analysis is that the single-photon indistinguishability can be estimated by applying a simple correction using the second-order intensity correlation, and that this correction is accurate for both a general separable noise model and a driven two-level system when the probability for multi-photon emission is small. This section concluded by generalizing the concept of indistinguishability measurements to a self-homodyne measurement setup, which can be used to measure other quantum properties of photonic states such as one- and two-photon number coherence.

In section 3.3, I applied the photon number decomposition to analyze the temporal density function of emitted photonic states from a purely dephased emitter. This allowed for a much more detailed perspective on the results obtained in section 3.2 on Hong-Ou-Mandel interference. I also showed how to use the photon number decomposition in the context of a self-homodyne measurement to probe the photon number coherence in time between specific photon-number subspaces. This section concluded with an exploration on how a two-level system can be manipulated to deterministically generate photon number entangled states encoded in discretized time bins. This included an analysis of the degrading effect of pure dephasing on the entanglement concurrence for photon number Bell states encoded in time.

In section 4.1, I built on the concept of self-homodyne measurements and applied the photon number decomposition to analyze joint photon counting measurements of two remote opticallyactive defects containing spin qubits. I presented an analysis of three popular photon-heralded entanglement generation protocols based on (1) spin-photon number entanglement, (2) spin-time bin entanglement, and (3) spin-polarization entanglement. These three protocols were found to have different upper bounds on the entanglement generation fidelity. Interestingly, the maximum fidelity values of the three protocols were found to satisfy a specific ordering and were also all bounded by the mean wavepacket overlap of photons from each emitter. The section concluded with a detailed comparison of the three schemes while taking into account photon losses, detector noise, and detector number resolution limitations in addition to optical and spin dephasing.

The discussion of upper bounds on fidelity was continued in section 4.2 in the context of local entanglement generation between spin qubits in optically-active defects mediated by an optical cavity. In this section, I explored how a cavity-emitter system operating in the bad-cavity regime can allow for high-fidelity entanglement even in the presence of pure emitter dephasing. This was emphasized by deriving a simple relation to estimate the entanglement fidelity from an effective cavity cooperativity. This analysis was then extended to cavities that are described by two interfering quasi-normal modes, which is relevant for cavities containing dissipative material such as plasmonics. This extension revealed that the subradiant mode of two interfering optical modes could mediate an adiabatic interaction between two emitters. By placing the two emitters in a Fano dip feature of the two-mode cavity, the maximum gate fidelity was shown to be dramatically higher than for a single mode cavity, even if the emitters experience severe dephasing.

5.2 Outlook

This thesis focused heavily on phenomenological pure dephasing as a non-trivial example requiring a master equation model and the superoperator formalism presented in chapter 2. I would like to emphasize that the methods I have demonstrated can be applied to a wide variety of Markovian master equations, including those with time-dependent parameters. For example, phonon interactions can cause an increased dephasing rate when a solid-state defect is driven by optical pulses. A rough approximation for this phenomenon can be made by taking the pure dephasing rate to be proportional to the square of the driving Rabi frequency [85]. Multi-level systems with many different dissipative and dephasing channels can also be easily handled in a phenomenological way, allowing for straightforward coarse estimates of figures of merit for complicated quantum information processing protocols. Moving beyond this phenomenological approach, weak-coupling and polaron Markovian master equations [85, 192, 249] may provide a natural extension to account for more detailed electron-phonon interactions for quantum dots. Alternatively, a quantum optical master equation with discrete damped phonon resonances may be more relevant to model atomic defects in diamond [250]. Lastly, there may be other interesting applications to analyze that involve the quasi-normal mode Markovian master equation model [103]. For these reasons, I hope that the methods and results presented in this thesis will benefit the study and development of a wide range of quantum devices based on light-matter interactions.

In many ways, the content of this thesis was motivated by a desire to understand and circumvent processes that may restrict solid-state optical devices to cryogenic temperatures. This original motivation is illustrated by the content of section 3.1 on the feasibility of room temperature indistinguishable single-photon sources. However, there are still many open questions regarding the feasibility of room-temperature solid-state devices. A promising direction to achieve this goal is to discover or engineer new optically-active solid-state defects that have a reduced electron-phonon coupling at room temperature. Current research in this direction suggests that defects in 2-dimensional materials, such as hexagonal boron-nitride, are promising candidates [251, 252]. These 2-dimensional materials may also be easily incorporated into small mode volume cavities or layered with plasmonic material to achieve the Purcell enhancements needed to overcome any remaining dephasing or emission inefficiency at room temperature. That said, the thermal limits to cavity devices operating at room temperature are still being explored [253].

An additional approach that could be applied to improve the quality of photonic state sources is to post process the emission. This approach is often only useful if the source quality is already quite high, but it could provide a last push needed to reach quality thresholds required for quantum information processing. The methods and results presented in sections 3.2 and 3.3 provide a detailed picture of the time dynamics and quantum properties of imperfect photonic states. This can be used to optimize post-processing methods such as temporal filtering [254] to minimize the effect of pure dephasing and unwanted multi-photon noise on the desired state. The content of section 3.3 also provides a basis to study the effect of emitter imperfections on more complicated processes, such as photonic graph state generation for all-optical quantum repeaters [255].

Imperfect emission also affects the quality of entanglement generation between nodes of quantum networks based on solid-state defects. A natural continuation of the content presented in section 4.1 would be to include realistic excitation pulses and study the effects of unwanted multi-photon emission on the entanglement generation fidelity. One of the imperfections that we did explore in section 4.1 was the relative detuning between remote emitters. Reducing this source of error for solid-state defects is quite difficult because optically-active defects are often quite spectrally distinct due to microscopic differences in their respective solid-state environments. If the emitters are close in frequency, it may be possible to use a Stark shift or strain tuning to bring two emitters into resonance [10, 151, 218]. For any remaining detuning, fast time-tagging measurements may still allow for successful spin-spin entanglement generation using spectrally distinct photons [256]. These time-tagging measurements could also be modeled using a photon number decomposition, which could extend the results in section 4.1 to compensate for finite emitter detuning.

The phase gate schemes presented in section 4.2 also require two emitters to be resonant. However, even if the emitters are brought fully into resonance, it may be difficult to individually excite just one of these two dipoles. This is a particularly large problem if the cavity has a very small mode volume, which would potentially require the defects to be closer than the diffraction limit for spatial addressability. As we studied in Ref. [vii] for the simple virtual photon scheme, this problem could be overcome by using a Raman-assisted approach where weak optical driving can allow for a virtual photon exchange interaction without needing the dipoles to be resonant. This approach could also be applied to the subradiant virtual photon exchange scheme in a similar way. Another modification that can improve these phase gate schemes is post-selection. As we studied in Ref. [viii], it is possible to monitor emission from the cavity mode(s) and reject any gate attempts where one or more photon is detected. As a result, the phase gate fidelity could be improved at the
cost of some efficiency. The photon number decomposition used in this thesis also provides the framework to analyze this type of post-selection as a function of the monitoring efficiency.

The fidelity limits on a subradiant virtual photon phase gate using a Fano resonance studied in section 4.2.3 suggest a significant improvement in the cavity cooperativity scaling versus a single-mode scheme. However, it remains to explore whether the derived ideal parameter regime for the quasi-normal mode model can be physically implemented. Fortunately, one of the strengths of this model is that the parameters of the master equation can be efficiently computed from classical electromagnetic simulations [103]. Guided by the phase, detuning, and mode linewidth conditions presented in section 4.2.3, it may be possible to engineer and optimize a cavity design that can perform fast and high-fidelity optically-mediated gates using a Fano dip. A promising approach is to combine a high-*Q* Fabry-Pérot cavity with a low-*Q* plasmonic resonator. Such hybrid cavities can give good optical Fano resonance spectra [166, 257]. That said, one challenge will be to engineer a Fano resonance that has a Fano dip arising on the exterior of the two main interfering modes. This type of Fano resonance can occur for plasmonic hybrid cavities [258] but it seems to be rare, with most studies showing examples with a dip between modes [103, 166, 257, 259]. Performing the gate fidelity analysis for more than two modes may also modify the phase conditions for a high-fidelity phase gate and allow for more flexibility to design an optimal cavity.

The development of efficient and robust quantum optical devices stands to make a tremendous impact on the technological progress of humanity. In particular, optically-active solid-state systems will undoubtedly continue pushing the frontier of the ongoing quantum revolution. These quantum systems may also provide the perfect platform to complement and enhance current classical computation and communication technology. This, in turn, could allow for a multitude of unprecedented advances that may fundamentally shape the future of our world. From clean energy production and agricultural practices to medical procedures and cryptography, the potential breadth of quantum technological applications is truly astonishing. Moreover, alongside all of these practical benefits, quantum physics will continue to allow humanity to ask and answer profound questions about the nature of our universe.

References

- M. D. Al-Amri, M. El-Gomati, and M. Suhail Zubairy, *Optics in Our Time* (Springer Nature, 2016).
- [2] A. W. Thackray, "The origin of Dalton's chemical atomic theory: Daltonian doubts resolved", Isis 57, 35 (1966).
- [3] J. C. Maxwell, "VIII. A dynamical theory of the electromagnetic field", Philosophical Transactions of the Royal Society of London 155, 459 (1865).
- [4] I. Falconer, "JJ Thomson and the discovery of the electron", Physics Education 32, 226 (1997).
- [5] A. Pais, "Einstein and the quantum theory", Reviews of Modern Physics 51, 863 (1979).
- [6] A. Einstein, B. Podolsky, and N. Rosen, "Can quantum-mechanical description of physical reality be considered complete?", Physical Review 47, 777 (1935).
- [7] J. S. Bell, "On the Einstein Podolsky Rosen paradox", Physics Physique Fizika 1, 195 (1964).
- [8] S. J. Freedman and J. F. Clauser, "Experimental test of local hidden-variable theories", Physical Review Letters 28, 938 (1972).
- [9] J.-Å. Larsson, "Loopholes in Bell inequality tests of local realism", Journal of Physics A: Mathematical and Theoretical 47, 424003 (2014).

- [10] B. Hensen, H. Bernien, A. E. Dréau, A. Reiserer, N. Kalb, M. S. Blok, J. Ruitenberg, R. F. Vermeulen, R. N. Schouten, C. Abellán, et al., "Loophole-free Bell inequality violation using electron spins separated by 1.3 kilometres", Nature 526, 682 (2015).
- [11] M. Giustina, M. A. Versteegh, S. Wengerowsky, J. Handsteiner, A. Hochrainer, K. Phelan,
 F. Steinlechner, J. Kofler, J.-Å. Larsson, C. Abellán, et al., "Significant-loophole-free test of Bell's theorem with entangled photons", Physical Review Letters 115, 250401 (2015).
- [12] L. K. Shalm, E. Meyer-Scott, B. G. Christensen, P. Bierhorst, M. A. Wayne, M. J. Stevens, T. Gerrits, S. Glancy, D. R. Hamel, M. S. Allman, et al., "Strong loophole-free test of local realism", Physical Review Letters 115, 250402 (2015).
- [13] A. Aspect, J. Dalibard, and G. Roger, "Experimental test of Bell's inequalities using timevarying analyzers", Physical Review Letters 49, 1804 (1982).
- [14] W. Tittel, J. Brendel, H. Zbinden, and N. Gisin, "Violation of Bell inequalities by photons more than 10 km apart", Physical Review Letters 81, 3563 (1998).
- [15] G. Weihs, T. Jennewein, C. Simon, H. Weinfurter, and A. Zeilinger, "Violation of Bell's inequality under strict Einstein locality conditions", Physical Review Letters 81, 5039 (1998).
- [16] P. G. Kwiat, K. Mattle, H. Weinfurter, A. Zeilinger, A. V. Sergienko, and Y. Shih, "New high-intensity source of polarization-entangled photon pairs", Physical Review Letters 75, 4337 (1995).
- [17] P. Benioff, "The computer as a physical system: a microscopic quantum mechanical Hamiltonian model of computers as represented by Turing machines", Journal of Statistical Physics 22, 563 (1980).
- [18] R. P. Feynman, "Simulating physics with computers", International Journal of Theoretical Physics 21, 467 (1982).

- [19] F. Arute, K. Arya, R. Babbush, D. Bacon, J. C. Bardin, R. Barends, R. Biswas, S. Boixo,
 F. G. S. L. Brandao, D. A. Buell, et al., "Quantum supremacy using a programmable superconducting processor", Nature 574, 505 (2019).
- [20] H.-S. Zhong, H. Wang, Y.-H. Deng, M.-C. Chen, L.-C. Peng, Y.-H. Luo, J. Qin, D. Wu, X. Ding, Y. Hu, et al., "Quantum computational advantage using photons", Science 370, 1460 (2020).
- [21] N. Gisin, G. Ribordy, W. Tittel, and H. Zbinden, "Quantum cryptography", Reviews of Modern Physics 74, 145 (2002).
- [22] R. Valivarthi, M. G. Puigibert, Q. Zhou, G. H. Aguilar, V. B. Verma, F. Marsili, M. D. Shaw,
 S. W. Nam, D. Oblak, and W. Tittel, "Quantum teleportation across a metropolitan fibre network", Nature Photonics 10, 676 (2016).
- [23] M. Aspelmeyer, T. Jennewein, M. Pfennigbauer, W. R. Leeb, and A. Zeilinger, "Longdistance quantum communication with entangled photons using satellites", IEEE Journal of Selected Topics in Quantum Electronics 9, 1541 (2003).
- [24] K. Boone, J.-P. Bourgoin, E. Meyer-Scott, K. Heshami, T. Jennewein, and C. Simon, "Entanglement over global distances via quantum repeaters with satellite links", Physical Review A 91, 052325 (2015).
- [25] S.-K. Liao, W.-Q. Cai, W.-Y. Liu, L. Zhang, Y. Li, J.-G. Ren, J. Yin, Q. Shen, Y. Cao, Z.-P. Li, et al., "Satellite-to-ground quantum key distribution", Nature 549, 43 (2017).
- [26] J. Yin, Y. Cao, Y.-H. Li, S.-K. Liao, L. Zhang, J.-G. Ren, W.-Q. Cai, W.-Y. Liu, B. Li, H. Dai, et al., "Satellite-based entanglement distribution over 1200 kilometers", Science 356, 1140 (2017).
- [27] H.-J. Briegel, W. Dür, J. I. Cirac, and P. Zoller, "Quantum repeaters: the role of imperfect local operations in quantum communication", Physical Review Letters 81, 5932 (1998).

- [28] S.-K. Liao, W.-Q. Cai, J. Handsteiner, B. Liu, J. Yin, L. Zhang, D. Rauch, M. Fink, J.-G. Ren, W.-Y. Liu, et al., "Satellite-relayed intercontinental quantum network", Physical Review Letters 120, 030501 (2018).
- [29] R. Valivarthi, S. I. Davis, C. Peña, S. Xie, N. Lauk, L. Narváez, J. P. Allmaras, A. D. Beyer, Y. Gim, M. Hussein, et al., "Teleportation systems toward a quantum internet", PRX Quantum 1, 020317 (2020).
- [30] M. Pompili, S. L. N. Hermans, S. Baier, H. K. C. Beukers, P. C. Humphreys, R. N. Schouten, R. F. L. Vermeulen, M. J. Tiggelman, L. d. S. Martins, B. Dirkse, S. Wehner, and R. Hanson, "Realization of a multi-node quantum network of remote solid-state qubits", arXiv:2102.04471 (2021).
- [31] H. J. Kimble, "The quantum internet", Nature 453, 1023 (2008).
- [32] C. Simon, "Towards a global quantum network", Nature Photonics 11, 678 (2017).
- [33] S. Wehner, D. Elkouss, and R. Hanson, "Quantum internet: a vision for the road ahead", Science 362, eaam9288 (2018).
- [34] D. Rideout, T. Jennewein, G. Amelino-Camelia, T. F. Demarie, B. L. Higgins, A. Kempf, A. Kent, R. Laflamme, X. Ma, and R. B. Mann, "Fundamental quantum optics experiments conceivable with satellites—reaching relativistic distances and velocities", Classical and Quantum Gravity 29, 224011 (2012).
- [35] P. W. Shor, "Polynomial-time algorithms for prime factorization and discrete logarithms on a quantum computer", SIAM Review 41, 303 (1999).
- [36] I. Kassal, J. D. Whitfield, A. Perdomo-Ortiz, M.-H. Yung, and A. Aspuru-Guzik, "Simulating chemistry using quantum computers", Annual Review of Physical Chemistry 62, 185 (2011).
- [37] M. Reiher, N. Wiebe, K. M. Svore, D. Wecker, and M. Troyer, "Elucidating reaction mechanisms on quantum computers", Proceedings of the National Academy of Sciences 114, 7555 (2017).

- [38] J. Preskill, "Quantum computing in the NISQ era and beyond", Quantum 2, 79 (2018).
- [39] J. Biamonte, P. Wittek, N. Pancotti, P. Rebentrost, N. Wiebe, and S. Lloyd, "Quantum machine learning", Nature 549, 195 (2017).
- [40] F. Gaitan, "Finding flows of a Navier–Stokes fluid through quantum computing", npj Quantum Information 6, 1 (2020).
- [41] R. Orús, S. Mugel, and E. Lizaso, "Quantum computing for finance: overview and prospects", Reviews in Physics 4, 100028 (2019).
- [42] T. Stollenwerk, B. O'Gorman, D. Venturelli, S. Mandrà, O. Rodionova, H. Ng, B. Sridhar,
 E. G. Rieffel, and R. Biswas, "Quantum annealing applied to de-conflicting optimal trajectories for air traffic management", IEEE Transactions on Intelligent Transportation Systems 21, 285 (2019).
- [43] S. Barz, E. Kashefi, A. Broadbent, J. F. Fitzsimons, A. Zeilinger, and P. Walther, "Demonstration of blind quantum computing", Science 335, 303 (2012).
- [44] V. Giovannetti, S. Lloyd, and L. Maccone, "Quantum metrology", Physical Review Letters 96, 010401 (2006).
- [45] V. Giovannetti, S. Lloyd, and L. Maccone, "Advances in quantum metrology", Nature Photonics 5, 222 (2011).
- [46] H. J. Mamin, M. Kim, M. H. Sherwood, C. T. Rettner, K. Ohno, D. D. Awschalom, and D. Rugar, "Nanoscale nuclear magnetic resonance with a nitrogen-vacancy spin sensor", Science 339, 557 (2013).
- [47] Y. Wang, Y. Zhang, and V. O. Lorenz, "Superresolution in interferometric imaging of strong thermal sources", arXiv:2012.14026 (2020).
- [48] C. M. Caves, "Quantum-mechanical noise in an interferometer", Physical Review D 23, 1693 (1981).

- [49] J. Aasi, J. Abadie, B. P. Abbott, R. Abbott, T. D. Abbott, M. R. Abernathy, C. Adams, T. Adams, P. Addesso, R. X. Adhikari, et al., "Enhanced sensitivity of the LIGO gravitational wave detector by using squeezed states of light", Nature Photonics 7, 613 (2013).
- [50] L. McCuller, C. Whittle, D. Ganapathy, K. Komori, M. Tse, A. Fernandez-Galiana, L. Barsotti, P. Fritschel, M. MacInnis, F. Matichard, et al., "Frequency-dependent squeezing for Advanced LIGO", Physical Review Letters 124, 171102 (2020).
- [51] M. Genovese, "Real applications of quantum imaging", Journal of Optics 18, 073002 (2016).
- [52] P. Komar, E. M. Kessler, M. Bishof, L. Jiang, A. S. Sørensen, J. Ye, and M. D. Lukin, "A quantum network of clocks", Nature Physics 10, 582 (2014).
- [53] D. Gottesman, T. Jennewein, and S. Croke, "Longer-baseline telescopes using quantum repeaters", Physical Review Letters 109, 070503 (2012).
- [54] P. Stankus, A. Nomerotski, A. Slosar, and S. Vintskevich, "Two-photon amplitude interferometry for precision astrometry", arXiv:2010.09100 (2020).
- [55] T. E. Northup and R. Blatt, "Quantum information transfer using photons", Nature Photonics 8, 356 (2014).
- [56] D. D. Awschalom, R. Hanson, J. Wrachtrup, and B. B. Zhou, "Quantum technologies with optically interfaced solid-state spins", Nature Photonics **12**, 516 (2018).
- [57] M. Atatüre, D. Englund, N. Vamivakas, S.-Y. Lee, and J. Wrachtrup, "Material platforms for spin-based photonic quantum technologies", Nature Reviews Materials 3, 38 (2018).
- [58] H. Häffner, C. F. Roos, and R. Blatt, "Quantum computing with trapped ions", Physics Reports 469, 155 (2008).
- [59] G. Wendin, "Quantum information processing with superconducting circuits: a review", Reports on Progress in Physics 80, 106001 (2017).
- [60] U. Woggon, Optical Properties of Semiconductor Quantum Dots Tracts in Modern Physics, Vol. 136 (Springer, Berlin, 1997).

- [61] M. W. Doherty, N. B. Manson, P. Delaney, F. Jelezko, J. Wrachtrup, and L. C. L. Hollenberg,"The nitrogen-vacancy colour centre in diamond", Physics Reports 528, 1 (2013).
- [62] T. B. Pittman, B. C. Jacobs, and J. D. Franson, "Probabilistic quantum logic operations using polarizing beam splitters", Physical Review A 64, 062311 (2001).
- [63] J. L. O'Brien, "Optical quantum computing", Science **318**, 1567 (2007).
- [64] D. P. DiVincenzo, "The physical implementation of quantum computation", Fortschritte der Physik: Progress of Physics 48, 771 (2000).
- [65] A. J. Fisher, "Quantum computing in the solid state: the challenge of decoherence", Philosophical Transactions of the Royal Society of London. Series A: Mathematical, Physical and Engineering Sciences 361, 1441 (2003).
- [66] L. Chirolli and G. Burkard, "Decoherence in solid-state qubits", Advances in Physics 57, 225 (2008).
- [67] D. P. S. McCutcheon and A. Nazir, "Quantum dot Rabi rotations beyond the weak excitonphonon coupling regime", New Journal of Physics 12, 113042 (2010).
- [68] N. Somaschi, V. Giesz, L. De Santis, J. C. Loredo, M. P. Almeida, G. Hornecker, S. L. Portalupi, T. Grange, C. Antón, J. Demory, et al., "Near-optimal single-photon sources in the solid state", Nature Photonics 10, 340 (2016).
- [69] T. Lutz, L. Veissier, C. W. Thiel, R. L. Cone, P. E. Barclay, and W. Tittel, "Modification of phonon processes in nanostructured rare-earth-ion-doped crystals", Physical Review A 94, 013801 (2016).
- [70] G. Balasubramanian, P. Neumann, D. Twitchen, M. Markham, R. Kolesov, N. Mizuochi, J. Isoya, J. Achard, J. Beck, J. Tissler, V. Jacques, P. R. Hemmer, F. Jelezko, and J. Wrachtrup, "Ultralong spin coherence time in isotopically engineered diamond", Nature Materials 8, 383 (2009).

- [71] L. Viola, E. Knill, and S. Lloyd, "Dynamical decoupling of open quantum systems", Physical Review Letters 82, 2417 (1999).
- [72] J. Iles-Smith, D. P. S. McCutcheon, A. Nazir, and J. Mørk, "Phonon scattering inhibits simultaneous near-unity efficiency and indistinguishability in semiconductor single-photon sources", Nature Photonics 11, 521 (2017).
- [73] W. Dür and H. J. Briegel, "Entanglement purification and quantum error correction", Reports on Progress in Physics 70, 1381 (2007).
- [74] W Dür, H.-J. Briegel, J. I. Cirac, and P Zoller, "Quantum repeaters based on entanglement purification", Physical Review A 59, 169 (1999).
- [75] H.-P. Breuer and F. Petruccione, *The Theory of Open Quantum Systems* (Oxford University Press, New York, 2002).
- [76] S. Kochen and E. P. Specker, "The problem of hidden variables in quantum mechanics", in *The Logico-Algebraic Approach to Quantum Mechanics* (Springer, 1975), p. 293.
- [77] M. F. Pusey, J. Barrett, and T. Rudolph, "On the reality of the quantum state", Nature Physics 8, 475 (2012).
- [78] A. Pechen, N. Il'in, F. Shuang, and H. Rabitz, "Quantum control by von Neumann measurements", Physical Review A 74, 052102 (2006).
- [79] K. Fischer, "Derivation of the quantum-optical master equation based on coarse-graining of time", Journal of Physics Communications 2, 091001 (2018).
- [80] R. Alicki, "Invitation to quantum dynamical semigroups", in *Dynamics of Dissipation* (Springer, 2002), p. 239.
- [81] D. Manzano, "A short introduction to the Lindblad master equation", AIP Advances 10, 025106 (2020).
- [82] D. J. Griffiths, Introduction to Electrodynamics (Prentice Hall, Upper Saddle River, 1999).

- [83] M. O. Scully and M. S. Zubairy, *Quantum Optics* (Cambridge University Press, New York, 1997).
- [84] R. Manson, K. Roy-Choudhury, and S. Hughes, "Polaron master equation theory of pulsedriven phonon-assisted population inversion and single-photon emission from quantum-dot excitons", Physical Review B 93, 155423 (2016).
- [85] A. Nazir and D. P. S. McCutcheon, "Modelling exciton-phonon interactions in optically driven quantum dots", Journal of Physics: Condensed Matter 28, 103002 (2016).
- [86] T. Grange, G. Hornecker, D. Hunger, J.-P. Poizat, J.-M. Gérard, P. Senellart, and A. Auffèves, "Cavity-funneled generation of indistinguishable single photons from strongly dissipative quantum emitters", Physical Review Letters 114, 193601 (2015).
- [87] V. Mlynárik, "Introduction to nuclear magnetic resonance", Analytical Biochemistry 529, 4 (2017).
- [88] A. Le Boité, "Theoretical methods for ultrastrong light–matter interactions", Advanced Quantum Technologies 3, 1900140 (2020).
- [89] A. Reiserer and G. Rempe, "Cavity-based quantum networks with single atoms and optical photons", Reviews of Modern Physics 87, 1379 (2015).
- [90] C. W. Gardiner and M. J. Collett, "Input and output in damped quantum systems: quantum stochastic differential equations and the master equation", Physical Review A 31, 3761 (1985).
- [91] H. Carmichael, An Open Systems Approach to Quantum Optics (Springer, Berlin, 1993).
- [92] A. Kiraz, M. Atatüre, and A Imamoğlu, "Quantum-dot single-photon sources: prospects for applications in linear optics quantum-information processing", Physical Review A 69, 032305 (2004).
- [93] K. A. Fischer, K. Müller, K. G. Lagoudakis, and J. Vučković, "Dynamical modeling of pulsed two-photon interference", New Journal of Physics 18, 113053 (2016).

- [94] A. Auffeves, J.-M. Gérard, and J.-P. Poizat, "Pure emitter dephasing: a resource for advanced solid-state single-photon sources", Physical Review A 79, 053838 (2009).
- [95] E. M. Purcell, "Spontaneous emission probabilities at radio frequencies", Physical Review 69, 681 (1946).
- [96] A. Auffèves, D. Gerace, J.-M. Gérard, M. F. Santos, L. C. Andreani, and J.-P. Poizat,
 "Controlling the dynamics of a coupled atom-cavity system by pure dephasing", Physical Review B 81, 245419 (2010).
- [97] K. Roy-Choudhury and S. Hughes, "Spontaneous emission from a quantum dot in a structured photonic reservoir: phonon-mediated breakdown of Fermi's golden rule", Optica 2, 434–437 (2015).
- [98] A. F. Kockum, A. Miranowicz, S. De Liberato, S. Savasta, and F. Nori, "Ultrastrong coupling between light and matter", Nature Reviews Physics 1, 19 (2019).
- [99] P. Forn-Díaz, L. Lamata, E. Rico, J. Kono, and E. Solano, "Ultrastrong coupling regimes of light-matter interaction", Reviews of Modern Physics **91**, 025005 (2019).
- [100] In the children's story of "The Three Bears", Goldilocks wanders into the house of a bear family and eats porridge from three bowls sitting on the table. She finds that one bowl is too cold, another is too hot, but the third is just right and so she eats it all. A Goldilocks condition is one that strikes the right balance. The Goldilocks principle is referenced in many fields, including physics.
- [101] A. Auffèves-Garnier, C. Simon, J.-M. Gérard, and J.-P. Poizat, "Giant optical nonlinearity induced by a single two-level system interacting with a cavity in the Purcell regime", Physical Review A 75, 053823 (2007).
- [102] V. Dorier, S. Guérin, and H.-R. Jauslin, "Critical review of quantum plasmonic models for finite-size media", Nanophotonics 9, 3899 (2020).

- [103] S. Franke, S. Hughes, M. K. Dezfouli, P. T. Kristensen, K. Busch, A. Knorr, and M. Richter, "Quantization of quasinormal modes for open cavities and plasmonic cavity quantum electrodynamics", Physical Review Letters 122, 213901 (2019).
- [104] J. Ren, S. Franke, A. Knorr, M. Richter, and S. Hughes, "Near-field to far-field transformations of optical quasinormal modes and efficient calculation of quantized quasinormal modes for open cavities and plasmonic resonators", Physical Review B 101, 205402 (2020).
- [105] S. Hughes, S. Franke, C. Gustin, M. K. Dezfouli, A. Knorr, and M. Richter, "Theory and limits of on-demand single-photon sources using plasmonic resonators: a quantized quasinormal mode approach", ACS Photonics 6, 2168 (2019).
- [106] S. Franke, M. Richter, J. Ren, A. Knorr, and S. Hughes, "Quantized quasinormal-mode description of nonlinear cavity-QED effects from coupled resonators with a Fano-like resonance", Physical Review Research 2, 033456 (2020).
- [107] C.-K. Hong, Z.-Y. Ou, and L. Mandel, "Measurement of subpicosecond time intervals between two photons by interference", Physical Review Letters 59, 2044 (1987).
- [108] C. Gustin and S. Hughes, "Pulsed excitation dynamics in quantum-dot-cavity systems: Limits to optimizing the fidelity of on-demand single-photon sources", Physical Review B 98, 045309 (2018).
- [109] K. A. Fischer, R. Trivedi, and D. Lukin, "Particle emission from open quantum systems", Physical Review A 98, 023853 (2018).
- [110] R. Trivedi, K. A. Fischer, J. Vučković, and K. Müller, "Generation of non-classical light using semiconductor quantum dots", Advanced Quantum Technologies 3, e1900007 (2020).
- [111] R. Jozsa, "Fidelity for mixed quantum states", Journal of Modern Optics 41, 2315 (1994).
- [112] W. K. Wootters, "Entanglement of formation and concurrence.", Quantum Information and Computation 1, 27 (2001).

- [113] K. Bartkiewicz, B. Horst, K. Lemr, and A. Miranowicz, "Entanglement estimation from Bell inequality violation", Physical Review A 88, 052105 (2013).
- [114] J. F. Clauser, M. A. Horne, A. Shimony, and R. A. Holt, "Proposed experiment to test local hidden-variable theories", Physical Review Letters 23, 880 (1969).
- [115] G. Guarnieri, A. Smirne, and B. Vacchini, "Quantum regression theorem and non-Markovianity of quantum dynamics", Physical Review A **90**, 022110 (2014).
- [116] A. J. Daley, "Quantum trajectories and open many-body quantum systems", Advances in Physics 63, 77 (2014).
- [117] M. Sanz, E. Solano, and Í. L. Egusquiza, "Beyond adiabatic elimination: effective Hamiltonians and singular perturbation", in *Applications+ Practical Conceptualization+ Mathematics= fruitful Innovation* (Springer, 2016), p. 127.
- [118] P. Zoller, M. Marte, and D. F. Walls, "Quantum jumps in atomic systems", Physical Review A 35, 198 (1987).
- [119] H. M. Wiseman, "Quantum trajectories and feedback", PhD thesis (University of Queensland, 1994).
- [120] H. J. Carmichael, S. Singh, R. Vyas, and P. R. Rice, "Photoelectron waiting times and atomic state reduction in resonance fluorescence", Physical Review A 39, 1200 (1989).
- [121] D. E. Browne, M. B. Plenio, and S. F. Huelga, "Robust creation of entanglement between ions in spatially separate cavities", Physical Review Letters 91, 067901 (2003).
- [122] A. V. Dodonov, S. S. Mizrahi, and V. V. Dodonov, "Microscopic models of quantum-jump superoperators", Physical Review A 72, 023816 (2005).
- [123] S. D. Barrett and P. Kok, "Efficient high-fidelity quantum computation using matter qubits and linear optics", Physical Review A 71, 060310(R) (2005).
- [124] X. H. Zhang and H. U. Baranger, "Quantum interference and complex photon statistics in waveguide QED", Physical Review A 97, 023813 (2018).

- [125] X. H. Zhang and H. U. Baranger, "Heralded bell state of dissipative qubits using classical light in a waveguide", Physical Review Letters 122, 140502 (2019).
- [126] L. Hanschke, K. A. Fischer, S. Appel, D. Lukin, J. Wierzbowski, S. Sun, R. Trivedi, J. Vučković, J. J. Finley, and K. Müller, "Quantum dot single-photon sources with ultra-low multi-photon probability", npj Quantum Information 4, 1 (2018).
- [127] R. Trivedi, K. Fischer, S. Xu, S. Fan, and J. Vučković, "Few-photon scattering and emission from low-dimensional quantum systems", Physical Review B 98, 144112 (2018).
- [128] K. A. Fischer, R. Trivedi, V. Ramasesh, I. Siddiqi, and J. Vučković, "Scattering into onedimensional waveguides from a coherently-driven quantum-optical system", Quantum 2, 69 (2018).
- [129] H. M. Wiseman and G. J. Milburn, *Quantum Measurement and Control* (Cambridge university press, New York, 2010).
- [130] G. Crowder, H. Carmichael, and S. Hughes, "Quantum trajectory theory of few-photon cavity-QED systems with a time-delayed coherent feedback", Physical Review A 101, 023807 (2020).
- [131] D. B. Horoshko and S. Y. Kilin, "Multimode unraveling of master equation and decoherence problem", Optics Express 2, 347 (1998).
- [132] M. B. Mensky, Continuous Quantum Measurements and Path Integrals (CRC Press, 1993).
- [133] R. H. Hadfield, "Single-photon detectors for optical quantum information applications", Nature Photonics 3, 696 (2009).
- [134] C. M. Natarajan, M. G. Tanner, and R. H. Hadfield, "Superconducting nanowire singlephoton detectors: physics and applications", Superconductor Science and Technology 25, 063001 (2012).
- [135] D. Rosenberg, A. E. Lita, A. J. Miller, and S. W. Nam, "Noise-free high-efficiency photonnumber-resolving detectors", Physical Review A 71, 061803(R) (2005).

- [136] A. E. Lita, A. J. Miller, and S. W. Nam, "Counting near-infrared single-photons with 95% efficiency", Optics Express 16, 3032 (2008).
- [137] E. Knill, R. Laflamme, and G. J. Milburn, "A scheme for efficient quantum computation with linear optics", Nature 409, 46 (2001).
- [138] P. Kok, W. J. Munro, K. Nemoto, T. C. Ralph, J. P. Dowling, and G. J. Milburn, "Linear optical quantum computing with photonic qubits", Reviews of Modern Physics **79**, 135 (2007).
- [139] I. M. Palstra, H. M. Doeleman, and A. F. Koenderink, "Hybrid cavity-antenna systems for quantum optics outside the cryostat?", Nanophotonics 8, 1513 (2019).
- [140] F. Peyskens, D. Chang, and D. Englund, "Integrated nanoplasmonic quantum interfaces for room-temperature single-photon sources", Physical Review B 96, 235151 (2017).
- [141] J. Iles-Smith, D. P. S. McCutcheon, J. Mørk, and A. Nazir, "Limits to coherent scattering and photon coalescence from solid-state quantum emitters", Physical Review B 95, 201305(R) (2017).
- [142] T. Grange, N. Somaschi, C. Antón, L. De Santis, G. Coppola, V. Giesz, A. Lemaître, I. Sagnes, A. Auffèves, and P. Senellart, "Reducing phonon-induced decoherence in solid-state single-photon sources with cavity quantum electrodynamics", Physical Review Letters 118, 253602 (2017).
- [143] J. N. Becker, J. Görlitz, C. Arend, M. Markham, and C. Becher, "Ultrafast all-optical coherent control of single silicon vacancy colour centres in diamond", Nature Communications 7, 13512 (2016).
- [144] A. Vagov, M. D. Croitoru, V. M. Axt, T. Kuhn, and F. M. Peeters, "Nonmonotonic field dependence of damping and reappearance of Rabi oscillations in quantum dots", Physical Review Letters 98, 227403 (2007).

- [145] T. Kaldewey, S. Lüker, A. V. Kuhlmann, S. R. Valentin, J.-M. Chauveau, A. Ludwig, A. D. Wieck, D. E. Reiter, T. Kuhn, and R. J. Warburton, "Demonstrating the decoupling regime of the electron-phonon interaction in a quantum dot using chirped optical excitation", Physical Review B 95, 241306(R) (2017).
- [146] J. B. Spring, B. J. Metcalf, P. C. Humphreys, W. S. Kolthammer, X.-M. Jin, M. Barbieri,
 A. Datta, N. Thomas-Peter, N. K. Langford, D. Kundys, J. C. Gates, B. J. Smith, P. G. R.
 Smith, and I. A. W. Walmsley, "Boson sampling on a photonic chip", Science 339, 798 (2013).
- [147] M. A. Broome, A. Fedrizzi, S. Rahimi-Keshari, J. Dove, S. Aaronson, T. C. Ralph, and A. G. White, "Photonic boson sampling in a tunable circuit", Science 339, 794 (2013).
- [148] N. Sangouard, C. Simon, J. Minář, H. Zbinden, H. De Riedmatten, and N. Gisin, "Longdistance entanglement distribution with single-photon sources", Physical Review A 76, 050301(R) (2007).
- [149] C. Joshi, A. Farsi, S. Clemmen, S. Ramelow, and A. L. Gaeta, "Frequency multiplexing for quasi-deterministic heralded single-photon sources", Nature Communications 9, 847 (2018).
- [150] F. Kaneda and P. G. Kwiat, "High-efficiency single-photon generation via large-scale active time multiplexing", Science Advances 5, eaaw8586 (2019).
- [151] H. Bernien, B. Hensen, W. Pfaff, G. Koolstra, M. S. Blok, L. Robledo, T. Taminiau, M. Markham, D. J. Twitchen, L. Childress, and R. Hanson, "Heralded entanglement between solid-state qubits separated by three metres", Nature 497, 86 (2013).
- [152] C. Santori, D. Fattal, J. Vučković, G. S. Solomon, and Y. Yamamoto, "Indistinguishable photons from a single-photon device", Nature 419, 594 (2002).
- [153] X. Ding, Y. He, Z.-C. Duan, N. Gregersen, M.-C. Chen, S. Unsleber, S. Maier, C. Schneider, M. Kamp, S. Höfling, C.-Y. Lu, and J.-W. Pan, "On-demand single photons with high

extraction efficiency and near-unity indistinguishability from a resonantly driven quantum dot in a micropillar", Physical Review Letters **116**, 020401 (2016).

- [154] I. Aharonovich, D. Englund, and M. Toth, "Solid-state single-photon emitters", Nature Photonics 10, 631 (2016).
- [155] S. I. Bozhevolnyi and J. B. Khurgin, "Fundamental limitations in spontaneous emission rate of single-photon sources", Optica **3**, 1418 (2016).
- [156] S. I. Bozhevolnyi and J. B. Khurgin, "The case for quantum plasmonics", Nature Photonics 11, 398 (2017).
- [157] G. Davies, "Vibronic spectra in diamond", Journal of Physics C: Solid State Physics 7, 3797 (1974).
- [158] K.-M. C. Fu, C. Santori, P. E. Barclay, L. J. Rogers, N. B. Manson, and R. G. Beausoleil, "Observation of the dynamic Jahn-Teller effect in the excited states of nitrogen-vacancy centers in diamond", Physical Review Letters 103, 256404 (2009).
- [159] J. T. Robinson, C. Manolatou, L. Chen, and M. Lipson, "Ultrasmall mode volumes in dielectric optical microcavities", Physical Review Letters 95, 143901 (2005).
- [160] M.-K. Seo, S.-H. Kwon, H.-S. Ee, and H.-G. Park, "Full three-dimensional subwavelength high-Q surface-plasmon-polariton cavity", Nano Letters 9, 4078 (2009).
- [161] M. Kuttge, F. J. García de Abajo, and A. Polman, "Ultrasmall mode volume plasmonic nanodisk resonators", Nano Letters 10, 1537 (2009).
- [162] K. J. Russell and E. L. Hu, "Gap-mode plasmonic nanocavity", Applied Physics Letters 97, 163115 (2010).
- [163] J.-H. Kang, Y.-S. No, S.-H. Kwon, and H.-G. Park, "Ultrasmall subwavelength nanorod plasmonic cavity", Optics Letters 36, 11 (2011).
- [164] S. Hu and S. M. Weiss, "Design of photonic crystal cavities for extreme light concentration", ACS Photonics 3, 1647 (2016).

- [165] H. Choi, M. Heuck, and D. Englund, "Self-similar nanocavity design with ultrasmall mode volume for single-photon nonlinearities", Physical Review Letters **118**, 223605 (2017).
- B. Gurlek, V. Sandoghdar, and D. Martín-Cano, "Manipulation of quenching in nanoantenna– emitter systems enabled by external detuned cavities: a path to enhance strong-coupling", ACS Photonics 5, 456 (2018).
- [167] K. Santhosh, O. Bitton, L. Chuntonov, and G. Haran, "Vacuum Rabi splitting in a plasmonic cavity at the single quantum emitter limit", Nature Communications 7, 11823 (2016).
- [168] R. Chikkaraddy, B. de Nijs, F. Benz, S. J. Barrow, O. A. Scherman, E. Rosta, A. Demetriadou, P. Fox, O. Hess, and J. J. Baumberg, "Single-molecule strong coupling at room temperature in plasmonic nanocavities", Nature 535, 127 (2016).
- [169] P. Peng, Y.-C. Liu, D. Xu, Q.-T. Cao, G. Lu, Q. Gong, and Y.-F. Xiao, "Enhancing coherent light-matter interactions through microcavity-engineered plasmonic resonances", Physical Review Letters 119, 233901 (2017).
- [170] L. C. Bassett, F. J. Heremans, D. J. Christle, C. G. Yale, G. Burkard, B. B. Buckley, and D. D. Awschalom, "Ultrafast optical control of orbital and spin dynamics in a solid-state defect", Science **345**, 1333 (2014).
- [171] C. Gardiner and P. Zoller, *Quantum noise: a handbook of Markovian and non-Markovian quantum stochastic methods with applications to quantum optics*, Vol. 56 (Springer Science & Business Media, 2004).
- [172] A. Delga, J. Feist, J. Bravo-Abad, and F. J. Garcia-Vidal, "Quantum emitters near a metal nanoparticle: strong coupling and quenching", Physical Review Letters 112, 253601 (2014).
- [173] E. Neu, D. Steinmetz, J. Riedrich-Möller, S. Gsell, M. Fischer, M. Schreck, and C. Becher, "Single photon emission from silicon-vacancy colour centres in chemical vapour deposition nano-diamonds on iridium", New Journal of Physics 13, 025012 (2011).

- [174] J. Benedikter, H. Kaupp, T. Hümmer, Y. Liang, A. Bommer, C. Becher, A. Krueger, J. M. Smith, T. W. Hänsch, and D. Hunger, "Cavity-enhanced single-photon source based on the silicon-vacancy center in diamond", Physical Review Applied 7, 024031 (2017).
- [175] C. Santori, P. E. Barclay, K.-M. C. Fu, R. G. Beausoleil, S. Spillane, and M. Fisch, "Nanophotonics for quantum optics using nitrogen-vacancy centers in diamond", Nanotechnology 21, 274008 (2010).
- [176] X. Brokmann, G. Messin, P. Desbiolles, E. Giacobino, M. Dahan, and J. P. Hermier,
 "Colloidal CdSe/ZnS quantum dots as single-photon sources", New Journal of Physics 6, 99 (2004).
- [177] A. M. Stoneham, A. H. Harker, and G. W. Morley, "Could one make a diamond-based quantum computer?", Journal of Physics: Condensed Matter **21**, 364222 (2009).
- [178] L. J. Rogers, S. Armstrong, M. J. Sellars, and N. B. Manson, "Infrared emission of the NV centre in diamond: Zeeman and uniaxial stress studies", New Journal of Physics 10, 103024 (2008).
- [179] M. W. Doherty, V. M. Acosta, A. Jarmola, M. S. J. Barson, N. B. Manson, D. Budker, and
 L. C. L. Hollenberg, "Temperature shifts of the resonances of the NV- center in diamond",
 Physical Review B 90, 041201(R) (2014).
- [180] L. J. Rogers, M. W. Doherty, M. S. J. Barson, S. Onoda, T. Ohshima, and N. B. Manson,"Singlet levels of the NV- centre in diamond", New Journal of Physics 17, 013048 (2015).
- B. L. Green, S. Mottishaw, B. G. Breeze, A. M. Edmonds, U. F. S. D'Haenens-Johansson, M. Doherty, S. D. Williams, D. J. Twitchen, and M. E. Newton, "Neutral silicon-vacancy center in diamond: spin polarization and lifetimes", Physical Review Letters 119, 096402 (2017).
- [182] B. C. Rose, D. Huang, Z.-H. Zhang, P. Stevenson, A. M. Tyryshkin, S. Sangtawesin, S. Srinivasan, L. Loudin, M. L. Markham, A. M. Edmonds, D. J. Twitchen, S. A. Lyon, and

N. P. de Leon, "Observation of an environmentally insensitive solid-state spin defect in diamond", Science **361**, 60 (2018).

- [183] A. Perrot, P. Goldner, D. Giaume, M. Lovrić, C. Andriamiadamanana, R. R. Gonçalves, and A. Ferrier, "Narrow optical homogeneous linewidths in rare earth doped nanocrystals", Physical Review Letters 111, 203601 (2013).
- [184] K. Binnemans, "Interpretation of europium (III) spectra", Coordination Chemistry Reviews 295, 1 (2015).
- [185] J. D. Plumhof, T. Stöferle, L. Mai, U. Scherf, and R. F. Mahrt, "Room-temperature Bose– Einstein condensation of cavity exciton–polaritons in a polymer", Nature Materials 13, 247 (2014).
- [186] S. Kumar, K. Boone, J. Tuszyński, P. Barclay, and C. Simon, "Possible existence of optical communication channels in the brain", Scientific Reports 6, 36508 (2016).
- [187] F. Liu, A. J. Brash, J. O'Hara, L. M. Martins, C. L. Phillips, R. J. Coles, B. Royall, E. Clarke, C. Bentham, N. Prtljaga, I. E. Itskevich, L. R. Wilson, M. S. Skolnick, and A. Mark Fox, "High Purcell factor generation of indistinguishable on-chip single photons", Nature Nanotechnology 13, 835–840 (2018).
- [188] P. Kumar and A. K. Sarma, "Ultrafast and selective coherent population transfer in four-level atoms by a single nonlinearly chirped femtosecond pulse", Physical Review A 88, 033823 (2013).
- [189] K. A. Fischer, L. Hanschke, J. Wierzbowski, T. Simmet, C. Dory, J. J. Finley, J. Vučković, and K. Müller, "Signatures of two-photon pulses from a quantum two-level system", Nature Physics 13, 649 (2017).
- [190] J. C. Loredo, C. Antón, B. Reznychenko, P. Hilaire, A. Harouri, C. Millet, H. Ollivier, N. Somaschi, L. De Santis, A. Lemaître, I. Sagnes, L. Lanco, A. Auffèves, O. Krebs, and P. Senellart, "Generation of non-classical light in a photon-number superposition", Nature Photonics 13, 803 (2019).

- [191] P. Hilaire, C. Millet, J. C. Loredo, C. Antón, A Harouri, A. Lemaître, I. Sagnes, N. Somaschi,
 O. Krebs, P. Senellart, and L. Lanco, "Deterministic assembly of a charged-quantum-dot– micropillar cavity device", Physical Review B 102, 195402 (2020).
- [192] C. Gustin and S. Hughes, "Efficient Pulse-Excitation Techniques for Single Photon Sources from Quantum Dots in Optical Cavities", Advanced Quantum Technologies 3, 1900073 (2020).
- [193] J. H. Quilter, A. J. Brash, F. Liu, M. Glässl, A. M. Barth, V. M. Axt, A. J. Ramsay, M. S. Skolnick, and A. M. Fox, "Phonon-assisted population inversion of a single InGaAs/GaAs quantum dot by pulsed laser excitation", Physical Review Letters 114, 137401 (2015).
- [194] M. Cosacchi, F. Ungar, M. Cygorek, A. Vagov, and V. M. Axt, "Emission-frequency separated high quality single-photon sources enabled by phonons", Physical Review Letters 123, 017403 (2019).
- [195] H. Wang, Y.-M. He, T.-H. Chung, H. Hu, Y. Yu, S. Chen, X. Ding, M.-C. Chen, J. Qin, X. Yang, et al., "Towards optimal single-photon sources from polarized microcavities", Nature Photonics 13, 770–775 (2019).
- [196] F. Rozpedek, R. Yehia, K. Goodenough, M. Ruf, P. C. Humphreys, R. Hanson, S. Wehner, and D. Elkouss, "Near-term quantum-repeater experiments with nitrogen-vacancy centers: overcoming the limitations of direct transmission", Physical Review A 99, 052330 (2019).
- [197] L.-M. Duan, M. D. Lukin, J. I. Cirac, and P. Zoller, "Long-distance quantum communication with atomic ensembles and linear optics", Nature 414, 413 (2001).
- [198] S. J. Van Enk, "Single-particle entanglement", Physical Review A 72, 064306 (2005).
- [199] P. Zarkeshian, C. Deshmukh, N. Sinclair, S. K. Goyal, G. H. Aguilar, P. Lefebvre, M. G. Puigibert, V. B. Verma, F. Marsili, M. D. Shaw, S. W. Nam, K. Heshami, D. Oblak, W. Tittel, and C. Simon, "Entanglement between more than two hundred macroscopic atomic ensembles in a solid", Nature Communications 8, 1 (2017).

- [200] I. Schwartz, D. Cogan, E. R. Schmidgall, Y. Don, L. Gantz, O. Kenneth, N. H. Lindner, and D. Gershoni, "Deterministic generation of a cluster state of entangled photons", Science 354, 434 (2016).
- [201] N. Sangouard, R. Dubessy, and C. Simon, "Quantum repeaters based on single trapped ions", Physical Review A 79, 042340 (2009).
- [202] N. Sangouard, C. Simon, H. De Riedmatten, and N. Gisin, "Quantum repeaters based on atomic ensembles and linear optics", Reviews of Modern Physics 83, 33 (2011).
- [203] Y. L. Lim, A. Beige, and L. C. Kwek, "Repeat-until-success linear optics distributed quantum computing", Physical Review Letters 95, 030505 (2005).
- [204] S. C. Benjamin, B. W. Lovett, and J. M. Smith, "Prospects for measurement-based quantum computing with solid state spins", Laser & Photonics Reviews 3, 556 (2009).
- [205] W. Yao, R.-B. Liu, and L. J. Sham, "Theory of control of the spin-photon interface for quantum networks", Physical Review Letters 95, 030504 (2005).
- [206] B. Korzh, C. C. W. Lim, R. Houlmann, N. Gisin, M. J. Li, D. Nolan, B. Sanguinetti, R. Thew, and H. Zbinden, "Provably secure and practical quantum key distribution over 307 km of optical fibre", Nature Photonics 9, 163 (2015).
- [207] W. K. Wootters and W. H. Zurek, "A single quantum cannot be cloned", Nature 299, 802 (1982).
- [208] D. Dieks, "Communication by EPR devices", Physics Letters A 92, 271 (1982).
- [209] C. Cabrillo, J. I. Cirac, P. García-Fernández, and P. Zoller, "Creation of entangled states of distant atoms by interference", Physical Review A 59, 1025 (1999).
- [210] C. Simon and W. T. M. Irvine, "Robust long-distance entanglement and a loophole-free Bell test with ions and photons", Physical Review Letters 91, 110405 (2003).
- [211] X.-L. Feng, Z.-M. Zhang, X.-D. Li, S.-Q. Gong, and Z.-Z. Xu, "Entangling distant atoms by interference of polarized photons", Physical Review Letters 90, 217902 (2003).

- [212] C.-W. Chou, H. de Riedmatten, D. Felinto, S. V. Polyakov, S. J. Van Enk, and H. J. Kimble,
 "Measurement-induced entanglement for excitation stored in remote atomic ensembles",
 Nature 438, 828 (2005).
- [213] C.-W. Chou, J. Laurat, H. Deng, K. S. Choi, H. De Riedmatten, D. Felinto, and H. J. Kimble,
 "Functional quantum nodes for entanglement distribution over scalable quantum networks",
 Science 316, 1316 (2007).
- [214] J. Hofmann, M. Krug, N. Ortegel, L. Gérard, M. Weber, W. Rosenfeld, and H. Weinfurter, "Heralded entanglement between widely separated atoms", Science 337, 72 (2012).
- [215] D. L. Moehring, P. Maunz, S. Olmschenk, K. C. Younge, D. N. Matsukevich, L.-M. Duan, and C. Monroe, "Entanglement of single-atom quantum bits at a distance", Nature 449, 68 (2007).
- [216] L. Slodička, G. Hétet, N. Röck, P. Schindler, M. Hennrich, and R. Blatt, "Atom-atom entanglement by single-photon detection", Physical Review Letters **110**, 083603 (2013).
- [217] A. Delteil, Z. Sun, W.-b. Gao, E. Togan, S. Faelt, and A. Imamoğlu, "Generation of heralded entanglement between distant hole spins", Nature Physics 12, 218 (2016).
- [218] R. Stockill, M. J. Stanley, L. Huthmacher, E. Clarke, M. Hugues, A. J. Miller, C. Matthiesen,
 C. Le Gall, and M. Atatüre, "Phase-tuned entangled state generation between distant spin qubits", Physical Review Letters 119, 010503 (2017).
- [219] R. Hanson, L. P. Kouwenhoven, J. R. Petta, S. Tarucha, and L. M. Vandersypen, "Spins in few-electron quantum dots", Reviews of Modern Physics 79, 1217 (2007).
- [220] V. N. Golovach, A. Khaetskii, and D. Loss, "Phonon-induced decay of the electron spin in quantum dots", Physical Review Letters 93, 016601 (2004).
- [221] T. Plakhotnik, M. W. Doherty, and N. B. Manson, "Electron-phonon processes of the nitrogen-vacancy center in diamond", Physical Review B 92, 081203(R) (2015).

- [222] X. Wu, A. Kolar, J. Chung, D. Jin, T. Zhong, R. Kettimuthu, and M. Suchara, "SeQUeNCe: a customizable discrete-event simulator of quantum networks", arXiv:2009.12000 (2020).
- [223] T. A. Brun, "Continuous measurements, quantum trajectories, and decoherent histories", Physical Review A 61, 042107 (2000).
- [224] J. C. Loredo, N. A. Zakaria, N. Somaschi, C. Anton, L. De Santis, V. Giesz, T. Grange, M. A. Broome, O. Gazzano, G. Coppola, et al., "Scalable performance in solid-state single-photon sources", Optica 3, 433 (2016).
- [225] A. Thoma, P. Schnauber, M. Gschrey, M. Seifried, J. Wolters, J.-H. Schulze, A. Strittmatter, S. Rodt, A. Carmele, A. Knorr, T. Heindel, and S. Reitzenstein, "Exploring dephasing of a solid-state quantum emitter via time-and temperature-dependent Hong-Ou-Mandel experiments", Physical Review Letters **116**, 033601 (2016).
- [226] M. E. Reimer, G. Bulgarini, A. Fognini, R. W. Heeres, B. J. Witek, M. A. M. Versteegh, A. Rubino, T. Braun, M. Kamp, S. Höfling, D. Dalacu, J. Lapointe, P. J. Poole, and V. Zwiller, "Overcoming power broadening of the quantum dot emission in a pure wurtzite nanowire", Physical Review B 93, 195316 (2016).
- [227] J. Calsamiglia and N. Lütkenhaus, "Maximum efficiency of a linear-optical Bell-state analyzer", Applied Physics B 72, 67 (2001).
- [228] L. S. Martin and K. B. Whaley, "Single-shot deterministic entanglement between noninteracting systems with linear optics", arXiv:1912.00067 (2019).
- [229] K. Mattle, H. Weinfurter, P. G. Kwiat, and A. Zeilinger, "Dense coding in experimental quantum communication", Physical Review Letters 76, 4656 (1996).
- [230] W. P. Grice, "Arbitrarily complete Bell-state measurement using only linear optical elements", Physical Review A 84, 042331 (2011).
- [231] J. Minář, H. De Riedmatten, C. Simon, H. Zbinden, and N. Gisin, "Phase-noise measurements in long-fiber interferometers for quantum-repeater applications", Physical Review A 77, 052325 (2008).

- [232] Y. Yu, F. Ma, X.-Y. Luo, B. Jing, P.-F. Sun, R.-Z. Fang, C.-W. Yang, H. Liu, M.-Y. Zheng, X.-P. Xie, et al., "Entanglement of two quantum memories via fibres over dozens of kilometres", Nature 578, 240 (2020).
- [233] M.-A. Lemonde, S. Meesala, A. Sipahigil, M. J. A. Schuetz, M. D. Lukin, M. Loncar, and P. Rabl, "Phonon networks with silicon-vacancy centers in diamond waveguides", Physical Review Letters 120, 213603 (2018).
- [234] N. Ohlsson, R. K. Mohan, and S. Kröll, "Quantum computer hardware based on rare-earthion-doped inorganic crystals", Optics Communications 201, 71 (2002).
- [235] L. You and M. S. Chapman, "Quantum entanglement using trapped atomic spins", Physical Review A 62, 052302 (2000).
- [236] L.-M. Duan, B. Wang, and H. J. Kimble, "Robust quantum gates on neutral atoms with cavity-assisted photon scattering", Physical Review A 72, 032333 (2005).
- [237] M. Feng, "Quantum computing with trapped ions in an optical cavity via Raman transition", Physical Review A 66, 054303 (2002).
- [238] J. Majer, J. M. Chow, J. M. Gambetta, J. Koch, B. R. Johnson, J. A. Schreier, L. Frunzio,
 D. I. Schuster, A. A. Houck, A. Wallraff, A. Blais, M. H. Devoret, S. M. Girvin, and R. J.
 Schoelkopf, "Coupling superconducting qubits via a cavity bus", Nature 449, 443 (2007).
- [239] M. A. Sillanpää, J. I. Park, and R. W. Simmonds, "Coherent quantum state storage and transfer between two phase qubits via a resonant cavity", Nature 449, 438 (2007).
- [240] J. P. Reithmaier, G. Sek, A. Löffler, C. Hofmann, S. Kuhn, S. Reitzenstein, L. V. Keldysh,
 V. D. Kulakovskii, T. L. Reinecke, and A. Forchel, "Strong coupling in a single quantum dot-semiconductor microcavity system", Nature 432, 197 (2004).
- [241] E. Peter, P. Senellart, D. Martrou, A. Lemaître, J. Hours, J. M. Gérard, and J. Bloch, "Excitonphoton strong-coupling regime for a single quantum dot embedded in a microcavity", Physical Review Letters 95, 067401 (2005).

- [242] D. Englund, A. Faraon, I. Fushman, N. Stoltz, P. Petroff, and J. Vučković, "Controlling cavity reflectivity with a single quantum dot", Nature 450, 857 (2007).
- [243] S. Welte, B. Hacker, S. Daiss, S. Ritter, and G. Rempe, "Photon-mediated quantum gate between two neutral atoms in an optical cavity", Physical Review X 8, 011018 (2018).
- [244] A. Blais, R.-S. Huang, A. Wallraff, S. M. Girvin, and R. J. Schoelkopf, "Cavity quantum electrodynamics for superconducting electrical circuits: an architecture for quantum computation", Physical Review A 69, 062320 (2004).
- [245] R. E. Evans, M. K. Bhaskar, D. D. Sukachev, C. T. Nguyen, A. Sipahigil, M. J. Burek,
 B. Machielse, G. H. Zhang, A. S. Zibrov, E. Bielejec, H. Park, M. Lončar, and M. D.
 Lukin, "Photon-mediated interactions between quantum emitters in a diamond nanocavity",
 Science 362, 662 (2018).
- [246] F. Nan, S.-J. Ding, L. Ma, Z.-Q. Cheng, Y.-T. Zhong, Y.-F. Zhang, Y.-H. Qiu, X. Li, L. Zhou, and Q.-Q. Wang, "Plasmon resonance energy transfer and plexcitonic solar cell", Nanoscale 8, 15071 (2016).
- [247] L.-Y. Hsu, W. Ding, and G. C. Schatz, "Plasmon-coupled resonance energy transfer", The Journal of Physical Chemistry Letters 8, 2357 (2017).
- [248] E. V. Denning, J. Iles-Smith, and J. Mork, "Quantum light-matter interaction and controlled phonon scattering in a photonic Fano cavity", Physical Review B **100**, 214306 (2019).
- [249] C. Gustin and S. Hughes, "Influence of electron-phonon scattering for an on-demand quantum dot single-photon source using cavity-assisted adiabatic passage", Physical Review B 96, 085305 (2017).
- [250] R. Betzholz, J. M. Torres, and M. Bienert, "Quantum optical master equation for solid-state quantum emitters", Physical Review A 90, 063818 (2014).
- [251] C. Li, Z.-Q. Xu, N. Mendelson, M. Kianinia, M. Toth, and I. Aharonovich, "Purification of single-photon emission from hBN using post-processing treatments", Nanophotonics 8, 2049 (2019).

- [252] A. Dietrich, M. W. Doherty, I. Aharonovich, and A. Kubanek, "Solid-state single photon source with Fourier transform limited lines at room temperature", Physical Review B 101, 081401(R) (2020).
- [253] C. Panuski, D. Englund, and R. Hamerly, "Fundamental thermal noise limits for optical microcavities", Physical Review X 10, 041046 (2020).
- [254] H. Abudayyeh, B. Lubotzky, S. Majumder, J. A. Hollingsworth, and R. Rapaport, "Purification of single photons by temporal heralding of quantum dot sources", ACS Photonics 6, 446 (2019).
- [255] D. Buterakos, E. Barnes, and S. E. Economou, "Deterministic generation of all-photonic quantum repeaters from solid-state emitters", Physical Review X 7, 041023 (2017).
- [256] T.-M. Zhao, H. Zhang, J. Yang, Z.-R. Sang, X. Jiang, X.-H. Bao, and J.-W. Pan, "Entangling different-color photons via time-resolved measurement and active feed forward", Physical Review Letters 112, 103602 (2014).
- [257] M. K. Dezfouli, R. Gordon, and S. Hughes, "Modal theory of modified spontaneous emission of a quantum emitter in a hybrid plasmonic photonic-crystal cavity system", Physical Review A 95, 013846 (2017).
- [258] H. M. Doeleman, C. D. Dieleman, C. Mennes, B. Ehrler, and A. F. Koenderink, "Observation of cooperative Purcell enhancements in antenna–cavity hybrids", ACS Nano 14, 12027 (2020).
- [259] M. Barth, S. Schietinger, S. Fischer, J. Becker, N. Nüsse, T. Aichele, B. Lochel, C. Sönnichsen, and O. Benson, "Nanoassembled plasmonic-photonic hybrid cavity for tailored light-matter coupling", Nano Letters 10, 891 (2010).

Appendix A

Properties of conditional propagation superoperators

In this appendix section, I will present three proofs related to conditional propagation superoperators in the interest of completion. The first proof is a validation that the integral form of the density operator solution satisfies the master equation. Although the form is commonly obtained by the method of variation of parameters, I validate it here in the notation of the thesis for consistency. The second is an interesting extension of the semi-group property to a set of conditional propagation superoperators. The last proof shows the invariance of time ordering for the jump operators composing a conditional propagator. Although I haven't found a proper citation for these last two proofs, I am quite confident that they are not novel.

A.1 Integral form of the master equation

The photon number decomposition relies on the solution

$$\hat{\rho}_{s}(t) = \mathcal{U}_{0}(t, t_{0})\hat{\rho}(t_{0}) + \int_{t_{0}}^{t} \mathcal{U}_{0}(t, t')\mathcal{J}\hat{\rho}_{s}(t')dt'$$
(A.1)

being the solution to the master equation $\dot{\rho} = \mathcal{L}_0 \hat{\rho}(t) + \mathcal{J} \hat{\rho}(t)$. I am using the notation $\hat{\rho}_s(t)$ to represent the proposed solution and then I will prove it is equal to the general solution $\hat{\rho}(t)$ for initial condition $\hat{\rho}(t_0)$. The solution $\hat{\rho}_s$ is constructed using the method of variation of parameters (which is equivalent to solving using an integrating factor for first-order differential equations) where the first term is the solution to the equation $\dot{\rho}_0 = \mathcal{L}_0 \hat{\rho}_0(t)$ and the second term is a particular solution to the total equation $\dot{\rho} = \mathcal{L}\hat{\rho}(t)$ where $\mathcal{L} = \mathcal{L}_0 + \mathcal{J}$. Here, I am using $\hat{\rho}_0(t)$ to again avoid confusion with the solution $\hat{\rho}(t)$ to the master equation. To prove that $\hat{\rho}_s(t)$ is the general solution to the master equation $\dot{\rho} = \mathcal{L}\hat{\rho}(t)$, we must first prove some properties of \mathcal{U}_0 . The propagator \mathcal{U}_0 is defined as $\hat{\rho}_0(t) = \mathcal{U}_0(t, t_0)\hat{\rho}_0(t_0)$ where $\dot{\rho}_0 = \mathcal{L}_0\hat{\rho}_0(t)$. The above definition of \mathcal{U}_0 implies that it also satisfies the same differential equation

$$\frac{d\mathcal{U}_0(t,t_0)}{dt} = \mathcal{L}_0 \mathcal{U}_0(t,t_0) \qquad [\mathbf{I}]$$
(A.2)

for any t_0 . In addition,

$$\mathcal{U}_{0}(t_{\rm f}, t_{0})\hat{\rho}_{0}(t_{0}) = \hat{\rho}_{0}(t_{\rm f}) = \mathcal{U}_{0}(t_{\rm f}, t)\hat{\rho}_{0}(t) = \mathcal{U}_{0}(t_{\rm f}, t)\mathcal{U}_{0}(t, t_{0})\hat{\rho}_{0}(t_{0})$$
(A.3)

implies $\mathcal{U}_0(t_f, t_0) = \mathcal{U}_0(t_f, t)\mathcal{U}_0(t, t_0)$ for any t [II]. Finally, $\mathcal{U}_0(t, t)\hat{\rho}_0(t) = \hat{\rho}_0(t)$ for all t implies $\mathcal{U}_0(t, t) = \mathcal{I}$ [III] for all t where \mathcal{I} is the identity superoperator. In addition to the above three properties, we need to use the product rule [IV] and the fundamental theorem of calculus [V].

Now, let us take the time derivative of the variation of parameters solution

$$\begin{split} \dot{\rho}_{s} &= \frac{d}{dt} \left[\mathcal{U}_{0}(t,t_{0})\hat{\rho}(t_{0}) + \int_{t_{0}}^{t} \mathcal{U}_{0}(t,t')\mathcal{J}\hat{\rho}_{s}(t')dt' \right] \\ &= \frac{d\mathcal{U}_{0}(t,t_{0})}{dt}\hat{\rho}(t_{0}) + \frac{d}{dt}\int_{t_{0}}^{t} \mathcal{U}_{0}(t,t')\mathcal{J}\hat{\rho}_{s}(t')dt' \\ &= \mathcal{L}_{0}\mathcal{U}_{0}(t,t_{0})\hat{\rho}(t_{0}) + \frac{d}{dt}\mathcal{U}_{0}(t,t'')\int_{t_{0}}^{t} \mathcal{U}_{0}(t'',t')\mathcal{J}\hat{\rho}_{s}(t')dt' \qquad \text{by [I] and [II]} \\ &= \mathcal{L}_{0}\mathcal{U}_{0}(t,t_{0})\hat{\rho}(t_{0}) + \frac{d\mathcal{U}_{0}(t,t'')}{dt}\int_{t_{0}}^{t} \mathcal{U}_{0}(t'',t')\mathcal{J}\hat{\rho}_{s}(t')dt' \\ &\quad + \mathcal{U}_{0}(t,t'')\frac{d}{dt}\int_{t_{0}}^{t} \mathcal{U}_{0}(t'',t')\mathcal{J}\hat{\rho}_{s}(t')dt' \qquad \text{by [IV]} \quad (A.4) \\ &= \mathcal{L}_{0}\mathcal{U}_{0}(t,t_{0})\hat{\rho}(t_{0}) + \mathcal{L}_{0}\mathcal{U}_{0}(t,t'')\int_{t_{0}}^{t} \mathcal{U}_{0}(t'',t')\mathcal{J}\hat{\rho}_{s}(t')dt' \end{split}$$

$$+ \mathcal{U}_0(t, t'')\mathcal{U}_0(t'', t)\mathcal{J}\hat{\rho}_s(t) \qquad \text{by [I] and [V]}$$
$$= \mathcal{L}_0\left[\mathcal{U}_0(t, t_0)\hat{\rho}(t_0) + \int_{t_0}^t \mathcal{U}_0(t, t')\mathcal{J}\hat{\rho}_s(t')dt'\right] + \mathcal{U}_0(t, t)\mathcal{J}\hat{\rho}_s(t) \qquad \text{by [II]}$$

$$= \mathcal{L}_0 \hat{\rho}_s(t) + \mathcal{J} \hat{\rho}_s(t)$$
by [III]
$$= \mathcal{L} \hat{\rho}_s(t).$$

Hence $\hat{\rho}_s(t)$ is a solution to $\dot{\rho} = \mathcal{L}\hat{\rho}(t)$. Finally, we can confirm that

$$\hat{\rho}_{s}(t_{0}) = \mathcal{U}_{0}(t_{0}, t_{0})\hat{\rho}(t_{0}) + \int_{t_{0}}^{t_{0}} \mathcal{U}_{0}(t_{0}, t')\mathcal{J}\hat{\rho}_{s}(t')dt'$$

$$= \mathcal{U}_{0}(t_{0}, t_{0})\hat{\rho}(t_{0}) + 0$$

$$= \hat{\rho}(t_{0}). \qquad \text{by [III]}$$

Thus $\hat{\rho}_s(t)$ satisfies the initial condition $\hat{\rho}(t_0)$. Therefore, by the uniqueness theorem for first-order differential equations, we have proven $\hat{\rho}(t) = \hat{\rho}_s(t)$.

A.2 Subintervals of time

The total propagation superoperator can be divided into subintervals

$$\mathcal{U}(t_{\rm f}, t_0) = \mathcal{U}(t_{\rm f}, t)\mathcal{U}(t, t_0) = \sum_{k,l} \mathcal{U}_k(t_{\rm f}, t)\mathcal{U}_l(t, t_0) = \sum_n \mathcal{U}_n(t_{\rm f}, t_0).$$
(A.6)

We can show that a similar relation

$$\mathcal{U}_n(t_{\rm f}, t_0) = \sum_{k,l} \mathcal{U}_k(t_{\rm f}, t) \mathcal{U}_l(t, t_0) \delta_{k+l,n}$$
(A.7)

holds for the conditional propagation superoperators. This relies on the fact that it holds for the homogeneous case

$$\mathcal{U}_0(t_{\rm f}, t_0) = \mathcal{U}_0(t_{\rm f}, t)\mathcal{U}_0(t, t_0) \tag{A.8}$$

as a consequence of U_0 being a propagator superoperator of a Markovian master equation $\dot{\rho} = \mathcal{L}_0 \hat{\rho}(t)$. First, consider the base case for n = 1

$$\begin{aligned} \mathcal{U}_{0}(t_{\rm f},t)\mathcal{U}_{1}(t,t_{0}) &+ \mathcal{U}_{1}(t_{\rm f},t)\mathcal{U}_{0}(t,t_{0}) \\ &= \int_{t_{0}}^{t} \mathcal{U}_{0}(t_{\rm f},t)\mathcal{U}_{0}(t,t')\mathcal{J}\mathcal{U}_{0}(t',t_{0})dt' + \int_{t}^{t_{\rm f}} \mathcal{U}_{0}(t_{\rm f},t')\mathcal{J}\mathcal{U}_{0}(t',t)\mathcal{U}_{0}(t,t_{0})dt' \\ &= \int_{t_{0}}^{t} \mathcal{U}_{0}(t_{\rm f},t')\mathcal{J}\mathcal{U}_{0}(t',t_{0})dt' + \int_{t}^{t_{\rm f}} \mathcal{U}_{0}(t_{\rm f},t')\mathcal{J}\mathcal{U}_{0}(t',t_{0})dt' \\ &= \mathcal{U}_{1}(t,t_{0}). \end{aligned}$$
(A.9)

Now, assume that for some $n\geq 0$ we have

$$\mathcal{U}_n(t,t_0) = \sum_{k,l} \mathcal{U}_k(t,t') \mathcal{U}_l(t',t_0) \delta_{k+l,n}.$$
(A.10)

Then this implies

$$\begin{aligned} \mathcal{U}_{n+1}(t_{\rm f},t_0) &= \int_{t'}^{t_{\rm f}} \mathcal{U}_0(t_{\rm f},t) \mathcal{J} \sum_{k,l} \mathcal{U}_k(t,t') \mathcal{U}_l(t',t_0) \delta_{k+l,n} dt + \int_{t_0}^{t'} \mathcal{U}_0(t_{\rm f},t) \mathcal{J} \mathcal{U}_n(t,t_0) dt \\ &= \sum_{k,l} \int_{t'}^{t_{\rm f}} \mathcal{U}_0(t_{\rm f},t) \mathcal{J} \mathcal{U}_k(t,t') dt \mathcal{U}_l(t',t_0) \delta_{k+l,n} + \mathcal{U}_0(t_{\rm f},t') \int_{t_0}^{t'} \mathcal{U}_0(t',t) \mathcal{J} \mathcal{U}_n(t,t_0) dt \\ &= \sum_{k=1,l} \mathcal{U}_k(t_{\rm f},t') \mathcal{U}_l(t',t_0) \delta_{k+l,n+1} + \mathcal{U}_0(t_{\rm f},t') \mathcal{U}_{n+1}(t',t_0) \\ &= \sum_{k,l} \mathcal{U}_k(t_{\rm f},t') \mathcal{U}_l(t',t_0) \delta_{k+l,n+1} \end{aligned}$$
(A.11)

which completes the proof. As a consequence of this proof, we know that dividing conditional propagators into subintervals of time conserves the total photon number, as expected.

A.3 Invariance of order

The definition of \mathcal{U}_n assumes that all n-1 photons are emitted before the integrated time

$$\mathcal{U}_n(t_{\rm f}, t_0) = \int_{t_0}^{t_{\rm f}} \mathcal{U}_0(t_{\rm f}, t) \mathcal{J} \mathcal{U}_{n-1}(t, t_0) dt.$$
(A.12)

However, we can show that the conditional propagator does not depend on the order of emission. That is, the more general case

$$\mathcal{U}_{k+l+1}(t_{\rm f}, t_0) = \int_{t_0}^{t_{\rm f}} \mathcal{U}_k(t_{\rm f}, t) \mathcal{J} \mathcal{U}_l(t, t_0) dt \tag{A.13}$$

holds as a consequence. First consider the base case where k = 0,

$$\begin{aligned} \mathcal{U}_{l+1}(t_{\rm f},t_0) &= \int_{t_0}^{t_{\rm f}} \mathcal{U}_0(t_{\rm f},t) \mathcal{J} \mathcal{U}_l(t,t_0) dt = \int_{t_0}^{t_{\rm f}} \int_{t_0}^{t} \mathcal{U}_0(t_{\rm f},t) \mathcal{J} \mathcal{U}_0(t,t') \mathcal{J} \mathcal{U}_{l-1}(t',t_0) dt' dt \\ &= \int_{t_0}^{t_{\rm f}} \int_{t'}^{t_{\rm f}} \mathcal{U}_0(t_{\rm f},t) \mathcal{J} \mathcal{U}_0(t,t') \mathcal{J} \mathcal{U}_{l-1}(t',t_0) dt dt' \quad (A.14) \\ &= \int_{t_0}^{t_{\rm f}} \mathcal{U}_1(t_{\rm f},t') \mathcal{J} \mathcal{U}_{l-1}(t',t_0) dt', \end{aligned}$$

where I have used the change of integration order relation

$$\int_{t_0}^{t_{\rm f}} \int_{t_0}^t f(t,t') dt' dt = \int_{t_0}^{t_{\rm f}} \int_{t'}^{t_{\rm f}} f(t,t') dt dt'.$$
(A.15)

Now, let us assume that for some $k\geq 0$ we have

$$\int_{t_0}^{t_{\rm f}} \mathcal{U}_k(t_{\rm f}, t) \mathcal{J} \mathcal{U}_l(t, t_0) dt = \int_{t_0}^{t_{\rm f}} \mathcal{U}_{k+1}(t_{\rm f}, t) \mathcal{J} \mathcal{U}_{l-1}(t, t_0) dt.$$
(A.16)

Then this implies

$$\int_{t_0}^{t_f} \mathcal{U}_{k+1}(t_f, t) \mathcal{J}\mathcal{U}_l(t, t_0) dt = \int_{t_0}^{t_f} \int_{t}^{t_f} \mathcal{U}_0(t_f, t') \mathcal{J}\mathcal{U}_k(t', t) dt' \mathcal{J}\mathcal{U}_l(t, t_0) dt
= \int_{t_0}^{t_f} \mathcal{U}_0(t_f, t') \mathcal{J} \int_{t_0}^{t'} \mathcal{U}_k(t', t) \mathcal{J}\mathcal{U}_l(t, t_0) dt dt'
= \int_{t_0}^{t_f} \mathcal{U}_0(t_f, t') \mathcal{J} \int_{t_0}^{t'} \mathcal{U}_{k+1}(t', t) \mathcal{J}\mathcal{U}_{l-1}(t, t_0) dt dt' \quad (A.17)
= \int_{t_0}^{t_f} \int_{t}^{t_f} \mathcal{U}_0(t_f, t') \mathcal{J}\mathcal{U}_{k+1}(t', t) dt' \mathcal{J}\mathcal{U}_{l-1}(t, t_0) dt
= \int_{t_0}^{t_f} \mathcal{U}_{k+2}(t_f, t) \mathcal{J}\mathcal{U}_{l-1}(t, t_0) dt.$$

Then for any k and l, we have

$$\int_{t_0}^{t_f} \mathcal{U}_k(t_f, t) \mathcal{J} \mathcal{U}_l(t, t_0) dt = \int_{t_0}^{t_f} \mathcal{U}_0(t_f, t) \mathcal{J} \mathcal{U}_{l+k}(t, t_0) dt$$

$$= \mathcal{U}_{k+l+1}(t_f, t_0).$$
(A.18)

Appendix B

Critical regime and phonon sideband corrections

B.1 Efficiency

To derive the indistinguishability and brightness in the critical coupling cavity QED regime, we begin by writing the optical Bloch equations in the single-excitation regime. By assuming that the cavity and emitter are resonant so that $\Delta = 0$, from section 1.3.4 we have

$$\frac{d}{dt} \begin{bmatrix} \langle \hat{a}^{\dagger} \rangle \\ \langle \hat{\sigma}^{\dagger} \rangle \end{bmatrix} = A_1 \begin{bmatrix} \langle \hat{a}^{\dagger} \rangle \\ \langle \hat{\sigma}^{\dagger} \rangle \end{bmatrix}, \qquad (B.1)$$

and

$$\frac{d}{dt} \begin{bmatrix} \langle \hat{a}^{\dagger} \hat{a} \rangle \\ \langle \hat{a}^{\dagger} \hat{\sigma} \rangle \\ \langle \hat{\sigma}^{\dagger} \hat{a} \rangle \\ \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle \end{bmatrix} = A_2 \begin{bmatrix} \langle \hat{a}^{\dagger} \hat{a} \rangle \\ \langle \hat{a}^{\dagger} \hat{\sigma} \rangle \\ \langle \hat{\sigma}^{\dagger} \hat{a} \rangle \\ \langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle \end{bmatrix}, \qquad (B.2)$$

where

$$A_1 = -\frac{1}{2} \begin{bmatrix} \kappa & -2ig \\ -2ig & \Gamma \end{bmatrix},$$
 (B.3)

and

$$A_{2} = -\frac{1}{2} \begin{bmatrix} 2\kappa & 2ig & -2ig & 0\\ 2ig & \kappa + \Gamma & 0 & -2ig\\ -2ig & 0 & \kappa + \Gamma & 2ig\\ 0 & -2ig & 2ig & 2\gamma \end{bmatrix}.$$
 (B.4)

The cavity emission efficiency, or brightness, is defined by $\beta = \kappa \int_0^\infty \langle \hat{a}^{\dagger}(t) \hat{a}(t) \rangle dt$ in section 1.4.2. An exact solution to β can be derived by integrating the solution to Eq. (B.4). Assuming instantaneous excitation of the system, we set the initial condition to $\rho(0) = |e\rangle \langle e|$. This implies that the only nonzero initial condition for the optical Bloch equations is $\langle \hat{\sigma}^{\dagger} \hat{\sigma} \rangle = 1$. With this initial condition, we have

$$\beta = \kappa \int_0^\infty \exp(tA_2)_{14} dt = -\kappa \left(A_2^{-1}\right)_{14}$$

= $\frac{4g^2\kappa}{4g^2(\gamma + \kappa) + \gamma\kappa(\kappa + \Gamma)},$ (B.5)

where the subscripts denote the element of the matrix. This expression for brightness holds in all parameter regimes for a Markovian system and is equal to the result given in [86]. In this section, as with section 3.1, we have defined $R = 4g^2/\kappa$ and $\Gamma = \gamma + \gamma^*$. Please note that γ^* is defined differently here by a factor of 2 compared to the rest of this thesis. By substituting $g = (R\kappa)^{1/2}/2$ into this expression, it can be arranged to the form given in section 3.1. Also, notice that $\gamma \ll \kappa, g, \gamma^*$ gives $\beta = 1$.

B.2 Indistinguishability

We can analytically solve for the indistinguishability in a way similar to that for the efficiency. We are interested in the regime where pure dephasing γ^* is small relative to g and κ . Since A_1 can be easily diagonalized, the evolution given by A_1 can be solved exactly by computing $U(t) = \exp(A_1 t)$. However, to simplify solving the propagator $W(t) = \exp(A_2 t)$, we treat it perturbatively for $\gamma^*/(\kappa + \gamma) < 1$. Interestingly, this condition is satisfied if we only assume $\gamma^* < \kappa$ and so we need not make assumptions about the relative magnitude of γ and γ^* or g and γ^* .

Let $A_2^{(0)} = A_2(\gamma^* = 0)$, and let $A_2^{(1)} = A_2 - A_2^{(0)}$. Then we can write $W(t) = W^{(0)} + W^{(1)} + O(\gamma^{*2})$ where

$$W^{(0)} = \exp\left(A_2^{(0)}t\right),$$

$$W^{(1)} = W^{(0)} \int_0^t \exp\left(-A_2^{(0)}t'\right) A_2^{(1)} \exp\left(A_2^{(0)}t'\right) dt',$$

(B.6)

using the same perturbation approach introduced in section 1.5.5 but applied to the optical Bloch equations. The indistinguishability from section 1.4.4 is

$$I = \frac{2\kappa^2}{\beta^2} \int_0^\infty \int_0^\infty |\langle \hat{a}^{\dagger}(t+\tau)\hat{a}(t)\rangle|^2 dt d\tau.$$
(B.7)

Using the quantum regression theorem (section 1.5.2), we can write

$$\langle \hat{a}^{\dagger}(t+\tau)\hat{a}(t)\rangle = U_{11}(\tau) \langle \hat{a}^{\dagger}(t)\hat{a}(t)\rangle + U_{12}(\tau) \langle \hat{\sigma}^{\dagger}(t)\hat{a}(t)\rangle, \qquad (B.8)$$

where the subscripts denote the element of the matrix propagator U. Taking the initial condition to be $\langle \hat{\sigma}^{\dagger}(0)\hat{\sigma}(0)\rangle = 1$, we have $\langle \hat{a}^{\dagger}(t)\hat{a}(t)\rangle = W_{14}$ and $\langle \hat{\sigma}^{\dagger}(t)\hat{a}(t)\rangle = W_{34}$. Then the correlation function becomes

$$\langle \hat{a}^{\dagger}(t+\tau)\hat{a}(t)\rangle = U_{11}(\tau)W_{14}(t) + U_{12}(\tau)W_{34}(t).$$
 (B.9)

From this, we can write the indistinguishability as:

$$I\beta^{2} = 2\kappa^{2} \left[h_{1} + 2\operatorname{Re}(h_{2}) + h_{3} \right], \qquad (B.10)$$
where

$$h_{1} = \int_{0}^{\infty} |U_{11}|^{2} d\tau \int_{0}^{\infty} |W_{14}|^{2} dt,$$

$$h_{2} = \int_{0}^{\infty} U_{11}^{\star} U_{12} d\tau \int_{0}^{\infty} W_{14}^{\star} W_{34} dt,$$

$$h_{3} = \int_{0}^{\infty} |U_{12}|^{2} d\tau \int_{0}^{\infty} |W_{34}|^{2} dt.$$

(B.11)

This expression can be split into two parts: $I = I^{(0)} + I^{(1)} + O(\gamma^{*2}/(\kappa + \gamma)^2)$ where $I^{(0)}$ is computed using terms with U and $W^{(0)}$. With a little help from Mathematica, we found $I^{(0)}$ to be:

$$I^{(0)}\beta^2 = \frac{R^2\kappa^2 \left[3\gamma^*(2\gamma + 3\kappa + \gamma^*) + \Gamma_1^2\right]}{(R+\gamma)(\kappa+\gamma)(R+\gamma+\gamma^*)(\kappa+\Gamma)\Gamma_1^2},$$
(B.12)

where $\Gamma_1^2 = (3\gamma + \kappa)(\gamma + 3\kappa) + 4\kappa R.$

The perturbation $I^{(1)}$ contains the correction required for I to be exact to first-order in $\gamma^*/(\kappa+\gamma)$. Since we are only interested in the first-order correction in the perturbation of W(t), we only compute the W cross-terms that are first-order in $\gamma^*/(\kappa+\gamma)$. This means, for example, only computing terms with $W^{(0)}{}_{14}^*W^{(1)}_{34} \propto \gamma^*$ but not those with $W^{(1)}{}_{14}^*W^{(1)}_{34} \propto \gamma^{*2}$. However, we keep U(t) terms exact because any expansion of U(t) in γ^* would require the assumption that $\gamma^*/\gamma < 1$, which is not the desired case. For this reason, the result for $I^{(1)}$ for arbitrary γ still contains some higher-order γ^* terms:

$$\frac{I^{(1)}}{\gamma^{\star}I^{(0)}} = \frac{(R - 2\gamma) \left[(\gamma + \gamma^{\star})^{2} + \gamma \kappa \right]}{\left[3\gamma^{\star}(2\gamma + 3\kappa + \gamma^{\star}) + \Gamma_{1}^{2} \right] \Gamma_{2}^{2}} - \frac{\gamma^{\star}(\gamma - \gamma^{\star})(4\gamma + R) + 2\gamma(\gamma + R)(2\gamma + R)}{2(\gamma + \kappa)(\gamma + R)\Gamma_{2}^{2}} - \frac{(\gamma + \kappa)(8\gamma + 5R)}{2(\gamma + R)\Gamma_{1}^{2}},$$
(B.13)

where $\Gamma_2^2 = 3\gamma^{\star}(\gamma - \gamma^{\star}) + 4\gamma(\gamma + R).$

We use this solution to compute estimations and generate plots in section 3.1. The expression can be simplified under the assumption that $\gamma \ll \gamma^*$, which arises when quenching is weak. In this case, the expression $I = I^{(0)} + I^{(1)}$ can be reduced to the form given by Eq. (3.1).

B.3 Phonon sideband corrections

To estimate the correction to $I\beta$ expected due to the presence of a phonon sideband (PSB) in the SiV⁻ spectrum, we first estimated the fraction F of PSB not removed by the cavity [141]. In terms of wavelength, this can be defined as:

$$F(Q) = \frac{\int_0^\infty S_{\text{cav}}(\lambda, Q) \times S_{\text{PSB}}(\lambda) d\lambda}{\int_0^\infty S_{\text{PSB}}(\lambda) d\lambda},$$
(B.14)

where

$$S_{\text{cav}}(\lambda, Q) = \frac{1}{1 + 4Q^2 \frac{(\lambda - \lambda_0)^2}{\lambda_0^2}},$$
 (B.15)

and

$$S_{\text{PSB}}(\lambda) = \sum_{i} \frac{a_i}{1 + \frac{(\lambda - \lambda_0 - c_i)^2}{b_i^2}}.$$
(B.16)

The coefficients (a_i, b_i, c_i) were determined by visually fitting the total spectrum $S(\lambda) = S_{ZPL}(\lambda) + S_{PSB}(\lambda)$ to the measured spectrum of samples reported by Neu *et al.* [173], while also maintaining that:

$$DW \simeq \frac{\int_0^\infty S_{ZPL}(\lambda) d\lambda}{\int_0^\infty S(\lambda) d\lambda},$$
(B.17)

for the associated Debye-Waller (DW) values reported for that sample. Here we use:

$$S_{\text{ZPL}}(\lambda) = \frac{1}{1 + \frac{(\lambda - \lambda_0)^2}{\delta^2}},$$
(B.18)

where λ_0 is the zero-phonon line (ZPL) resonance of the sample and δ is the ZPL width in wavelength. We chose to fit the sample with the smallest δ (sample 3), and the sample with the largest DW (sample 5). The PSB coefficients that we determined are given in Table B.1.

The estimated spectra for samples 3 and 5 are illustrated in Fig. B.1. The corresponding DW factors were calculated to be $DW_3 = 0.791$ and $DW_5 = 0.884$, which closely match the measured values of 0.79 and 0.88 for samples 3 and 5, respectively. For a cavity with Q = 60, we estimated

	Sample 3			Sample 5		
i	$a_i \times 10^3$	b_i (nm)	$c_i (\mathrm{nm})$	$a_i \times 10^3$	b_i (nm)	$c_i (nm)$
1	1.4	1.3	4.0	1.4	1.7	7.5
2	6.7	6.5	10.5	1.8	8.0	11.0
3	2.0	6.0	20.5	2.6	2.9	17.5
4	2.2	6.0	32.0	2.0	3.3	22.5
5	2.4	0.9	39.0	0.9	2.5	27.0
6	1.0	20.0	47.0	1.2	5.0	33.0
7				1.0	8.0	39.0
8				0.3	1.1	41.5
9				0.7	18.0	49.0

Table B.1: Coefficients for $S_{PSB}(\lambda)$ used to represent the nanodiamond negatively-charged silicon vacancy centre spectrum of samples 3 and 5 from Neu *et al.* [173].

the fraction F to be $F_3(60) = 0.19$ and $F_5(60) = 0.15$. Using $B^2 = DW$ and the estimated values for F(Q), we applied the PSB corrections using Eqs. (3.3) and (3.4) to our results for indistinguishability I_0 and intrinsic cavity efficiency β_0 under the Markovian approximation.

The narrower linewidth of sample 3 would provide a higher $I_0\beta_0$ than sample 5. However, the larger PSB of sample 3 would restrict the indistinguishability for lower Q values. Hence the maximum $I\beta$ for sample 3 is attained at Q = 100 rather than Q = 60. For Q = 100, $R/\gamma_r = 2.7 \times 10^5$, and $\Delta_q (1 - \eta_r)^{-1/2} = 2\pi \times 30$ THz, sample 3 reaches I = 0.85 and $\beta = 0.99$ giving $I\beta = 0.84$, which is comparable to sample 5 at Q = 60 (I = 0.87, $\beta = 0.97$, $I\beta = 0.84$).

The regime in which the Markovian approximation is valid depends strongly on the shape and size of the emitter's PSB. In general, the approximation becomes less accurate as the cavity quality factor is decreased (see Fig. B.2 (a)). In addition, when computing I_0 to use in Eq. (3.3), we assume that the phonon-induced pure-dephasing rate is not significantly altered by the coupling between the PSB and the ZPL via the cavity mode. A sufficient condition to ensure that this assumption is valid is given when the pure-dephasing rate is much larger than the total effective rate between the PSB and the cavity mode: $\gamma^* \gg RS_{PSB}(\lambda_0)$. In this case, the cavity-PSB interaction can be considered as predominantly a filtering effect. For samples 3 and 5, we can estimate $S_0 = S_{PSB}(\lambda_0)$ as 2.4×10^{-3} and 9.6×10^{-4} , respectively. For $R > \gamma^*/S_0$, it may be necessary to consider the



Figure B.1: Estimated nanodiamond negatively-charged silicon-vacancy (SiV⁻) center emission spectrum of sample 3 (a,b) and sample 5 (c,d) from Neu *et al.* [173]. (b,d) Intensity log-scale plot illustrating the reduction of the PSB due to a cavity with quality factor Q = 60 on resonance with the zero-phonon line (ZPL) of the sample.

influence of the cavity-PSB interaction on the ZPL dephasing rate.

We also note that we have not considered interactions between the PSB and the higher-order plasmon modes. However, the PSB is predominantly Stokes-shifted to lower energy whereas the higher-order plasmon modes are generally of higher energy than the cavity resonance. Hence, the higher-order plasmon modes should not enhance the PSB and so this additional interaction can be safely neglected in the same regime where quenching does not dominate.

For narrow emitters with a spectrally-separated PSB, such as the nanodiamond SiV⁻, the $I\beta$ maximum can exist in a parameter regime where the detrimental effects of the PSB are small, leaving the maximum to be primarily limited by Markovian processes (see Fig. B.2 (b)). This leaves a cavity-parameter range where efficient indistinguishable emission should be attainable at room temperature.



Figure B.2: Error estimates for the Markovian approximation using a phonon sideband correction. (a) The relative error $2(I_0\beta_0 - I\beta)/(I_0\beta_0 + I\beta)$ for the PSB correction plotted in the critical regime and in the mode-detuned case with $\Delta_q(1 - \eta_r)^{-1/2} = 2\pi \times 30$ THz, where Δ_q is the effective detuning parameter for higher-order non-radiative plasmon modes and η_r is the bare cavity quantum efficiency. Here $R = 4g^2/\kappa$ where g is the cavity coupling rate and κ is the bare cavity linewidth. $I_0\beta_0$ is the derived estimation under the Markovian approximation and $I\beta$ is the value after including the correction due to the phonon sideband (PSB). This small correction could be inaccurate in the strong-coupling regime $(2g > \kappa + \gamma^* + \gamma)$ and when the PSB might begin to enhance the ZPL dephasing rate $R > \gamma^*/S_0$ (blue shaded regions). Parameters used for the SiV⁻: $\omega = 2\pi \times 405$ THz, $1/\gamma_r = 8.3$ ns [174], $\gamma^* = 2\pi \times 500$ GHz [173], DW = 0.88 [173], and $S_0 = 9.6 \times 10^{-4}$. See the Appendix text for the PSB spectrum used to calculate the correction. The blue dots represent the plasmonic bowtie [167] and a plasmonic-Fabry-Pérot hybrid cavity [166]. For the bowtie, a $R/\gamma_r = 1.7 \times 10^6$ ratio is determined from g = 60 meV, Q = 7.3 [167], and $1/\gamma_r = 20$ ns [176]. The dashed line marks $R/\gamma_r = 2.7 \times 10^5$ expected for the hybrid cavity (at Q = 986) from the enhancement of the local density of states (LDOS) [166]. The white star shows the proposed single-photon source. (b) Absolute difference $I_0\beta_0 - I\beta$ plotted to illustrate the magnitude of the correction.

Appendix C

Copyright permissions

In this appendix section, I give the copyright permissions to fully include papers [iii] and [ix] in this thesis in their published form.

C.1 Permissions from journals

Paper [iii] is published in Physical Review B and paper [ix] is published in Physical Review A. Both journals are of the American Physical Society (APS), which grants an author the right to include published papers in their thesis (see Fig. C.1).



Figure C.1: Permission to include APS papers in this thesis.

C.2 Permissions from co-authors

Each co-author of papers [iii] and [ix] have given their written permission by email to include these papers in this thesis.

- Christoph Simon, see Fig. C.2
- Roohollah (Farid) Ghobadi, see Fig. C.3
- Nikolai Lauk, see Fig. C.4
- Faezeh Kimiaee Asadi, see Fig. C.5
- Jia-Wei Ji, see Fig. C.6
- Yu-Feng Wu, see Fig. C.7

Christoph Simon

Tue 3/9/2021 1:35 PM To: Stephen Christopher Wein

You have my permission! Christoph

On Tue, Mar 9, 2021 at 1:20 PM Stephen Christopher Wein wrote: Hi Christoph,

Could you please send me a short email permitting me to publish the following two papers in my thesis?

S, Wein, N. Lauk, R. Ghobadi, and C. Simon, "Feasibility of efficient room-temperature solid-state sources of indistinguishable single photons using ultrasmall mode volume cavities" Physical Review B 97, 205418 (2018).

S. C. Wein, J. W. Ji, Y. F. Wu, F. K. Asadi, R. Ghobadi, and C. Simon, "Analyzing photon-count heralded entanglement generation between solid-state spin qubits by decomposing the master-equation dynamics" Physical Review A, 102, 033701 (2020).

Thanks, Stephen

Figure C.2: Permission from Christoph Simon to include Ref. [iii] and Ref. [ix] in this thesis.

Farid Ghobadi Tue 3/9/2021 2:01 PM To: Stephen Christopher Wein

Hi Stephen,

I hereby give you permission to publish the following two papers in your thesis:

S Wein, N Lauk, R. Ghobadi, and C Simon, "Feasibility of efficient room-temperature solid-state sources of indistinguishable single photons using ultrasmall mode volume cavities" Physical Review B 97, 205418 (2018).

S. C. Wein, J. W. Ji, Y. F. Wu, F. K. Asadi, R. Ghobadi, and C. Simon, "Analyzing photon-count heralded entanglement generation between solid-state spin qubits by decomposing the master-equation dynamics" Physical Review A, 102, 033701 (2020).

Cheers, Farid

On Tue, Mar 9, 2021 at 1:18 PM Stephen Christopher Wein wrote: Sorry, I actually need permission for both the PRB and PRA!

Could you please send me a short email permitting me to publish the following two papers in my thesis?

S Wein, N Lauk, R. Ghobadi, and C Simon, "Feasibility of efficient room-temperature solid-state sources of indistinguishable single photons using ultrasmall mode volume cavities" Physical Review B 97, 205418 (2018).

S. C. Wein, J. W. Ji, Y. F. Wu, F. K. Asadi, R. Ghobadi, and C. Simon, "Analyzing photon-count heralded entanglement generation between solid-state spin qubits by decomposing the master-equation dynamics" Physical Review A, 102, 033701 (2020).

Thanks, Stephen

Figure C.3: Permission from Roohollah (Farid) Ghobadi to include Ref. [iii] and Ref. [ix] in this thesis.

Lauk, Nikolai Wed 3/10/2021 11:40 AM To: Stephen Christopher Wein

Hi Stephen,

I'm doing good thank you. I am glad to see that you are in the final stage of your graduation and I have no objections for publishing this paper in your thesis.

Best, Nikolai

On Mar 10, 2021, at 12:10 AM, Stephen Christopher Wein wrote:

Hi Nikolai, I hope you are well!

Could you please send me a short email permitting me to publish the following paper in my thesis?

S Wein, N Lauk, R. Ghobadi, and C Simon,

"Feasibility of efficient room-temperature solid-state sources of indistinguishable single photons using ultrasmall mode volume cavities" Physical Review B 97, 205418 (2018).

Thanks, Stephen

Figure C.4: Permission from Nikolai Lauk to include Ref. [iii] in this thesis.

Faezeh Kimiaee Asadi

Tue 3/9/2021 1:33 PM To: Stephen Christopher Wein

Hi Stephen,

I absolutely permit you to do this.

Cheers, Faezeh

On Mar 9, 2021, at 1:21 PM, Stephen Christopher Wein wrote:

Hi Faezeh,

Could you please send me a short email permitting me to publish the following paper in my thesis?

S. C. Wein, J. W. Ji, Y. F. Wu, F. K. Asadi, R. Ghobadi, and C. Simon, "Analyzing photon-count heralded entanglement generation between solid-state spin qubits by decomposing the masterequation dynamics" Physical Review A, 102, 033701 (2020).

Thanks, Stephen

Figure C.5: Permission from Faezeh Kimiaee Asadi to include Ref. [ix] in this thesis.

Jiawei Ji Tue 3/9/2021 1:25 PM To: Stephen Christopher Wein

Hi Stephen, Absolutely! I grant my permission to you for publishing the following paper in your thesis!

S. C. Wein, J. W. Ji, Y. F. Wu, F. K. Asadi, R. Ghobadi, and C. Simon, "Analyzing photon-count heralded entanglement generation between solid-state spin qubits by decomposing the master-equation dynamics" Physical Review A, 102, 033701 (2020).

Cheers, Jiawei

From: Stephen Christopher Wein Date: Tuesday, March 9, 2021 at 4:21 PM To: Jiawei Ji Subject: Permission to publish our PRA in my thesis

Hi Jiawei,

Could you please send me a short email permitting me to publish the following paper in my thesis?

S. C. Wein, J. W. Ji, Y. F. Wu, F. K. Asadi, R. Ghobadi, and C. Simon, "Analyzing photon-count heralded entanglement generation between solid-state spin qubits by decomposing the master-equation dynamics" Physical Review A, 102, 033701 (2020).

Thanks, Stephen

Figure C.6: Permission from Jia-Wei Ji to include Ref. [ix] in this thesis.

Yufeng Wu Tue 3/9/2021 1:31 PM To: Stephen Christopher Wein

Hi Stephen,

Absolutely. I agree to include the publication (Physical Review A, 102, 033701 (2020).) in your thesis.

Best regards, Yufeng

From: Stephen Christopher Wein Sent: March 9, 2021 1:22 PM To: Yufeng Wu Subject: Permission to publish our PRA in my thesis

Hi Yufeng,

Could you please send me a short email permitting me to publish the following paper in my thesis?

S. C. Wein, J. W. Ji, Y. F. Wu, F. K. Asadi, R. Ghobadi, and C. Simon, "Analyzing photon-count heralded entanglement generation between solid-state spin qubits by decomposing the master-equation dynamics" Physical Review A, 102, 033701 (2020).

Thanks, Stephen

Figure C.7: Permission from Yu-Feng Wu to include Ref. [ix] in this thesis.